
SECURITIES AND EXCHANGE COMMISSION WASHINGTON, DC 20549

FORM 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE FISCAL YEAR ENDED SEPTEMBER 30, 2003

Commission file number 1-13692 Commission file number 33-92734-01 Commission file number 333-72986-02 Commission file number 333-72986-01

AMERIGAS PARTNERS, L.P. AMERIGAS FINANCE CORP. AMERIGAS EAGLE FINANCE CORP. AP EAGLE FINANCE CORP.

(EXACT NAME OF REGISTRANTS AS SPECIFIED IN THEIR CHARTERS)

 Delaware
 23-2787918

 Delaware
 23-2800532

 Delaware
 23-3074434

 Delaware
 23-3077318

(STATE OR OTHER JURISDICTION OF INCORPORATION OR ORGANIZATION)

(I.R.S. EMPLOYER IDENTIFICATION NO.)

460 North Gulph Road, King of Prussia, PA 19406 (ADDRESS OF PRINCIPAL EXECUTIVE OFFICES) (ZIP CODE)

(610) 337-7000

(REGISTRANTS' TELEPHONE NUMBER, INCLUDING AREA CODE)

SECURITIES REGISTERED PURSUANT TO SECTION 12(b) OF THE ACT:

TITLE OF CLASS

NAME OF EACH EXCHANGE ON WHICH REGISTERED

Common Units representing limited partner interests

New York Stock Exchange, Inc.

SECURITIES REGISTERED PURSUANT TO SECTION 12(g) OF THE ACT: None

INDICATE BY CHECK MARK WHETHER EACH REGISTRANT (1) HAS FILED ALL REPORTS REQUIRED TO BE FILED BY SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 DURING THE PRECEDING 12 MONTHS (OR FOR SUCH SHORTER PERIOD THAT THE REGISTRANT WAS REQUIRED TO FILE SUCH REPORTS) AND (2) HAS BEEN SUBJECT TO SUCH FILING REQUIREMENTS FOR THE PAST 90 DAYS. YES [X]. NO [].

INDICATE BY CHECK MARK IF DISCLOSURE OF DELINQUENT FILERS PURSUANT TO ITEM 405 OF REGULATION S-K IS NOT CONTAINED HEREIN, AND WILL NOT BE CONTAINED, TO THE BEST OF REGISTRANT'S KNOWLEDGE, IN DEFINITIVE PROXY OR INFORMATION STATEMENTS INCORPORATED BY REFERENCE IN PART III OF THIS FORM 10-K OR ANY AMENDMENT TO THIS FORM 10-K. [X]

The aggregate market value of AmeriGas Partners, L.P. Common Units held by nonaffiliates of AmeriGas Partners, L.P. on March 31, 2003 was approximately \$596,142,873. At December 15, 2003, there were outstanding 52,347,708 Common Units representing limited partner interests.

DOCUMENTS INCORPORATED BY REFERENCE: Portions of the AmeriGas Partners, L.P. Annual Report for the year ended September 30, 2003 are incorporated by reference in Part II of this Form 10-K.

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Act). Yes [X] No $[\]$

TABLE OF CONTENTS

				PAGE
PART	I: Ttem	1	BUSINESS. Business.	1
	Item		Properties	8
	Item	3.	Legal Proceedings	9
	Item	4.	Submission of Matters to a Vote of Security Holders	10
PART	II:		SECURITIES AND FINANCIAL INFORMATION	11
	Item	5.	Market for Registrant's Common Units and Related Security Holder Matters	11
	Item	6.	Selected Financial Data	13
	Item	7.	Management's Discussion and Analysis of Financial Condition and Results of	
			Operations	14
	Item	7A.	Quantitative and Qualitative Disclosures About Market Risk	27
	Item	8.	Financial Statements and Supplementary Data	27
	Item	9.	Changes in and Disagreements with Accountants on Accounting and Financial	
			Disclosure	27
	Item	9A.	Controls and Procedures	27
PART	III:		MANAGEMENT AND SECURITY HOLDERS	28
	Item	10.	Directors and Executive Officers of the General Partner	28
	Item	11.	Executive Compensation	32
	Item	12.	Security Ownership of Certain Beneficial Owners and Management and Related	
			Security Holder Matters	43

Item 13. Item 14.	Certain Relationships and Related TransactionsPrincipal Accountant Fees and Services	
PART IV: Item 15.	ADDITIONAL EXHIBITS, SCHEDULES AND REPORTS	48 57

PART T: BUSTNESS

ITEM 1. BUSINESS

GENERAL

AmeriGas Partners, L.P. ("AmeriGas Partners" or the "Partnership") is a publicly traded limited partnership formed under Delaware law on November 2, 1994. We are the largest retail propane distributor in the United States, based on retail sales volume. As of September 30, 2003, we served approximately 1.3 million residential, commercial, industrial, agricultural and motor fuel customers from approximately 650 district locations in 46 states.

We conduct our business principally through our subsidiary, AmeriGas Propane, L.P. ("AmeriGas OLP") and its subsidiary, AmeriGas Eagle Propane, L.P. ("Eagle OLP" and together with AmeriGas OLP, the "Operating Partnership"), both Delaware limited partnerships. Eagle OLP has a less-than-one percent minority limited partner. Our common units ("Common Units"), which represent limited partner interests, are traded on the New York Stock Exchange under the symbol "APU." Our executive offices are located at 460 North Gulph Road, King of Prussia, Pennsylvania 19406, and our telephone number is (610) 337-7000. In this report, the terms "Partnership" and "AmeriGas Partners," as well as the terms "our," "we," and "its," are used sometimes as abbreviated references to AmeriGas Partners, L.P. itself or AmeriGas Partners, L.P. and its consolidated subsidiaries, including the Operating Partnership.

On October 1, 2003, AmeriGas OLP acquired substantially all of the retail propane distribution assets and business of Horizon Propane LLC ("Horizon Propane") for approximately \$31 million. During its 2003 fiscal year, Horizon Propane sold over 30 million gallons of propane from ninety locations in twelve states.

AmeriGas Propane, Inc. is our general partner (the "General Partner"). The General Partner is a wholly owned subsidiary of UGI Corporation ("UGI"), a public company listed on the New York Stock Exchange and the Philadelphia Stock Exchange. Through various subsidiaries, UGI has been in the propane distribution business for over 40 years. The General Partner and its subsidiary Petrolane Incorporated have an effective 48% ownership interest in the Partnership. See Note 1 to the Partnership's Consolidated Financial Statements. The General Partner is responsible for managing our operations.

We have three co-registrants: AmeriGas Finance Corp., formed on March 13, 1995; AmeriGas Eagle Finance Corp., formed on February 22, 2001; and AP Eagle Finance Corp., formed on April 12, 2001 (each, a "Finance Corp." and together the "Finance Corp."). Each Finance Corp. was formed to serve as co-obligor for one of our series of senior notes. Each Finance Corp. has nominal assets and conducts no business operations. Accordingly, this report contains no discussion of the results of operations, liquidity or capital resources of any Finance Corp. The Finance Corp. executive offices are located at 460 North Gulph Road, King of Prussia, Pennsylvania 19406; telephone number (610) 337-7000.

BUSINESS STRATEGY

Our strategy is to increase market share through acquisitions and $% \left(1\right) =\left(1\right) \left(1\right) \left$ internal growth, leverage our national and local economies of scale, and achieve operating efficiencies through productivity improvements. We regularly consider and evaluate opportunities for growth through the acquisition of local, regional and national propane distributors. Acquisitions are an important part of our strategy, because the retail propane industry is mature, with only modest growth in total demand for the product foreseen. We may choose to finance future acquisitions with debt, equity, cash or a combination of the three. We compete for acquisitions with others engaged in the propane distribution business. Although we believe there are numerous potential acquisition candidates in the industry, there can be no assurance that we will find attractive candidates in the future, or that we will be able to acquire such candidates on economically acceptable terms. Internal growth will be provided in part from expansion of our PPX Prefilled Propane Xchange(R) and strategic accounts programs. In addition, we believe opportunities exist to grow our business internally through saleS and marketing programs designed to attract and retain customers.

GENERAL INDUSTRY INFORMATION

Propane is separated from crude oil during the refining process and also extracted from natural gas or oil wellhead gas at processing plants. Propane is normally transported and stored in a liquid state under moderate pressure or refrigeration for economy and ease of handling in shipping and distribution. When the pressure is released or the temperature is increased, it is usable as a flammable gas. Propane is colorless and odorless; an odorant is added to allow its detection. Propane is clean burning, producing negligible amounts of pollutants when properly consumed.

The primary customers for propane are residential, commercial, agricultural, motor fuel and industrial users to whom natural gas is not readily available. Propane is typically more expensive than natural gas, competitive with fuel oil when operating efficiencies are taken into account and, in most areas, cheaper than electricity on an equivalent energy basis.

PRODUCTS, SERVICES AND MARKETING

As of September 30, 2003, the Partnership distributed propane to approximately 1.3 million customers from approximately 650 district locations in 46 states. The Partnership also sells, installs and services propane appliances, including heating systems. In certain markets, the Partnership also installs and services propane fuel systems for motor vehicles. Typically, district locations are found in suburban and rural areas where natural gas is not available. Districts generally consist of an office, appliance showroom, warehouse and service facilities, with one or more 18,000 to 30,000 gallon storage tanks on the premises. As part of its overall transportation and distribution infrastructure, the Partnership operates as an interstate carrier in 48 states throughout the United States. It is also licensed as a carrier in Canada.

The Partnership sells propane primarily to five markets: residential, commercial/industrial, motor fuel, agricultural and wholesale. Approximately 84% of the Partnership's fiscal year 2003 sales (based on gallons sold) were to retail accounts and

approximately 16% were to wholesale customers. Sales to residential customers in fiscal 2003 represented approximately 42% of retail gallons sold; industrial/commercial customers 34%; motor fuel customers 11%; and agricultural customers 7%. Transport gallons, which are large-scale deliveries to retail customers other than residential, accounted for 6% of 2003 retail gallons. No single customer accounts for 5% or more of the Partnership's consolidated revenues

In the residential market, which includes both conventional and manufactured housing, propane is used primarily for home heating, water heating and cooking purposes. Commercial users, which include motels, hotels, restaurants and retail stores, generally use propane for the same purposes as residential customers. The Partnership continues to expand its PPX Prefilled Propane Xchange program ("PPX(R)"). At September 30, 2003, PPX was available at approximately 20,000 retail locations throughout the United States. Sales of our PPX grill cylinders to retailers are included in the commercial/industrial market. The PPX program enables consumers to exchange their empty 20-pound propane grill cylinders for filled cylinders at various retail locations such as home centers, mass merchandisers and grocery and convenience stores. During fiscal year 2002, the Partnership introduced PPX Plus. PPX Plus cylinders are equipped with a special overfill protection device ("OPD") required by the National Fire Protection Association ("NFPA").

Industrial customers use propane to fire furnaces, as a cutting gas and in other process applications. Other industrial customers are large-scale heating accounts and local gas utility customers who use propane as a supplemental fuel to meet peak load deliverability requirements. As a motor fuel, propane is burned in internal combustion engines that power over-the-road vehicles, forklifts and stationary engines. Agricultural uses include tobacco curing, chicken brooding and crop drying. In its wholesale operations, the Partnership principally sells propane to large industrial end-users and other propane distributors.

Retail deliveries of propane are usually made to customers by means of bobtail and rack trucks. Propane is pumped from the bobtail truck, which generally holds 2,400 to 3,000 gallons of propane, into a stationary storage tank on the customer's premises. The Partnership owns most of these storage tanks and leases them to its customers. The capacity of these tanks ranges from approximately 120 gallons to approximately 1,200 gallons.

The Partnership also delivers propane to retail customers in portable cylinders with capacities of 4 to 24 gallons. Some of these deliveries are made to the customer's location, where empty cylinders are either picked up for replenishment or filled in place.

PROPANE SUPPLY AND STORAGE

The Partnership has over 200 domestic and international sources of supply, including the spot market. Supplies of propane from the Partnership's sources historically have been readily available. During the year ended September 30, 2003, over 90% of the Partnership's propane supply was purchased under supply agreements with terms of 1 to 3 years. Approximately 79% of the volumes purchased under those agreements were from 10 suppliers, including BP roducts North America Inc. and its affiliate BP Marketing Inc. (approximately 20%); Dynegy Midstream Services (approximately 18%); and Enterprise Products Operating LP and its affiliate Canadian

Enterprises Gas Products Ltd. (approximately 15%). The availability of propane supply is dependent upon, among other things, the severity of winter weather and the price and availability of competing fuels such as natural gas and crude oil. Although no assurance can be given that supplies of propane will be readily available in the future, management currently expects to be able to secure adequate supplies during fiscal year 2004. If supply from major sources were interrupted, however, the cost of procuring replacement supplies and transporting those supplies from alternative locations might be materially higher and, at least on a short-term basis, margins could be affected. Aside from BP, Dynegy and Enterprise Products, no single supplier provided more than 10% of the Partnership's total propane supply in fiscal year 2003. In certain market areas, however, some suppliers provide 70% to 80% of the Partnership's requirements. Disruptions in supply in these areas could also have an adverse impact on the Partnership's margins.

During fiscal year 2003, 92% of the Partnership's supply contracts provided for pricing based upon posted prices at the time of delivery or index formulas based on the current prices established at major storage points such as Mont Belvieu, Texas, or Conway, Kansas. In addition, some agreements provided maximum and minimum seasonal purchase volume guidelines. The percentage of contract purchases, and the amount of supply contracted for at fixed prices, will vary from year to year as determined by the General Partner. The Partnership uses a number of interstate pipelines, as well as railroad tank cars, delivery trucks and barges, to transport propane from suppliers to storage and distribution facilities. The Partnership stores propane at facilities in Arizona, Pennsylvania, Virginia and several other states.

Because the Partnership's profitability is sensitive to changes in wholesale propane costs, the Partnership generally seeks to pass on increases in the cost of propane to customers. There is no assurance, however, that the Partnership will always be able to pass on product cost increases fully, particularly when product costs rise rapidly. Product cost increases can be triggered by periods of severe cold weather, supply interruptions, or other unforeseen events. The General Partner has adopted supply acquisition and product price risk management practices to reduce the effect of price volatility on product costs. These practices currently include the use of summer storage, forward purchases and derivative commodity instruments such as options and propane price swaps. See "Management's Discussion and Analysis of Financial Condition and Results of Operations - Market Risk Disclosures."

The following graph shows the average prices of propane on the propane spot market during the last five fiscal years at Mont Belvieu, Texas and Conway, Kansas, two major storage areas.

Conway

[AVERAGE PROPANE SPOT MARKET PRICES GRAPH]

	HOHE BETVICE	Conway
Oct-96	51.57	51.53
Nov-96	58.05	63.41
Dec-96	61.04	84.29
Jan-97	47.45	63.39
Feb-97	38.71	39.02
Mar-97	38.50	37.26
Apr-97	34.88	35.26
May-97	35.31	36.48
Jun-97	34.43	35.86
Jul-97	34.91	34.63
Aug-97	37.03	36.53
Sep-97	38.68	37.95
Oct-97	39.83	37.32
Nov-97	35.95	35.00
Dec-97	33.57	31.36
Jan-98	30.07	28.21
Feb-98	29.79	28.32
Mar-98	27.39	27.84
Apr-98	29.06	29.47
May-98	27.42	27.82
Jun-98	24.42	24.84
Jul-98	24.54	24.55
Aug-98	24.12	23.87
Sep-98	24.83	24.04
Oct-98	25.72	24.57
Nov-98	24.79	23.20
Dec-98	20.89	18.72
Jan-99	21.75	19.61
Feb-99	22.43	20.58
Mar-99	24.10	23.40
Apr-99	28.26	27.58
May-99	28.31	26.88
Jun-99	30.95	28.68
Jul-99	37.26	34.62
Aug-99	40.51	37.56
Sep-99	43.18	42.40
Oct-99	45.46	43.39
Nov-99	43.44	38.78
Dec-99	42.83	35.10
Jan-00	56.11	42.32
Feb-00	59.72	47.26
Mar-00	51.13	47.65
Apr-00	46.88	43.64
May-00	51.31	50.81
• • •		

Mont Belvieu

Jun-00	55.47	56.22
Jul-00	54.88	56.29
Aug-00	58.54	63.52
Sep-00	64.21	70.95
Oct-00	61.82	64.05
Nov-00	60.71	60.45
Dec-00	77.63	79.75
Jan-01	77.27	83.03
Feb-01	59.39	63.03
Mar-01	54.94	57.12
Apr-01	54.37	60.26
May-01	51.20	56.90
Jun-01	43.17	47.70
Jul-01	38.87	43.27
Aug-01	41.54	45.71
Sep-01	41.67	46.53
Oct-01	39.48	44.19
Nov-01	33.04	35.19
Dec-01	30.43	30.34
Jan-02	29.05	26.60
Feb-02	31.20	27.92
Mar-02	37.95	35.93
Apr-02	41.52	40.07
May-02	40.69	38.09
Jun-02	37.51	35.25
Jul-02	37.19	35.47
Aug-02	41.49	41.53
Sep-02	47.17	45.93
0ct-02	47.95	47.12
Nov-02	47.26	48.01
Dec-02	52.40	52.32
Jan-03	60.38	57.70
Feb-03	77.30	73.03
Mar-03	62.77	57.09
Apr-03	50.42	50.28
May-03	54.09	55.41
Jun-03	55.98	59.71
Jul-03	53.01	58.90
Aug-03	54.84	63.63 59.44
Sep-03	52.00	59.44

COMPETITION

Propane competes with other sources of energy, some of which are less costly for equivalent energy value. Propane distributors compete for customers against suppliers of electricity, fuel oil and natural gas, principally on the basis of price, service, availability and portability. Electricity is a major competitor of propane, but propane generally enjoys a competitive price advantage over electricity for space heating, water heating and cooking. Fuel oil is also a major competitor of propane and is generally less expensive than propane. Operating efficiencies and other factors such as air quality and environmental advantages, however, generally make propane competitive with fuel oil as a heating source. Furnaces and appliances that burn propane will not operate on fuel oil, and vice versa, and, therefore, a conversion from one fuel to the other requires the installation of new equipment. Propane serves as an alternative to natural gas in rural and suburban areas where natural gas is unavailable or portability of product is required. Natural gas is generally a less expensive source of energy than propane, although in areas where natural gas is available, propane is used for certain industrial and commercial applications and as a standby fuel during interruptions in natural gas service. The gradual expansion of the nation's natural gas distribution systems has resulted in the availability of natural gas in some areas that previously depended upon propane. However, natural gas pipelines are not present in many regions of the country where propane is sold for heating and cooking purposes.

In the motor fuel market, propane competes with gasoline and diesel fuel as well as electric batteries and fuel cells. Wholesale propane distribution is a highly competitive, low margin business. Propane sales to other retail distributors and large-volume, direct-shipment industrial end-users are price sensitive and frequently involve a competitive bidding process.

The retail propane industry is mature, with only modest growth in total demand for the product foreseen. Therefore, the Partnership's ability to grow within the industry is dependent on its ability to acquire other retail distributors and to achieve internal growth, which includes expansion of the PPX(R)program (through which consumers can exchange an empty propane grill cylinder for a filled one) and strategic accounts program (formerly called "national accounts," through which the Partnership encourages large, multi-location propane users to enter into a supply agreement with it rather than with many small suppliers), as well as the success of its sales and marketing programs designed to attract and retain customers. The failure of the Partnership to retain and grow its customer base would have an adverse effect on its results.

The domestic propane retail distribution business is highly competitive. The Partnership competes in this business with other large propane marketers, including other full-service marketers, and thousands of small independent operators. In recent years, some rural electric cooperatives and fuel oil distributions have expanded their businesses to include propane distribution and the Partnership competes with them as well. The ability to compete effectively depends on providing high quality customer service, maintaining competitive retail prices and controlling operating expenses.

Based on the most recent annual survey by the American Petroleum Institute, the 2002 domestic retail market for propane (annual sales for other than chemical uses) was approximately 11.9 billion gallons and, based on LP-GAS magazine rankings, 2002 sales volume of the ten

largest propane companies (including AmeriGas Partners) represented approximately 32% of domestic retail sales. Management believes the Partnership's 2003 retail volume represents 9% of the domestic retail market.

TRADE NAMES, TRADE AND SERVICE MARKS

The Partnership markets propane principally under the "AmeriGas(R)," "America's Propane Company(R)" and "PPX Prefilled Propane Xchange(R)" trade names and related service marks. UGI owns, directly or indirectly, all the right, title and interest in the "AmeriGas" and related trade and service marks. The General Partner owns all right, title and interest in the "America's Propane Company" and "PPX Prefilled Propane Xchange" trade names and related service marks. The Partnership has an exclusive (except for use by UGI, AmeriGas, Inc. and the General Partner), royalty-free license to use these names and trade and service marks. UGI and the General Partner each have the option to terminate its respective license agreement (on 12 months prior notice in the case of UGI), without penalty, if the General Partner is removed as general partner of the Partnership other than for cause. If the General Partner ceases to serve as the general partner of the Partnership for cause, the General Partner has the option to terminate its license agreement upon payment of a fee equal to the fair market value of the licensed trade names. UGI has a similar termination option, however, UGI must provide 12 months prior notice in addition to paying the fee.

SEASONALITY

Because many customers use propane for heating purposes, the Partnership's retail sales volume is seasonal, with approximately 59% of the Partnership's fiscal year 2003 retail sales volume occurring during the five-month peak heating season from November through March. As a result of this seasonality, sales are concentrated in the Partnership's first and second fiscal quarters (October 1 through March 31). Cash receipts are greatest during the second and third fiscal quarters when customers pay for propane purchased during the winter heating season.

Sales volume for the Partnership traditionally fluctuates from year-to-year in response to variations in weather, prices, competition, customer mix and other factors, such as conservation efforts and general economic conditions. For historical information on national weather statistics, see "Management's Discussion and Analysis of Financial Condition and Results of Operations."

GOVERNMENT REGULATION

The Partnership is subject to various federal, state and local environmental, safety and transportation laws and regulations governing the storage, distribution and transportation of propane. These laws include, among others, the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA" or, the "Superfund Law"), the Clean Air Act, the Occupational Safety and Health Act, the Homeland Security Act of 2002, the Emergency Planning and Community Right to Know Act, the Clean Water Act and comparable state statutes. CERCLA imposes joint and several liability on certain classes of persons considered to have contributed to the release or threatened release of a "hazardous substance" into the environment without regard to fault or the legality of the original conduct. Propane is not a hazardous substance within the meaning of federal and state environmental laws. However, the Partnership owns and operates real property where such

hazardous substances may exist. See Notes 1 and 11 to the Partnership's Consolidated Financial Statements.

All states in which the Partnership operates have adopted fire safety codes that regulate the storage and distribution of propane. In some states these laws are administered by state agencies, and in others they are administered on a municipal level. The Partnership conducts training programs to help ensure that its operations are in compliance with applicable governmental regulations. With respect to general operations, NFPA Pamphlets No. 54 and No. 58, which establish a set of rules and procedures governing the safe handling of propane, or comparable regulations, have been adopted as the industry standard in a majority of the states in which the Partnership operates. Effective April 1, 2002, NFPA Pamphlet No. 58 requires portable propane cylinders to be equipped with an OPD. Although NFPA Pamphlet No. 58 has not yet been adopted in all states, the Partnership complies with the OPD requirements throughout the United States. Effective July 2004, NFPA Pamphlet No. 58 requires stationary cylinders that are filled in place to be re-qualified periodically. The Partnership maintains various permits under environmental laws that are necessary to operate certain of its facilities, some of which may be material to the operations of the Partnership. Management believes that the procedures currently in effect at all of its facilities for the handling, storage and distribution of propane are consistent with industry standards and are in compliance in all material respects with applicable environmental, health and safety laws.

With respect to the transportation of propane by truck, the Partnership is subject to regulations promulgated under the Federal Motor Carrier Safety Act and the Homeland Security Act of 2002. These regulations cover the security of and transportation of hazardous materials and are administered by the United States Department of Transportation ("DOT"). The Natural Gas Safety Act of 1968 required the DOT to develop and enforce minimum safety regulations for the transportation of gases by pipeline. The DOT's pipeline safety code applies to, among other things, a propane gas system which supplies 10 or more customers from a single source and a propane gas system any portion of which is located in a public place. The code requires operators of all gas systems to provide training and written instructions for employees, establish written procedures to minimize the hazards resulting from gas pipeline emergencies, and keep records of inspections and testing.

EMPL OYEES

The Partnership does not directly employ any persons responsible for managing or operating the Partnership. The General Partner provides these services and is reimbursed for its direct and indirect costs and expenses, including all compensation and benefit costs. At September 30, 2003, the General Partner had approximately 6,200 employees, including approximately 400 temporary and part-time employees, working on behalf of the Partnership. UGI also performs certain financial and administrative services for the General Partner on behalf of the Partnership and is reimbursed by the Partnership for its direct and indirect costs and expenses.

ITEM 2. PROPERTIES

As of September 30, 2003, the Partnership owned approximately 87% of its district locations. In addition, the Partnership subleases three one-million barrel underground storage caverns in Arizona to store propane and butane for itself and third parties. The Partnership also owns a 600,000 barrel refrigerated, above-ground storage facility located on leased property in California, which could be used in connection with waterborne imports or exports of propane or butane. The California facility, which the Partnership operates, is currently leased to several refiners for the storage of butane. In Virginia, the Partnership has a 50% indirect equity interest in a 476,000 barrel refrigerated, above-ground import terminal.

The transportation of propane requires specialized equipment. The trucks and railroad tank cars utilized for this purpose carry specialized steel tanks that maintain the propane in a liquefied state. As of September 30, 2003, the Partnership operated a transportation fleet with the following assets:

APPROXIM	MATE QUANTITY & EQUIPMENT TYPE	% OWNED	% LEASED
400	Trailers	90	10
200	Tractors	45	55
150	Railroad tank cars	0	100
2,500	Bobtail trucks	20	80
350	Rack trucks	25	75
2,200	Service and delivery trucks	25	75

Other assets owned at September 30, 2003 included approximately 900,000 stationary storage tanks with typical capacities of 121 to 2,000 gallons and approximately 2.4 million portable propane cylinders with typical capacities of 1 to 120 gallons. The Partnership also owned approximately 6,100 large volume tanks which are used for its own storage requirements. AmeriGas OLP has debt secured by liens and mortgages on its real and personal property. AmeriGas OLP owns approximately 67% of the Partnership's property, plant and equipment.

ITEM 3. LEGAL PROCEEDINGS

With the exception of the matter set forth below, no material legal proceedings are pending involving the Partnership, any of its subsidiaries, or any of their properties, and no such proceedings are known to be contemplated by governmental authorities other than claims arising in the ordinary course of the Partnership's business.

Swiger, et al. v. UGI/AmeriGas, Inc. et al. Plaintiffs Samuel and Brenda Swiger and their son (the "Swigers") sustained personal injuries and property damage as a result of a fire that occurred when propane that leaked from an underground line ignited. In July 1998, the Swigers filed a class action lawsuit against AmeriGas Propane, L.P. (named incorrectly as "UGI/AmeriGas, Inc."), in the circuit Court of Monongalia County, West Virginia (Civil Action No. 98-C-298), in which they sought to recover an unspecified amount of compensatory and punitive damages and attorney's fees, for themselves and on behalf of persons in West Virginia for whom the defendants had installed propane gas lines, allegedly resulting from the defendants' failure to install underground propane lines at depths required by applicable safety standards. Plaintiffs have filed various motions with the court, which seek to broaden the scope of their claims and to expand the size of the class to include customers whose lines were installed by other propane suppliers. These motions are currently pending before the court and defendants cannot predict the outcome of those motions. Beginning in 2001, the defendants voluntarily undertook to inspect and replace underground lines of its current customers that may not be in compliance with applicable safety standards. The General Partner expects to complete the line replacement project by late fiscal 2004 or early fiscal 2005. In 2003, the defendants settled the individual personal injury and property damage claims of the Swigers. The defendants believe

they have defenses to the claims of the class members and intend to vigorously defend against the remaining claims in this lawsuit.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

First Quarter

ITEM 5. MARKET FOR REGISTRANT'S COMMON UNITS AND RELATED SECURITY HOLDER MATTERS

Each Common Unit represents a limited partner interest in the Partnership. Common Units are listed on the New York Stock Exchange, which is the principal trading market for such securities, under the symbol "APU." The following table sets forth, for the periods indicated, the high and low sale prices per Common Unit, as reported on the New York Stock Exchange ("NYSE") Composite Transactions tape, and the amount of cash distributions paid per Common Unit.

2003 FISCAL YEAR	PRICE	RANGE	CASH
	HIGH	LOW	DISTRIBUTION
Fourth Quarter	\$27.13	\$22.50	\$0.55
Third Quarter	27.25	24.00	0.55
Second Quarter	25.09	23.30	0.55
First Quarter	24.73	20.25	0.55
2002 FISCAL YEAR	PRICE	RANGE	CASH
	HIGH	LOW	DISTRIBUTION
Fourth Quarter	\$23.79	\$17.11	\$0.55
Third Quarter	23.85	21.32	0.55
Second Quarter	22.74	19.36	0.55

As of December 15, 2003, there were 1,588 record holders of the Partnership's Common Units. Effective November 18, 2002, the Partnership's subordinated units, representing limited partner interests, were converted to Common Units.

21.40

0.55

25.35

The Partnership makes quarterly distributions to its partners in an aggregate amount equal to its Available Cash, as defined in the Second Amended and Restated Agreement of Limited Partnership of AmeriGas Partners, L.P., which is filed as an exhibit to this report. Available Cash generally means, with respect to any fiscal quarter of the Partnership, all cash on hand at the end of such quarter, plus all additional cash on hand as of the date of determination resulting from borrowings subsequent to the end of such quarter, less the amount of cash reserves established by the General Partner in its reasonable discretion for future cash requirements. Certain reserves are maintained to provide for the payment of principal and interest under the terms of the Partnership's debt agreements and other reserves may be maintained to provide for the proper conduct of the Partnership's business, and to provide funds for distribution during the next four fiscal quarters. The information concerning restrictions on distributions required by Item 5 of this report is incorporated herein by reference to Notes 5 and 6 to the Partnership's Consolidated Financial Statements which are incorporated herein by reference. Prior to the termination of the subordination period effective November 18, 2002, Distributions of Available Cash to the

General Partner on its subordinated units, representing limited partner interests ("Subordinated Units") were subject to the prior rights of holders of the Common Units to receive the Minimum Quarterly Distribution ("MQD") for each quarter during the subordination period, and to receive any arrearages in the distribution of the MQD on the Common Units for prior quarters during the subordination period. In December 2002, the General Partner determined that the cash-based performance and distribution requirements for termination of the subordination period were achieved in respect of the quarter ended September 30, 2003. As a result, effective November 18, 2002, the subordinated units held by the General Partner were converted to Common Units.

Year Ended September 30,

	September 30,									
		2003		2002		2001 (a)	2000 (a)			1999 (a)
			(Thousands of	do	llars, exce	ept	per unit)		
FOR THE PERIOD: INCOME STATEMENT DATA:										
Revenues	\$ 1	1,628,424	\$:	L,307,880	\$	1,418,364	\$	1,120,056	\$	872,535
Income before accounting changes Cumulative effect of accounting changes (b)	\$	71,958	\$	55,366	\$	53,015 12,494	\$	15,196	\$	25, 635
Net income (c) (d)	\$	71,958	\$	55,366	\$	65,509	\$	15,196	\$	25,635
Limited partners' interest in net income Income per limited partner unit - basic and diluted:	\$	71,238	\$	54,812	\$	64,854	\$	15,044	\$	25,379
Income before accounting changes Cumulative effect of accounting changes	\$	1.42	\$	1.12	\$	1.18 0.28	\$	0.36	\$	0.61
Net income (c) (d)	\$	1.42	\$	1.12	\$	1.46	\$	0.36	\$	0.61
Cash distributions declared per limited partner unit	\$	2.20	\$	2.20	\$	2.20	\$	2.20	\$	2.20
AT PERIOD END: BALANCE SHEET DATA:										
Current assets	\$	236,332	\$	224,843	\$	230,260	\$	188,845	\$	140,569
Total assets	~	1,482,176		L,472,618		1,496,422		1,258,220		1,196,461
Current liabilities (excluding debt)	_	239,343		237,426		238,512		172,501		148,513
Total debt		927,286		955,784		1,005,904		887,234		766,725
Minority interests		7,005		6,232		5,641		2,587		3,380
Partners' capital		253,683		228,366		203,505		155, 971		234, 041
OTHER DATA:	•	224 264	•	200 640	•	222 222	•	457 000	•	457 450
EBITDA (e) Capital expenditures (including	\$	234,364	\$	209,649	\$	220,338	\$	157,326	\$	157,156
capital leases)	\$	53,429	\$	53,472	\$	39,204	\$	30,427	\$	34,577
Retail propane gallons sold (millions) (f) Degree days - % (warmer) colder		1,074.9		987.5		866.8		817.7		829.5
than normal (g)		0.2		(10.0)		2.6		(13.7)		(9.9)

- (a) Arthur Andersen LLP audited the Partnership's consolidated financial statements for 2001, 2000, and 1999. See Item 15 - Notice Regarding Arthur Andersen LLP.
- (b) Includes cumulative effect of accounting changes associated with the Partnership's changes in accounting for tank fee revenue and tank installation costs and the adoption of Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities" (see Notes 2 and 18 to Consolidated Financial Statements).
- (c) Pro forma net income and net income per limited partner unit after applying retroactively the changes in accounting for tank installation costs and tank fee revenue are as follows: 2000 \$14,989 and \$0.35; 1999 \$26,091 and \$0.62, respectively.
- (d) SFAS No. 142, "Goodwill and Other Intangible Assets," was adopted effective October 1, 2001. Net income and net income per limited partner unit adjusted to reflect the impact of SFAS No. 142 as if it had been adopted at the beginning of the periods presented are as follows: 2001 -\$89,079 and \$1.98; 2000 - \$38,313 and \$0.90; 1999 - \$48,336 and \$1.14, respectively.
- (e) EBITDA (earnings before interest expense, income taxes, depreciation and amortization) should not be considered as an alternative to net income (as an indicator of operating performance) or as an alternative to cash flow (as a measure of liquidity or ability to service debt obligations) and is not a measure of performance or financial condition under accounting principles generally accepted in the United States. Management believes EBITDA is a meaningful non-GAAP measure used by investors to compare the Partnership's operating performance with other companies within the propane industry. Weather significantly impacts demand for propane and profitability because many customers use propane for heating purposes.
- (f) Retail gallons sold in 2003 include certain bulk gallons previously considered wholesale gallons. Prior-year gallon amounts have been adjusted to conform to the current-year classification.
- (g) Deviation from average heating degree days based upon national weather statistics provided by the National Oceanic and Atmospheric Administration (NOAA) for 335 airports in the United States, excluding Alaska.

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Our Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") relates to AmeriGas Partners and the Operating Partnerships. AmeriGas Finance Corp., AmeriGas Eagle Finance Corp. and AP Eagle Finance Corp. have nominal assets and do not conduct any operations but act as co-obligors of certain AmeriGas Partners' debt securities. Accordingly, discussions of the results of operations and financial condition and liquidity of these entities are not presented. Our MD&A should be read in conjunction with our consolidated financial statements and related notes thereto included elsewhere in this Annual Report on Form 10-K.

ANALYSIS OF RESULTS OF OPERATIONS

The following analysis compares the Partnership's results of operations for (1) the year ended September 30, 2003 ("Fiscal 2003") with the year ended September 30, 2002 ("Fiscal 2002") and (2) Fiscal 2002 with the year ended September 30, 2001 ("Fiscal 2001").

The following table provides gallon, weather and certain financial information for the Partnership should be read in conjunction with "Fiscal 2003 Compared to Fiscal 2002" and "Fiscal 2002 Compared to Fiscal 2001":

(Dollars in millions)

	Year Ended September 30,			
	2003	2002	2001	
Gallons sold (millions) (a):				
Retail	1,074.9	987.5	866.8	
Wholesale	209.8	195.3	246.8	
	1,284.7	1,182.8	1,113.6	
	=======	=======	=======	
Revenues:				
Retail propane	\$ 1,375.5	\$ 1,102.8	\$ 1,147.3	
Wholesale propane	127.1	88.8	175.6	
Other	125.8	116.3	95.5	
	\$ 1,628.4	\$ 1,307.9	\$ 1,418.4	
	\$ 1,020.4 =======	φ 1,307.9 =======	\$ 1,410.4 =======	
Total margin (b)	\$ 718.1	\$ 654.8	\$ 571.3	
EBITDA (c)	\$ 234.4	\$ 209.6	\$ 220.3	
Operating income (d)	\$ 164.0	\$ 145.3	\$ 133.8	
Degree days - % colder (warmer)				
than normal (e)	0.2%	(10.0)%	2.6%	

- (a) Retail gallons sold in Fiscal 2003 include certain bulk gallons previously considered wholesale gallons. Prior-year gallon amounts have been adjusted to conform to the current year classification.
- (b) Total margin represents total revenues less total cost of sales.

(c) EBITDA (earnings before interest expense, income taxes, depreciation and amortization) should not be considered as an alternative to net income (as an indicator of operating performance) or as an alternative to cash flow (as a measure of liquidity or ability to service debt obligations) and is not a measure of performance or financial condition under accounting principles generally accepted in the United States of America. Management believes EBITDA is a meaningful non-GAAP financial measure used by investors to compare the Partnership's operating performance with other companies within the propane industry and to evaluate our ability to meet loan covenants. Weather significantly impacts demand for propane and profitability because many customers use propane for heating purposes. The following table includes reconciliations of net income to EBITDA for the fiscal years presented:

Year E	inded	Septemb	oer	30,
--------	-------	---------	-----	-----

	2003	2002	2001
Net income	\$ 72.0	\$ 55.4	\$ 65.5
Income tax expense (benefit)	0.6	0.3	(0.3)
Interest expense	87.2	87.8	80.4
Depreciation	70.4	62.0	48.1
Amortization	4.2	4.1	26.6
EBITDA	\$ 234.4	\$ 209.6	\$ 220.3
	======	======	======

- (d) Results for Fiscal 2003 and 2002 reflect the elimination of goodwill amortization resulting from the adoption of Statement of Financial Accounting Standards No. 142, "Goodwill and Other Intangible Assets." Pro forma operating income for Fiscal 2001 as if the adoption of SFAS 142 had occurred as of October 1, 2000 is \$157.6 million.
- (e) Deviation from average heating degree days based upon national weather statistics provided by the National Oceanic and Atmospheric Administration ("NOAA") for 335 airports in the United States, excluding Alaska.

FISCAL 2003 COMPARED WITH FISCAL 2002

Weather based upon heating degree days was essentially normal during Fiscal 2003 compared to weather that was 10.0% warmer than normal in Fiscal 2002. Although temperatures nationwide averaged near normal during Fiscal 2003, our overall results reflect weather that was significantly warmer in the West and generally colder than normal in the East. Retail propane volumes sold increased 87.4 million gallons in Fiscal 2003 due principally to the effects of the colder weather and, to a much lesser extent, volume growth from acquisitions and customer growth. These increases were achieved notwithstanding the effects of price-induced customer conservation and, with respect to commercial and industrial customers, continuing economic weakness.

Retail propane revenues increased \$272.7 million reflecting (1) a \$175.1 million increase due to higher average selling prices and (2) a \$97.6 million increase due to the higher retail volumes sold. Wholesale propane revenues increased \$38.3 million reflecting (1) a \$31.7 million increase due to higher average selling prices and (2) a \$6.6 million increase due to the higher volumes sold. The higher retail and wholesale selling prices reflect significantly higher propane product costs during Fiscal 2003 resulting from, among other things, higher crude oil and natural gas

prices and lower propane inventories. Other revenues from ancillary sales and services were \$125.8 million in Fiscal 2003 and \$116.3 million in Fiscal 2002. Total cost of sales increased \$257.3 million reflecting the higher propane product costs and higher volumes sold.

The \$63.3 million increase in total margin is principally due to the higher propane gallons sold and, to a lesser extent, slightly higher average retail propane unit margins. Notwithstanding the previously mentioned significant increase in the commodity price of propane, retail propane unit margins were slightly higher than the prior year reflecting the effects of the higher average selling prices and the benefits of favorable propane product cost management activities. Beginning in Fiscal 2002 and continuing in Fiscal 2003, unit margins associated with the Partnership's Prefilled Propane Xchange program ("PPX(R)") were higher than historical levels reflecting increases in PPX(R) sales prices to fund cylinder valve replacement capital expenditures. These capital expenditures resulted from National Fire Protection Association ("NFPA") guidelines enacted in Fiscal 2002 requiring propane grill cylinders be fitted with overfill protection devices ("OPDs"). The extent to which this level of PPX(R) margin is sustainable in the future will depend upon a number of factors including the continuing rate of OPD valve replacement and competitive market conditions.

EBITDA increased \$24.8 million in Fiscal 2003 reflecting the previously mentioned increase in total margin and a \$4.6 million increase in other income partially offset by a \$40.6 million increase in operating and administrative expenses and a \$2.3 million increase in losses associated with early extinguishments of long-term debt. Although EBITDA is not a measure of performance or financial condition under accounting principles generally accepted in the United States of America, management believes EBITDA is a meaningful non-GAAP financial measure used by investors to compare the Partnership's operating performance with other companies within the propane industry and to evaluate our ability to meet loan covenants. Operating and administrative expenses increased principally due to higher medical and general insurance expenses, higher distribution expenses as a result of the previously mentioned greater retail volumes, and higher incentive compensation and uncollectible accounts expenses. In addition, the Partnership incurred \$3.8 million of costs during Fiscal 2003 associated with a realignment of the Partnership's management structure announced in June 2003. Other income in Fiscal 2003 includes a gain of \$1.1 million from the settlement of certain hedge contracts and greater income from finance charges and asset sales while other income in the prior year was reduced by a \$2.1 million loss from declines in the value of propane commodity option contracts. Operating income in Fiscal 2003 increased less than the increase in EBITDA due to higher depreciation expense principally associated with PPX(R) partially offset by the previously mentioned increase in losses associated with early extinguishments of long-term debt.

FISCAL 2002 COMPARED WITH FISCAL 2001

The Partnership's Fiscal 2002 operating results were negatively impacted by significantly warmer than normal heating-season weather. Fiscal 2002 temperatures based upon heating degree day data provided by NOAA were approximately 10.0% warmer than normal and 12.3% warmer than Fiscal 2001. Notwithstanding the impact of the warmer weather on heating-related sales and the effects of a sluggish U.S. economy on commercial sales, retail gallons sold increased 120.7 million gallons principally as a result of the full-year effect of the Partnership's August 21, 2001 acquisition of Columbia Propane and, to a much lesser extent, greater volumes from our PPX(R) grill cylinder exchange business. The increase in PPX(R) sales principally reflects the effect on Fiscal 2002 grill cylinder exchanges resulting from the previously mentioned NFPA

guidelines requiring grill cylinders be fitted with OPDs and, to a lesser extent, the full-year effects of Fiscal 2001 increases in the number of PPX(R) distribution outlets.

Retail propane revenues were \$1,102.8 million in Fiscal 2002, a decrease of \$44.5 million from Fiscal 2001, reflecting a \$204.3 million decrease as a result of lower average selling prices partially offset by a \$159.8 million increase as a result of the greater retail volumes sold. Wholesale propane revenues were \$88.8 million in Fiscal 2002, a decrease of \$86.8 million, reflecting a \$50.2 million decrease due to lower average selling prices and a \$36.6 million decrease as a result of lower wholesale volumes sold. The lower Fiscal 2002 retail and wholesale selling prices resulted from lower Fiscal 2002 propane product costs. Revenues from other sales and services increased \$20.8 million primarily due to the full-year impact of the Columbia Propane acquisition. Total cost of sales declined \$193.9 million in Fiscal 2002 reflecting lower average propane product costs and the lower wholesale sales partially offset by the higher retail gallons sold.

Total margin increased \$83.5 million reflecting the full-year volume impact of the Columbia Propane acquisition and a \$25.5 million increase in total margin from PPX(R) reflecting higher volumes and unit margins. PPX(R) propane unit margins in Fiscal 2002 were higher than in Fiscal 2001 reflecting increases in sales prices to fund OPD valve replacement capital expenditures on out-of-compliance grill cylinders.

EBITDA increased \$1.8 million (excluding the \$12.5 million cumulative effect of the Partnership's changes in accounting for tank fee revenue and tank installation costs and the adoption of SFAS 133 in Fiscal 2001) as the significant increase in total margin was substantially offset by a \$78.9 million increase in Partnership operating and administrative expenses and a decrease in other income. EBITDA of PPX(R) increased approximately \$21 million in Fiscal 2002 partially offsetting the effects of the significantly warmer winter weather on our heating-related volumes. The greater operating and administrative expenses in Fiscal 2002 resulted primarily from the full-year impact of the Columbia Propane acquisition and higher volume-driven PPX(R) expenses. During Fiscal 2002, the Partnership completed its planned blending of 90 Columbia Propane distribution locations with existing AmeriGas Propane locations. As a result of these district consolidations and other cost reduction activities, management believes that by September 30, 2002 it achieved its anticipated \$24 million reduction in annualized operating cost savings subsequent to the acquisition of Columbia Propane. Operating income increased \$11.5 million principally due to the cessation of goodwill amortization in Fiscal 2002 as a result of the adoption of SFAS 142 partially offset by higher depreciation and intangible asset amortization associated with Columbia Propane and higher PPX(R) depreciation. Fiscal 2001 operating income includes \$23.8 million of goodwill amortization.

FINANCIAL CONDITION AND LIQUIDITY

CAPITALIZATION AND LIQUIDITY

The Partnership's debt outstanding at September 30, 2003 totaled \$927.3 million. There were no amounts outstanding under AmeriGas OLP's Credit Agreement at September 30, 2003.

AmeriGas OLP's Credit Agreement expires on October 15, 2006 and consists of (1) a \$100 million Revolving Credit Facility and (2) a \$75 million Acquisition Facility. The Revolving

Credit Facility may be used for working capital and general purposes of AmeriGas OLP. The Acquisition Facility provides AmeriGas OLP with the ability to borrow up to \$75 million to finance the purchase of propane businesses or propane business assets or, to the extent it is not so used, may be used for working capital and general purposes. Issued and outstanding letters of credit under the Revolving Credit Facility, which reduce the amount available for borrowings, totaled \$33.4 million at September 30, 2003. AmeriGas OLP's short-term borrowing needs are seasonal and are typically greatest during the fall and winter heating-season months due to the need to fund higher levels of working capital.

AmeriGas OLP also has a credit agreement with the General Partner to borrow up to \$20 million on an unsecured, subordinated basis, for working capital and general purposes. UGI has agreed to contribute up to \$20 million to the General Partner to fund such borrowings.

AmeriGas Partners periodically issues debt and equity securities and expects to continue to do so. It has effective debt and equity shelf registration statements with the U.S. Securities and Exchange Commission ("SEC") under which it may issue up to an additional (1) \$28 million principal amount of 8.875% Senior Notes due 2011, (2) 1.4 million AmeriGas Partners Common Units and (3) up to \$500 million of debt or equity pursuant to an unallocated shelf registration statement.

AmeriGas OLP must maintain certain financial ratios in order to borrow under its Credit Agreement including a minimum interest coverage ratio and a maximum debt to EBITDA ratio, as defined. AmeriGas OLP's ratios calculated as of September 30, 2003 permit it to borrow up to the maximum amount available. For a more detailed discussion of the Partnership's credit facilities, see Note 6 to Consolidated Financial Statements. Based upon existing cash balances, cash expected to be generated from operations, borrowings available under its Credit Agreement, and the expected refinancing of its maturing long-term debt, the Partnership's management believes that the Partnership will be able to meet its anticipated contractual commitments and projected cash needs in Fiscal 2004.

PARTNERSHIP DISTRIBUTIONS

Since its formation in 1995, the Partnership has paid the Minimum Quarterly Distribution of \$0.55 ("MQD") on all limited partner units outstanding. The amount of Available Cash needed annually to pay the MQD on all units and the general partner interests in Fiscal 2003, 2002 and 2001 was approximately \$112 million, \$109 million and \$99 million, respectively. Based upon the number of Partnership units outstanding on September 30, 2003, the amount of Available Cash needed annually to pay the MQD on all units and the general partner interests is approximately \$117 million. A reasonable proxy for the amount of cash available for distribution that is generated by the Partnership can be calculated by subtracting from the Partnership's EBITDA (1) interest expense and (2) capital expenditures needed to maintain operating capacity. Partnership distributable cash as calculated under this method for Fiscal 2003, 2002 and 2001 is as follows:

Year Ended September 30,	2003	2002	2001	
(Millions of dollars)				
Net income	\$ 72.0	\$ 55.4	\$ 65.5	
Income tax expense (benefit)	0.6 87.2	0.3 87.8	(0.3) 80.4	
Interest expense				
Depreciation Amortization	70.4 4.2	62.0 4.1	48.1 26.6	
EBITDA	234.4	209.6	220.3	
Interest expense	(87.2)	(87.8)	(80.4)	
Maintenance capital expenditures	(22.0)	(20.7)	(17.8)	
Distributable cash	\$ 125.2	\$ 101.1	\$ 122.1	

Although distributable cash is a reasonable estimate of the amount of cash generated by the Partnership, it does not reflect, among other things, the impact of changes in working capital, which can significantly affect cash available for distribution. Distributable cash should not be considered as an alternative to net income (as an indicator of operating performance) or as an alternative to cash flow (as a measure of liquidity or ability to service debt obligations) and is not a measure of performance or financial condition under accounting principles generally accepted in the United States of America. Management believes distributable cash is a meaningful non-GAAP measure for evaluating the Partnership's ability to declare and pay the MQD pursuant to the terms of the Partnership Agreement. The Partnership's definition of distributable cash may be different from that used by other companies. Although the level of distributable cash in Fiscal 2002 was less than the full MQD, other sources of cash, including cash from equity offerings and borrowings and changes in working capital, was more than sufficient to permit the Partnership to pay the full MQD. The ability of the Partnership to pay the MQD on all units depends upon a number of factors. These factors include (1) the level of Partnership earnings; (2) the cash needs of the Partnership's operations (including cash needed for maintaining and increasing operating capacity); (3) changes in operating working capital; and (4) the Partnership's ability to borrow under its bank credit agreement, to refinance maturing debt and to increase its long-term debt. Some of these factors are affected by conditions beyond our control including weather, competition in markets we serve, the cost of propane and changes in capital market conditions.

CONVERSION OF SUBORDINATED UNITS

In December 2002, the General Partner determined that the cash-based performance and distribution requirements for the conversion of the remaining 9,891,072 Subordinated Units, all of which were held by the General Partner, had been met in respect of the quarter ended September 30, 2002. As a result, these Subordinated Units were converted to a like number of Common Units effective November 18, 2002. The conversion of the Subordinated Units did not result in an increase in the total number of AmeriGas Partners limited partner units outstanding.

CONTRACTUAL CASH OBLIGATIONS AND COMMITMENTS

The Partnership has certain contractual cash obligations that extend beyond Fiscal 2003 including obligations associated with long-term debt, lease obligations and propane supply contracts. The

following table presents significant contractual cash obligations as of September 30, 2003 (in millions).

Payments Due by Period

	Total	Less than 1 year	2 - 3 years	4 - 5 years	After 5 years
Long-term debt Operating leases Propane supply contracts	\$ 911.3 177.5 16.7	\$ 55.6 37.2 16.7	\$ 169.9 59.4	\$ 108.2 41.3	\$ 577.6 39.6
Total	\$ 1,105.5	\$ 109.5	\$ 229.3	\$ 149.5	\$ 617.2

CASH FLOWS

OPERATING ACTIVITIES. Due to the seasonal nature of the Partnership's business, cash flows from operating activities are generally strongest during the second and third fiscal quarters when customers pay for propane consumed during the heating season months. Conversely, operating cash flows are generally at their lowest levels during the first and fourth fiscal quarters when the Partnership's investment in working capital, principally accounts receivable and inventories, is generally greatest. The Partnership uses its Revolving Credit Facility to satisfy its seasonal operating cash flow needs. Cash flow from operating activities was \$139.3 million in Fiscal 2001. Cash flow from operating activities before changes in operating working capital was \$153.3 million in Fiscal 2003, \$122.8 million in Fiscal 2002 and \$130.7 million in Fiscal 2001. Changes in operating working capital used \$14.0 million of operating cash flow in Fiscal 2003, and provided \$36.7 million and \$22.3 million in Fiscal 2002 and Fiscal 2001, respectively. The increase in Fiscal 2003 working capital cash needs primarily reflects the effects of higher propane product costs on customer accounts receivable and inventories.

INVESTING ACTIVITIES. Cash flow used in investing activities was \$72.5 million in Fiscal 2003, \$44.4 million in Fiscal 2002, and \$238.1 million in Fiscal 2001 (which included the acquisition of Columbia Propane). We spent \$52.9 million for property, plant and equipment (comprising maintenance capital expenditures of \$22.0 million and growth capital expenditures of \$30.9 million) in Fiscal 2003 compared to expenditures of \$53.5 million (comprising \$20.7 million of maintenance capital expenditures and \$32.8 million of growth capital expenditures) in Fiscal 2002. During Fiscal 2003, the Partnership acquired several propane distribution businesses for total cash consideration of \$27.0 million.

FINANCING ACTIVITIES. Cash flow used by financing activities was \$68.3 million in Fiscal 2003 and \$100.3 million in Fiscal 2002. Cash provided by financing activities was \$106.8 million in Fiscal 2001 which includes debt and equity issued in conjunction with the acquisition of Columbia Propane. Financing activity cash flow is primarily the result of repayments and issuances of long-term debt, borrowings under our Credit Agreement, distributions on limited partner units and proceeds from issuances of Common Units.

In June 2003, AmeriGas Partners sold 2,900,000 Common Units in an underwritten public offering at a public offering price of \$27.12 per unit. The net proceeds of the public offering totaling \$75.0 million, and associated capital contributions from the General Partner totaling \$1.5 million, were contributed to AmeriGas OLP and used to reduce indebtedness under its bank

credit agreement and for general partnership purposes. The underwriters' overallotment option expired unexercised.

The Partnership also completed a number of debt transactions during Fiscal 2003. In December 2002, AmeriGas Partners issued \$88 million face amount of 8.875% Senior Notes due 2011 at an effective interest rate of 8.30%. The net proceeds of \$89.1 million were used in January 2003 to redeem prior to maturity AmeriGas Partners' \$85 million face amount of 10.125% Senior Notes due April 2007 at a redemption price of 102.25%, plus accrued interest. The Partnership recognized a loss of \$3.0 million relating to the redemption premium and other associated costs and expenses. In April 2003, AmeriGas OLP repaid \$53.8 million of maturing First Mortgage Notes. In conjunction with this repayment, in April 2003 AmeriGas Partners issued \$32 million face amount of 8.875% Senior Notes due 2011 at an effective interest rate of 7.72% and contributed the net proceeds of \$33.7 million, including debt premium, to AmeriGas OLP.

RELATED PARTY TRANSACTIONS

Pursuant to the Partnership Agreement and a Management Services Agreement among AmeriGas Eagle Holdings, Inc. ("AEH"), the general partner of AmeriGas Eagle Propane, L.P. ("Eagle OLP"), and AmeriGas Propane, Inc. (the "General Partner"), the General Partner is entitled to reimbursement for all direct and indirect expenses incurred or payments it makes on behalf of the Partnership. These costs, which totaled \$284.3 million in 2003, \$262.4 million in 2002 and \$208.9 million in 2001, include employee compensation and benefit expenses of employees of the General Partner and general and administrative expenses.

UGI Corporation ("UGI") provides certain financial and administrative services to the General Partner. UGI bills the General Partner for these direct and indirect corporate expenses and the General Partner is reimbursed by the Partnership for these expenses. Such corporate expenses totaled \$8.3 million in 2003, \$6.3 million in 2002, and \$5.3 million in 2001. In addition, UGI and certain of its subsidiaries provide office space and automobile liability insurance to the Partnership. These expenses totaled \$1.7 million in 2003, \$1.5 million in 2002 and \$1.3 million in 2001.

Subsequent to the Columbia Propane acquisition, the Partnership purchases propane on behalf of Atlantic Energy, Inc. ("Atlantic Energy"). Atlantic Energy reimburses AmeriGas OLP for its purchases plus interest as Atlantic Energy sells such propane to third parties or to AmeriGas OLP itself. The total dollar value of propane purchased on behalf of Atlantic Energy was \$17.2 million and \$11.4 million in 2003 and 2002, respectively. Purchases of propane by AmeriGas OLP from Atlantic Energy during 2003 and 2002 totaled \$23.9 million and \$12.1 million, respectively.

The General Partner also provides other services to Atlantic Energy including accounting, insurance and other administrative services and is reimbursed for the related costs. Such costs were not material during 2003 or 2002. In addition, AmeriGas OLP enters into product cost hedging contracts on behalf of Atlantic Energy. When these contracts are settled, AmeriGas OLP is reimbursed the cost of any losses, or distributes the proceeds of any gains, to Atlantic Energy.

Amounts due from Atlantic Energy at September 30, 2003 and 2002 totaled \$2.0 million and \$5.2 million, respectively, which amounts are included in accounts receivable - related parties in the Consolidated Balance Sheets.

OFF-BALANCE SHEET ARRANGEMENTS

We lease various buildings and other facilities, transportation, computer and office equipment. We account for these arrangements as operating leases. These off-balance sheet arrangements enable us to lease facilities and equipment from third parties rather than, among other options, purchasing the equipment and facilities using on-balance sheet financing. For a summary of scheduled future payments under these lease arrangements, see "Contractual Cash Obligations and Commitments."

MARKET RISK DISCLOSURES

Our primary financial market risks include commodity prices for propane and interest rates on borrowings.

The risk associated with fluctuations in the prices the Partnership pays for propane are principally a result of market forces reflecting changes in supply and demand for propane and other energy commodities. The Partnership's profitability is sensitive to changes in propane supply costs, and the Partnership generally attempts to pass on increases in such costs to customers. The Partnership may not, however, always be able to pass through product cost increases fully, particularly when product costs rise rapidly. In order to reduce the volatility of the Partnership's propane market price risk, we use contracts for the forward purchase or sale of propane, propane fixed-price supply agreements, and over-the-counter derivative commodity instruments including price swap and option contracts. Over-the-counter derivative commodity instruments utilized by the Partnership to hedge forecasted purchases of propane are generally settled at expiration of the contract. In order to minimize credit risk associated with derivative commodity contracts, we carefully monitor established credit limits with the contract counterparties. Although we use derivative financial and commodity instruments to reduce market price risk associated with forecasted transactions, we do not use derivative financial and commodity instruments for speculative or trading purposes.

The Partnership has both fixed-rate and variable-rate debt. Changes in interest rates impact the cash flows of variable-rate debt but generally do not impact its fair value. Conversely, changes in interest rates impact the fair value of fixed-rate debt but do not impact their cash flows.

Our variable rate debt includes borrowings under AmeriGas OLP's Credit Agreement. This agreement has interest rates that are generally indexed to short-term market interest rates. At September 30, 2003, there were no borrowings outstanding under this agreement. At September 30, 2002, borrowings outstanding under the prior bank credit agreement totaled \$10 million. Based upon weighted average borrowings outstanding under these agreements during Fiscal 2003 and Fiscal 2002, an increase in short-term interest rates of 100 basis points (1%) would have increased our interest expense by \$0.2 million and \$0.1 million, respectively.

The remainder of our debt outstanding is subject to fixed rates of interest. A 100 basis point increase in market interest rates would result in decreases in the fair value of this fixed rate debt of \$42.6 million and \$41.2 million at September 30, 2003 and 2002, respectively. A 100 basis point decrease in market interest rates would result in increases in the fair market value of this debt of \$45.5 million and \$44.1 million at September 30, 2003 and 2002, respectively.

Our long-term debt is typically issued at fixed rates of interest based upon market rates for debt having similar terms and credit ratings. As these long-term debt issues mature, we may refinance such debt with new debt having interest rates reflecting then-current market conditions. This debt may have an interest rate that is more or less than the refinanced debt. In order to reduce interest rate risk associated with near-term forecasted issuances of fixed-rate debt, from time to time we enter into interest rate protection agreements.

The following table summarizes the fair values of unsettled market risk sensitive derivative instruments held at September 30, 2003 and 2002. It also includes the changes in fair value that would result if there were an adverse change in (1) the market price of propane of 10 cents per gallon and (2) interest rates on ten-year U.S. treasury notes of 50 basis points:

	Fair Value	Change in Fair Value
	(Millions of dollars)	
September 30, 2003 Propane commodity price risk Interest rate risk	\$ (0.6) (0.2)	\$ (24.3) (2.1)
September 30, 2002 Propane commodity price risk Interest rate risk	\$ 9.8 (2.8)	\$ (10.9) (4.4)

Because the Partnership's derivative instruments generally qualify as hedges under SFAS 133, we expect that changes in the fair value of derivative instruments used to manage propane price or interest rate risk would be substantially offset by gains or losses on the associated underlying transactions.

CRITICAL ACCOUNTING POLICIES AND ESTIMATES

The preparation of financial statements and related disclosures in compliance with generally accepted accounting principles requires the selection and application of appropriate accounting principles to the relevant facts and circumstances of the Partnership's operations and the use of estimates made by management. The Partnership has identified the following critical accounting policies that are most important to the portrayal of the Partnership's financial condition and results of operations. Changes in these policies could have a material effect on the financial statements. The application of these accounting policies necessarily requires management's most subjective or complex judgments regarding estimates and projected outcomes of future events which could have a material impact on the financial statements. Management has reviewed these critical accounting policies, and the estimates and assumptions associated with them, with its Audit Committee. In addition, management has reviewed the following disclosures regarding the application of these critical accounting policies with the Audit Committee.

LITIGATION ACCRUALS. The Partnership is involved in litigation regarding pending claims and legal actions that arise in the normal course of its business. In accordance with accounting principles generally accepted in the United States of America, the Partnership establishes reserves for

pending claims and legal actions when it is probable that a liability exists and the amount or range of amounts can be reasonably estimated. Reasonable estimates involve management judgments based on a broad range of information and prior experience. These judgments are reviewed quarterly as more information is received and the amounts reserved are updated as necessary. Such estimated reserves may differ materially from the actual liability, and such reserves may change materially as more information becomes available and estimated reserves are adjusted.

DEPRECIATION AND AMORTIZATION OF LONG-LIVED ASSETS. We compute depreciation on property, plant and equipment on a straight-line basis over estimated useful lives generally ranging from 2 to 40 years. We also use amortization methods and determine asset values of intangible assets other than goodwill using reasonable assumptions and projections. Changes in the estimated useful lives of property, plant and equipment and changes in intangible asset amortization methods or values could have a material effect on our results of operations.

RECENTLY ISSUED ACCOUNTING PRONOUNCEMENTS

In December 2002, the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation - Transition and Disclosure" ("SFAS 148"). SFAS 148 provides alternative methods of transition for an entity that voluntarily changes to a fair value based method of accounting for stock-based employee compensation. In addition, SFAS 148 amends SFAS No. 123, "Accounting for Stock-Based Compensation" ("SFAS 123") to require more prominent disclosure about the effects on reported net income of stock-based employee compensation. As permitted by SFAS 148 and SFAS 123, the Partnership expects to continue to account for stock-based compensation in accordance with Accounting Principles Board Opinion No. 25, "Accounting for Stock Issued to Employees," and will continue to provide the required disclosures in its annual and interim financial statements

In April 2003, the FASB issued SFAS No. 149, "Amendment of Statement 133 on Derivative Instruments and Hedging Activities" ("SFAS 149"). SFAS 149 is effective for contracts entered into or modified after June 30, 2003 and for hedging relationships designated after June 30, 2003. SFAS 149 (i) clarifies under what circumstances a contract with an initial net investment meets the characteristic of a derivative, (ii) clarifies when a derivative contains a financing component, (iii) amends the definition of an underlying- rate, price or index to conform it to language used in FASB Interpretation No. 45, "Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness of Others," and (iv) amends certain other existing pronouncements. SFAS 149 did not change the methods the Partnership uses to account for and report its derivatives and hedging activities.

In May 2003, the FASB issued SFAS No. 150, "Accounting for Certain Financial Instruments with Characteristics of Both Liabilities and Equity" ("SFAS 150"). SFAS 150 is effective at the beginning of the first interim period beginning after June 15, 2003. SFAS 150 establishes guidelines on how an issuer classifies and measures certain financial instruments with characteristics of both liabilities and equity. SFAS 150 further defines and requires that certain instruments within its scope be classified as liabilities on the financial statements. The adoption of SFAS 150 did not affect the Partnership's financial position or results of operations.

In January 2003, the FASB issued Financial Interpretation No. 46, "Consolidation of Variable Interest Entities" ("FIN 46"), which clarifies Accounting Research Bulletin No. 51, "Consolidated Financial Statements." FIN 46 is effective immediately for variable interest entities created or obtained after January 31, 2003. For variable interests created or acquired before February 1, 2003, FIN 46 is effective for the first fiscal or interim period beginning after December 15, 2003. If certain conditions are met, FIN 46 requires the primary beneficiary to consolidate certain variable interest entities in which the other equity investors lack the essential characteristics of a controlling financial interest or their investment at risk is not sufficient to permit the variable interest entity to finance its activities without additional subordinated financial support from other parties. The Partnership has not created or obtained any variable interest entities after January 31, 2003, and is currently in the process of evaluating the impact of FIN 46, which is not expected to have a material effect on its financial position or results of operations.

FORWARD-LOOKING STATEMENTS

Information contained above in this Management's Discussion and Analysis of Financial Condition and Results of Operations and elsewhere in this Report on Form 10-K may contain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Such statements use forward-looking words such as "believe," "plan," "anticipate," "continue," "estimate," "expect," "may," "will," or other similar words. These statements discuss plans, strategies, events or developments that we expect or anticipate will or may occur in the future.

A forward-looking statement may include a statement of the assumptions or bases underlying the forward-looking statement. We believe that we have chosen these assumptions or bases in good faith and that they are reasonable. However, we caution you that actual results almost always vary from assumed facts or bases, and the differences between actual results and assumed facts or bases can be material, depending on the circumstances. When considering forward-looking statements, you should keep in mind the following important factors which could affect our future results and could cause those results to differ materially from those expressed in our forward-looking statements: (1) adverse weather conditions resulting in reduced demand; (2) price volatility and availability of propane, and the capacity to transport it to our market areas; (3) changes in laws and regulations, including safety, tax and accounting matters; (4) large supplier, counterparty or customer defaults; (5) competitive pressures from the same and alternative energy sources; (6) failure to acquire new customers thereby reducing or limiting any increase in revenues; (7) liability for environmental claims; (8) customer conservation measures and improvements in energy efficiency and technology resulting in reduced demand; (9) adverse labor relations; (10) inability to make business acquisitions on economically acceptable terms resulting in failure to acquire new customers thereby limiting any increase in revenues; (11) liability for personal injury and property damage arising from explosions and other catastrophic events, including acts of terrorism, resulting from operating hazards and risks incidental to transporting, storing and distributing propane, butane and ammonia, including liability in excess of insurance coverage; (12) political, regulatory and economic conditions in the United States and in foreign countries; and (13) interest rate fluctuations and other capital market conditions.

These factors are not necessarily all of the important factors that could cause actual results to differ materially from those expressed in any of our forward-looking statements. Other unknown or unpredictable factors could also have material adverse effects on future results. We undertake no obligation to update publicly any forward-looking statement whether as a result of new information or future events except as required by federal securities laws.

ITEM 7A. OUANTITATIVE AND OUALITATIVE DISCLOSURES ABOUT MARKET RISK

"Quantitative and Qualitative Disclosures About Market Risk" are contained in Management's Discussion and Analysis of Financial Condition and Results of Operations under the caption "Market Risk Disclosures" and are incorporated here by reference.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

The financial statements and financial statement schedules referred to in the index contained on pages F-2 and F-3 of this report are incorporated herein by reference.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

During fiscal year 2002, the registrants engaged a new independent auditor, PricewaterhouseCoopers LLP. The information required by Item 9 is incorporated in this Report by reference to the registrants' Current Report on Form 8-K dated May 21, 2002, which report was amended on July 2, 2002.

ITEM 9A. CONTROLS AND PROCEDURES

(a) Evaluation of Disclosure Controls and Procedures

The General Partner's management, with the participation of the Chief Executive Officer and Chief Financial Officer, evaluated the effectiveness of the Partnership's disclosure controls and procedures as of the end of the period covered by this report. Based on that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the Partnership's disclosure controls and procedures as of the end of the period covered by this report were designed and functioning effectively to provide reasonable assurance that the information required to be disclosed by the Partnership in reports filed under the Securities Exchange Act of 1934, as amended, is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms. The General Partner believes that a controls system, no matter how well designed and operated, cannot provide absolute assurance that the objectives of the controls system are met, and no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within a company have been detected.

(b) Change in Internal Control over Financial Reporting

No change in the Partnership's internal control over financial reporting occurred during the Partnership's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Partnership's internal control over financial reporting.

PART III: MANAGEMENT AND SECURITY HOLDERS

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS OF THE GENERAL PARTNER

We do not directly employ any persons responsible for managing or operating the Partnership. The General Partner and UGI provide such services and are reimbursed for direct and indirect costs and expenses including all compensation and benefit costs. See "Certain Relationships and Related Transactions" and Note 12 to the Partnership's Consolidated Financial Statements.

The Board of Directors of the General Partner established a committee (the "Audit Committee") consisting of three individuals, currently, Messrs. Marrazzo, Van Dyck and Vincent, who are neither officers nor employees of the General Partner or any affiliate of the General Partner. The Board of Directors of the General Partner has determined that all members of the audit committee qualify as "audit committee financial experts" within the meaning of the Securities and Exchange Commission regulations. The Board of Directors considered Mr. Vincent's professional experience to be "other relevant experience" within the meaning of the applicable regulations. See "Directors and Executive Officers of the General Partner" below, for a description of Mr. Vincent's professional experience. Each member of the Audit Committee is "independent" as defined by the New York Stock Exchange listing standards currently in effect and those approved by the SEC on November 4, 2003. The Audit Committee has the authority to (i) make determinations or review determinations made by management in transactions that require special approval by the Committee under the terms of the Partnership Agreement and (ii) at the request of the General Partner, review specific matters as to which the General Partner believes there may be a conflict of interest, in order to determine if the resolution of such conflict is fair and reasonable to the Partnership. In addition, the Audit Committee acts on behalf of the Board of Directors in fulfilling its responsibility to:

- oversee the financial reporting process and the adequacy of controls relative to financial and business risk;
- monitor the independence of the Partnership's independent accountants and the performance of the independent accountants and internal audit staff; and
- provide a means for open communication among the independent accountants, management, internal audit staff and the Board of Directors.

The Audit Committee has sole authority to appoint, retain, fix the compensation of and oversee the work of the independent auditors. A copy of the current charter of the Audit Committee is posted on the Partnership's website, www.amerigas.com; see "Investor Relations - Corporate Governance."

The General Partner has adopted a Code of Ethics for the Chief Executive Officer and Senior Financial Officers that applies to the General Partner's Chief Executive Officer, Chief Financial Officer and Chief Accounting Officer. The Code of Ethics is included as an exhibit to this Report and is posted on the Partnership's website, www.amerigas.com; see "Investor Relations - Corporate Governance."

DIRECTORS AND EXECUTIVE OFFICERS OF THE GENERAL PARTNER

The following table sets forth certain information with respect to the directors and executive officers of the General Partner. Directors are elected annually by AmeriGas, Inc. as the sole shareholder of the General Partner. AmeriGas, Inc. is a wholly owned subsidiary of UGI. Executive officers are elected for one-year terms. There are no family relationships between any of the directors or any of the executive officers or between any of the executive officers and any of the directors.

NAME	AGE	POSITION WITH THE GENERAL PARTNER	
Lon R. Greenberg	53	Chairman, Director	
Eugene V. N. Bissell	50	President, Chief Executive Officer and Director	
Thomas F. Donovan	70	Director	
Richard C. Gozon	65	Director	
William J. Marrazzo	54	Director	
James W. Stratton	67	Director	
Stephen A. Van Dyck	60	Director	
Roger B. Vincent	58	Director	
Richard R. Eynon	56	Controller and Chief Accounting Officer	
William D. Katz	50	Vice President - Human Resources	
Robert H. Knauss	50	Vice President and General Counsel and Corporate Secretary	
Martha B. Lindsay	51	Vice President - Finance and Chief Financial Officer	
David L. Lugar	46	Vice President - Supply and Logistics	
Carey M. Monaghan	52	Vice President - Sales and Marketing	

Mr. Greenberg is a director (since 1994) and Chairman of the General Partner. He previously served as President and Chief Executive Officer of the General Partner from 1996 until July 2000. He is also a director (since 1994) and Chairman (since 1996), Chief Executive Officer (since 1995), and President (since 1994) of UGI Corporation, having previously been Senior Vice President Legal and Corporate Development of UGI (1989 to 1994). Mr.

Greenberg previously served as Vice President and General Counsel of AmeriGas, Inc. (1984 to 1994). He also serves as a director of UGI Utilities, Inc.

Mr. Bissell is President, Chief Executive Officer and a director of the General Partner (since July 2000). He previously served as Senior Vice President - - Sales and Marketing of the General Partner (October 1999 to July 2000), having served as Vice President - Sales and Operations (1995 to 1999). Previously, he was Vice President - Distributors and Fabrication, BOC Gases (1995), having been Vice President - National Sales (1993 to 1995) and Regional Vice President (Southern Region) for Distributor and Cylinder Gases Division, BOC Gases (1989 to 1993). From 1981 to 1987, Mr. Bissell held various positions with UGI Corporation and its subsidiaries, including Director, Corporate Development. He is immediate past president and a member of the Board of Directors of the National Propane Gas Association.

Mr. Donovan was elected a director of the General Partner on April 25, 1995. He retired as Vice Chairman of Mellon Bank on January 31, 1997, a position he had held since 1988. He continues to serve as a director of UGI Corporation, UGI Utilities, Inc. and Nuclear Electric Insurance Ltd.

Mr. Gozon was elected a director of the General Partner on February 24, 1998. He retired as Executive Vice President of Weyerhaeuser Company in 2002 (an integrated forest products company) and Chairman of Norpac (North Pacific Paper Company, a joint venture with Nippon Paper Industries), positions he had held since 1994. Mr. Gozon was formerly a director (1984 to 1993), President and Chief Operating Officer of Alco Standard Corporation (a provider of paper and office products) (1988 to 1993); Executive Vice President and Chief Operating Officer (1987); Vice President (1982 to 1988); and President (1979 to 1987) of Paper Corporation of America. He also serves as a director of UGI Corporation, UGI Utilities, Inc., AmeriSource Bergen Corp., and Triumph Group, Inc.

Mr. Marrazzo was elected a director of the General Partner on April 23, 2001. He is Chief Executive Officer and President of WHYY, Inc., a public television and radio company in the nation's fourth largest market (since 1997). Previously, he was Chief Executive Officer and President of Roy F. Weston, Inc. (1988-1997); Water Commissioner for the Philadelphia Water Department (1971-1988) and Managing Director for the City of Philadelphia (1983-1984). He also serves as a director of Tenet Health Corp. - Hahnemann University Hospital and American Water Corporation.

Mr. Stratton was elected a director of the General Partner on April 25, 1995. He has been the Chairman, Chief Executive Officer and a director of Stratton Management Company (investment advisory and financial consulting firm) since 1972. In addition, Mr. Stratton is a director of UGI Corporation, UGI Utilities, Inc., Stratton Growth Fund, Inc., Stratton Monthly Dividend REIT Shares, Inc., Stratton Small-Cap Value Fund, Teleflex, Inc. and BE&K, Inc.

Mr. Van Dyck was elected a director of the General Partner on June 15, 1995. He is Chairman of the Board of Maritrans Inc. (since 1987) having served as Chief Executive Officer (1987-2003). Maritrans is one of the nation's largest independent marine transporters of petroleum. He also serves as a director of the Board of West of England Mutual Insurance Association, the American Petroleum Institute, the Chamber of Shipping of America and Seamans' Church Institute.

Mr. Vincent was elected a director of the General Partner on January 8, 1998. He is President of Springwell Corporation, a corporate finance advisory firm located in New York (since 1989). Mr. Vincent served in various capacities at Bankers Trust Company (1971 to 1989), including positions with direct and oversight responsibilities for credit management. Mr. Vincent is also a director of the ING Funds.

Mr. Eynon is Controller and Chief Accounting Officer of the General Partner (since 1998). Prior to his election, Mr. Eynon was Controller of the General Partner (March 1997 to January 1998) and Assistant Controller of UGI Corporation (1985 to 1997). Previously, he was a Senior Manager with Price Waterhouse.

Mr. Katz is Vice President - Human Resources of the General Partner (since December 1999), having served as Vice President - Corporate Development (1996 to 1999). Previously, he was Vice President - Corporate Development of UGI Corporation (1995 to 1996). Prior to joining UGI Corporation, Mr. Katz was Director of Corporate Development with Campbell Soup Company for over five years. He also practiced law for approximately 10 years, first with the firm of Jones, Day, Reavis & Pogue, and later in the Legal Department at Campbell Soup Company.

Mr. Knauss is Vice President and General Counsel of the General Partner (since October 2003) and UGI Corporation (since September 2003). He is also Corporate Secretary of the General Partner (since 1994). Prior to October 2003, Mr. Knauss served as Vice President - Law and Associate General Counsel of the General Partner (1996 -2003). Previously he was Group Counsel - Propane (1989 to 1996) of UGI Corporation. He joined UGI Corporation as Associate Counsel in 1985. Before joining UGI Corporation, Mr. Knauss was an associate at the firm of Ballard, Spahr, Andrews & Ingersoll in Philadelphia, Pennsylvania.

Ms. Lindsay is Vice President - Finance and Chief Financial Officer of the General Partner (since 1998). She previously served as Vice President and Treasurer (1994 to 1997) and as Treasurer (1994) of Tambrands Inc., a manufacturer of personal products. Prior to 1994, Ms. Lindsay held the positions of Director of Business Development (1987 to 1989) and Assistant Treasurer (1990 to 1993) at Tambrands Inc.

Mr. Lugar is Vice President - Supply and Logistics of the General Partner (since September 2000). Previously, he served as Director - NGL Marketing for Conoco, Inc., where he spent 20 years in increasingly responsible positions in propane marketing, operations, and supply.

Mr. Monaghan is Vice President - Sales and Marketing of the General Partner (since May 2000). Prior to joining AmeriGas Partners, he was Vice President-General Manager, Dry Soup for Campbell Soup Company (since 1997), where he also served as a Business Director and General Manager of a number of Campbell Soup Divisions for the 10 prior years.

SECTION 16(a) - BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Securities Exchange Act of 1934 requires the directors and certain officers of the General Partner and any 10% beneficial owners of the Partnership to send reports of their beneficial ownership of Common Units and changes in beneficial ownership to the Securities and Exchange Commission. Based on our records, we believe that during Fiscal 2003

ITEM 11. EXECUTIVE COMPENSATION

The following table shows cash and other compensation paid or accrued to the General Partner's Chief Executive Officer and each of its six other most highly compensated executive officers, (collectively, the "Named Executives") for the last three fiscal years.

			ANNUAL COMPENSATION		LONG TERM COMPENSATION										
											AWAR	DS	PAY0I	JTS	
NAME AND PRINCIPAL POSITION	FISCAL YEAR	SALARY	В	ONUS	(1)	ANN COM	HER UAL PEN- ION	(2)	UN:	STRI(IT/ST ARDS	госк	SECURITIES UNDERLYING OPTIONS / SARS	LT:	IP OUTS	ALL OTHER COMPENSATION (4)
Eugene V. N. Bissell, President and Chief Executive Officer	2003	\$372,080	\$	245,	281	\$ 2	, 520		\$ \$ \$	238, 238, 238,	500	52,500	\$	0	\$ 60,277
OTTICET	2002	\$352,656	\$	109,	941	\$	585		\$	190,	145	35,000	\$	0	\$ 42,717
	2001	\$329,415	\$	236,	313	\$	300		\$		263 725	0	\$	0	\$ 55,648
Lon R. Greenberg,	2003	\$757,008	\$1	,075,	981	\$12	, 824		\$	972,	140	180,000	\$	0	\$ 28,757
Chairman (5)	2002	\$705,015	\$	521,	092	\$15	, 342		\$ \$ \$	785, 785, 785,	200	120,000	\$	0	\$ 28,033
	2001	\$667,799	\$	595,	010	\$14	, 849			323, ,000,		0	\$	0	\$ 20,939
R. Paul Grady, Senior Vice President - Operations and Chief	2003	\$256,133	\$	112,	529	\$ 4	, 283		\$ \$ \$	131, 131, 131,	175	27,000	\$	0	\$ 37,857
Operating Officer	2002	\$246,088	\$	54,	825	\$ 4	,041		\$	104,	580	18,000	\$	0	\$ 25,544
	2001	\$223,850	\$	120,	313	\$ 3	, 845		\$		196 315	0	\$	0	\$ 34,426
Brendan P. Bovaird,	2003	\$241,949	\$	181,	986	\$ 7	, 313		\$	112,	170	21,750	\$	0	\$284,678
Vice President and General Counsel (5)	2002	\$232,683	\$	95,	459	\$ 5	, 449		\$ \$ \$	90,	600 600 600	14,500	\$	0	\$ 7,411
	2001	\$222, 283	\$	96,	708	\$ 5	,012		\$	38, 120,	813 105	0	\$	0	\$ 6,112
Robert H. Knauss, Vice President and General Counsel	2003	\$202,143	\$	71,	032	\$ 1	, 838		\$ \$	47,	700 700 700	12,000	\$	0	\$ 27,346
	2002	\$196,172	\$	34,	935	\$ 1	, 612		\$	38,	029	8,000	\$	0	\$ 19,117
	2001	\$186,819	\$	78,	115	\$ 1	, 793		\$	14, 123,	131 255	0	\$	0	\$ 26,659
Martha B. Lindsay, Vice President - Finance and Chief	2003	\$196,698	\$	69,	129	\$ 1	, 200		\$ \$ \$	47,	700 700 700	12,000	\$	0	\$ 26,774
Financial Officer	2002	\$190,060	\$	33,	840	\$ 1	, 908		\$	38,	029	8,000	\$	0	\$ 18,577
	2001	\$183,087	\$	87,	460	\$	664		\$ \$		131	0	\$	0	\$ 26,958
William D. Katz, Vice President - Human Resources	2003	\$194,038	\$	68,	211	\$ 1	, 350		\$ \$ \$	47,	700 700 700	12,000	\$	0	\$ 26,288
	2002	\$184,860	\$	32,	915	\$ 1	, 237		\$	38,	029	8,000	\$	0	\$ 18,008

\$177,843 \$ 74,355 \$ 577 \$ 14,131 0 \$ 0 \$ 25,203 \$ 62,748

⁽¹⁾ Messrs. Greenberg and Bovaird participate in the UGI Annual Bonus Plan. All other Named Executives participate in the AmeriGas Propane, Inc. Annual Bonus Plan. Awards under both Plans are for the year reported, regardless of the year paid. Awards under both Plans are based on the achievement of business and/or financial performance objectives which support business plans and goals. Bonus opportunities vary by position and for the Fiscal Year 2003 ranged from 0 to 225% of base salary for Mr. Bissell, 0 to 196% of base salary for Mr. Greenberg, 0 to 150% of base salary for Mr. Grady, 0 to 104% of base salary for Mr. Bovaird, and 0 to 120% of base salary for Mr. Knauss, Mr. Katz and Ms. Lindsay.

- (2) Amounts represent tax payment reimbursements for certain benefits and, for Mr. Bovaird, above-market interest on deferred compensation during Fiscal 2002 and 2003.
- Effective January 1, 2003, the Board of Directors of AmeriGas Propane, (3) Inc. approved three phantom performance-contingent restricted Common Unit awards ("Restricted Units") to the Named Executives, other than Messrs. Greenberg and Bovaird, under the 2000 AmeriGas Propane, Inc. Long-Term Incentive Plan. Distribution equivalents will accumulate on the Restricted Units awarded. These distribution equivalents may be leveraged based on performance described below. Each award has a separate performance measurement period as follows: January 1, 2003 through December 31, 2003; January 1, 2003 through December 31, 2004; and January 1, 2003 through December 31, 2005. The performance period for all three awards will end on December 31, 2005. If the recipient ceases to be employed by the General Partner before December 31, 2005 other than by reason of retirement, disability or death, all awards of Restricted Units and distribution equivalents will be forfeited. The performance requirement is that the Partnership's total unitholder return ("TR") during the relevant measurement period equals the median TR of a peer group of publicly traded limited partnerships. The actual amount of the award may be higher or lower than the original grant, or even zero, based on the Partnership's TR percentile rank relative to that of the partnerships in the peer group. The maximum payout potential is 200% of the original award. At the discretion of the . General Partner, Restricted Unit awards may be paid out in Common Units, in cash, or in a combination of Units and cash.

Effective January 1, 2002, the Board of Directors of AmeriGas Propane, Inc. approved Restricted Unit awards to the Named Executives, other than Messrs. Greenberg and Bovaird, under the 2000 AmeriGas Propane, Inc. Long-Term Incentive Plan. Distribution equivalents will accumulate on the Restricted Units awarded. The performance requirement was evidence of AmeriGas' meaningful progress toward the achievement of its strategic objectives during 2002 including the Partnership's acquisition integration, productivity improvement, internal growth, and cash generation goals. The Restricted Units reported for fiscal year 2002 are equal to 85% of the original award based on achievement of goals at that level.

Effective January 1, 2003, the Board of Directors of UGI approved phantom performance-contingent restricted stock awards ("Restricted ') to Messrs. Greenberg and Bovaird under the UGI Corporation 2000 Stock Incentive Plan. Dividend equivalents will accumulate on the Restricted Shares. These dividend equivalents will also be leveraged based on UGI's total shareholder return ("TSR") performance as described below and distributed when the performance period on the Restricted Shares ends on December 31, 2005. If the recipient ceases to be employed by the Company before December 31, 2005, other than by reason of retirement, death or disability, awards of Restricted Shares and dividend equivalents will be forfeited. The performance requirement is that UGI's TSR during the performance period equals the median of a peer group. The peer group is the group of companies that comprises the S&P Utilities Index. The share priced used for determining the TSR at the beginning and the end of the performance period is the average for the 90-day period preceding each December 31st. The actual amount of the award may be higher or lower than the original grant, or even zero, based on UGI's TSR percentile rank relative to the companies in the S&P Utilities Index. The maximum payout potential is 200% of the original award. The maximum number of shares to be issued in respect of awards of Restricted Shares will be the target number of shares originally awarded. All leverage on Restricted Share awards will be paid in cash.

Effective January 1, 2002, the Board of Directors of UGI approved three Restricted Share awards to Messrs. Greenberg and Bovaird under the UGI Corporation 2000 Stock Incentive Plan. Dividend equivalents will accumulate on the Restricted Shares awarded. These dividend equivalents will also be leveraged based on UGI's TSR performance and distributed when the performance period on the Restricted Shares ends on December 31, 2004. Each award has a separate performance measurement period as follows: January 1, 2002 through December 31, 2002; January 1, 2002 through December 31, 2003; and January 1, 2002 through December 31, 2004. The performance period for all three awards will end on December 31, 2004. If the recipient ceases to be employed by the Company before December 31, 2004, other than by reason of retirement, disability or death, awards of Restricted Shares and dividend equivalents will be forfeited. The performance requirement is that UGI's TSR during the relevant performance measurement period equals the median of a peer group. The peer group is the group of companies that comprises the S&P Utilities Index. The share price used for determining the TSR at the beginning and the end of each performance measurement period is the average price for the 90-day period preceding each December 31st. The actual amount of the award may be higher or lower than the original grant, or even zero, based on UGI's TSR percentile rank relative to the companies in the S&P Utilities Index. The maximum payout potential is 200% of the original

award. The maximum number of shares to be issued in respect of awards of Restricted Shares will be the target number of shares originally awarded. All leverage on Restricted Share awards will be paid in cash. Awards of dividend equivalents will be paid in cash.

The dollar values shown in the restricted stock awards column of the table above for all years represent the aggregate value of each award on the date of grant, determined by multiplying the number of Restricted Units awarded by the closing price of a Common Unit of AmeriGas Partners, or in the case of Messrs. Greenberg and Bovaird, the number of Restricted Shares awarded by the closing price of UGI Common Stock, on the New York Stock Exchange on the effective dates of the respective grants.

Based on the closing unit price of AmeriGas Partners, L.P. Common Units on the New York Stock Exchange on September 30, 2003, Mr. Bissell's 44,650 Restricted Units had a market value of \$1,120,715; Mr. Grady's 25,900 Restricted Units had a market value of \$650,090; Mr. Knauss' 14,050 Restricted Units had a market value of \$352,655; Ms. Lindsay's 11,550 Restricted Units had a market value of \$289,905; and Mr. Katz's 11,350 Restricted Units had a market value of \$284,885. Based on the closing stock price of UGI Common Stock on the New York Stock Exchange on September 30, 2003, Mr. Greenberg's 231,000 Restricted Shares had a market value of \$6,682,830. Mr. Bovaird's Restricted Shares were forfeited on September 30, 2003.

- (4) The amounts represent contributions by the General Partner or UGI in accordance with the provisions of the AmeriGas Propane, Inc. Employee 401(k) Savings Plan (the "AmeriGas Employee Savings Plan"), the UGI Utilities, Inc. Employee 401(k) Savings Plan (the "UGI Employee Savings Plan"), allocations under the UGI Corporation Senior Executive Retirement Plan (the "UGI Executive Retirement Plan"), and/or allocations under the AmeriGas Propane, Inc. Supplemental Executive Retirement Plan (the "AmeriGas Executive Retirement Plan"). During fiscal years 2003, 2002 and 2001, the following contributions were made to the Named Executives: (i) under the AmeriGas Employee Savings Plan: Mr. Bissell, \$8,541, \$4,957, and \$7,576; Mr. Grady, \$10,991, \$3,953 and \$8,509; Mr. Knauss, \$10,029, \$4,506 and \$8,666; Ms. Lindsay, \$10,192, \$4,687 and \$8,403; Mr. Katz, \$10,064, \$4,730 and \$8,484; (ii) under the UGI Employee Savings Plan: Mr. Greenberg, \$4,500, \$3,825 and \$3,825; and Mr. Bovaird, \$4,500, \$3,825, and \$3,825; (iii) under the UGI Executive Retirement Plan: Mr. Greenberg, \$24,257, \$24,208 and \$17,114; and Mr. Bovaird, \$3,092, \$3,586 and \$2,287; (iv) as a result of his resignation effective September 30, 2003, Mr. Bovaird received \$277,086 under the UGI Severance Pay Plan for Senior Executives; (v) under the AmeriGas Executive Retirement Plan: Mr. Bissell, \$51,736, \$37,760 and \$48,072; Mr. Grady, \$26,866, \$21,591 and \$25,916; Mr. Knauss, \$17,317, \$14,611 and \$17,993; Ms. Lindsay, \$16,582, \$13,890 and \$18,555; and Mr. Katz, \$16,224, \$13,278, and \$16,719.
- (5) Compensation reported for Messrs. Greenberg and Bovaird is attributable to their respective positions of Chairman, President and Chief Executive Officer, and Vice President and General Counsel of UGI Corporation. Mr. Bovaird resigned on September 30, 2003. Compensation for these individuals is also reported in the UGI Proxy Statement for the 2004 Annual Meeting of Shareholders and is not additive. The General Partner does not compensate Mr. Greenberg or Mr. Bovaird.

OPTION EXERCISES IN LAST FISCAL YEAR

UGI STOCK OPTION EXERCISES IN LAST FISCAL YEAR AND FISCAL YEAR-END OPTION VALUES

NUMBER OF SECURITIES UNDERLYING UNEXERCISED OPTIONS AT FISCAL YEAR END (#)

VALUE OF UNEXERCISED IN-THE-MONEY OPTIONS AT FISCAL YEAR END (2)

NAME	SHARES ACQUIRED ON EXERCISE (#)	VALUE REALIZED (1	1) EXERCISABLE	UNEXERCISABLE	EXERCISABLE	UNEXERCISABLE
Eugene V. N. Bissell	43,500	\$ 755,168	23,876	87,500	\$ 241,270	\$ 498,046
Lon R. Greenberg	300,000	\$4,748,335	660,000	300,000	\$9,664,425	\$1,707,600
R. Paul Grady	51,000	\$ 818,112	0	45,000	\$ 0	\$ 256,140
Brendan P. Bovaird	25,000	\$ 445,528	55,251	36,250	\$ 763,891	\$ 206,335
Robert H. Knauss	22,500	\$ 306,728	4,001	20,000	\$ 34,124	\$ 113,836
Martha B. Lindsay	0	\$ 0	26,501	20,000	\$ 375,674	\$ 113,836
William D. Katz	6,000	\$ 81,500	13,001	20,000	\$ 170,744	\$ 113,836

- (1) Value realized is calculated on the difference between the option exercise price and the closing market price of UGI's Common Stock on the date of exercise multiplied by the number of shares to which the exercise relates.
- (2) The closing price of UGI's Common Stock as reported on the New York Stock Exchange Composite tape on September 30, 2003 was \$28.93 and is used in calculating the value of unexercised options.

OPTION GRANTS IN LAST FISCAL YEAR (1)

	IN	IDIVIDUAL GRANTS			GRANT DATE VALUE
NAME	NUMBER OF SECURITIES UNDERLYING OPTIONS GRANTED	% OF TOTAL OPTIONS GRANTED TO EMPLOYEES IN FISCAL YEAR (2)	EXERCISE OR BASE PRICE	EXPIRATION DATE	GRANT DATE PRESENT VALUE (3)
Eugene V. N. Bissell	52,500	8.05%	\$25.13	12/31/2012	\$141,750
Lon R. Greenberg	180,000	27.59%	\$25.13	12/31/2012	\$486,000
R. Paul Grady	27,000	4.14%	\$25.13	12/31/2012	\$ 72,900
Brendan P. Bovaird	21,750	3.33%	\$25.13	12/31/2012	\$ 58,725
Robert H. Knauss	12,000	1.84%	\$25.13	12/31/2012	\$ 32,400
Martha B. Lindsay	12,000	1.84%	\$25.13	12/31/2012	\$ 32,400
William D. Katz	12,000	1.84%	\$25.13	12/31/2012	\$ 32,400

- (1) Option grants reflect 3-for-2 stock split effective April 1, 2003.
- (2) A total of 652,500 options were granted to employees and executive officers of UGI and its subsidiaries, including the General Partner, during fiscal year 2003 under the UGI 1997 Stock Option and Dividend Equivalent Plan, the UGI 2000 Stock Incentive Plan and the UGI 2002 Non-Qualified Stock Option Plan. Under each Plan, the option exercise price is not less than 100% of the fair market value of UGI's Common Stock on the effective date of the grant. These options become exercisable in three equal annual installments beginning on the first anniversary of the grant date. All options are nontransferable and generally exercisable only while the optionee is employed by UGI or an affiliate, with exceptions for exercise following retirement, disability and death. Options are subject to adjustment in the event of recapitalizations, stock splits, mergers, and other similar corporate transactions affecting UGI's Common Stock.
- (3) Based on the Black-Scholes options pricing model. The assumptions used in calculating the grant date present value are as follows:
 - Three years of closing monthly stock price and dividend observations were used to calculate the stock volatility and dividend yield assumptions.

-	Stock volatility	21.56%
-	Stock's dividend yield	6.08%
-	Length of option term	10 years
-	Annualized risk-free interest rate	4.23%
-	Discount of risk of forfeiture	3% per year

All options were granted at fair market value. The actual value, if any, the executive may realize will depend on the excess of the stock price on the date the option is exercised over the exercise price. There is no assurance that the value realized by the executive will be at or near the value estimated by the Black-Scholes model.

EQUITY COMPENSATION PLAN INFORMATION

The following table sets forth information as of the end of the Partnership's 2003 fiscal year with respect to compensation plans under which equity securities of the Partnership are authorized for issuance.

PLAN CATEGORY	NUMBER OF SECURITIES TO BE ISSUED UPON EXERCISE OF OUTSTANDING OPTIONS, WARRANTS AND RIGHTS (a)	WEIGHTED AVERAGE EXERCISE PRICE OF OUTSTANDING OPTIONS, WARRANTS AND RIGHTS (b)	NUMBER OF SECURITIES REMAINING AVAILABLE FOR FUTURE ISSUANCE UNDER EQUITY COMPENSATION PLANS (EXCLUDING SECURITIES REFLECTED IN COLUMN (a)) (c)
Equity compensation plans approved by security holders (1)	171,086	0	328,914
Equity compensation plans not approved by security holders	0	0	0
TOTAL	171,086	0	328,914

(1) These plans include the AmeriGas Propane, Inc. 2000 Long-Term Incentive Plan and the AmeriGas Propane, Inc. Discretionary Long-Term Incentive Plan for Non-Executive Key Employees.

RETIREMENT BENEFITS

The following Pension Plan Benefits Table shows the annual benefits payable upon retirement to Messrs. Greenberg and Bovaird under the Retirement Income Plan for Employees of UGI Utilities, Inc. (the "Retirement Plan") and the UGI Corporation Supplemental Executive Retirement Plan. The amounts shown assume the executive retires in 2003 at age 65, and that the aggregate benefits are not subject to statutory maximums. Messrs. Greenberg and Bovaird had, respectively, 23 years and 8 years of credited service under these Plans at September 30, 2003. Messrs Bissell, Knauss and Grady previously accumulated more than 6, 12 and 5 years of credited service with UGI and its subsidiaries before joining AmeriGas Propane, Inc. Messrs. Bissell and Grady do not currently participate in the Retirement Plan.

PENSTON PLAN BENEFITS TABLE

ANNUAL PLAN BENEFIT FOR YEARS CREDITED SERVICE SHOWN (2)

FINAL 5-YEAR AVERAGE ANNUAL EARNINGS (1)	5 YEARS	10 YEARS	15 YEARS	20 YEARS	25 YEARS	30 YEARS	35 YEARS	40 YEARS
\$ 200,000	\$ 19,000	\$ 38,000	\$ 57,000	\$ 76,000	\$ 95,000	\$ 114,000	\$ 133,000	\$ 136,800(3)
\$ 400,000	\$ 38,000	\$ 76,000	\$ 114,000	\$ 152,000	\$ 190,000	\$ 228,000	\$ 266,000	\$ 273,600(3)
\$ 600,000	\$ 57,000	\$ 114,000	\$ 171,000	\$ 228,000	\$ 285,000	\$ 342,000	\$ 399,000	\$ 410,400(3)
\$ 800,000	\$ 76,000	\$ 152,000	\$ 228,000	\$ 304,000	\$ 380,000	\$ 456,000	\$ 532,000	\$ 547,200(3)
\$1,000,000	\$ 95,000	\$ 190,000	\$ 285,000	\$ 380,000	\$ 475,000	\$ 570,000	\$ 665,000	\$ 684,000(3)
\$1,200,000	\$ 114,000	\$ 228,000	\$ 342,000	\$ 456,000	\$ 570,000	\$ 684,000	\$ 798,000	\$ 820,800(3)
\$1,400,000	\$ 133,000	\$ 266,000	\$ 399,000	\$ 532,000	\$ 665,000	\$ 798,000	\$ 931,000	\$ 957,600(3)
\$1,600,000	\$ 152,000	\$ 304,000	\$ 456,000	\$ 608,000	\$ 760,000	\$ 912,000	\$1,064,000	\$1,094,400(3)
\$1,800,000	\$ 171,000	\$ 342,000	\$ 513,000	\$ 684,000	\$ 855,000	\$1,026,000	\$1,197,000	\$1,231,200(3)
\$2,000,000	\$ 190,000	\$ 380,000	\$ 570,000	\$ 760,000	\$ 950,000	\$1,140,000	\$1,330,000	\$1,368,000(3)

- (1) Consists of (i) base salary, commissions and cash payments under the Annual Bonus Plan, and (ii) deferrals thereof permitted under the Internal Revenue Code.
- (2) Annual benefits are computed on the basis of straight life annuity amounts. These amounts include pension benefits, if any, to which a participant may be entitled as a result of participation in a pension plan of a subsidiary during previous periods of employment. The amounts shown do not take into account exclusion of up to 35% of the estimated primary Social Security benefit. The Retirement Plan provides a minimum benefit equal to 25% of a participant's final 12-months' earnings, reduced proportionately for less than 15 years of credited service at retirement. The minimum Retirement Plan benefit is not subject to Social Security offset.
- (3) The maximum benefit under the Retirement Plan and the UGI Supplemental Executive Retirement Plan is equal to 60% of a participant's highest consecutive 12 months' earnings during the last 120 months.

SEVERANCE PAY PLAN FOR SENIOR EXECUTIVE EMPLOYEES

Named Executives Employed by UGI Corporation. The UGI Corporation Senior Executive Employee Severance Pay Plan (the "UGI Severance Plan") assists certain senior level employees of UGI, including Messrs. Greenberg and Bovaird, in the event their employment is terminated without fault on their part. Benefits are payable to a senior executive covered by the UGI Severance Plan if the senior executive's employment is involuntarily terminated for any reason other than for cause or as a result of the senior executive's death or disability.

The UGI Severance Plan provides for cash payments equal to a participant's compensation for a period of time ranging from 3 months to 15 months (30 months in the case of Mr. Greenberg), depending on length of service. In addition, a participant receives the cash equivalent of his or her target bonus under the Annual Bonus Plan, pro-rated for the number of months served in the fiscal year. However, if the termination occurs in the last two months of the fiscal year, the Chief Executive Officer has the discretion to determine whether the participant will receive a pro-rated target bonus, or the actual annual bonus which would have been paid

after the end of the fiscal year, assuming that the participant's entire bonus was contingent on meeting the applicable financial performance goal. Certain employee benefits are continued under the Plan for a period of up to 15 months (30 months in the case of Mr. Greenberg). UGI has the option to pay a participant the cash equivalent of those employee benefits.

In order to receive benefits under the UGI Severance Plan, a senior executive is required to execute a release which discharges UGI and its subsidiaries from liability for any claims the senior executive may have against any of them, other than claims for amounts or benefits due to the executive under any plan, program or contract provided by or entered into with UGI or its subsidiaries. The senior executive is also required to cooperate in attending to matters pending at the time of his or her termination of employment.

Named Executives Employed by AmeriGas Propane. The AmeriGas Propane, Inc. Executive Employee Severance Pay Plan (the "AmeriGas Severance Plan") assists certain senior level employees of the General Partner including Messrs. Bissell, Grady, Katz, Knauss and Ms. Lindsay in the event their employment is terminated without fault on their part. Specified benefits are payable to a senior executive covered by the AmeriGas Severance Plan if the senior executive's employment is involuntarily terminated for any reason other than for cause or as a result of the senior executive's death or disability.

The AmeriGas Severance Plan provides for cash payments equal to a participant's compensation for three months (6 months in the case of the Chief Executive Officer). In addition, a participant receives the cash equivalent of his or her target bonus under the Annual Bonus Plan, pro-rated for the number of months served in the fiscal year. However, if the termination occurs in the last two months of the fiscal year, the Chief Executive Officer has the discretion to determine whether the participant will receive a pro-rated target bonus, or the actual annual bonus which would have been paid after the end of the fiscal year, assuming that the participant's entire bonus was contingent on meeting the applicable financial performance goal. The Plan also provides for separation pay equal to one day's pay per month of service, not to exceed 12 months' compensation. Minimum separation pay ranges from six to twelve months' base salary, depending on the executive's employment grade. Certain employee benefits are continued under the Plan for a period not exceeding 15 months (18 months in the case of the Chief Executive Officer). This period is called the "Employee Benefit Period." The General Partner has the option to pay a participant the cash equivalent of those employee benefits.

In order to receive benefits under the AmeriGas Severance Plan, a senior executive is required to execute a release which discharges the General Partner and its affiliates from liability for any claims the senior executive may have against any of them, other than claims for amounts or benefits due to the executive under any plan, program or contract provided by or entered into with the General Partner or its affiliates. The senior executive is also required to cooperate in attending to matters pending at the time of his or her termination of employment.

CHANGE OF CONTROL ARRANGEMENTS

Named Executives Employed By UGI Corporation. Mr. Greenberg has an agreement with UGI Corporation (the "Agreement") which provides certain benefits in the event of a change of control. The Agreements operate independently of the UGI Severance Plan, continue through

July 2004, and are automatically extended in one-year increments thereafter unless, prior to a change of control, UGI terminates an Agreement. In the absence of a change of control, each Agreement will terminate when, for any reason, the executive terminates his or her employment with UGI or its subsidiaries.

A change of control is generally deemed to occur if: (i) any person $% \left(1\right) =\left(1\right) \left(1\right)$ (other than the executive, his or her affiliates and associates, UGI or any of its subsidiaries, any employee benefit plan of UGI or any of its subsidiaries, or any person or entity organized, appointed, or established by UGI or its subsidiaries for or pursuant to the terms of any such employee benefit plan) together with all affiliates and associates of such person, acquires securities representing 20% or more of either (x) the then outstanding shares of common stock of UGI or (y) the combined voting power of UGI's then outstanding voting securities; (ii) individuals who at the beginning of any 24-month period constitute the Board of Directors (the "Incumbent Board") and any new director whose election by the Board, or nomination for election by UGI's shareholders, was approved by a vote of at least a majority of the Incumbent Board, cease for any reason to constitute a majority thereof; (iii) UGI is reorganized, merged or consolidated with or into, or sells all or substantially all of its assets to, another corporation in a transaction in which former shareholders of UGI do not own more than 50% of the outstanding common stock and the combined voting power, respectively, of the then outstanding voting securities of the surviving or acquiring corporation after the transaction; or (iv) UGI is liquidated or

Severance benefits are payable under the Agreements if there is a termination of the executive's employment without cause at any time within three years after a change of control. In addition, following a change of control, the executive may elect to terminate his or her employment without loss of severance benefits in certain specified contingencies, including termination of officer status; a significant adverse change in authority, duties, responsibilities or compensation; the failure of UGI to comply with and satisfy any of the terms of the Agreement; or a substantial relocation or excessive travel requirements.

An executive who is terminated with rights to severance compensation under an Agreement will be entitled to receive an amount equal to 1.0 or 1.5 (2.5 in the case of Mr. Greenberg) times his or her average total cash remuneration for the preceding five calendar years. If the severance compensation payable under the Agreement, either alone or together with other payments to an executive, would constitute "excess parachute payments," as defined in Section 280G of the Internal Revenue Code of 1986, as amended (the "Code"), the executive will also receive an amount to satisfy the executive's additional tax burden.

Named Executives Employed by the General Partner. Messrs. Bissell, Grady, Katz, Knauss and Ms. Lindsay each have an agreement with the General Partner (the "Agreement") which provides certain benefits in the event of a change of control. The Agreements operate independently of the AmeriGas Severance Plan, continue through July 2004, and are automatically extended in one-year increments thereafter unless, prior to a change of control, the General Partner terminates an Agreement. In the absence of a change of control, each Agreement will terminate when, for any reason, the executive terminates his or her employment with the General Partner or any of its subsidiaries.

A change of control is generally deemed to occur if: (i) a change of control of UGI, as defined above, occurs, (ii) the General Partner, AmeriGas Partners or the Operating Partnership is reorganized, merged or consolidated with or into, or sells all or substantially all of its assets to,

another corporation or partnership in a transaction in which the former shareholders of the General Partner, or former limited partners, as the case may be, do not own more than 50% of the outstanding common stock and combined voting power, or the outstanding common units of such partnership, after the transaction, (iii) the General Partner, AmeriGas Partners or the Operating Partnership is liquidated or dissolved, (iv) UGI and its subsidiaries fail to own more than fifty percent of the general partnership interests of AmeriGas Partners or the Operating Partnership, (v) UGI and its subsidiaries fail to own more than fifty percent of the combined voting power of the General Partner's then outstanding voting securities, or (vi) AmeriGas Propane, Inc. is removed as the general partner of AmeriGas Partners or AmeriGas Propane, Inc. is removed as the general partner of AmeriGas Partners or the Operating Partnership as a result of judicial or administrative proceedings.

Severance benefits are payable under the Agreements if there is a termination of the executive's employment without cause at any time within three years after a change of control. In addition, following a change of control, the executive may elect to terminate his or her employment without loss of severance benefits in certain specified contingencies, including termination of officer status; a significant adverse change in authority, duties, responsibilities or compensation; the failure of the General Partner to comply with and satisfy any of the terms of the Agreement; or a substantial relocation or excessive travel requirements.

An executive who is terminated with rights to severance compensation under an Agreement will be entitled to receive an amount equal to 1.0 (1.5 in the case of Mr. Bissell) times his or her average total cash remuneration for the preceding five calendar years. If the severance compensation payable under the Agreement, either alone or together with other payments to an executive, would constitute "excess parachute payments," as defined in Section 280G of the Code, the executive will also receive an amount to satisfy the executive's additional tax burden.

BOARD OF DIRECTORS

Officers of the General Partner receive no additional compensation for service on the Board of Directors or on any Committee of the Board. For fiscal year 2003, the General Partner paid an annual retainer of \$22,000 to all other directors and an attendance fee of \$1,000 for each Board meeting. For service on Committees, the General Partner paid an annual retainer of (1) \$5,000 to members of the Audit Committee and (2) \$2,000 to each Committee Chairman. Directors were paid an attendance fee of \$1,000 for each Committee meeting. The Directors are also offered employee rates on propane purchases. The General Partner reimburses directors for expenses incurred by them (such as travel expenses) in serving on the Board and Committees. The General Partner determines all expenses allocable to the Partnership, including expenses allocable to the services of directors.

COMPENSATION/PENSION COMMITTEE

The members of the General Partner's Compensation/Pension Committee are Richard C. Gozon (Chairman), Thomas F. Donovan and William J. Marrazzo.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED SECURITY HOLDER MATTERS

OWNERSHIP OF LIMITED PARTNERSHIP UNITS BY CERTAIN BENEFICIAL OWNERS

The following table sets forth certain information regarding each person known by the Partnership to have been the beneficial owner of more than 5% of the Partnership's voting securities representing limited partner interests as of December 1, 2003. AmeriGas Propane, Inc. is the sole general partner of the Partnership.

	TITLE OF CLASS	NAME AND ADDRESS (1) OF BENEFICIAL OWNER	AMOUNT AND NATURE OF BENEFICIAL OWNERSHIP OF PARTNERSHIP UNITS	PERCENT OF CLASS
Common	Units	UGI Corporation AmeriGas, Inc. AmeriGas Propane, Inc. Petrolane Incorporated	24,525,004 (2) 24,525,004 (3) 24,525,004 (4) 7,839,911 (4)	47.0% 47.0% 47.0% 15.0%

- (1) The address of each of UGI, AmeriGas, Inc., AmeriGas Propane, Inc. and Petrolane Incorporated is 460 North Gulph Road, King of Prussia, PA 19406.
- (2) Based on the number of units held by its indirect, wholly-owned subsidiaries, Petrolane Incorporated ("Petrolane") and AmeriGas Propane, Inc.
- (3) Based on the number of units held by its direct and indirect, wholly-owned subsidiaries, AmeriGas Propane, Inc. and Petrolane.
- (4) AmeriGas Propane, Inc.'s beneficial ownership includes 7,839,911 Common Units held by its subsidiary, Petrolane. Beneficial ownership of those Common Units is shared with UGI Corporation and AmeriGas, Inc.

OWNERSHIP OF PARTNERSHIP COMMON UNITS BY THE DIRECTORS AND EXECUTIVE OFFICERS OF THE GENERAL PARTNER

The table below sets forth as of October 31, 2003 the beneficial ownership of Partnership Common Units by each director and each of the Named Executives currently serving the General Partner, as well as by the directors and all of the executive officers of the General Partner as a group. No director, Named Executive or executive officer beneficially owns more than 1% of the Partnership's Common Units. The total number of Common Units beneficially owned by the directors and executive officers of the General Partner as a group represents less than 1% of the Partnership's outstanding Common Units.

Lon R. Greenberg 6,500(2) Thomas F. Donovan 1,000 Richard C. Gozon 5,000 James W. Stratton 1,000(3) Stephen A. Van Dyck Roger B. Vincent 6,000 William J. Marrazzo Eugene V. N. Bissell 12,750(5) Robert H. Knauss Martha B. Lindsay 5,488(6) William D. Katz 7,875 Directors and executive officers as a group (14 62,113 persons)

- (1) Sole voting and investment power unless otherwise specified.
- (2) Of the Units shown, 4,500 are held by Mr. Greenberg's adult children.
- (3) Mr. Stratton's Units are held jointly with his spouse.
- (4) Mr. Marrazzo's Units are held jointly with his spouse.
- (5) Mr. Bissell's Units are held jointly with his spouse.
- (6) Of the Units shown for Ms. Lindsay, 400 are held jointly with her children.

The General Partner is a wholly owned subsidiary of AmeriGas, Inc. which is a wholly owned subsidiary of UGI. The table below sets forth, as of October 31, 2003, the beneficial ownership of UGI Common Stock by each director and each of the Named Executives, as well as by the directors and the executive officers of the General Partner as a group. Including the number of shares of stock underlying exercisable options, Mr. Greenberg is the beneficial owner of approximately 2% of UGI's Common Stock. All other directors, Named Executives and executive officers own less than 1% of UGI's outstanding shares. The total number of shares beneficially owned by the directors and executive officers as a group (including 840,032 shares subject to exercisable options), represents approximately 3% of UGI's outstanding shares.

NUMBER OF UGI SHARES AND NATURE OF BENEFICIAL OWNERSHIP

BENEFICIAL OWNERSHIP NUMBER OF
EXCLUDING EXERCISABLE UGI
ST STOCK OPTIONS (1)(2) STOCK OPTIONS

BENEFICIAL OWNER	UGI STOCK OPTIONS (1)(2)	STOCK OPTIONS	TOTAL
Lon R. Greenberg	158,902 (3)	660,000	828,407
Thomas F. Donovan	13,474 (2)	12,000	25,474
Richard C. Gozon	41,256 (2)	29,700	70,956
James W. Stratton	28,090 (2)(4)	29,700	57,790
Stephen A. Van Dyck	0	0	0
Roger B. Vincent	0	0	0
William J. Marrazzo	0	0	0
Eugene V.N. Bissell	56,190 (5)	23,876	80,066
Robert H. Knauss	6,756	4,001	10,757
Martha B. Lindsay	8,917 (6)	26,501	35,418
William D. Katz	13,755 (7)	13,001	26,756
Directors and executive officers as a group (14 persons)	351,834	840,032	1,191,866

(1) Sole voting and investment power unless otherwise specified.

NAME OF

- (2) Included in the number of shares shown are Deferred Units ("Units") acquired through the UGI Corporation 1997 Directors' Equity Compensation Plan, Amended and Restated as of April 29, 2003. Units are neither actual shares nor other securities, but each Unit will be converted to one share of UGI common stock and paid out to directors upon their retirement or termination of service. The number of Units included for the directors is as follows: Messrs. Donovan (6,494), Gozon (30,452) and Stratton (21,786).
- (3) Mr. Greenberg holds 132,330 shares jointly with his spouse and 9,505 shares are represented by units held in the UGI Stock Fund of the 401(k) Employee Savings Plan.
- (4) Mr. Stratton holds 6,304 shares jointly with his spouse.
- (5) Mr. Bissell holds these shares jointly with his spouse.
- (6) Of the shares shown for Ms. Lindsay, 750 are held jointly with her children.
- (7) Mr. Katz holds 1,990 shares jointly with his spouse and 3,589 shares are represented by units held in the UGI Stock Fund of the 401(k) Employee Savings Plan.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

The General Partner employs persons responsible for managing and operating the Partnership. The Partnership reimburses the General Partner for the direct and indirect costs of providing these services, including all compensation and benefit costs. For fiscal year 2003, these costs totaled approximately \$284.0 million.

The Operating Partnership has a revolving line of credit up to a maximum of \$20 million from the General Partner available until October 15, 2006, the termination date of the Revolving Credit Facility. Any loans under this agreement will be unsecured and subordinated to all senior debt of the Operating Partnership. The commitment fees for this line of credit are computed on the same basis as the facility fees under the Revolving Credit Facility, and totaled \$70,972 in fiscal year 2003. Interest rates are based on one-month offshore interbank borrowing rates. The interest rate for a recent Credit Facility borrowing from October 24, 2002 to October 25, 2002 was 4.75%. See Note 6 to the Partnership's Consolidated Financial Statements, which are filed as an exhibit to this report.

The Partnership and the General Partner also have extensive, ongoing relationships with UGI and its affiliates. UGI performs certain financial and administrative services for the General Partner on behalf of the Partnership. UGI does not receive a fee for such services, but is reimbursed for all direct and indirect expenses incurred in connection with providing these services, including all compensation and benefit costs. A wholly owned subsidiary of UGI provides the Partnership with excess automobile liability insurance with limits of \$500,000 per occurrence and in the aggregate excess of \$500,000 per occurrence. Another wholly owned subsidiary of UGI leases office space to the General Partner for its headquarters staff. In addition, a UGI master policy provides accidental death and business travel and accident insurance coverage for employees of the General Partner. The General Partner is billed directly by the insurer for this coverage. As discussed under "Business--Trade Names; Trade and Service Marks," UGI and the General Partner have licensed the trade names "AmeriGas" and "America's Propane Company" and the related service marks and trademark to the Partnership on a royalty-free basis. Finally, the Partnership obtains management information services from the General Partner, and reimburses the General Partner for its direct and indirect expenses related to those services. The rental payments and insurance premiums charged to the Partnership by UGI and its affiliates are comparable to amounts charged by unaffiliated parties. In fiscal year 2003, the Partnership paid UGI and its affiliates, including the General Partner, approximately \$10 million for the services and expense reimbursements referred to in this paragraph.

The Partnership purchases propane on behalf of Atlantic Energy, Inc. ("Atlantic Energy"), a 50% owned joint venture with Conoco, Inc. Atlantic Energy reimburses AmeriGas OLP for its purchases plus interest at the rate of 8% as Atlantic Energy sells such propane to third parties or to the Partnership itself. The total dollar value of propane purchased on behalf of Atlantic Energy was \$17.2 million and \$11.4 million in fiscal years 2003 and 2002, respectively. Purchases of propane by AmeriGas OLP from Atlantic Energy during fiscal years 2003 and 2002 totaled \$23.9 million and \$12.1 million, respectively.

AmeriGas OLP also provides other services to Atlantic Energy including marketing, billing, accounting, insurance and other administrative services and is reimbursed for the related costs. In addition, AmeriGas OLP enters into product cost hedging contracts on behalf of Atlantic Energy. When these contracts are settled, AmeriGas OLP is reimbursed the cost of any losses by, or distributes the proceeds of any gains to, Atlantic Energy. Conoco Inc. has agreed to indemnify AmeriGas OLP for one-half of any losses arising from its hedging activities on behalf of Atlantic Energy to the extent that AmeriGas OLP is not indemnified for such losses by Atlantic Energy.

The highest amounts due from Atlantic Energy during fiscal year 2003 and at November 30, 2003 were \$12.3\$ million and \$1.2\$ million, respectively.

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

The aggregate fees billed by PricewaterhouseCoopers LLP, the Partnership's independent public accountants, in fiscal 2003 were as follows:

AUDIT FEES

TAX FEES

In the course of its meetings, the Audit Committee considered whether the provision by PricewaterhouseCoopers LLP of the professional services described under "All Other Fees" is compatible with PricewaterhouseCoopers LLP's independence.

ITEM 15. EXHIBITS, FINANCIAL STATEMENT SCHEDULES, AND REPORTS ON FORM 8-K

- (a) DOCUMENTS FILED AS PART OF THIS REPORT:
- (1) AND (2) FINANCIAL STATEMENTS AND FINANCIAL STATEMENT SCHEDULES

The financial statements and financial statement schedules incorporated by reference or included in this report are listed in the accompanying Index to Financial Statements and Financial Statement Schedules set forth on pages F-2 and F-3 of this report, which is incorporated herein by reference.

NOTICE REGARDING ARTHUR ANDERSEN LLP

Arthur Andersen LLP audited our consolidated financial statements for the three years in the period ended September 30, 2001, and issued a report thereon dated November 16, 2001. Arthur Andersen LLP has not reissued its report or consented to the incorporation by reference of such report into the Partnership's prospectuses for the offer and sale of Common Units and debt securities. On June 15, 2002, Arthur Andersen LLP was convicted of obstruction of justice by a federal jury in Houston, Texas in connection with Arthur Andersen LLP's work for Enron Corp. On September 15, 2002, a federal judge upheld this conviction. Arthur Andersen LLP ceased its audit practice before the SEC on August 31, 2002. Effective May 21, 2002, we terminated the engagement of Arthur Andersen LLP as our independent accountants and engaged PricewaterhouseCoopers LLP to serve as our independent accountants for the fiscal year ending September 30, 2002. Because of the circumstances currently affecting Arthur Andersen LLP, as a practical matter it may not be able to satisfy any claims arising from the provision of auditing services to us, including claims available to security holders under federal and state securities laws.

(3) LIST OF EXHIBITS:

The exhibits filed as part of this report are as follows (exhibits incorporated by reference are set forth with the name of the registrant, the type of report and registration number or last date of the period for which it was filed, and the exhibit number in such filing):

HIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
2.1	Merger and Contribution Agreement among AmeriGas Partners, L.P., AmeriGas Propane, L.P., New AmeriGas Propane, Inc., AmeriGas Propane, Inc., AmeriGas Propane-2, Inc., Cal Gas Corporation of America, Propane Transport, Inc. and NORCO Transportation Company	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 33-92734)	10.21

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
2.2	Conveyance and Contribution Agreement among AmeriGas Partners, L.P., AmeriGas Propane, L.P. and Petrolane Incorporated	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 33-92734)	10.22
3.1	Second Amended and Restated Agreement of Limited Partnership of AmeriGas Partners, L.P. dated as of September 30, 2000	AmeriGas Partners, L.P.	Form 8-K (9/30/00)	1
3.2	Certificate of Incorporation of AmeriGas Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 33-92734)	3.3
3.3	Certificate of Incorporation of AmeriGas Eagle Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 333-72986)	3.2
3.4	Certificate of Incorporation of AP Eagle Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 333-72986)	3.3
3.5	Bylaws of AmeriGas Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 33-92734)	3.4
3.6	By-Laws of AmeriGas Eagle Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-3 (No. 333-72986)	3.4
3.7	By-Laws of AP Eagle Finance Corp.	AmeriGas Partners, L.P.	Registration Statement on Form S-3 (No. 333-72986)	3.5
3.8	Amended and Restated Agreement of Limited Partnership of AmeriGas Eagle Propane, L.P. dated July 19, 1999	AmeriGas Partner, L.P.	Form 10-K (9/30/01)	3.8

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBI ⁻
3.9	Amended and Restated Agreement of Limited Partnership of AmeriGas Propane, L.P. dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	3.9
4	Instruments defining the rights of security holders, including indentures. (The Partnership agrees to furnish to the Commission upon request a copy of any instrument defining the rights of holders of long-term debt not required to be fil pursuant to Item 601(b)(4) of Regulation S-K)			
4.1	[Intentionally omitted]			
4.2	[Intentionally omitted]			
4.3	Note Agreement dated as of April 12, 1995 among The Prudential Insurance Company of America, Metropolitan Life Insurance Company, and certain other institutional investors and AmeriGas Propane, L.P., New AmeriGas Propane, Inc. and Petrolane Incorporated	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.8
4.4	First Amendment dated as of September 12, 1997 to Note Agreement dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-Q (9/30/97)	4.5
4.5	Second Amendment dated as of September 15, 1998 to Note Agreement dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/98)	4.6
4.6	Third Amendment dated as of March 23, 1999 to Note Agreement dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-Q (3/31/99)	10.2
4.7	Fourth Amendment dated as of March 16, 2000 to Note Agreement dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-Q (6/30/00)	10.2
4.8	Fifth Amendment dated as of August 1, 2001 to Note Agreement dated as of April 12, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	4.8
4.9	Indenture dated April 4, 2001 among AmeriGas Partners, L.P., AmeriGas Eagle Finance Corp., and First Union National Bank, as Trustee	AmeriGas Partners, L.P.	Form 10-Q (6/30/01)	4
4.10	Indenture dated August 21, 2001 among AmeriGas Partners, L.P., AP Eagle Finance Corp. and First Union National Bank as trustee	AmeriGas Partners, L.P.	Registration Statement on Form S-4 (No. 333-72986)	4.2

KHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
*10.1	Credit Agreement dated as of August 28, 2003 among AmeriGas Propane, L.P., AmeriGas Propane, Inc., Petrolane Incorporated, Citicorp USA, Inc., Credit Suisse First Boston, Wachovia Bank, National Association, as Issuing Bank and certain financial institutions			
10.2**	AmeriGas Propane, Inc. Discretionary Long-Term Incentive Plan for Non-Executive Key Employees	AmeriGas Partners, L.P.	Form 10-K (9/30/02)	10.2
10.3	[Intentionally omitted]			
10.4	[Intentionally omitted]			
10.5	[Intentionally omitted]			
*10.6	Notice of appointment of Wachovia Bank National Association as Collateral Agent effective as of August 28, 2003, pursuant to Intercreditor and Agency Agreement dated as of April 19, 1995			
10.7	Intercreditor and Agency Agreement dated as of April 19, 1995 among AmeriGas Propane, Inc., Petrolane Incorporated, AmeriGas Propane, L.P., Bank of America National Trust and Savings Association ("Bank of America") as Agent, Mellon Bank, N.A. as Cash Collateral Sub-Agent, Bank of America as Collateral Agent and certain creditors of AmeriGas Propane, L.P.	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.2
10.8	First Amendment dated as of July 31, 2001 to Intercreditor and Agency Agreement dated as of April 19, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.8
10.9	General Security Agreement dated as of April 19, 1995 among AmeriGas Propane, L.P., Bank of America National Trust and Savings Association and Mellon Bank, N.A.	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.3
10.10	First Amendment dated as of July 31, 2001 to General Security Agreement dated as of April 19, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.10
10.11	Subsidiary Security Agreement dated as of April 19, 1995 among AmeriGas Propane, L.P., Bank of America National Trust and Savings Association as Collateral Agent and Mellon Bank, N.A. as Cash Collateral Agent	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.4

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
10.12	First Amendment dated as of July 31, 2001 to Subsidiary Security Agreement dated as of April 19, 1995	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.12
10.13	Restricted Subsidiary Guarantee dated as of April 19, 1995 by AmeriGas Propane, L.P. for the benefit of Bank of America National Trust and Savings Association, as Collateral Agent	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.5
10.14	Trademark License Agreement dated April 19, 1995 among UGI Corporation, AmeriGas, Inc., AmeriGas Propane, Inc., AmeriGas Partners, L.P. and AmeriGas Propane, L.P.	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.6
10.15	Trademark License Agreement dated April 19, 1995 among AmeriGas Propane, Inc., AmeriGas Partners, L.P. and AmeriGas Propane, L.P.	AmeriGas Partners, L.P.	Form 10-Q (3/31/95)	10.7
10.16	Stock Purchase Agreement dated May 27, 1989, as amended and restated July 31, 1989, between Texas Eastern Corporation and QFB Partners	Petrolane Incorporated/ AmeriGas, Inc.	Registration on Form S-1 (No. 33-69450)	10.16(a)
10.17	Amended and Restated Sublease Agreement dated April 1, 1988, between Southwest Salt Co. and AP Propane, Inc. (the "Southwest Salt Co. Agreement")	UGI Corporation	Form 10-K (9/30/94)	10.35
10.18	Letter dated July 8, 1998 pursuant to Article 1, Section 1.2 of the Southwest Salt Co. Agreement re: option to renew for period of June 1, 2000 to May 31, 2005	UGI Corporation	Form 10-K (9/30/99)	10.5
*10.19	Financing Agreement dated as of August 28, 2003 between AmeriGas Propane, Inc. and AmeriGas Propane, L.P.			
10.20	Agreement by Petrolane Incorporated and certain of its subsidiaries parties thereto ("Subsidiaries") for the Sale of the Subsidiaries' Inventory and Assets to the Goodyear Tire & Rubber Company and D.C.H., Inc., as Purchaser, dated as of December 18, 1985	Petrolane Incorporated	Form 10-K (9/23/94)	10.13
10.21	[Intentionally omitted]			
10.22**	UGI Corporation Annual Bonus Plan dated March 8, 1996	UGI Corporation	Form 10-Q (6/30/96)	10.4

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
10.23**	AmeriGas Propane, Inc. Annual Bonus Plan effective October 1, 1998	AmeriGas Partners, L.P.	Form 10-K (9/30/99)	10.17
10.24**	1997 Stock Purchase Loan Plan	UGI Corporation	Form 10-K (9/30/97)	10.16
10.25**	UGI Corporation Senior Executive Employee Severance Pay Plan effective January 1, 1997	UGI Corporation	Form 10-K (9/30/97)	10.12
10.26**	AmeriGas Propane, Inc. Executive Employee Severance Pay Plan effective January 1, 1997	AmeriGas Partners, L.P.	Form 10-Q (12/31/96)	10.1
10.27**	Amendment No. 1 to AmeriGas Propane, Inc. Executive Employee Severance Pay Plan	AmeriGas Partners, L.P.	Form 10-Q (6/30/98)	10
10.28**	UGI Corporation 1992 Non-Qualified Stock Option Plan, Amended and Restated as of April 29, 2003	UGI Corporation	Form 10-Q (3/31/03)	10.6
10.29**	Form of AmeriGas Propane, Inc. 2000 Long-Term Incentive Plan	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.29
10.30**	UGI Corporation 2000 Stock Incentive Plan Amended and Restated as of April 29, 2003	UGI Corporation	Form 10-Q (3/31/03)	10.5
10.31**	AmeriGas Propane, Inc. Supplemental Executive Retirement Plan effective October 1, 1996	AmeriGas Partners, L.P.	Form 10-K (9/30/97)	10.27
10.32**	UGI Corporation 1997 Stock Option and Dividend Equivalent Plan Amended and Restated as of April 29, 2003	UGI Corporation	Form 10-Q (3/31/03)	10.4
10.33**	UGI Corporation Supplemental Executive Retirement Plan Amended and Restated effective October 1, 1996	UGI Corporation	Form 10-Q (6/30/98)	10
10.34**	Description of Change of Control arrangements for Mr. Greenberg	UGI Corporation	Form 10-K (9/30/99)	10.33
10.35**	Description of Change of Control arrangements for Messrs. Bissell, Grady, Katz and Knauss and Ms. Lindsay	AmeriGas Partners, L. P.	Form 10-K (9/30/99)	10.31
10.36	Purchase Agreement by and among Columbia Energy Group, Columbia Propane Corporation, CP Holdings, Inc., Columbia Propane, L.P., AmeriGas Propane, L.P., AmeriGas Partners, L.P. and AmeriGas Propane, Inc. dated as of January 30, 2001 and amended and restated August 7, 2001	AmeriGas Partners, L.P.	Form 8-K (8/8/01)	10.1

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
10.37	Purchase Agreement by and among Columbia Propane, L.P., CP Holdings, Inc., Columbia Propane Corporation, National Propane Partners, L.P., National Propane Corporation, National Propane SPG, Inc., and Triarc Companies, Inc. dated as of April 5, 1999	National Propane Partners, L.P.	Form 8-K (4/19/99)	10.5
10.38	Capital Contribution Agreement dated as of August 21, 2001 by and between Columbia Propane, L.P. and AmeriGas Propane, L.P. acknowledged and agreed to by CP Holdings, Inc.	AmeriGas Partners, L.P.	Form 8-K (8/21/01)	10.2
10.39	Promissory Note by National Propane L.P., a Delaware limited partnership in favor of Columbia Propane Corporation dated July 19, 1999	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.39
10.40	Loan Agreement dated July 19, 1999, between National Propane, L.P. and Columbia Propane Corporation	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.40
10.41	First Amendment dated August 21, 2001 to Loan Agreement dated July 19, 1999 between National Propane, L.P. and Columbia Propane Corporation	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.41
10.42	Columbia Energy Group Payment Guaranty dated April 5, 1999	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.42
10.43	Ground Lease dated August 13, 2001 by and between Reading Terminals Corporation and Columbia Propane Corporation	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.43
10.44	Master Lease dated August 20, 2001 between AmeriGas Propane, L.P. and Columbia Propane, L.P.	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.44
10.45	Master Sublease dated August 20, 2001 between AmeriGas Propane, L.P. and Columbia Propane, L.P.	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.45
10.46	Keep Well Agreement by and between AmeriGas Propane, L.P. and Columbia Propane Corporation dated August 21, 2001	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.46
10.47	Management Services Agreement effective as of August 21, 2001 between AmeriGas Propane, Inc. and AmeriGas Eagle Holdings, Inc., the general partner of AmeriGas Eagle Propane, L.P.	AmeriGas Partners, L.P.	Form 10-K (9/30/01)	10.47

EXHIBIT NO.	EXHIBIT	REGISTRANT	FILING	EXHIBIT
*13	Pages 10 through 27 of the AmeriGas Partners, L.P. Annual Report for the year ended September 30, 2003			
*14	Code of Ethics for principal executive, financial and accounting officers			
18	Letter of Arthur Andersen LLP regarding change in accounting principles	AmeriGas Partners, L.P.		18
*21	Subsidiaries of AmeriGas Partners, L.P.			
*23	Consent of PricewaterhouseCoopers LLP			
*31.1	Certification by the Chief Executive Officer relating to the Registrants' Report on Form 10-K for the year ended September 30, 2003 pursuant to Section 302 of the Sarbanes-Oxley Act of 2002)		
*31.2	Certification by the Chief Financial Officer relating to the Registrants' Report on Form 10-K for the year ended September 30, 2003 pursuant to Section 302 of the Sarbanes-Oxley Act of 2002)		
*32	Certification by the Chief Executive Officer and the Chief Financial Officer relating to the Registrants' Report on Form 10-K for the fiscal year ended September 30, 2003, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002			

^{*} Filed herewith.

(b) REPORTS ON FORM 8-K:

The Company furnished information in a Current Report on Form 8-K during the fourth quarter of fiscal year 2003 as follows:

Date of Report	<pre>Item Number(s)</pre>	Content
7/30/03	7, 12	Press Release reporting financial results for the third fiscal quarter ended June 30, 2003

 $^{^{\}star\star}$ As required by Item 14(a)(3), this exhibit is identified as a compensatory plan or arrangement.

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized.

AMERIGAS PARTNERS, L.P.

Date: December 15, 2003 By: AmeriGas Propane, Inc.

SIGNATURE

its General Partner

By: Martha B.Lindsay

Martha B. Lindsay Vice President - Finance and Chief Financial Officer

TITLE

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below on December 15, 2003 by the following persons on behalf of the Registrant and in the capacities with AmeriGas Propane, Inc., General Partner, indicated.

Eugene V.N. Bissell Eugene V.N. Bissell	President, and Chief Executive Officer (Principal Executive Officer) and Director
Lon R. Greenberg	Chairman and Director
Lon R. Greenberg	
Martha B. Lindsay Martha B. Lindsay	Vice President - Finance and Chief Financial Officer (Principal Financial Officer)
Richard R. Eynon Richard R. Eynon	Controller and Chief Accounting Officer (Principal Accounting Officer)

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below on December 15, 2003 by the following persons on behalf of the Registrant and in the capacities with AmeriGas Propane, Inc., General Partner, indicated.

SIGNATURE	TITLE
Thomas F. Donovan	Director
Thomas F. Donovan	
Richard C. Gozon	Director
Richard C. Gozon	
William J. Marrazzo	Director
William J. Marrazzo	
James W. Stratton	Director
James W. Stratton	
Stephen A. Van Dyck	Director
Stephen A. Van Dyck	
Roger B. Vincent	Director
Roger B. Vincent	

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized.

AMERIGAS FINANCE CORP.

TITLE

Date: December 15, 2003 By: Martha B.Lindsay

Martha B.Lindsay Vice President - Finance and Chief Financial Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below on December 16, 2003 by the following persons on behalf of the Registrant and in the capacities indicated.

Eugene V.N. Bissell Eugene V.N. Bissell	President (Principal Executive Officer) and Director
Martha B. Lindsay 	Vice President - Finance and Chief Financial Officer (Principal Financial Officer) and Director
Richard R. Eynon 	Controller and Chief Accounting Officer (Principal Accounting Officer)
Robert H. Knauss	Director

Robert H. Knauss

SIGNATURE

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized.

AMERIGAS EAGLE FINANCE CORP.

Date: December 15, 2003

By: Martha B.Lindsay

Martha B.Lindsay Vice President - Finance and Chief Financial Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below on December 15, 2003 by the following persons on behalf of the Registrant and in the capacities indicated.

SIGNATURE	TITLE
Eugene V.N. Bissell Eugene V.N. Bissell	President (Principal Executive Officer) and Director
Martha B. Lindsay Martha B. Lindsay	Vice President - Finance and Chief Financial Officer (Principal Financial Officer) and Director
Richard R. Eynon 	Controller and Chief Accounting Officer (Principal Accounting Officer)
Robert H. Knauss	Director

Robert H. Knauss

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned, thereunto duly authorized.

AP EAGLE FINANCE CORP.

TITLE

Director

Date: December 15, 2003

SIGNATURE

By: Martha B.Lindsay

Martha B.Lindsay Vice President - Finance and Chief Financial Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below on December 15, 2003 by the following persons on behalf of the Registrant and in the capacities indicated.

Eugene V.N. Bissell	President (Principal Executive Officer) and Director
Eugene V.N. Bissell	officer) and birector
Martha B. Lindsay	Vice President - Finance and Chief Financial Officer
Martha B. Lindsay	(Principal Financial Officer) and Director
Richard R. Eynon	Controller and Chief Accounting Officer
Richard R. Eynon	(Principal Accounting Officer)

Robert H. Knauss

Robert H. Knauss

AMERIGAS PARTNERS, L.P. AMERIGAS FINANCE CORP. AMERIGAS EAGLE FINANCE CORP. AP EAGLE FINANCE CORP.

FINANCIAL INFORMATION

FOR INCLUSION IN ANNUAL REPORT ON

FORM 10-K FOR THE FISCAL

YEAR ENDED SEPTEMBER 30, 2003

F-1

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES

INDEX TO FINANCIAL STATEMENTS AND FINANCIAL STATEMENT SCHEDULES

The consolidated financial statements of AmeriGas Partners, L.P. and subsidiaries, together with the reports thereon of PricewaterhouseCoopers LLP dated November 17, 2003 and Arthur Andersen LLP dated November 16, 2001, listed in the following index, are included in AmeriGas Partners' 2003 Annual Report to Unitholders and are incorporated herein by reference. With the exception of the pages listed in this index and information incorporated in Items 5 and 8, the 2003 Annual Report to Unitholders is not to be deemed filed as part of this Report.

	Form 10-K (page)	Annual Report to Unitholders (page)
AmeriGas Partners, L.P. and Subsidiaries		
Financial Statements:		
Reports of Independent Auditors	Exhibit 13	26
Consolidated Balance Sheets as of September 30, 2003 and 2002	Exhibit 13	10
Consolidated Statements of Operations for the years ended September 30, 2003, 2002 and 2001	Exhibit 13	11
Consolidated Statements of Cash Flows for the years ended September 30, 2003, 2002 and 2001	Exhibit 13	12
Consolidated Statements of Partners' Capital for the years ended September 30, 2003, 2002 and 2001	Exhibit 13	13
Notes to Consolidated Financial Statements		14-25
Supplementary Data (unaudited):		
Quarterly Data for the years ended September 30, 2003 and 2002	Exhibit 13	25
Financial Statements Schedules:		
I - Condensed Financial Information of Registrant (Parent Company)	S-1 to S-3	
II - Valuation and Qualifying Accounts	S-4 to S-5	
Reports of Independent Accountants on Financial Statement Schedules	S-6 to S-7	

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES

INDEX TO FINANCIAL STATEMENTS AND FINANCIAL STATEMENT SCHEDULES (continued)

Form 10-K (page)
F-5 to F-6
F-7
F-8
F-9
F-11 to F-12
F-13
F-14
F-15
F-17 to F-18
F-19
F-20
F-21

We have omitted all other financial statement schedules because the required information is either (1) not present; (2) not present in amounts sufficient to require submission of the schedule; or (3) the information required is included elsewhere in the financial statements or related notes.

AMERIGAS FINANCE CORP.

FINANCIAL STATEMENTS for the years ended September 30, 2003, 2002 and 2001

F-4

Report of Independent Auditors

To the Board of Directors of AmeriGas Propane, Inc.:

In our opinion, the accompanying balance sheets and the related statements of stockholder's equity, present fairly, in all material respects, the financial position of AmeriGas Finance Corp. at September 30, 2003 and 2002, in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the management of AmeriGas Propane, Inc.; our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion. The financial statements of AmeriGas Finance Corp. as of and for the year ended September 30, 2001 were audited by other independent accountants who have ceased operations. Those independent accountants expressed an unqualified opinion on those financial statements in their report dated November 16, 2001.

PricewaterhouseCoopers LLP Philadelphia, Pennsylvania

November 17, 2003

THIS REPORT IS A COPY OF THE PREVIOUSLY ISSUED ACCOUNTANT'S REPORT OF ARTHUR ANDERSEN LLP AND HAS NOT BEEN REISSUED BY ARTHUR ANDERSEN LLP.

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To AmeriGas Finance Corp.:

We have audited the accompanying balance sheets of AmeriGas Finance Corp. (a Delaware corporation and a wholly owned subsidiary of AmeriGas Partners, L.P.) as of September 30, 2001 and 2000, and the related statements of stockholder's equity for each of the three years in the period ended September 30, 2001. These financial statements are the responsibility of the management of AmeriGas Propane, Inc. Our responsibility is to express an opinion on these financial statements based upon our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AmeriGas Finance Corp. as of September 30, 2001 and 2000, in conformity with accounting principles generally accepted in the United States.

ARTHUR ANDERSEN LLP

Philadelphia, Pennsylvania November 16, 2001

AMERIGAS FINANCE CORP. (A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

BALANCE SHEETS

	September 30,	
	2003	2002
ASSETS		
Cash	\$1,000	\$1,000
Total assets	\$1,000 =====	\$1,000 =====
STOCKHOLDER'S EQUITY		
Common stock, without par value; 100 shares authorized, issued and outstanding Additional paid-in capital	\$ - 1,000	\$ - 1,000
Total stockholder's equity	\$1,000 =====	\$1,000 =====

See accompanying note to financial Statements.

F-7

AMERIGAS FINANCE CORP. (A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

STATEMENTS OF STOCKHOLDER'S EQUITY

	Common Stock	Additional Paid-in Capital	Retained Earnings
BALANCE SEPTEMBER 30, 2001	\$ - 	\$1,000 	\$ -
BALANCE SEPTEMBER 30, 2002	-	1,000	-
BALANCE SEPTEMBER 30, 2003	\$ - =====	\$1,000 =====	\$ - ======

See accompanying note to financial statements.

F-8

NOTE TO FINANCIAL STATEMENTS

SEPTEMBER 30, 2003 AND 2002

AmeriGas Finance Corp. (AmeriGas Finance), a Delaware corporation, was formed on March 13, 1995 and is a wholly owned subsidiary of AmeriGas Partners, L.P. (AmeriGas Partners).

On April 19, 1995, AmeriGas Partners and AmeriGas Finance jointly and severally issued \$100,000,000 face value of 10.125% Senior Notes due April 2007 (Notes). In November 2001, AmeriGas Partners redeemed prior to maturity \$15,000,000 face value of the Notes at a redemption price of 103.375% and redeemed the remaining \$85,000,000 face value of the Notes in January 2003 at a redemption premium of 102.25%.

As of November 21, 2003, AmeriGas Partners and AmeriGas Finance have an effective unallocated shelf registration statement with the Securities and Exchange Commission under the Securities Act of 1933 under which AmeriGas Partners may issue up to \$500,000,000 of debt or equity. AmeriGas Finance will be the co-obligor of the debt securities, if any, issued pursuant to the registration statement.

AmeriGas Partners owns all 100 shares of AmeriGas Finance common stock outstanding.

FINANCIAL STATEMENTS for the year ended September 30, 2003 and 2002 and for the period from February 22, 2001 (inception) through September 30, 2001

F-10

REPORT OF INDEPENDENT AUDITORS

To the Board of Directors of AmeriGas Propane, Inc.:

In our opinion, the accompanying balance sheets and the related statements of stockholder's equity, present fairly, in all material respects, the financial position of Americas Eagle Finance Corp. at September 30, 2003 and 2002, in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the management of AmeriGas Propane, Inc.; our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion. The financial statements of AmeriGas Eagle Finance Corp. as of September 30, 2001 and for the period from February 22, 2001 (inception) through September 30, 2001 were audited by other independent accountants who have ceased operations. Those independent accountants expressed an unqualified opinion on those financial statements in their report dated November 16, 2001.

PricewaterhouseCoopers LLP Philadelphia, Pennsylvania November 17, 2003 THIS REPORT IS A COPY OF THE PREVIOUSLY ISSUED ACCOUNTANT'S REPORT OF ARTHUR ANDERSEN LLP AND HAS NOT BEEN REISSUED BY ARTHUR ANDERSEN LLP.

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To AmeriGas Eagle Finance Corp.:

We have audited the accompanying balance sheet of AmeriGas Eagle Finance Corp. (a Delaware corporation and a wholly owned subsidiary of AmeriGas Partners, L.P.) as of September 30, 2001, and the related statements of stockholder's equity for the period from February 22, 2001 (inception) through September 30, 2001. These financial statements are the responsibility of the management of AmeriGas Propane, Inc. Our responsibility is to express an opinion on these financial statements based upon our audit.

We conducted our audit in accordance with auditing standards generally accepted in the United States. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe our audit provides a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AmeriGas Eagle Finance Corp. as of September 30, 2001, in conformity with accounting principles generally accepted in the United States.

ARTHUR ANDERSEN LLP

Philadelphia, Pennsylvania November 16, 2001

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

BALANCE SHEETS

	September 30,	
	2003	2002
ASSETS		
Cash	\$1,000	\$1,000
Total assets	\$1,000	\$1,000
	=====	=====
STOCKHOLDER'S EQUITY		
Common stock, without par value; 100 shares authorized,		
issued and outstanding	\$ -	\$ -
Additional paid-in capital	1,000	1,000
Total stockholder's equity	\$1,000	\$1,000
	=====	=====

See accompanying note to financial statements.

F-13

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

STATEMENTS OF STOCKHOLDER'S EQUITY

	Additional				
	Common	Paid-in	Retained		
	Stock	Capital	Earnings		
BALANCE FEBRUARY 22, 2001	\$ -	\$ -	\$ -		
Issuance of AmeriGas Eagle Finance Corp.					
Common Stock	-	1,000	-		
BALANCE SEPTEMBER 30, 2001	-	1,000	_		
,					
BALANCE SEPTEMBER 30, 2002	-	1,000	-		
,					
BALANCE SEPTEMBER 30, 2003	\$ -	\$1,000	\$ -		
·	=======	======	========		

See accompanying note to financial statements.

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

NOTE TO FINANCIAL STATEMENTS

SEPTEMBER 30, 2003 AND 2002

AmeriGas Eagle Finance Corp. (Eagle Finance), a Delaware corporation, was formed on February 22, 2001 and is a wholly owned subsidiary of AmeriGas Partners, L.P. (AmeriGas Partners).

On April 4, 2001, AmeriGas Partners and Eagle Finance jointly and severally issued \$60,000,000 face amount of 10% Senior Notes due April 2006.

AmeriGas Partners owns all 100 shares of Eagle Finance common stock outstanding.

FINANCIAL STATEMENTS for the year ended September 30, 2003 and 2002 and for the period from April 12, 2001 (inception) through September 30, 2001

F-16

REPORT OF INDEPENDENT AUDITORS

To the Board of Directors of AmeriGas Propane, Inc.:

In our opinion, the accompanying balance sheets and the related statements of stockholder's equity, present fairly, in all material respects, the financial position of AP Eagle Finance Corp. at September 30, 2003 and 2002, in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the management of AmeriGas Propane, Inc.; our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion. The financial statements of AP Eagle Finance Corp. as of September 30, 2001 and for the period from April 12, 2001 (inception) through September 30, 2001 were audited by other independent accountants who have ceased operations. Those independent accountants expressed an unqualified opinion on those financial statements in their report dated November 16, 2001.

PricewaterhouseCoopers LLP Philadelphia, Pennsylvania November 17, 2003 THIS REPORT IS A COPY OF THE PREVIOUSLY ISSUED ACCOUNTANT'S REPORT OF ARTHUR ANDERSEN LLP AND HAS NOT BEEN REISSUED BY ARTHUR ANDERSEN LLP.

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To AP Eagle Finance Corp.:

We have audited the accompanying balance sheet of AP Eagle Finance Corp. (a Delaware corporation and a wholly owned subsidiary of AmeriGas Partners, L.P.) as of September 30, 2001, and the related statements of stockholder's equity for the period from April 12, 2001 (inception) through September 30, 2001. These financial statements are the responsibility of the management of AmeriGas Propane, Inc. Our responsibility is to express an opinion on these financial statements based upon our audit.

We conducted our audit in accordance with auditing standards generally accepted in the United States. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe our audit provides a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AP Eagle Finance Corp. as of September 30, 2001, in conformity with accounting principles generally accepted in the United States.

ARTHUR ANDERSEN LLP

Philadelphia, Pennsylvania November 16, 2001

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

BALANCE SHEETS

	September 30,	
	2003	2002
ASSETS		
Cash	\$1,000	\$1,000
Total assets	\$1,000	\$1,000
	=====	=====
STOCKHOLDER'S EQUITY		
Common stock, without par value; 100 shares authorized,		
issued and outstanding	\$ -	\$ -
Additional paid-in capital	1,000	1,000
Total stockholder's equity	\$1,000	\$1,000
	=====	=====

See accompanying note to financial statements.

F-19

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

STATEMENTS OF STOCKHOLDER'S EQUITY

	Common Stock	Additional Paid-in Capital	Retained Earnings
BALANCE APRIL 12, 2001	\$ -	\$ -	\$ -
Issuance of AP Eagle Finance Corp. Common Stock	-	1,000	-
BALANCE SEPTEMBER 30, 2001	-	1,000	-
BALANCE SEPTEMBER 30, 2002		1,000	-
BALANCE SEPTEMBER 30, 2003	\$ - =====	\$1,000 =====	\$ - =====

See accompanying note to financial statements.

(A WHOLLY OWNED SUBSIDIARY OF AMERIGAS PARTNERS, L.P.)

NOTE TO FINANCIAL STATEMENTS

SEPTEMBER 30, 2003 AND 2002

AP Eagle Finance Corp. (AP Eagle Finance), a Delaware corporation, was formed on April 12, 2001 and is a wholly owned subsidiary of AmeriGas Partners, L.P. (AmeriGas Partners).

On August 21, 2001, AmeriGas Partners and AP Eagle Finance jointly and severally issued \$200,000,000 face amount of 8.875% Series A Senior Notes due May 2011. On December 20, 2001, AmeriGas Partners and AP Eagle Finance exchanged \$199,985,000 face amount of 8.875% Series A Senior Notes due May 2011 for a like amount of AmeriGas Partners and AP Eagle Finance 8.875% Series B Senior Notes due May 2011 pursuant to a registered exchange offer. On May 3, 2002, AmeriGas Partners and AP Eagle Finance jointly and severally issued \$40,000,000 face amount of 8.875% Series B Senior Notes due May 2011. On December 3, 2002, AmeriGas Partners and AP Eagle Finance jointly and severally issued \$88,000,000 face amount of 8.875% Senior Notes due May 2011. On April 4, 2003, AmeriGas Partners and AP Eagle Finance exchanged (1) \$15,000 face amount of 8.875% Series A Senior Notes due May 2011 and (2) \$88,000,000 face amount of 8.875% Series A Senior Notes due May 2011 for like amounts of AmeriGas Partners and AP Eagle Finance 8.875% Series B Senior Notes due May 2011 pursuant to a registered exchange offer.

In April 2003, AmeriGas Partners and AP Eagle Finance jointly and severally issued \$32,000,000 face amount of 8.875% Series B Senior Notes due May 2011.

AmeriGas Partners owns all 100 shares of AP Finance common stock outstanding

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES SCHEDULE I - CONDENSED FINANCIAL INFORMATION OF REGISTRANT (PARENT COMPANY)

BALANCE SHEETS (Thousands of dollars)

	Septembe		
		2002	
ASSETS			
Current assets:			
Cash Accounts receivable	\$ 4,258 - - 4,258	\$	593 7,303
Total current assets	 4,258		7,896
Investment in AmeriGas Propane, L.P. Deferred charges	683,251 8,131		615,067 7,792
Total assets	\$ 695,640	\$ ==	630,755
LIABILITIES AND PARTNERS' CAPITAL			
Current liabilities:			
Accounts payable Accrued interest	\$ 1,192 14,506	\$	1,534 14,528
Total current liabilities	15,698		16,062
Long-term debt	426,259		386,327
Commitments and contingencies			
Partners' capital:	255,423 2,577 (4,317) 253,683		17,846 2,218 6,646
Total liabilities and partners' capital	695,640		630,755

Commitments and Contingencies:

Scheduled principal repayments of long-term debt for each of the next five fiscal years ending September 30 are as follows:

2004 - \$0; 2005 - \$0; 2006 - \$60,000; 2007 - \$0; 2008 - \$0.

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES SCHEDULE I - CONDENSED FINANCIAL INFORMATION OF REGISTRANT (PARENT COMPANY)

STATEMENTS OF OPERATIONS (Thousands of dollars)

Year Ended September 30,

	' '		
	2003	2002	2001
Operating expenses Loss on extinguishments of debt Interest expense	\$ (151) (3,023) (38,384)	\$ (49) (752) (35,171)	\$ (4) (15,722)
Loss before income taxes Income tax expense	(41,558) 61	(35,972)	
Loss before equity in income of AmeriGas Propane, L.P. Equity in income before accounting changes of AmeriGas Propane, L.P.	(41,619)	(36,060)	(15,726)
Income before equity in accounting changes of unconsolidated subsidiaries		55,366	53,015
Equity in accounting changes of AmeriGas Propane, L.P.	-	-	12,494
Net income	\$ 71,958 ======	\$ 55,366 ======	\$ 65,509 ======
General partner's interest in net income	\$ 720 ======	\$ 554 =======	\$ 655 ======
Limited partners' interest in net income	\$ 71,238	\$ 54,812	\$ 64,854
Income per limited partner unit - basic and diluted:	=======	=======	=======
Income before accounting changes Cumulative effect of accounting changes, net	\$ 1.42 -	\$ 1.12	\$ 1.18 0.28
Net income	\$ 1.42 ======	\$ 1.12 =======	\$ 1.46 ======
Average limited partner units outstanding - basic (thousands)	50,267	48,909 ======	44,453
Average limited partner units outstanding - diluted (thousands)	======= 50,337 =======		44,453 =======

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES SCHEDULE I - CONDENSED FINANCIAL INFORMATION OF REGISTRANT (PARENT COMPANY)

STATEMENTS OF CASH FLOWS (Thousands of dollars)

Year Ended September 30, -----2001 2003 2002 \$ 99,169 NET CASH PROVIDED BY OPERATING ACTIVITIES (a) \$ 112,010 \$ 123,761 CASH FLOWS FROM INVESTING ACTIVITIES: Contributions to AmeriGas Propane, L.P. (108, 513)(97,693) (294,357) Net cash used by investing activities (108, 513)(97,693) (294, 357) CASH FLOWS FROM FINANCING ACTIVITIES: Distributions (111, 462)(108, 504)(98, 435)Issuance of long-term debt 122,780 40,900 252,833 Repayments of long-term debt (86,913) (15,000) Proceeds from issuance of Common Units 75,005 56,556 39,836 571 Capital contribution from General Partner 758 956 Net cash provided (used) by financing activities 168 (25,477) 195,190 Increase in cash and cash equivalents \$ 3,665 591 -----CASH AND CASH EQUIVALENTS: End of year 4,258 593 2 Beginning of year 593 2 Increase \$ 3,665 \$ 591 \$ 2 =======

⁽a) Includes distributions received from AmeriGas Propane, L.P. of \$142,935, \$152,004, and \$111,744, for the years ended September 30, 2003, 2002, and 2001, respectively.

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES

SCHEDULE II - VALUATION AND QUALIFYING ACCOUNTS (Thousands of dollars)

	Balance at beginning of year	Charged (credited) to costs and expenses		(credited) to costs and		(credited) to costs and		(credited) to costs and		(credited) to costs and		Other	Balance at end of year
YEAR ENDED SEPTEMBER 30, 2003													
Reserves deducted from assets in the consolidated balance sheet: Allowance for doubtful accounts	\$ 7,588 ======	\$	9,046	\$ (7,442)(1)	\$ 9,192 ======								
Other reserves:													
Self-insured property and casualty liability	\$ 37,395 ======	\$	20,488	\$(12,027)(2)	\$ 45,856 ======								
Insured property and casualty liability	\$ 3,500 =====	\$	(2,805)	\$ (68)(3)	\$ 627 ======								
Environmental, litigation and other	\$ 12,999 ======	\$	2,525	\$ (3,610)(2)	\$ 12,358 ======								
				444 (3)									
YEAR ENDED SEPTEMBER 30, 2002													
Reserves deducted from assets in the consolidated balance sheet: Allowance for doubtful accounts	\$ 10,792 ======	\$	7,171	\$(10,375)(1)	\$ 7,588 ======								
Other reserves: Self-insured property and casualty liability	\$ 31,668 ======	\$	16,739	\$(11,012)(2)	\$ 37,395 ======								
Insured property and casualty liability	\$ 1,466 ======	\$	-	\$ 2,034 (3)	\$ 3,500 ======								
Environmental, litigation and other	\$ 10,629 ======	\$	3,468	\$ (2,387)(2)	\$ 12,999 ======								
				1,289 (3)									

AMERIGAS PARTNERS, L.P. AND SUBSIDIARIES

SCHEDULE II - VALUATION AND QUALIFYING ACCOUNTS (CONTINUED) (Thousands of dollars)

	Balance at beginning of year	Charged (credited) to costs and expenses	Other	Balance at end of year
YEAR ENDED SEPTEMBER 30, 2001				
Reserves deducted from assets in the consolidated balance sheet:				
Allowance for doubtful accounts	\$ 6,529 ======	\$ 7,497	\$ (3,234) (1)	\$ 10,792 ======
Other reserves:				
Self-insured property and casualty liability	\$ 32,036 ======	\$ 12,837	\$(13,205) (2)	\$ 31,668 ======
Insured property and casualty liability	\$ 2,068 ======	\$ (602)	\$ -	\$ 1,466 ======
Environmental, litigation and other	\$ 11,366 ======	\$ 1,389	\$ (2,126) (2)	\$ 10,629 ======

⁽¹⁾ Uncollectible accounts written off, net of recoveries.

⁽²⁾ Payments, net of any refunds

⁽³⁾ Other adjustments, primarily reclasses

REPORT OF INDEPENDENT AUDITORS

To the Partners of AmeriGas Partners, L.P. and the Board of Directors of AmeriGas Propane, Inc.:

In our opinion, the consolidated financial statements listed in the index appearing under Item 15a (1) and (2) present fairly, in all material respects, the financial position of AmeriGas Partners, L.P. and its subsidiaries at September 30, 2003 and 2002, and the results of their operations and their cash flows for each of the two years in the period ended September 30, 2003 in conformity with accounting principles generally accepted in the United States of America. In addition, in our opinion, the financial statement schedules listed in the index appearing under Item 15a (1) and (2) present fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements. These financial statements and financial statement schedules are the responsibility of the management of AmeriGas Propane, Inc.; our responsibility is to express an opinion on these financial statements and financial statement schedules based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion. The consolidated financial statements of AmeriGas Partners, L.P. and its subsidiaries as of and for the year ended September 30, 2001 were audited by other independent accountants who have ceased operations. Those independent accountants expressed an unqualified opinion on those financial statements in their report dated November 16, 2001.

As discussed in Note 2 to the consolidated financial statements, the Partnership adopted Statement of Financial Accounting Standards No. 142, "Goodwill and Other Intangible Assets" in fiscal 2002.

As discussed above, the consolidated financial statements of AmeriGas Partners, L.P. and its subsidiaries for the year ended September 30, 2001, were audited by other independent auditors who have ceased operations. As described in Note 2, these financial statements have also been revised to include the transitional disclosures required by Statement of Financial Accounting Standards No. 142, "Goodwill and Other Intangible Assets" which was adopted by the Partnership as of October 1, 2001. We audited the transitional disclosures described in Note 2. In our opinion, the transitional disclosures for 2001 in Note 2 are appropriate. However, we were not engaged to audit, review or apply procedures to the 2001 consolidated financial statements of the Partnership other than with respect to such adjustments and, accordingly, we do not express an opinion or any other form of assurance on the 2001 consolidated financial statements taken as a whole.

PricewaterhouseCoopers LLP Philadelphia, Pennsylvania November 17, 2003 THIS REPORT IS A COPY OF THE PREVIOUSLY ISSUED ACCOUNTANT'S REPORT OF ARTHUR ANDERSEN LLP AND HAS NOT BEEN REISSUED BY ARTHUR ANDERSEN LLP.

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

To the Partners of AmeriGas Partners, L.P. and the Board of Directors of AmeriGas Propane, Inc.:

We have audited, in accordance with auditing standards generally accepted in the United States, the consolidated financial statements included in the AmeriGas Partners, L.P. annual report to unitholders for the year ended September 30, 2001, incorporated by reference in this Form 10-K, and have issued our report thereon dated November 16, 2001. Our report on the financial statements includes an explanatory paragraph with respect to the changes in the method of accounting for tank installation costs and nonrefundable tank fees and the adoption of the provisions of SFAS No. 133 as discussed in Notes 2 and 4 to the financial statements. Our audits were made for the purpose of forming an opinion on those basic consolidated financial statements taken as a whole. The schedules listed in the index on pages F-2 and F-3 are the responsibility of the management of AmeriGas Propane, Inc. and are presented for purposes of complying with the Securities and Exchange Commission's rules and are not part of the basic financial statements. These schedules have been subjected to the auditing procedures applied in the audits of the basic financial statements and, in our opinion, fairly state in all material respects the financial data required to be set forth therein in relation to the basic financial statements taken as a

ARTHUR ANDERSEN LLP

Philadelphia, Pennsylvania November 16, 2001

EXHIBIT INDEX

EXHIBIT NO.	DESCRIPTION
10.1	Credit Agreement dated as of August 28, 2003 among AmeriGas Propane, L.P., AmeriGas Propane, Inc., Petrolane Incorporated, Citicorp USA, Inc., Credit Suisse First Boston, Wachovia Bank, National Association, as Issuing Bank and certain financial institutions
10.6	Notice of appointment of Wachovia Bank, National Association as Collateral Agent effective as of August 28, 2003, pursuant to Intercreditor and Agency Agreement dated as of April 19, 1995
10.19	Financing Agreement dated as of August 28, 2003 between AmeriGas Propane, Inc. and AmeriGas Propane, L.P.
13	Pages 10 to 27 of the AmeriGas Partners, L.P. 2003 Annual Report for the year ended September 30, 2003
14	Code of Ethics for principal executive, financial and accounting officers
21	Subsidiaries of AmeriGas Partners, L.P.
23	Consent of PricewaterhouseCoopers LLP
31.1	Certification by the Chief Executive Officer pursuant to Section 302 the Sarbanes-Oxley Act
31.2	Certification by the Chief Financial Officer pursuant to Section 302 the Sarbanes-Oxley Act
32	Certification of Chief Executive Officer and Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act

CREDIT AGREEMENT

Dated as of August 28, 2003

among

AMERIGAS PROPANE, L.P., as Borrower,

AMERIGAS PROPANE, INC., as a Guarantor,

PETROLANE INCORPORATED, as a Guarantor,

CITICORP USA, INC., as Syndication Agent,

CREDIT SUISSE FIRST BOSTON, acting through its Cayman Islands Branch, as Documentation Agent,

WACHOVIA BANK, NATIONAL ASSOCIATION, as Agent, Issuing Bank and Swing Line Bank,

and

THE OTHER FINANCIAL INSTITUTIONS PARTY HERETO

WACHOVIA CAPITAL MARKETS, LLC, Lead Arranger and Book Manager

Credit Agreement Table of Contents

			Page
ARTICLE I	DEFINIT	IONS	1
1	1.1	Certain Defined Terms	1
1	1.2	Other Interpretive Provisions	30
1	1.3	Accounting Principles	31
ARTICLE TI	T THE CRE	EDITS	31
	2.1	Amounts and Terms of Commitments	31
_	2.2	Loan Accounts.	32
_	2.3	Procedure for Borrowing.	33
	2.4	Conversion and Continuation Elections	34
	2.5	Voluntary Termination or Reduction of Commitments.	35
=	2.6	Optional Prepayments	35
=	2.0	Mandatory Prepayments of Loans; Mandatory Commitment Reductions	36
_	2. <i>1</i> 2.8	Page was the payments of Loans, Manuatory Committeent Reductions	36
_		Repayment	
	2.9	Interest	37
_	2.10	Fees.	38
	2.11	Computation of Fees and Interest	39
_	2.12	Payments by the Borrower	40
_	2.13	Payments by the Banks to the Agent, etc	40
_	2.14	Sharing of Payments, etc	41
_	2.15	Revolving Termination Date	41
2	2.16	Swing Line Loans	42
ARTICLE I	II THE LE	TTERS OF CREDIT	45
3	3.1	The Letter of Credit Subfacility	45
3	3.2	Issuance, Amendment and Renewal of Letters of Credit	46
3	3.3	Risk Participations, Drawings and Reimbursements	48
3	3.4	Repayment of Participations	49
3	3.5	Role of the Issuing Bank	49
3	3.6	Obligations Absolute	50
3	3.7	Cash Collateral Pledge	51
3	3.8	Letter of Credit Fees	51
3	3.9	Applicability of ISP98 and UCP	52
	3.10	Conflict with L/C Application	52
ADTTOLE TV	/ TAVES	YIELD PROTECTION AND ILLEGALITY	52
	4.1	Taxes	52
	4.1 4.2		54
	4.2 4.3	Illegality	54 54
	4.4	Funding Losses	55
	4.5	Inability to Determine Rates	56
	4.6	Certificates of Banks	56
4	4.7	Substitution of Banks	57

4.8	Survival	57
V CONDITT	ONS PRECEDENT	57
		57
		60
5.2	CONDITIONS to ALL BUTTOWINGS	00
VI REPRES	ENTATIONS AND WARRANTIES	61
		61
		61
		62
		62
		62
		63
		63
6.8		63
6.9	Litigation, etc	64
6.10	Compliance with Other Instruments, etc	64
6.11	Governmental Consent	65
		65
		65
	Chief Eventive Office	65
		65
		66
		66
		66
	Environmental Warranties	66
6.20	Copyrights, Patents, Trademarks and Licenses, etc	68
6.21	Insurance	68
6.22	Full Disclosure	68
6.23	Solvency	68
6 24		69
		69
		69
		70
		70
6.29	Tax Disclosure.	70
VII AFFIR	MATIVE COVENANTS	71
		71
		75
		75
	Partnership or Corporate Evictonece Ducinese: Compliance with Laws	75
		75
		76
		76
		76
7.9	Guarantors	76
	V CONDITI 5.1 5.2 VI REPRES 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17 6.18 6.19 6.20 6.21 6.22 6.23 6.24 6.25 6.26 6.27 6.28 6.29	VCONDITIONS PRECEDENT. 5.1 Conditions to Effectiveness. 5.2 Conditions to All Borrowings. VI REPRESENTATIONS AND WARRANTIES. 6.1 Organization, Standing, etc. 6.2 Partnership Interests and Subsidiaries. 6.3 qualification; Corporate or Partnership Authorization. 6.4 Financial Statements. 6.5 Changes, etc. 6.6 Tax Returns and Payments. 6.7 Indebtedness. 6.8 Transfer of Assets and Business. 6.9 Litigation, etc. 6.10 Compliance with Other Instruments, etc. 6.11 Governmental Consent. 6.12 Investment Company Act. 6.13 Public Utility Holding Company Act. 6.14 Chief Executive Office. 6.15 Matters Relating to Petrolane. 6.16 Matters Relating to the General Partner. 6.17 ERISA Compliance. 6.18 Use of Proceeds; Margin Regulations. 6.19 Environmental Warranties. 6.20 Copyrights, Patents, Trademarks and Licenses, etc. 6.21 Insurance. 6.22 Full Disclosure. 6.23 Solvency. 6.24 Title Insurance. 6.25 Mortgages. 6.26 Copyrights, Patents, Trademarks and Licenses, etc. 6.27 PPD/GP Debt Contributions. 6.28 State Mortgage acc. 6.29 Tax Disclosure. 6.20 Tax Disclosure. 6.21 A Transurance. 6.22 Full Disclosure. 6.23 Solvency. 6.24 Title Insurance. 6.25 Mortgages. 6.26 Security Interest; Parity Debt. 6.27 PPD/GP Debt Contributions. 6.28 State Mortgage Taxes. 6.29 Tax Disclosure. 6.29 Tax Disclosure. 6.20 Reserved. 7.1 Financial Statements. 7.2 Reserved. 7.3 Adequate Reserves. 7.4 Partnership or Corporate Existence; Business; Compliance with Laws. 7.5 Payment of Taxes and Claims. 7.6 Maintenance of Properties; Insurance. 7.7 License Agreements. 7.8 Chief Executive Office.

	7.10	New Mortgages; Recordation	76
	7.11	Further Assurances	77
	7.12	Covenant to Secure Notes Equally	77
	7.13	Designations With Respect to Subsidiaries	77
	7.14	Covenants of the General Partner and Petrolane	78
	7.15	Books and Records	79
	7.16	Environmental Covenant	80
	7.17	Post Closing.	80
	7.18	Tax Disclosure	80
ARTICLE	VIII NEGA	TIVE COVENANTS	80
	8.1	Indebtedness	81
	8.2	Minimum Interest Coverage	83
	8.3	Liens, etc	83
	8.4	Investments, Contingent Obligations, etc	86
	8.5	Restricted Payments	88
	8.6	Transactions with Affiliates	89
	8.7	Subsidiary Stock and Indebtedness	89
	8.8	Consolidation, Merger, Sale of Assets, etc.	90
	8.9	Use of Proceeds.	94
	8.10	Change in Business	94
	8.11	Accounting Changes.	94
	8.12	Clean Down.	95
	8.13	Receivables.	95
	8.14	Leverage Ratio	95
	8.15	Acquisitions	95
	8.16	Limitation on Restricted Agreements.	95
	8.17	AEPLP	96
	0.17	ALI LI	30
ARTICLE	TX EVENTS	OF DEFAULT	98
WITOLL	9.1	Event of Default	98
	9.2	Remedies	101
	9.3	Rights Not Exclusive	102
	9.4	Application of Funds	102
	3.4	Application of funds	102
APTTCLE	Y THE AGE	NT	103
TITULL	10.1	Appointment and Authorization	103
	10.2	Delegation of Duties	103
	10.2	Liability of Agent	103
	10.3	Reliance by Agent	103
	10.4	Notice of Default	104
	10.5	Credit Decision	104
			104
	10.7 10.8	Indemnification	105
		Agent in Individual Capacity	
	10.9	Successor Agent	106
	10.10	Agent May File Proofs of Claim	106
	10.11	Collateral and Guaranty Matters	107

	10.12 10.13	Other Agents; Arrangers and Managers	107 108
ARTICLE	XT GHARAN	TEE	109
AUCTOLL	11.1	Each Guaranteed Obligation	109
	11.2	Obligations Exclusive	109
	11.3	Obligations Independent	110
	11.4	Waiver of Notice	110
	11.5	Guarantee of Payment	110
	11.6	Obligations Unconditional	111
	11.7	Continuing Guarantee	111
	11.8	Subordination	112
	11.9	Exhaustion of Remedies	112
	11.10	Reinstatement	114
ARTICLE	XII MISCEI	LLANEOUS	114
	12.1	Amendments and Waivers	114
	12.2	Notices and Other Communications; Facsimile Copies	116
	12.3	No Waiver; Cumulative Remedies	116
	12.4	Costs and Expenses	117
	12.5	Indemnity	117
	12.6	Liability	118
	12.7	Payments Set Aside	119
	12.8	Successors and Assigns	120
	12.9	Assignments, Participations. etc	120
	12.10	Confidentiality	122
	12.11	Set-off	123
	12.12	Notification of Addresses; etc	123
	12.13	Counterparts	124
	12.14	Severability	124
	12.15	No Third Parties Benefited	124
	12.16	Governing Law and Jurisdiction	124
	12.17	Waiver of Jury Trial	124
	12.18	Entire Agreement	125
	12.19	Collateral Agency Agreement	125
	12.20	Ratification and Confirmation of the Security Documents	125
	12.21	Interest Rate Limitation	125
	12.22	Survival of Representations and Warranties	126

Table of Contents

Schedules

Schedule 2.1: Commitments and Percentages

Schedule 3.1(a): Existing Letters of Credit

Schedule 5.1(p): Mortgage Amendment Sites

Schedule 6.2: Partnership Interests and Subsidiaries

Schedule 6.3: Foreign Qualifications Schedule 6.7: Permitted Indebtedness

Schedule 6.8(a): Permits and Consents

Schedule 6.9: Litigation

Schedule 6.15: Matters Relating to Petrolane

Schedule 6.16: Matters Relating to General Partner

Schedule 6.19: **Environmental Liabilities**

Schedule 6.20: Copyrights, Patents, Trademarks and Licenses, Etc.

Schedule 6.24: Real Property with Title Insurance or Value in Excess

of \$1,000,000

Real Property Owned by the Borrower or a Restricted Subsidiary and Subject to a Mortgage $\,$ Schedule 6.25:

Schedule 8.4(i): Investments in Atlantic Energy, Inc.

Schedule 12.2: Offshore and Domestic Lending Office, Addresses for

Notices

Exhibits

Exhibit A-1: Notice of Borrowing (Swing Line Loans)

Exhibit A-2: Notice of Borrowing (Revolving Loans/Acquisition

Loans)

Exhibit B: Notice of Conversion/ Continuation

Form of Commitment Termination Date Extension Request Exhibit C:

Exhibit D: Form of Compliance Certificate

Exhibit E: Form of Mortgage

Exhibit F: Form of Assignment and Acceptance Agreement

Exhibit G-1: Form of Promissory Note (Acquisition Loans)

Exhibit G-2: Form of Promissory Note (Revolving Loans)

Exhibit H: Form of Post-Closing Agreement

Exhibit I: Form of Subordination Provisions

CREDIT AGREEMENT

This CREDIT AGREEMENT (as the same may be amended, supplemented, assigned or otherwise modified from time to time in accordance with the terms hereof, this "AGREEMENT"), dated as of August 28, 2003, among AMERIGAS PROPANE, L.P., a belaware limited partnership (the "BORROWER"), AMERIGAS PROPANE, INC., a Pennsylvania corporation (the "GENERAL PARTNER"), PETROLANE INCORPORATED, a Pennsylvania corporation ("PETROLANE"; the General Partner and Petrolane are, on a joint and several basis, the "GUARANTORS"; the Borrower, the General Partner and Petrolane are, on a joint and several basis, the "OBLIGORS"), CITICORP USA INC., as Syndication Agent, CREDIT SUISSE FIRST BOSTON, acting through its Cayman Islands Branch, as Documentation Agent, the several financial institutions from time to time party to this Agreement (collectively, the "BANKS"; individually, a "BANK") and WACHOVIA BANK, NATIONAL ASSOCIATION, as administrative agent for the Banks (the "AGENT"), issuing bank and swing line bank.

WHEREAS, the Obligors have requested that (i) the \$100,000,000 of Revolving Commitments under (and as defined in) the Existing Credit Agreement and the related Revolving Loans outstanding under (and as defined in) the Existing Credit Agreement be terminated and replaced with the \$100,000,000 of Revolving Commitments and Revolving Loans under this Agreement, the proceeds of which are to be used by the Borrower for working capital and general purposes of the Borrower; (ii) the \$75,000,000 of Acquisition Commitments under (and as defined in) the Existing Credit Agreement and the related Acquisition Loans and Specified Acquisition Loans outstanding under (and as defined in) the Existing Credit Agreement be terminated and replaced with the \$75,000,000 of Acquisition Commitments and Acquisition Loans and Specified Acquisition Loans under this Agreement, the proceeds of which are to be used by the Borrower to finance acquisitions, or with respect to Specified Acquisition Loans, for working capital and general purposes of the Borrower and (iii) the Existing Credit Agreement otherwise be paid in full and the commitments thereunder be terminated; and

WHEREAS, the Banks are willing, on the terms and subject to the conditions set forth in this Agreement, to enter into, and to extend credit under, this Agreement as more particularly hereinafter set forth.

NOW, THEREFORE, in consideration of the mutual agreements, provisions and covenants contained herein, the parties agree as follows:

ARTICLE I

DEFINITIONS

- "ACQUIRED DEBT" means with respect to any specified Person, (i) Indebtedness of any other Person existing at the time such other Person merged with or into or became a Subsidiary of such specified Person, including Indebtedness incurred in connection with, or in contemplation of, such other Person merging with or into or becoming a Subsidiary of such specified Person and (ii) Indebtedness encumbering any asset acquired by such specified Person.

"ACQUISITION" means, as to any Person, any acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of capital stock or other securities of another Person, (b) a loan, advance or capital contribution to, guaranty of debt of, or purchase or other acquisition of any other debt or equity participation or interest in, another Person, including any partnership or joint venture interest in such other Person, or (c) an Asset Acquisition.

"ACQUISITION COMMITMENT" has the meaning specified in Section 2.1(a).

"ACQUISITION LOAN" has the meaning specified in Section

2.1(a).

of:

"ACQUISITION LOAN TERMINATION DATE" means the earlier to occur

(a) October 15, 2006; and

(b) the date on which the Acquisition Commitments terminate in accordance with the provisions of this Agreement.

"AEPH" means AmeriGas Eagle Holdings, Inc. (formerly CP Holdings, Inc.), a Delaware corporation.

"AEPI" means AmeriGas Eagle Propane, Inc. (formerly Columbia Propane Corporation), a Delaware corporation.

"AEPLP" means AmeriGas Eagle Propane, L.P. (formerly Columbia Propane, L.P.), a Delaware limited partnership.

"AEPLP ACQUISITIONS" has the meaning specified in Section

8.15.

"AEPLP AVAILABLE DATE" means the earliest of (i) 180 days after the expiration of the Debt Indemnity provided under the National Propane Purchase Agreement, (ii) the purchase by AEPLP of the partnership interest of the Special Limited Partner (as defined in the AEPLP Partnership Agreement) in AEPLP pursuant to the Special Limited Partner's put option under Section 4.5 of the AEPLP Partnership Agreement and (iii) the purchase by AEPLP of the partnership interest of the Special Limited Partner in AEPLP pursuant to AEPLP's call option under Section 4.5 of the AEPLP Partnership Agreement.

"AEPLP PARTNERSHIP AGREEMENT" means that certain Amended and Restated Agreement of Limited Partnership of National Propane, L.P. (renamed AEPLP), dated as of July 19, 1999, by and among AEPI, AEPH, and National Propane Corporation, as amended, supplemented, or otherwise modified from time to time.

"AEPLP SECURITY DATE" has the meaning specified in Section $% \left(1\right) =\left(1\right) \left(1\right) \left$

Section 8.17.

8.17.

"AEPLP SUBSIDIARY GUARANTY" has the meaning specified in

"AEPLP TAXES" means all federal, state, local or foreign taxes, governmental fees or like charges of any kind whatsoever, whether disputed or not.

"AFFECTED BANK" has the meaning specified in Section 4.7.

"AFFILIATE" means, as to any Person, any other Person which, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. A Person shall be deemed to control another Person if the controlling Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of the other Person, whether through the ownership of voting securities, by contract, or otherwise.

"AGENT" means Wachovia in its capacity as administrative agent for the Banks hereunder, and any successor agent arising under Section 10.9.

"AGENT-RELATED PERSONS" means the Agent, together with its Affiliates (including, in the case of Wachovia in its capacity as the Agent, the Arranger), and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.

"AGENT'S PAYMENT OFFICE" means the address for payments set forth on Schedule 12.2 hereto in relation to the Agent, or such other address as the Agent may from time to time specify by written notice to the Borrower and the Banks.

"AGREEMENT" has the meaning specified in the introductory clause hereto. $% \left(1\right) =\left(1\right) \left(1\right) \left$

"AMERIGAS EAGLE PARTS & SERVICE" means AmeriGas Eagle Parts & Service, Inc., a Delaware corporation.

"ANNUAL LIMIT" has the meaning specified in Section 8.4(c).

"APPLICABLE MARGIN" means

- (i) with respect to Base Rate Loans, 0%; and
- (ii) with respect to Eurodollar Rate Loans,

the applicable margin set forth below at such time as a Pricing Tier (the "PRICING TIER") set forth below is applicable:

Pricing Tier	Funded Debt Ratio	Margin
I	Less than or equal to 1.75x	1.0000%
II	Greater than 1.75x but less than or equal to 2.75x	1.2500%
III	Greater than 2.75x but less than or equal to 3.25x	1.3750%
IV	Greater than 3.25x but less than or equal to 3.75x	1.6250%

V	Greater than 3.75x but less than or equal to 4.25x	1.7500%
VI	Greater than 4.25x but less than or equal to 4.75x	2.0000%
VII	Greater than 4.75	2.2500%

For the purpose of determining the applicable Pricing Tier above and subject to the last sentence of this paragraph, EBITDA shall be determined as at the end of each fiscal quarter for the four fiscal quarters then ending and Funded Debt shall be determined as at the end of each fiscal quarter for which such calculation is being determined. Pricing changes shall be effective on the later of (i) 45 days after the end of each of the first three fiscal quarters of each fiscal year and 90 days after each fiscal year end and (ii) the Agent's receipt of financial statements hereunder for such fiscal quarter or fiscal year; provided, however, that if the financial statements are not delivered when due in accordance with Section 7.1, then Pricing Tier VII shall apply as of the first Business Day after the date on which such financial statements were required to have been delivered until the date upon which such financial statements are delivered to the Agent. For the period from the Closing Date through December 29, 2003, the applicable Pricing Tier shall be determined by reference to the Compliance Certificate delivered pursuant to Section 5.1(k).

"ARRANGER" means Wachovia Capital Markets, LLC.

"ASSET ACQUISITION" means (a) an Investment by the Borrower or any Restricted Subsidiary in any other Person pursuant to which such Person shall become a Restricted Subsidiary or shall be merged with or into the Borrower or any Restricted Subsidiary, (b) the acquisition by the Borrower or any Restricted Subsidiary of the assets of any Person (other than a Restricted Subsidiary) which constitute all or substantially all of the assets of such Person or (c) the purchase or other acquisition by the Borrower or any Restricted Subsidiary (in one or a series of transactions) of any division or line of business of any Person (other than a Restricted Subsidiary).

"ASSET SALE" has the meaning specified in Section 8.8(c).

"ASSETS" means the assets owned by, licensed to, leased or otherwise used in the business by the Borrower and its Subsidiaries.

"ASSIGNEE" has the meaning specified in Section 12.9(a).

"ATLANTIC ENERGY" means Atlantic Energy, Inc., a Delaware

corporation.

"ATTORNEY COSTS" means and includes all reasonable fees and disbursements of any law firm or other external counsel, the reasonable allocated cost of internal legal services and all reasonable disbursements of internal counsel.

"AUTO-RENEWAL LETTER OF CREDIT" has the meaning specified in Section 3.2(d).

- (a) the sum of (i) all cash of the Borrower and the Restricted Subsidiaries on hand at the end of such quarter and (ii) all additional cash of the Borrower and the Restricted Subsidiaries on hand on the date of determination of Available Cash with respect to such quarter resulting from borrowings subsequent to the end of such quarter, less
- the amount of cash reserves that is necessary or appropriate in the reasonable discretion of the General Partner to (i) provide for the proper conduct of the business of the Borrower and the Restricted Subsidiaries (including reserves for future capital expenditures) subsequent to such quarter, (ii) provide funds for distributions under Sections 5.3(a), (b) and (c) or 5.4(a) of the partnership agreement of the Public Partnership (such Sections as in effect on the Closing Date, together with all related definitions, being hereby incorporated herein in the form included in such partnership agreement on the Closing Date and without regard to any subsequent amendments or waivers of the provisions of, or any termination of, such partnership agreement) in respect of any one or more of the next four quarters, or (iii) comply with applicable law or any debt instrument or other agreement or obligation to which the Borrower or any Restricted Subsidiary is a party or its assets are subject; provided, however, that Available Cash attributable to any Restricted Subsidiary shall be excluded to the extent dividends or distributions of such Available Cash by such Restricted Subsidiary are not at the date of determination permitted by the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or other regulation.

In addition, without limiting the foregoing, Available Cash shall reflect a reserve equal to 50% of the interest to be paid on the First Mortgage Notes and the Acquisition Loans in the next fiscal quarter and, beginning with a date three fiscal quarters before a scheduled principal payment date on the First Mortgage Notes, the Revolving Loans or the Acquisition Loans, 25% of the aggregate principal amount thereof due on any such payment date in the third succeeding fiscal quarter, 50% of the aggregate principal amount due on any such quarterly payment date in the second succeeding fiscal quarter and 75% of the aggregate principal amount due on any quarterly payment date in the next succeeding fiscal quarter on such notes and facilities. The foregoing reserves for principal amounts to be paid shall be reduced by the aggregate amount of advances available to the Borrower from responsible financial institutions under binding, irrevocable credit facility commitments (and which are subject to no conditions which the Borrower is unable to meet) and letters of credit to be used to refinance such principal (so long as no repayment obligations under such credit facilities and no reimbursement obligation with respect to any such letter of credit would come due within three quarters).

"AVERAGE CONSOLIDATED PRO FORMA DEBT SERVICE" means as of any date of determination, the average amount payable by the Borrower and the Restricted Subsidiaries on a consolidated basis during all periods of four consecutive calendar quarters, commencing with the calendar quarter in which such date of determination occurs and ending June 30, 2010, in respect of scheduled interest (but not principal) payments with respect to all Indebtedness of the Borrower and the Restricted Subsidiaries outstanding on such date of determination, after giving effect to any Indebtedness proposed on such date to be incurred and to the substantially concurrent repayment of any other Indebtedness (a) including actual payments of Capitalized Lease Liabilities, (b) assuming, in the case of Indebtedness (other than Indebtedness referred to

in clause (c) below) bearing interest at fluctuating interest rates which cannot be determined in advance, that the rate actually in effect on such date will remain in effect throughout such period, and (c) including only actual interest payments associated with the Indebtedness incurred pursuant to Section 8.1(e) during the most recent four consecutive calendar quarters.

 $$\operatorname{\mathtt{"BANK"}}$$ has the meaning specified in the introductory clause hereto.

"BANKRUPTCY CODE" means the Federal Bankruptcy Reform Act of 1978, as amended (11 U.S.C. Section 101, et seq.).

"BASE RATE" means for any day a fluctuating rate per annum equal to the higher of (a) the Federal Funds Rate plus 1/2 of 1% and (b) Prime Rate.

"BASE RATE LOAN" means a Loan that bears interest based on the Base Rate.

"BOFA" means Bank of America, N.A.

"BORROWER" has the meaning specified in the introductory

clause hereto.

"BORROWER FINANCIALS" has the meaning specified in Section

7.1.

"BORROWER'S ACCOUNT" means the account maintained by the Borrower with Mellon Bank, N.A. and designated as account number 094-0764 or such other account designated by the Borrower in writing.

"BORROWING" means a borrowing hereunder consisting of Loans of the same Type made to the Borrower on the same day by the Banks (or in the case of Swing Line Loans, by the Swing Line Bank) and, in the case of Eurodollar Rate Loans, having the same Interest Period, in either case under Article II.

"BORROWING DATE" means any date on which a Borrowing occurs under Section 2.3. $\,$

"BUSINESS" means the business of wholesale and retail sales, distribution and storage of propane gas and related petroleum derivative products and the retail sale of propane related supplies and equipment, including home appliances.

"BUSINESS DAY" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the laws of, or are in fact closed in, the state where the Agent's Payment Office is located and, if such day relates to any Eurodollar Rate Loan, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"CAPITAL ADEQUACY REGULATION" means any guideline, request or directive of any central bank or other Governmental Authority, or any other law, rule or regulation, whether or not having the force of law, in each case, regarding capital adequacy of any bank or of any corporation controlling a bank.

"CAPITAL STOCK" means with respect to any Person, any and all shares, interests, participations, rights in or other equivalents (however designated) of such Person's capital stock, including, with respect to partnerships and limited liability companies, partnership interests (whether general or limited) or membership interests and any other interest or participation that confers upon a Person the right to receive a share of the profits and losses of, or distributions of assets of, such partnership or limited liability company, and any rights (other than debt securities convertible into capital stock), warrants or options exchangeable for or convertible into such capital stock.

"CAPITALIZED LEASE LIABILITIES" means all monetary obligations of the Borrower or any of its Restricted Subsidiaries under any leasing or similar arrangement which, in accordance with GAAP, would be classified as capitalized leases, and, for purposes of this Agreement and each other Loan Document, the amount of such obligations shall be the principal components thereof.

"CAPPED INVESTMENTS" has the meaning specified in Section

8.17(a).

"CARRYOVER THRESHOLD" has the meaning specified in Section

8.15.

"CASH COLLATERALIZE" means to pledge and deposit with or deliver to the Agent, for the benefit of the Agent, the Issuing Bank and the Banks, as additional collateral for the L/C Obligations, cash or deposit account balances pursuant to documentation in form and substance reasonably satisfactory to the Agent and the Issuing Bank (which documents are hereby consented to by the Banks). Derivatives of such term shall have corresponding meanings.

"CASH EQUIVALENTS" has the meaning specified in Section

8.4(a).

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.

"CHANGE OF CONTROL" means (i) UGI shall fail to own directly or indirectly 100% of the general partnership interests in the Borrower, or, if the Borrower shall have been converted to a corporate form, at least 51% of the voting shares of the Borrower; or (ii) UGI shall fail to own directly or indirectly at least a 30% ownership interest in the Borrower.

"CLOSING DATE" means the date on which all conditions precedent set forth in Section $5.1\ \mathrm{are}$ satisfied or waived by the Banks.

"CODE" means the Internal Revenue Code of 1986, as amended, and regulations promulgated thereunder, in each case as in effect from time to time.

"COLLATERAL AGENCY AGREEMENT" means the Intercreditor and Agency Agreement dated as of April 19, 1995 among the Obligors, the Restricted Subsidiaries, the Agent, the holders of the Series A-C First Mortgage Notes and the Collateral Agent, as the same may be amended, supplemented, assigned or otherwise modified from time to time.

"COLLATERAL AGENT" means Wachovia and its successors and/or assigns, in its capacity as Collateral Agent under the Security Documents.

"COLUMBIA ACQUISITION" means the acquisition by the Borrower of the propane distribution business of Columbia Energy Group, a Delaware corporation, pursuant to the Columbia Purchase Agreement.

"COLUMBIA PURCHASE AGREEMENT" means that certain Purchase Agreement, dated as of January 30, 2001, and amended and restated on August 7, 2001 by and among Columbia Energy Group, a Delaware corporation, AEPI, AEPLP, the Borrower, the Public Partnership and the General Partner, as amended, supplemented or otherwise modified from time to time.

 $\mbox{\tt "COMMITMENT"},$ as to each Bank, means its Revolving Commitment and its Acquisition Commitment.

"COMMITMENT LETTER" shall mean that certain letter, dated June 16, 2003, among the Borrower, the Agent and the Arranger.

"COMMITMENT TERMINATION DATE EXTENSION REQUEST" means a request substantially in the form of Exhibit C.

"COMPLIANCE CERTIFICATE" means a certificate substantially in the form of Exhibit D.

"CONSOLIDATED CASH FLOW" means with respect to the Borrower and the Restricted Subsidiaries for any period, (1) the sum of, without duplication, the amounts for such period, taken as a single accounting period, of (a) Consolidated Net Income, (b) Consolidated Non-Cash Charges, (c) Consolidated Interest Expense and (d) Consolidated Income Tax Expense less (2) any non-cash items increasing Consolidated Net Income for such period that had previously been added to Consolidated Net Income when incurred as a Consolidated . Non-Cash Charge. Consolidated Cash Flow shall be calculated after giving effect, on a pro forma basis for the four full fiscal quarters immediately preceding the date of the transaction giving rise to the need to calculate Consolidated Cash Flow, to, without duplication, any Asset Sales or Asset Acquisitions (including without limitation any Asset Acquisition giving rise to the need to make such calculation as a result of the Borrower or one of its Restricted Subsidiaries (including any Person who becomes a Restricted Subsidiary as a result of the Asset Acquisition) incurring, assuming or otherwise being liable for Acquired Debt) occurring during the period commencing on the first day of such period to and including the date of the transaction (the "REFERENCE PERIOD"), as if such Asset Sale or Asset Acquisition occurred on the first day of the Reference Period; provided, however, that Consolidated Cash Flow generated by an acquired business or asset shall be determined by the actual gross profit (revenues minus cost of goods sold) of such acquired business or asset during the immediately preceding four full fiscal quarters in the Reference Period minus the pro forma expenses that would have been incurred by the Borrower and the Restricted Subsidiaries in the operation of such acquired business or asset during such period computed on the basis of personnel expenses for employees retained or to be retained by the Borrower and the Restricted Subsidiaries in the operation of such acquired business or asset and non-personnel costs and expenses incurred by the Borrower and the Restricted Subsidiaries in the operation of the Borrower's business at similarly situated Borrower facilities or Restricted Subsidiary facilities.

"CONSOLIDATED INCOME TAX EXPENSE" means with respect to the Borrower and the Restricted Subsidiaries for any period, the provision for federal, state, local and foreign income taxes of the Borrower and the Restricted Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP.

"CONSOLIDATED INTEREST EXPENSE" means, with respect to the Borrower and the Restricted Subsidiaries for any period, without duplication, the sum of (i) the interest expenses of the Borrower and the Restricted Subsidiaries for such period as determined on a consolidated basis in accordance with GAAP, including without limitation (a) any amortization of debt discount, (b) the net cost under Interest Rate Agreements, (c) the interest portion of any deferred payment obligation, (d) all commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing and (e) all accrued interest plus (ii) the interest component of capital leases paid, accrued or scheduled to be paid or accrued by the Borrower and the Restricted Subsidiaries during such period as determined on a consolidated basis in accordance with GAAP.

"CONSOLIDATED NET INCOME" means the net income of the Borrower and the Restricted Subsidiaries, as determined on a consolidated basis in accordance with GAAP and as adjusted to exclude (i) net after-tax extraordinary gains or losses, (ii) net after-tax gains or losses attributable to Asset Sales, (iii) the net income or loss of any Person which is not a Restricted Subsidiary and which is accounted for by the equity method of accounting, provided, that Consolidated Net Income shall include the amount of dividends or distributions actually paid to the Borrower or any Restricted Subsidiary, (iv) the net income of any Restricted Subsidiary to the extent that dividends or distributions of such net income are not at the date of determination permitted by the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or other regulation and (v) the cumulative effect of any changes in accounting principles.

"CONSOLIDATED NET WORTH" means, of any Person, at any date of determination, the total partners' equity (in the case of a partnership), total stockholders' equity (in the case of a corporation) or total membership interests (in the case of a limited liability company) of such Person at such date, as would be shown on a balance sheet (consolidated, if applicable) of such Person and, if applicable, its Subsidiaries (Restricted Subsidiaries in the case of the Borrower) prepared in accordance with GAAP (less, in the case of the Borrower, the Net Amount of Unrestricted Investment as of such date).

"CONSOLIDATED NON-CASH CHARGES" means, with respect to the Borrower and the Restricted Subsidiaries for any period, the aggregate depreciation, amortization and any other non-cash charges resulting in write downs in non-current assets, in each case reducing Consolidated Net Income of the Borrower and the Restricted Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP.

"CONSOLIDATED PRO FORMA DEBT SERVICE" means as of any date of determination, the total amount payable by the Borrower and the Restricted Subsidiaries on a consolidated basis during the four consecutive calendar quarters next succeeding the date of determination, in respect of scheduled interest (but not principal) payments with respect to Indebtedness of the Borrower and the Restricted Subsidiaries outstanding on such date of determination, after giving

effect to any Indebtedness proposed on such date to be incurred and to the substantially concurrent repayment of any other Indebtedness (a) including actual payments of Capitalized Lease Liabilities, (b) assuming, in the case of Indebtedness (other than Indebtedness referred to in clause (c) below) bearing interest at fluctuating interest rates which cannot be determined in advance, that the rate actually in effect on such date will remain in effect throughout such period, and (c) including only actual interest payments associated with the Indebtedness incurred pursuant to Section 8.1(e) during the most recent four consecutive calendar quarters.

"CONTINGENT OBLIGATION" means, as to any Person, any direct or indirect liability of that Person, whether or not contingent, with or without recourse (otherwise than for collection or deposit in the ordinary course of business), (a) with respect to any Indebtedness, lease, dividend, letter of credit or other obligation (the "primary obligations) of another Person (the "primary obligor"), including any obligation of that Person (i) to purchase, repurchase or otherwise acquire such primary obligations or any security therefor, (ii) to advance or provide funds for the payment or discharge of any such primary obligation, or to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency or any balance sheet item, level of income or financial condition of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation, or (iv) otherwise to assure or hold harmless the holder of any such primary obligation against loss in respect thereof (each, a "GUARANTY OBLIGATION"); (b) with respect to any Surety Instrument (other than any Letter of Credit) issued for the account of that Person or as to which that Person is otherwise liable for reimbursement of drawings or payments; (c) to purchase any materials, supplies or other property from, or to obtain the services of, another Person if the relevant contract or other related document or obligation requires that payment for such materials, supplies or other property, or for such services, shall be made regardless of whether delivery of such materials, supplies or other property is ever made or tendered, or such services are ever performed or tendered, or (d) in respect of any Swap Contract. The amount of any Contingent Obligation shall, in the case of Guaranty Obligations, be deemed equal to the stated or determinable amount of the primary obligation in respect of which such Guaranty Obligation is made or, if not stated or if indeterminable, the maximum reasonably anticipated liability in respect thereof, and in the case of other Contingent Obligations, shall be equal to the maximum reasonably anticipated liability in respect thereof.

"CONTROL AFFILIATE" means UGI, the Public Partnership, the General Partner and any Person controlling or controlled by, or under common control with, UGI, the Public Partnership or the General Partner (other than the Borrower or any of its Subsidiaries).

"CONVERSION/CONTINUATION DATE" means any date on which, under Section 2.4, any Borrower (a) converts Loans of one Type to the other Type, or (b) continues as Eurodollar Rate Loans, but with a new Interest Period, Eurodollar Rate Loans having Interest Periods expiring on such date.

"COVERED PERSONS" shall have the meaning specified in the definition of Restricted Payment. $\,$

"CREDIT EXTENSION" means and includes (a) the making of any Loan hereunder, and (b) the Issuance of any Letters of Credit hereunder.

"CREDIT PARTIES" means the Obligors and any Restricted Subsidiary party to any Security Document.

7.13(a).

United States.

"DEBT INDEMNITY" means the indemnity provided by Triarc Companies, Inc. under Section 5.9 of the National Propane Purchase Agreement.

"DEFAULT" means any event or circumstance which, with the giving of notice, the lapse of time, or both, would (if not cured or otherwise remedied during such time) constitute an Event of Default.

"DESIGNATION AMOUNTS" has the meaning specified in Section

"DESIGNEE" has the meaning specified in Section 7.13(d).

"DISINTERESTED DIRECTORS" means, with respect to any transaction or series of transactions with Affiliates, a member of the Board of Directors of the General Partner who has no financial interest, and whose employer has no financial interest, in such transaction or series of transactions.

"DOLLARS", "DOLLARS" and "\$" each mean lawful money of the

"EBIT" means, for any period, the Borrower's and its Restricted Subsidiaries' Consolidated Net Income (not including extraordinary gains or losses, other than losses arising from reserves established in connection with the Tax Indemnity Provisions (as defined in the National Propane Purchase Agreement)) plus Consolidated Interest Expense and Consolidated Income Tax Expense in each case for such period, as determined in accordance with GAAP.

"EBITDA" means, for any period, EBIT plus the Borrower's and its Restricted Subsidiaries' depreciation and amortization of property, plant and equipment and intangible assets, in each case as taken into account in calculating Consolidated Net Income, in each case for such period, as determined in accordance with GAAP.

For the purposes of calculating the Applicable Margin, the rate of the facility fees payable pursuant to Section 2.10(b) and the Leverage Ratio, EBITDA for any period (the "APPLICABLE PERIOD") shall be adjusted by the addition of the EBITDA of any Asset Acquisitions made during the Applicable Period, as if such Asset Acquisitions occurred on the first day of the Applicable Period, plus the addition of the "Savings Factor"; provided, however, that in the case of calculating the Applicable Margin or the rate of the facility fees payable pursuant to Section 2.10(b), the Savings Factor shall be added only for the purpose of causing the Pricing Tier to remain the same (or to limit its increase) and not to decrease it (i.e. the Savings Factor may be used to maintain pricing or limit any increase in pricing, not to decrease pricing).

The "SAVINGS FACTOR" shall equal, with respect to any Asset Acquisition, an amount equal to 50% of the difference between (a) Actual Acquisition Expense minus (b) Pro Forma Acquisition Expense. "ACTUAL ACQUISITION EXPENSE" means an amount equal to the

personnel expenses and non personnel costs and expenses (which would be deducted from gross profits in calculating costs and EBITDA) related to the operation of any Asset Acquisition from the beginning of the Applicable Period to the date of the purchase of the Asset Acquisition. "PRO FORMA ACQUISITION EXPENSE" means an amount equal to the personnel and non-personnel costs and expenses (which would be deducted from gross profits in calculating costs and EBITDA) that would have been incurred with respect to the operation of any Asset Acquisition for the period from the beginning of the Applicable Period to the date of purchase of the Asset Acquisition, on the assumption that the ongoing personnel and non personnel cost and expense savings projected as of the date of the Asset Acquisition had been realized on the first day of the Applicable Period. In no event shall the aggregate Savings Factor for any Applicable Period exceed 10% of EBITDA, before taking into effect the EBITDA relating to such Asset Acquisition, for the Borrower and its Restricted Subsidiaries for such Applicable Period.

"EFFECTIVE AMOUNT" means: (a) with respect to any Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any Borrowings and prepayments or repayments of Loans occurring on such date; and (b) with respect to any outstanding L/C Obligations on any date, the amount of such L/C Obligations on such date after giving effect to any Issuances of Letters of Credit occurring on such date and any other changes in the aggregate amount of the L/C Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Letters of Credit or any reductions in the maximum amount available for drawing under Letters of Credit taking effect on such date.

"ELIGIBLE ASSIGNEE" means (i) a commercial bank organized under the laws of the United States, or any state thereof, and having a combined capital and surplus of at least \$100,000,000; and (ii) a commercial bank organized under the laws of any other country which is a member of the Organization for Economic Cooperation and Development (the "OECD"), or a political subdivision of any such country, and having a combined capital and surplus of at least \$100,000,000, provided, that such bank is acting through a branch or agency located in the United States.

"ENVIRONMENTAL LAWS" means all federal, state or local laws, statutes, common law duties, rules, regulations, ordinances and codes, together with all administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authorities, in each case relating to environmental matters.

"ERISA" means the Employee Retirement Income Security Act of 1974 and the regulations thereunder. References to sections of ERISA also refer to any successor sections.

"ERISA AFFILIATE" means any trade or business (whether or not incorporated) under common control with the Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"ERISA EVENT" means (a) a Reportable Event with respect to a Pension Plan; (b) the failure to make a required contribution to a Pension Plan if such failure is sufficient to give rise to a Lien under Section 302(f) of ERISA; (c) a withdrawal by the Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in

which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations which is treated as such a withdrawal under Section 4062(e) of ERISA; (d) a complete or partial withdrawal by the Borrower or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization; (e) the filing of a notice of intent to terminate, the treatment of a Plan amendment as a termination under Section 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan; (f) an event or condition which might reasonably be expected to constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; or (g) the imposition of any liability under Title IV of ERISA, other than PBGC premiums due but not delinquent under Section 4007 of ERISA, upon the Borrower or any ERISA Affiliate.

"EURODOLLAR BASE RATE" has the meaning set forth in the definition of Eurodollar Rate. $\,$

"EURODOLLAR RATE" means for any Interest Period with respect to any Eurodollar Rate Loan, a rate per annum determined by the Agent pursuant to the following formula:

Eurodollar Rate = Eurodollar Rate = 1.00 - Eurodollar Reserve Percentage

Where,

"EURODOLLAR BASE RATE" means, for such Interest Period:

- (a) the rate per annum (rounded upward, if necessary, to the nearest 1/16 of 1%) equal to the rate determined by the Agent to be the offered rate that appears on the page of the Telerate screen (or any successor thereto) that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (b) if the rate referenced in the preceding clause (a) does not appear on such page or service or such page or service shall not be available, the rate per annum (rounded upward, if necessary, to the nearest 1/16 of 1%) equal to the rate determined by the Agent to be the offered rate on such other page or other service that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum (rounded upward, if necessary, to the nearest 1/16 of 1%) determined by the Agent as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in

same day funds in the approximate amount of the Eurodollar Rate Loan being made, continued or converted by Wachovia and with a term equivalent to such Interest Period would be offered by Wachovia's London branch to major banks in the London interbank eurodollar market at their request at approximately 4:00 p.m. (London time) two Business Days prior to the first day of such Interest Period.

"EURODOLLAR RESERVE PERCENTAGE" means, for any day during any Interest Period, the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day, whether or not applicable to any Bank, under regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities"). The Eurodollar Rate for each outstanding Eurodollar Rate Loan shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"EURODOLLAR RATE LOAN" means a Loan that bears interest at a rate based on the Eurodollar Rate.

"EVENT OF DEFAULT" has the meaning specified in Section 9.1.

"EXCESS SALE PROCEEDS" has the meaning specified in Section 8.8(c)(ii)(B).

"EXCHANGE ACT" means the Securities Exchange Act of 1934, and regulations promulgated thereunder.

"EXISTING CREDIT AGREEMENT" means that certain Second Amended and Restated Credit Agreement, dated as of August 22, 2002, among the Obligors, the financial institutions party thereto and BofA as administrative agent, as amended from time to time in accordance with its terms.

"EXISTING LETTERS OF CREDIT" means all letters of credit listed and identified by letter or credit number on Schedule 3.1(a) and outstanding on the Closing Date.

"FDIC" means the Federal Deposit Insurance Corporation, and any Governmental Authority succeeding to any of its principal functions.

"FEDERAL FUNDS RATE" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole

multiple of 1/100 of 1%) charged to Wachovia on such day on such transactions as determined by the Agent.

"FEE LETTER" has the meaning specified in Section 2.10(a).

"FIRST MORTGAGE NOTE AGREEMENTS" means, collectively, (a) the separate Note Agreements, each dated as of April 12, 1995, among the Obligors and the holders of the Series A-C First Mortgage Notes, as the same may be amended, supplemented or otherwise modified from time to time, (b) the Note Agreement, dated as of March 15, 1999, among the Borrower, the General Partner and the holders of the Series D First Mortgage Notes, as the same may be amended, supplemented or otherwise modified from time to time and (c) the Note Agreement, dated as of March 15, 2000, among the Borrower, the General Partner and the holders of the Series E First Mortgage Note, as the same may be amended, supplemented or otherwise modified from time to time.

"FIRST MORTGAGE NOTES" means the Series A-C First Mortgage Notes, the Series D First Mortgage Notes and the Series E First Mortgage Notes.

"FOREIGN BANK" has the meaning specified in Section 10.13(a).

"FRB" means the Board of Governors of the Federal Reserve System, and any Governmental Authority succeeding to any of its principal functions.

"FUNDED DEBT" means, as of any date of determination, (i) indebtedness of the Borrower and/or its Restricted Subsidiaries for borrowed money or for the deferred purchase price of property or services, other than indebtedness for trade payables and non-recourse indebtedness which is not required by GAAP to be classified as a liability on the balance sheet of the debtor, (ii) Capitalized Lease Liabilities, and (iii) Contingent Obligations. For purposes of this definition, undrawn letters of credit shall not constitute Funded Debt.

"FUNDED DEBT RATIO" means the ratio of (a) Funded Debt to (b)

EBITDA.

"GAAP" has the meaning specified in Section 1.3(a).

"GUARANTORS" has the meaning specified in the introductory

clause hereto.

"GUARANTOR FINANCIALS" has the meaning specified in Section $\,$

7.14(b).

"GENERAL COLLATERAL" means, collectively, the Mortgaged Property, and the properties referred to as the "Collateral" in the General Security Agreement and each Subsidiary Security Agreement and as the "Security" in the Collateral Agency Agreement.

"GENERAL PARTNER" has the meaning specified in the introductory clause hereto. $% \left(1\right) =\left(1\right) \left(1\right) \left($

"GENERAL SECURITY AGREEMENT" means the General Security
Agreement, dated as of April 19, 1995, among the Borrower, the Collateral Agent,
and Mellon Bank, N.A., as Cash Collateral Sub-Agent, as the same may be amended,
supplemented, assigned or otherwise modified from time to time.

"GOVERNMENTAL AUTHORITY" means any nation or government, any state or other political subdivision thereof, any central bank (or similar monetary or regulatory authority) thereof, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, and any corporation or other entity owned or controlled, through stock or capital ownership or otherwise, by any of the foregoing.

 $\tt "GUARANTY\ OBLIGATION"\ has\ the\ meaning\ specified\ in\ the\ definition\ of\ "Contingent\ Obligation".$

"HAZARDOUS MATERIAL" means:

- (a) any "hazardous substance", as defined by CERCLA;
- - (c) any petroleum product other than propane; or
- (d) any pollutant or contaminant or hazardous, dangerous or toxic chemical, material or substance within the meaning of any other applicable federal, state or local law, regulation, ordinance or requirement (including consent decrees and administrative orders) relating to or imposing liability or standards of conduct concerning any hazardous, toxic or dangerous waste, substance or material, all as amended or hereafter amended.

"HONOR DATE" has the meaning specified in Section 3.3(b).

"ICC" has the meaning specified in Section 3.9.

"INCORPORATED COVENANT" has the meaning specified in Section

7.14(d).

"INDEBTEDNESS" of any Person means, without duplication, (a) all indebtedness for borrowed money; (b) all obligations issued, undertaken or assumed as the deferred purchase price of property or services (other than trade payables and accrued expenses arising in the ordinary course of business on ordinary terms); (c) all non-contingent reimbursement or payment obligations with respect to Surety Instruments; (d) all obligations evidenced by notes, bonds, debentures or similar instruments, including obligations so evidenced incurred in connection with the acquisition of property, assets or businesses; (e) all indebtedness created or arising under any conditional sale or other title retention agreement, or incurred as financing, in either case with respect to property acquired by the Person (even though the rights and remedies of the seller or bank under such agreement in the event of default are limited to repossession or sale of such property); (f) all Capitalized Lease Liabilities; (g) all indebtedness referred to in clauses (a) through (f) above secured by (or otherwise, to be secured by) any Lien upon or in property (including accounts and contracts rights) owned by such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness; (h) all Redeemable Capital Stock of such Person valued at the greater of its voluntary or involuntary maximum fixed repurchase price plus accrued dividends; (i) any Preferred Stock of any Subsidiary of such Person valued at the sum of the liquidation preference thereof or any mandatory redemption payment obligations in respect

thereof plus, in either case, accrued dividends thereon and (j) all Guaranty Obligations in respect of indebtedness or obligations of others of the kinds referred to in clauses (a) through (i) above.

"INDEMNIFIED LIABILITIES" has the meaning specified in Section $\,$

12.5.

"INDEMNIFIED PARTIES" has the meaning specified in Section

12.5.

"INSOLVENCY PROCEEDING" means (a) any case, action or proceeding before any court or other Governmental Authority relating to bankruptcy, reorganization, insolvency, liquidation, receivership, dissolution, winding-up or relief of debtors, or (b) any general assignment for the benefit of creditors, composition, marshaling of assets for creditors, or other, similar arrangement in respect of a Person's creditors generally or any substantial portion of a Person's creditors; in each case undertaken under U.S. Federal, state or foreign law, including in each case the Bankruptcy Code.

"INTERCOMPANY LOAN AGREEMENT" means that certain Loan Agreement, dated July 19, 1999, between National Propane, L.P. (renamed AEPLP) and Columbia Propane Corporation (renamed AEPI), as amended, supplemented or otherwise modified from time to time.

"INTERCOMPANY NOTE" means that certain Promissory Note, dated July 19, 1999, by AEPLP in favor of the Borrower by endorsement from AEPI in the original principal amount of \$137,997,000, as amended, supplemented or otherwise modified from time to time.

"INTEREST PAYMENT DATE" means, as to any Eurodollar Rate Loan, the last day of each Interest Period applicable to such Loan and, as to any Base Rate Loan, the last Business Day of each calendar quarter; provided, however, that if any Interest Period for a Eurodollar Rate Loan exceeds three months, the date that falls three months after the beginning of such Interest Period is also an Interest Payment Date.

"INTEREST PERIOD" means, as to any Eurodollar Rate Loan, the period commencing on the Borrowing Date of such Loan or on the Conversion/Continuation Date on which the Loan is converted into or continued as a Eurodollar Rate Loan, and ending on the date two weeks or one, two, three or six months thereafter as selected by the Borrower in its Notice of Borrowing or Notice of Conversion/Continuation;

provided, that:

- (i) if any Interest Period would otherwise end on a day that is not a Business Day, that Interest Period shall be extended to the following Business Day unless the result of such extension would be to carry such Interest Period into another calendar month, in which event such Interest Period shall end on the preceding Business Day;
- (ii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period:

(iii) no Interest Period for any Loan shall extend

beyond its Maturity Date; and

(iv) no Interest Period applicable to an Acquisition Loan or portion thereof shall extend beyond any date upon which is due any scheduled principal payment in respect of Acquisition Loans unless the aggregate principal amount of Acquisition Loans represented by Base Rate Loans or Eurodollar Rate Loans having Interest Periods that will expire on or before such date, equals or exceeds the amount of such principal payment.

"INTEREST RATE AGREEMENT" means any interest rate swap agreement, interest rate cap agreement, interest rate collar agreement or other similar agreement or arrangement designed to protect the Borrower against fluctuations in interest rates on Parity Debt.

"INVESTMENT" means, as applied to any Person, any direct or indirect purchase or other acquisition by such Person of stock or other securities of any other Person, or any direct or indirect loan, advance or capital contribution by such Person to any other Person, and any other item which would be classified as an "investment" on a balance sheet of such Person prepared in accordance with GAAP, including without limitation any direct or indirect contribution of such Person of property or assets to a joint venture, partnership or other business entity in which such Person retains an interest (it being understood that a direct or indirect purchase or other acquisition by such Person of assets of any other Person (other than stock or other securities) shall not constitute an "Investment" for purposes of this Agreement). For purposes of Section 8.4(c), the amount involved in Investments made during any period shall be the aggregate cost to the Borrower and its Restricted Subsidiaries of all such Investments made during such period, determined in accordance with GAAP, but without regard to unrealized increases or decreases in value, or write-ups, write-downs or write-offs, of such Investments and without regard to the existence of any undistributed earnings or accrued interest with respect thereto accrued after the respective dates on which such Investments were made, less any net return of capital realized during such period upon the sale, repayment or other liquidation of such Investments (determined in accordance with GAAP, but without regard to any amounts received during such period as earnings (in the form of dividends not constituting a return of capital, interest or otherwise) on such Investments or as loans from any Person in whom such Investments have been made).

"INVESTMENT CONDITION" has the meaning specified in Section

7.13(a).

"INVESTMENT LIMIT" has the meaning specified in Section

8.4(c).

"IRS" means the Internal Revenue Service, and any Governmental Authority succeeding to any of its principal functions under the Code.

"ISP98" has the meaning specified in Section 3.9.

"ISSUANCE DATE" has the meaning specified in Section 3.1(a).

"ISSUE" means, with respect to any Letter of Credit, to issue or to extend the expiry of, or to renew or increase the amount of, such Letter of Credit; and the terms ISSUED, ISSUING and ISSUANCE have corresponding meanings.

"ISSUING BANK" means Wachovia in its capacity as issuer of one or more Letters of Credit hereunder, together with any replacement letter of credit issuer arising under Section 10.1(b) or Section 10.9.

"KEEP WELL AGREEMENT" means that certain Keep Well Agreement, dated as of August 21, 2001, between the Borrower and Columbia Propane Corporation (renamed AEPI).

"L/C ADVANCE" means each Bank's participation in any L/C Borrowing in accordance with its Pro Rata Share.

"L/C AMENDMENT APPLICATION" means an application form for amendment or renewal of outstanding standby letters of credit as shall at any time be in use at the Issuing Bank, as the Issuing Bank shall request.

 $\,$ "L/C APPLICATION" means an application form for issuances of standby letters of credit as shall at any time be in use at the Issuing Bank, as the Issuing Bank shall request.

"L/C BORROWING" means an extension of credit resulting from a drawing under any Letter of Credit which shall not have been reimbursed on the date when made or converted into a Borrowing of Revolving Loans under Section 3.3(c).

"L/C COMMITMENT" means the commitment of the Issuing Bank to Issue, and the commitment of the Banks severally to participate in, Letters of Credit from time to time Issued or outstanding under Article III, in an aggregate amount not to exceed on any date the amount of the Revolving Commitment, it being understood that the L/C Commitment is a part of the Revolving Commitments rather than a separate, independent commitment.

"L/C OBLIGATIONS" means at any time the sum of (a) the aggregate undrawn amount of all Letters of Credit then outstanding, plus (b) the amount of all outstanding L/C Borrowings.

"L/C-RELATED DOCUMENTS" means the Letters of Credit, the L/C Applications, the L/C Amendment Applications and any other document executed by the Borrower relating to any Letter of Credit, including any of the Issuing Bank's standard form documents for letter of credit issuances, as the same may be amended, supplemented, assigned or otherwise modified from time to time.

"LENDING OFFICE" means, as to any Bank, the office or offices of such Bank specified as its "Lending Office" or "Domestic Lending Office" or "Eurodollar Lending Office", as the case may be, on Schedule 12.2, or such other office or offices as such Bank may from time to time notify the Borrower and the Agent.

"LETTERS OF CREDIT" means any standby letters of credit issued by the Issuing Bank pursuant to Article III.

"LEVERAGE RATIO" means, as of any date of determination, the ratio of (i) Total Debt to (ii) EBITDA.

"LICENSE AGREEMENTS" means, collectively, (a) the Software License Agreement, dated April 19, 1995, by and among the General Partner, the Public Partnership, and the Borrower relating to the FAST and Stars I and II proprietary software systems, (b) the Trademark License Agreement, dated April 19, 1995, by and among Petrolane, the General Partner, the Public Partnership and the Borrower, (c) the Trademark License Agreement, dated April 19, 1995, by and among UGI, AmeriGas, Inc., the General Partner, the Public Partnership and the Borrower and (d) the Trademark License Agreement, dated April 19, 1995, by and among the General Partner, the Public Partnership and the Borrower, as the same may be amended, modified or supplemented from time to time.

"LIEN" means any security interest, mortgage, deed of trust, pledge, hypothecation, assignment, charge or deposit arrangement, encumbrance, lien (statutory or other) or preferential arrangement of any kind or nature whatsoever in respect of any property (including those created by, arising under or evidenced by any conditional sale or other title retention agreement, the interest of a lessor under a capital lease, any financing lease having substantially the same economic effect as any of the foregoing, or the filing of any financing statement naming the owner of the asset to which such lien relates as debtor, under the Uniform Commercial Code or any comparable law) and any contingent or other agreement to provide any of the foregoing, but not including the interest of a lessor under an operating lease.

"LOAN" means an extension of credit by a Bank to the Borrower under Article II, and may be a Base Rate Loan or a Eurodollar Rate Loan (each, a "TYPE" of Loan), and includes any Revolving Loan, Acquisition Loan or Swing Line Loan; provided, that no Swing Line Loan may be a Eurodollar Rate Loan.

"LOAN DOCUMENTS" means this Agreement, any Notes, the Fee Letter, the Security Documents, the L/C Related Documents, the Post-Closing Agreement, each Notice of Borrowing, each Notice of Conversion/Continuation and each Compliance Certificate.

"LONG TERM FUNDED DEBT" means, as applied to any Person, all Indebtedness of such Person which by its terms or by the terms of any instrument or agreement relating thereto matures one year or more from the date of execution of the instruments governing any such Indebtedness or, if applicable, the execution of any instrument extending the maturity date of such Indebtedness, provided, that Long Term Funded Debt shall include any Indebtedness which does not otherwise come within the foregoing definition but which is directly or indirectly renewable or extendible at the option of the debtor to a date one year or more (including an option of the debtor under a revolving credit or similar agreement obligating the lender or lenders to extend credit over a period of one year or more) from the date of execution of the instruments governing any such Indebtedness or, if applicable, the execution of any instrument extending the maturity date of such Indebtedness.

"MARGIN STOCK" means "margin stock" as such term is defined in Regulation U or X of the FRB.

"MATERIAL ADVERSE EFFECT" means (a) a material adverse effect on the business, Assets or financial condition of the Borrower and its Restricted Subsidiaries taken as a whole; (b) a material impairment of the ability of the Borrower or any Restricted Subsidiary to perform any of its obligations under this Agreement, the Notes or the Security Documents to which it is a party; or (c) a material impairment of the legal, valid, binding and enforceable nature of the Security Documents or of the validity and priority of the Liens created thereby on the General Collateral.

"MATURITY DATE" means the Revolving Termination Date in respect of the Revolving Loans, and the Acquisition Loan Termination Date in respect of the Acquisition Loans.

"MORTGAGED PROPERTY" means the real properties referred to as the "Mortgaged Property" in the granting clauses of the Mortgages.

 ${\tt "MORTGAGES"}$ shall have the meaning specified in the definition of Security Documents.

"MULTIEMPLOYER PLAN" means a "multiemployer plan", within the meaning of Section 4001(a)(3) of ERISA, with respect to which the Borrower or any ERISA Affiliate may have any liability.

"NATIONAL PROPANE PURCHASE AGREEMENT" means that certain Purchase Agreement, dated April 5, 1999, by and among AEPLP, AEPH, AEPI, National Propane Partners, L.P., National Propane Corporation, National Propane SGP, Inc. and Triarc Companies, Inc., as amended, supplemented or otherwise modified from time to time.

"NET AMOUNT OF UNRESTRICTED INVESTMENT" means the sum of, without duplication, (x) the aggregate amount of all Investments made after the date hereof pursuant to Section 8.4(h) (computed as provided in the last sentence of the definition of Investment) and (y) the aggregate of all Designation Amounts in connection with the designation of Unrestricted Subsidiaries pursuant to the provisions of Section 7.13 less all Designation Amounts in respect of Unrestricted Subsidiaries which have been designated as Restricted Subsidiaries in accordance with the provisions of Section 7.13 and otherwise reduced in a manner consistent with the provisions of the last sentence of the definition of Investment.

"NET PROCEEDS" means with respect to any Asset Sale, the proceeds thereof in the form of cash or Cash Equivalents including payments in respect of deferred payment obligations when received in the form of cash or Cash Equivalents net of (i) reasonable brokerage commissions and other reasonable fees and expenses (including without limitation reasonable fees and expenses of legal counsel and accountants and reasonable fees, expenses and discounts or commissions of underwriters, placement agents and investment bankers) related to such Asset Sale; (ii) provisions for all taxes payable as a result of such Asset Sale; (iii) amounts required to be paid to any Person (other than the Borrower or any Restricted Subsidiary) owning a beneficial interest in the assets subject to such Asset Sale; (iv) appropriate amounts to be provided by the Borrower or any Restricted Subsidiary, as the case may be, as a reserve required in accordance with GAAP against any liabilities associated with such Asset Sale and retained by the Borrower or any Restricted Subsidiary, as the case may be, after such Asset Sale, including, without limitation, pension and other post-employment benefit liabilities, liabilities related to environmental matters and liabilities under any indemnification obligations associated with such

Asset Sale; and (v) amounts required to be applied to the repayment of Indebtedness (other than the Notes and the Parity Debt) secured by a Lien on the asset or assets sold in such Asset Sale.

"NEW BANKS" shall have the meaning specified in Section 2.15(b). $\label{eq:banks}$

"NON-AEPLP RESTRICTED SUBSIDIARY" has the meaning specified in Section 8.17(a).

"NON-PP&E ASSETS" has the meaning specified in Section

8.17(b).

"NOTE" means a promissory note executed by the Borrower in favor of a Bank pursuant to Section 2.2(d), substantially in the form of Exhibit G-1 (in the case of Acquisition Loans) or Exhibit G-2 (in the case of Revolving Loans), as the same may be amended, supplemented or otherwise modified from time to time in accordance with the terms hereof.

"NOTICE DATE" shall have the meaning specified in Section

2.15(b).

"NOTICE OF BORROWING" means a notice in substantially the form

"NOTICE OF BORROWING" means a notice in substantially the form of Exhibit A-1, in the case of a Swing Line Loan, and Exhibit A-2 in the case of any other Loan.

"NOTICE OF CONVERSION/CONTINUATION" means a notice in substantially the form of Exhibit B.

"OBLIGATIONS" means all advances, debts, liabilities, obligations, covenants and duties arising under any Loan Document owing by any of the Obligors or other Credit Parties to any Bank, the Agent or any Indemnified Party, whether direct or indirect (including those acquired by assignment), absolute or contingent, due or to become due, now existing or hereafter arising.

"OBLIGORS" has the meaning specified in the introductory

clause hereto.

"OFFICERS' CERTIFICATE" means as to any corporation, a certificate executed on its behalf by the Chairman of the Board of Directors (if an officer) or its President or one of its Vice Presidents, and its Treasurer, or Controller, or one of its Assistant Treasurers or Assistant Controllers, and, as to any partnership, a certificate executed on behalf of such partnership by its general partner in a manner which would qualify such certificate (a) if such general partner is a corporation, as an Officers' Certificate of such general partner hereunder or (b) if such general partner is a partnership or other entity, as a certificate executed on its behalf by Persons authorized to do so pursuant to the constituting documents of such partnership or other entity.

"ORGANIZATION DOCUMENTS" means, for any corporation, the certificate or articles of incorporation, the bylaws, any certificate of determination or instrument relating to the rights of preferred shareholders of such corporation, any shareholder rights agreement, and all applicable resolutions of the board of directors (or any committee thereof) of such corporation and as to any partnership, its partnership agreement, certificate of partnership and related agreements and as to any other entity, such other entity's analogous organizational documents, as the same may be amended, supplemented or otherwise modified from time to time.

"ORIGINAL CLOSING DATE" means April 19, 1995.

"OTHER TAXES" means (i) any present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or from the execution, delivery or registration of, or otherwise with respect to, this Agreement or any other Loan Documents and (ii) any documentary stamp taxes and nonrecurring intangible personal property taxes payable under Chapters 199 and 201 of the Florida Statutes.

"PARITY DEBT" means the Series A-C First Mortgage Notes, the Series D First Mortgage Notes, the Series E First Mortgage Notes, Indebtedness under this Agreement and the Notes and other Indebtedness of the Borrower incurred in accordance with Section 8.1(a), (b), (e), (f) and (l) and secured by the respective Liens of the Security Documents in accordance with Section 8.3(j), (k), (l) and (m).

"PARTICIPANT" has the meaning specified in Section 12.9(e).

"PARTNERSHIP AGREEMENT" means the Amended and Restated Agreement of Limited Partnership of the Borrower, as in effect on the Closing Date, and as the same may from time to time be amended, supplemented or otherwise modified.

"PARTNERSHIP UNRESTRICTED SUBSIDIARIES" means the Unrestricted Subsidiaries of the Public Partnership as defined in the Public Partnership Indenture as in effect on the Closing Date.

"PBGC" means the Pension Benefit Guaranty Corporation, or any Governmental Authority succeeding to any of its principal functions under ERISA.

"PENSION PLAN" means a "pension plan", as such term is defined in section 3(2) of ERISA, which is subject to Title IV of ERISA (other than a multiemployer plan as defined in section 4001(a)(3) of ERISA) with respect to which the Borrower or any ERISA Affiliates may have any liability.

"PERMITTED BANKS" has the meaning specified in Section 8.4(a).

"PERMITTED INSURERS" means (a) insurers with ratings of A or better according to Best's Insurance Reports, or with comparable ratings from a comparable rating agency for insurance companies whose principal offices are located outside of the United States and Canada, and with assets of no less than \$500 million and (b) the underwriters at Lloyd's, London.

"PERSON" means an individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture or Governmental Authority or other entity.

clause hereto.

"PETROLANE" has the meaning specified in the introductory

"PLAN" means an employee benefit plan (as defined in Section 3(3) of ERISA) which the Borrower sponsors or maintains or to which the Borrower makes, is making, or is obligated to make contributions and includes any Pension Plan.

"POST-CLOSING AGREEMENT" has the meaning specified in Section

"PP&E ACQUISITION/INVESTMENT/TRANSFER LIMIT" has the meaning specified in Section 8.15.

5.1(e).

"PP&E ASSETS" means assets that would, in accordance with GAAP, be classified and accounted for as "property, plant and equipment" on the consolidated balance sheet of the Borrower and the Restricted Subsidiaries.

"PP&E TRANSFER" has the meaning specified in Section 8.17(b).

"PPD/GP DEBT CONTRIBUTION" means the amount of aggregate net cash proceeds previously received by the Borrower from time to time from the Public Partnership as a capital contribution made with the proceeds of Public Partnership Indebtedness and the General Partner in connection with its related and contemporaneous capital contribution and designated as such by such Persons at the time of contribution in the corporate or other records of such Persons.

"PREFERRED STOCK", as applied to the Capital Stock of any Person, means Capital Stock of any class or classes (however designated), which is preferred as to the payment of distributions or dividends, or upon any voluntary or involuntary liquidation or dissolution of such Person, over shares or units of Capital Stock of any other class of such Person.

"PRIME RATE" means, at any time, the rate of interest in effect for such day as publicly announced from time to time by Wachovia as its "prime rate" (which is not necessarily the lowest rate charged to any customer). Any change in such rate announced by Wachovia shall take effect at the opening of business on the day specified in the public announcement of such change.

"PRO RATA SHARE" means, as to any Bank at any time, the percentage equivalent (expressed as a decimal, rounded to the ninth decimal place) at such time of such Bank's Commitment divided by the combined Commitments of all Banks.

"PUBLIC PARTNERSHIP" means AmeriGas Partners, L.P., a Delaware limited partnership.

"PUBLIC PARTNERSHIP INDENTURE" means each of the Indentures among the Public Partnership, its financing subsidiaries, and Wachovia (formerly known as First Union National Bank), as trustee, with respect to the Public Partnership Notes, as the same may be amended, supplemented or otherwise modified from time to time.

"PUBLIC PARTNERSHIP NOTES" means the notes issued, from time to time, jointly and severally, by the Public Partnership and its financing subsidiaries, as the same may be amended, supplemented or otherwise modified from time to time.

"PURCHASE MONEY LIEN" has the meaning specified in Section 8.3(n).

"REDEEMABLE CAPITAL STOCK" means any shares of any class or series of Capital Stock, that, either by the terms thereof, by the terms of any security into which it is convertible or exchangeable or by contract or otherwise, is or upon the happening of an event or passage of time would be, required to be redeemed prior to the date of the last scheduled payment of any Loan then outstanding or is redeemable at the option of the holder thereof at any time prior to such date, or is convertible into or exchangeable for Indebtedness at any time prior to such date.

"REFERENCE PERIOD" shall have the meaning specified in the definition of Consolidated Cash Flow.

"REPLACEMENT BANK" has the meaning specified in Section 4.7.

"REPORTABLE EVENT" means, any of the events set forth in Section 4043(b) of ERISA or the regulations thereunder, other than any such event for which the 30-day notice requirement under ERISA has been waived in regulations issued by the PBGC.

"REQUIRED BANKS" means at any time Banks then holding at least 66-2/3% of the then aggregate unpaid principal amount of the Loans (assuming that any outstanding Swing Line Loans were converted into Revolving Loans or participated in by the Banks pursuant to Section 2.16) and L/C Borrowings and risk participations in outstanding Letters of Credit (or in the case of the Issuing Bank, the amount of the outstanding Letters of Credit minus risk participations of the other Banks therein), or, if no amounts are outstanding, Banks then having at least 66-2/3% of the aggregate amount of the Commitments.

"REQUIREMENT OF LAW" means, as to any Person, any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or of a Governmental Authority, in each case applicable to or binding upon the Person or any of its property or to which the Person or any of its property is subject.

"RESOURCE CONSERVATION AND RECOVERY ACT" means the Resource Conservation and Recovery Act, 42 U.S.C. Section 690, et seq., as in effect from time to time.

"RESPONSIBLE OFFICER" means the chief executive officer or the president of the Borrower, or any other officer having substantially the same authority and responsibility; or, with respect to compliance with financial covenants, the chief financial officer or the treasurer of the Borrower, or any other officer having substantially the same authority and responsibility.

"RESTRICTED PAYMENT" means with respect to the Borrower and its Restricted Subsidiaries (the "COVERED PERSONS"), (a) in the case of any Covered Person that is a partnership, (i) any payment or other distribution, direct or indirect, in respect of any partnership interest in such Covered Person, except a distribution payable solely in additional partnership interests in such Covered Person, and (ii) any payment, direct or indirect, by such Covered

Person on account of the redemption, retirement, purchase or other acquisition of any partnership interest in such Covered Person, except to the extent that such payment consists of additional partnership interests in such Covered Person; or (b) in the case of any Covered Person that is a corporation, (i) any dividend or other distribution, direct or indirect, on account of any shares of any class of stock of such Covered Person then outstanding, except a dividend payable solely in shares of stock of such Covered Person, and (ii) any payment, direct or indirect, by such Covered Person on account of the redemption, retirement, purchase or other acquisition of any shares of any class of stock of such Covered Person then outstanding, or of any warrants, rights or options, to acquire any such shares, except to the extent that such payment consists of shares of Capital Stock of such Covered Person; (c) in the case of any Covered Person that is a limited liability company, (i) any payment or other distribution, direct or indirect, in respect of any membership interest in such Covered Person, except a distribution payable solely in additional membership interests in such Covered Person, and (ii) any payment, direct or indirect, by such Covered Person on account of the redemption, retirement, purchase or other acquisition of any membership interest in such Covered Person, except to the extent that such payment consists of additional membership interests in such Covered Person; or (d) any indemnification payment made by AEPLP, AEPH or AEPI pursuant to the Tax Indemnity Provisions (as defined in the National Propane Purchase Agreement), including any payment made by the Borrower to AEPI pursuant to the Keep Well Agreement.

"RESTRICTED SUBSIDIARY" means any Subsidiary of the Borrower organized under the laws of the United States or any state thereof or Canada or any province thereof or the District of Columbia, none of the Capital Stock or ownership interests of which is owned by Unrestricted Subsidiaries and substantially all of the operating assets of which are located in, and substantially all of the business of which is conducted within, the United States or Canada and which is designated as a Restricted Subsidiary in Schedule 6.2 or which shall be designated as a Restricted Subsidiary by the General Partner at a subsequent date as provided in Section 7.13; provided, however, that (a) to the extent a newly formed or acquired Subsidiary is not declared either a Restricted Subsidiary or an Unrestricted Subsidiary within 90 days of its formation or acquisition, such Subsidiary shall be deemed a Restricted Subsidiary and (b) a Restricted Subsidiary may be designated as an Unrestricted Subsidiary in accordance with the provisions of Section 7.13.

"REVOLVING COMMITMENT" has the meaning specified in Section 2.1(b).

"REVOLVING LOAN" has the meaning specified in Section 2.1(b).

"REVOLVING TERMINATION DATE" means the earlier to occur of:

- (a) October 15, 2006, as such date may be extended pursuant to Section 2.15 hereof; and $\,$
- (b) the date on which the Revolving Commitments terminate in accordance with the provisions of this Agreement.

"ROUTINE PERMITS" has the meaning specified in Section 6.8(a).

"SALE AND LEASE-BACK TRANSACTION" of a Person (a "TRANSFEROR") means any arrangement (other than between the Borrower and a Wholly-Owned Restricted Subsidiary or between Wholly-Owned Restricted Subsidiaries) whereby (a) property (the "SUBJECT PROPERTY") has been or is to be disposed of by such Transferor to any other Person with the intention on the part of such Transferor of taking back a lease of such Subject Property pursuant to which the rental payments are calculated to amortize the purchase price of such Subject Property substantially over the useful life of such Subject Property, and (b) such Subject Property is in fact so leased by such Transferor or an Affiliate of such Transferor.

"SALE CONDITION" has the meaning specified in Section 7.13(a).

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"SECURITY DOCUMENTS" means (a) the Collateral Agency Agreement, (b) each of (i) the mortgage, assignment of leases and rents, security agreement, financing statement and fixture filings made by the Borrower in favor of the Collateral Agent, (ii) the deed of trust, assignment of leases and rents, security agreement, financing statement and fixture filings made by the Borrower in favor of the Collateral Agent and (iii) the deed to secure debt, assignment of leases and rents, security agreement, financing statement and fixture filings made by the Borrower in favor of the Collateral Agent, each dated on or before the Closing Date and covering one or more of the Mortgaged Properties located in the counties listed on Schedule 6.8(b), those executed after the Closing Date by the Borrower as required by Section 7.10 and those executed by Restricted Subsidiaries after the Closing Date as required by Section 7.9, in each case substantially in the form of Exhibit E (as each of the same may be amended, supplemented, assigned or otherwise modified from time to time, collectively, the "MORTGAGES"), (c) the General Security Agreement, (d) the Subsidiary Security Agreement and (e) the Subsidiary Guarantee.

"SENIOR INDEBTEDNESS" means the Obligations, the obligations of the Borrower, the General Partner and Petrolane under the Series A-C First Mortgage Notes, the obligations of the Borrower and the General Partner under the Series D First Mortgage Notes and the Series E First Mortgage Notes and the obligations of the Borrower with respect to Parity Debt.

"SERIES A-C FIRST MORTGAGE NOTES" means the First Mortgage Notes, Series A through C, in an aggregate principal amount not exceeding \$518,000,000, issued pursuant to that certain Note Agreement, dated as of April 12, 1995, among the Borrower, the General Partner, Petrolane and the purchasers named in Schedule I thereto, as amended, supplemented, assigned or otherwise modified from time to time.

"SERIES D FIRST MORTGAGE NOTES" means the First Mortgage
Notes, Series D, in aggregate principal amount not exceeding \$70,000,000, issued
pursuant to that certain Note Agreement, dated as of March 15, 1999, among the
Borrower, the General Partner and the purchasers named in Schedule I thereto, as
amended, supplemented, assigned or otherwise modified from time to time.

"SERIES E FIRST MORTGAGE NOTES" means the First Mortgage Notes, Series E, in an aggregate principal amount of \$80,000,000, issued pursuant to that certain Note Agreement, dated as of March 15, 2000, among the Borrower, the General Partner and the purchasers named in Schedule I thereto, as amended, supplemented, assigned or otherwise modified from time to time.

"SIGNIFICANT SUBSIDIARY GROUP" means any Subsidiary of the Borrower which is, or any group of Subsidiaries of the Borrower all of which are, at any time of determination, subject to one or more of the proceedings or conditions described in subsection (f) or (g) of Section 9.1 and which Subsidiary or group of Subsidiaries accounted for (or in the case of a recently formed or acquired Subsidiary would have so accounted for on a pro forma basis) more than 1% of consolidated operating revenues of the Borrower for the fiscal year most recently ended or more than 1% of consolidated Total Assets of the Borrower as of the end of the most recently ended fiscal quarter, in each case computed in accordance with GAAP.

"SPECIAL RATING" means a risk-based capital factor attributable to Indebtedness for purposes of generally applicable state insurance regulations for life, health and disability insurance companies, substantially equivalent to an investment grade rating issued by a nationally recognized credit rating agency.

"SPECIFIED ACQUISITION LOANS" means the Acquisition Loans used solely for the purposes described in Section 8.9(d)(ii).

"SPECIFIED MORTGAGE" means any Mortgage covering the Mortgaged Property identified on Schedule 9.1(d).

"SUBSIDIARY" means, with respect to any Person, any corporation, limited liability company, partnership, joint venture, association, trust or other entity of which (or in which) more than 50% of (a) the issued and outstanding Capital Stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time Capital Stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (b) the interests in the capital or profits of such partnership, limited liability company, joint venture or association with ordinary voting power to elect a majority of the board of directors (or Persons performing similar functions) of such partnership, limited liability company, joint venture or association, or (c) the beneficial interests in such trust or other entity with ordinary voting power to elect a majority of the board of trustees (or Persons performing similar functions) of such trust or other entity, is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries, or by one or more of such Person's other Subsidiaries.

"SUBSIDIARY GUARANTEE" means that certain Restricted Subsidiary Guarantee, dated as of April 19, 1995, by all of the Restricted Subsidiaries (other than AEPLP and any Subsidiary of AEPLP) for the benefit of the Collateral Agent, as the same may be amended, supplemented, assigned or otherwise modified from time to time.

"SUBSIDIARY SECURITY AGREEMENT" means that certain Subsidiary Security Agreement, dated as of April 19, 1995, among all of the Restricted Subsidiaries (other than AEPLP and any Subsidiary of AEPLP), the Collateral Agent and Mellon Bank, N.A., as Cash Collateral Sub-Agent, as the same may be amended, supplemented, assigned or otherwise modified from time to time.

"SURETY INSTRUMENTS" means all letters of credit (including standby and commercial), bankers acceptances, bank guaranties, shipside bonds, surety bonds and similar instruments.

"SWAP CONTRACT" means any agreement (including any master agreement and any agreement, whether or not in writing, relating to any single transaction) that is an interest rate swap agreement, basis swap, forward rate agreement, commodity swap, commodity option, equity or equity index swap or option, bond option, interest rate option, forward foreign exchange agreement, rate cap, collar or floor agreement, currency swap agreement, cross-currency rate swap agreement, swaption, currency option or any other, similar agreement (including any option to enter into any of the foregoing).

"SWING LINE BANK" means Wachovia in its capacity as provider of Swing Line Loans, or any successor swing line bank hereunder.

"SWING LINE LOAN" has the meaning specified in Section 2.16(a).

"SWING LINE SUBLIMIT" means an amount equal to the lesser of (a) 20,000,000 and (b) the Revolving Commitments. The Swing Line Sublimit is part of, and not in addition to, the Revolving Commitments.

"TAXES" means any and all present or future taxes, levies, imposts or withholdings, and all penalties, interest and additions to taxes with respect thereto, excluding, in the case of each Bank and the Agent, such taxes (including income taxes or franchise taxes) as are imposed on or measured by each Bank's net income or capital by the jurisdiction (or any political subdivision thereof) under the laws of which such Bank or the Agent, as the case may be, is organized or maintains a lending office.

"TOTAL ASSETS" means as of any date of determination, the consolidated total assets of the Borrower and the Restricted Subsidiaries as would be shown on a consolidated balance sheet of the Borrower and the Restricted Subsidiaries prepared in accordance with GAAP as of that date.

"TOTAL DEBT" means as of any date of determination, the aggregate principal amount of all Indebtedness of the Borrower and the Restricted Subsidiaries at the time outstanding (other than Indebtedness permitted by Section 8.1(c)). For purposes of computing the Leverage Ratio pursuant to Section 8.14, Total Debt shall also include the obligations described in clause (c) of the definition of "Contingent Obligation."

"TRANSFER" has the meaning specified in Section 8.17(b).

 $$\tt "TRANSFEROR"$$ shall have the meaning specified in the definition of Sale and Lease-Back Transaction.

"TYPE" has the meaning specified in the definition of "Loan."

"UGI" means UGI Corporation, a Pennsylvania corporation.

"UNITED STATES" and "U.S." each means the United States of

America.

"UNRESTRICTED SUBSIDIARY" means a Subsidiary of the Borrower which is not a Restricted Subsidiary.

"WACHOVIA" means Wachovia Bank, National Association and its

 ${\it successors.}$

"WHOLLY-OWNED RESTRICTED SUBSIDIARY" means any Restricted Subsidiary that is also a Wholly-Owned Subsidiary of the Borrower.

"WHOLLY-OWNED SUBSIDIARY" means, as applied to any Subsidiary of any Person, a Subsidiary in which (other than directors' qualifying shares required by law) 100% of the Capital Stock of each class having ordinary voting power, and 100% of the Capital Stock of every other class, in each case, at the time as of which any determination is being made, is owned, beneficially and of record, by such Person, or by one or more of such Person's other Wholly-Owned Subsidiaries, or both; provided, that for the purposes of this Agreement, (a) AEPLP shall be deemed a "Wholly-Owned Subsidiary" of the Borrower for so long as the Borrower directly or indirectly owns at least 99% of the Capital Stock of AEPLP and 100% of the general partnership interests therein, and (b) AmeriGas Eagle Parts & Service shall be deemed a "Wholly-Owned Subsidiary" of the Borrower for so long as (i) AEPLP remains a Restricted Subsidiary and a "Wholly-Owned Subsidiary" of the Borrower and (ii) AEPLP directly or indirectly owns at least 100% of the Capital Stock of AmeriGas Eagle Parts & Service.

"YEARLY THRESHOLD" has the meaning specified in Section 8.15.

- $1.2\,$ Other Interpretive Provisions. (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) The words "hereof", "herein", "hereunder" and similar words refer to this Agreement as a whole and not to any particular provision of this Agreement; and subsection, Section, Schedule and Exhibit references are to this Agreement unless otherwise specified.
- (c) (i) The term "including" is not limiting and means "including without limitation."
- (ii) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding", and the word "through" means "to and including."

- (d) Unless otherwise expressly provided herein, (i) references to agreements (including this Agreement) and other contractual instruments shall be deemed to include all subsequent amendments and other modifications thereto, but only to the extent such amendments and other modifications are not prohibited by the terms of any Loan Document, and (ii) references to any statute or regulation are to be construed as including all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting the statute or regulation.
- (e) The captions and headings of this Agreement are for convenience of reference only and shall not affect the interpretation of this Agreement.
- (f) This Agreement and other Loan Documents may use several different limitations, tests or measurements to regulate the same or similar matters. All such limitations, tests and measurements are independent and shall each be performed in accordance with their terms.
- (g) This Agreement and the other Loan Documents are the result of negotiations among and have been reviewed by counsel to the Agent, the Borrower and the other parties, and are the products of all parties. Accordingly, they shall not be construed against the Banks or the Agent merely because of the Agent's or Banks' involvement in their preparation.
- accounting Principles. (a) Unless otherwise specified, all accounting terms used herein or in any other Loan Document shall be interpreted, all accounting determinations and computations hereunder or thereunder shall be made, and all financial statements required to be delivered hereunder or thereunder shall be prepared in accordance with, those generally accepted accounting principles in effect in the United States of America from time to time ("GAAP"). Notwithstanding the foregoing, if the Borrower, the Required Banks or the Agent determines that a change in GAAP from that in effect on the date hereof has altered the treatment of certain financial data to its detriment under this Agreement, such party may seek of the others a renegotiation of any financial covenant affected thereby. If the Borrower, the Required Banks and Agent cannot agree on renegotiated covenants, then, for the purposes of this Agreement, GAAP will refer to generally accepted accounting principles on the date just prior to the date on which the change that gave rise to the renegotiation occurred.
- (b) References herein to "fiscal year" and "fiscal quarter" refer to such fiscal periods of the Borrower.

ARTICLE II

THE CREDITS

2.1 Amounts and Terms of Commitments.

(a) The Acquisition Credit. Each Bank severally agrees, on the terms and conditions set forth herein, to make loans to the Borrower (each such loan, an "ACQUISITION LOAN") from time to time on any Business Day during the period from the Closing Date to the Acquisition Loan Termination Date in an aggregate principal amount not to exceed at any time outstanding the amount set forth opposite such Bank's name on Schedule 2.1 (such amount as

the same may be reduced under Section 2.5 or Section 2.7 or as reduced or increased as a result of one or more assignments under Section 12.9, the Bank's "ACQUISITION COMMITMENT"). Within the limits of each Bank's Acquisition Commitment and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.1(a), prepay under Section 2.6 and reborrow under this Section 2.1(a).

- The Revolving Credit. Each Bank severally agrees, on the terms and conditions set forth herein, to make loans to the Borrower (each such loan, a "REVOLVING LOAN") from time to time on any Business Day during the period from the Closing Date to the Revolving Termination Date, in an aggregate principal amount not to exceed at any time outstanding the amount set forth opposite such Bank's name on Schedule 2.1 (such amount as the same may be reduced under Section 2.5 or Section 2.7 or reduced or increased as a result of one or more assignments under Section 12.9, the Bank's "REVOLVING COMMITMENT"); provided, that after giving effect to any Borrowing of Revolving Loans, the Effective Amount of all outstanding Revolving Loans plus the Effective Amount of all L/C Obligations plus the Effective Amount of all Swing Line Loans shall not exceed the Revolving Commitments. Within the limits of each Bank's Revolving Commitment, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.1(b), prepay under Section 2.6 and reborrow under this Section 2.1(b). As a subfacility of the Banks' Revolving Commitments, the Borrower may request the Issuing Bank to Issue Letters of Credit from time to time pursuant to Article III. In addition, the Borrower may request the Swing Line Bank to make Swing Line Loans to the Borrower from time to time pursuant to Section 2.16.
- 2.2 Loan Accounts. (a) The Loans made by each Bank shall be evidenced by one or more loan accounts or records maintained by such Bank in the ordinary course of business. Each Bank will make reasonable efforts to maintain the accuracy of its loan account or accounts and to update promptly its loan account or accounts from time to time, as necessary.
- (b) The Agent shall maintain the Register pursuant to Section 12.9(d) and a loan subaccount for each Bank, in which Register and loan subaccount (taken together) shall be recorded (i) the date, amount, and Interest Period, if applicable, of each Loan, and whether such Loan is a Base Rate Loan, a Eurodollar Rate Loan or a Swing Line Loan, (ii) the amount of any principal or interest due and payable or to become due and payable to each Bank hereunder and (iii) the amount of any sum received by the Agent hereunder from or for the loan account of the Borrower and each Bank's percentage share thereof. The Agent will make reasonable efforts to maintain the accuracy of the subaccounts referred to in the preceding sentence and to update promptly such loan subaccounts from time to time, as necessary.
- (c) The entries made in the Register and loan subaccounts maintained pursuant to subsection (b) of this Section 2.2, to the extent permitted by applicable law, shall be prima facie evidence of the existence and amounts of such obligations of the Borrower therein recorded; provided, however, that the failure of the Agent or any Bank to maintain any such Register, loan subaccount or loan account, as applicable, or any error therein, shall not in any manner affect the obligations of the Borrower to repay the Loans in accordance with the terms thereof.

- (d) Upon the request of any Bank made through the Agent, and at the expense of the Borrower, the Loans made by such Bank may be evidenced by one or more Notes, instead of loan accounts. Each such Bank shall endorse on the schedules annexed to its Note(s) the date, amount and maturity of each Loan made by it and the amount of each payment of principal made by the Borrower with respect thereto. Each such Bank is irrevocably authorized by the Borrower to so endorse its Note(s) and each Bank's record shall be rebuttable presumptive evidence of the amount of the Loans made by such Bank to the Borrower and the interest and principal payments thereof; provided, however, that the failure of a Bank to make, or an error in making, a notation thereon with respect to any Loan shall not limit or otherwise affect the obligations of the Borrower hereunder or under any such Note to pay any amount owing with respect to the Loans made by such Bank.
- (e) Each Bank represents that at no time shall any part of the funds used to make any Loan constitute, or deemed under ERISA, the Code or any other applicable law, or any ruling or regulation issued thereunder, or any court decision, to constitute, the assets of any employee benefit plan (as defined in section 3(3) of ERISA) or any plan (as defined in section 4975(e)(1) of the Code).
- 2.3 Procedure for Borrowing. (a) Each Borrowing of Loans (other than Swing Line Loans) shall be made upon the Borrower's irrevocable written notice delivered to the Agent in the form of a Notice of Borrowing (which notice must be received by the Agent prior to 1:00 p.m.) (New York City time) (i) three Business Days prior to the requested Borrowing Date, in the case of Eurodollar Rate Loans; and (ii) one Business Day prior to the requested Borrowing Date, in the case of Base Rate Loans, specifying:
 - (A) the amount of the Borrowing, which shall be in an aggregate minimum amount of \$5,000,000 in the case of Eurodollar Rate Loans or \$1,000,000 in the case of Base Rate Loans, or any multiple of \$1,000,000 in excess thereof; provided, however, that the Borrower may request (x) up to two Borrowings of Base Rate Loans in a minimum amount of \$500,000 in any fiscal quarter and (y) Borrowings of Base Rate Loans in such amount as is necessary to pay to the Agent the amounts required by the last sentence of Section 2.13(a);
 - (B) the requested Borrowing Date, which shall be a Business Day;
 - (C) the Type of Loans comprising the Borrowing;
 - (D) whether the Loans comprising the Borrowing shall be Acquisition Loans or Revolving Loans and, if the Loans comprising the Borrowing shall be Acquisition Loans, whether the Acquisition Loans comprising the Borrowing shall be Specified Acquisition Loans; and
 - (E) other than in the case of Base Rate Loans, the duration of the Interest Period applicable to the Loans included in such notice. If the Notice of Borrowing fails to specify the duration of the Interest Period for any Borrowing comprised of Eurodollar Rate Loans, such Interest Period shall be one month.

provided, however, that with respect to any Borrowing to be made on the Closing Date, the Notice of Borrowing shall be delivered to the Agent no later than 10:00 a.m. (New York City time) on the Closing Date and such Borrowing will consist of Base Rate Loans only.

- (b) The Agent will promptly notify each Bank of its receipt of any Notice of Borrowing (or a deemed notice of Borrowing under Section 2.16(b)) and of the amount of such Bank's Pro Rata Share of that Borrowing.
- (c) Each Bank will make the amount of its Pro Rata Share of each Borrowing available to the Agent for the account of the Borrower at the Agent's Payment Office by 1:00 p.m. (New York City time) on the Borrowing Date requested by the Borrower in funds immediately available to the Agent. The proceeds of all such Loans will then be made available to the Borrower by the Agent on the Borrowing Date by crediting the Borrower's Account with the aggregate of such amounts made available to the Agent by the Banks and in like funds as received by the Agent.
- (d) After giving effect to any Borrowing, there may not be more than ten different Interest Periods in effect.
- 2.4 Conversion and Continuation Elections. (a) The Borrower may, upon irrevocable written notice to the Agent in accordance with Section 2.4(b):
- (i) elect, as of any Business Day, in the case of Base Rate Loans, or as of the last day of the applicable Interest Period, in the case of Eurodollar Rate Loans, to convert any such Loans (or any part thereof in an amount not less than \$5,000,000 in the case of a conversion to a Eurodollar Rate Loan or \$1,000,000 in the case of a conversion to a Base Rate Loan, or that is in an integral multiple of \$1,000,000 in excess thereof) into Loans of the other Type; or
- (ii) elect, as of the last day of the applicable Interest Period, to continue as Eurodollar Rate Loans any Eurodollar Rate Loans having Interest Periods expiring on such day (or any part thereof in an amount not less than \$5,000,000, or that is in an integral multiple of \$1,000,000 in excess thereof);

provided, that if at any time the aggregate amount of Eurodollar Rate Loans in respect of any Borrowing is reduced, by payment, prepayment, or conversion of part thereof to be less than \$5,000,000, such Eurodollar Rate Loans shall automatically convert into Base Rate Loans, and on and after such date the right of the Borrower to continue such Loans as, and convert such Loans into, Eurodollar Rate Loans shall terminate.

- (b) The Borrower shall deliver a Notice of Conversion/Continuation to be received by the Agent not later than 1:00 p.m. (New York City time) (i) three Business Days in advance of the Conversion/Continuation Date, if the Loans are to be converted into or continued as Eurodollar Rate Loans; and (ii) one Business Day in advance of the Conversion/Continuation Date, if the Loans are to be converted into Base Rate Loans, specifying:
 - (A) the proposed Conversion/Continuation Date;

- (B) the aggregate amount of Loans to be converted or continued;
- (C) the Type of Loans resulting from the proposed conversion or continuation; and $% \left(1\right) =\left(1\right) \left(1\right)$
- (D) other than in the case of conversions into Base Rate Loans, the duration of the requested Interest Period.
- (c) If upon the expiration of any Interest Period, the Borrower has failed to select timely a new Interest Period to be applicable to the Eurodollar Rate Loans having the expired Interest Period or if any Default or Event of Default then exists, the Borrower shall be deemed to have elected to convert such Eurodollar Rate Loans into Base Rate Loans effective as of the expiration date of such Interest Period.
- (d) The Agent will promptly notify each Bank of its receipt of a Notice of Conversion/Continuation, or, if no timely notice is provided by the Borrower, the Agent will promptly notify each Bank of the details of any automatic conversion. All conversions and continuations shall be made ratably according to the respective outstanding principal amounts of the Loans with respect to which the notice was given held by each Bank.
- (e) Unless the Required Banks otherwise agree, during the existence of a Default or unless all the Banks otherwise agree, during the existence of an Event of Default, the Borrower may not elect to have a Loan converted into or continued as a Eurodollar Rate Loan.
- (f) After giving effect to any conversion or continuation of Loans, there may not be more than ten different Interest Periods in effect.
- Voluntary Termination or Reduction of Commitments. The Borrower may, upon prior notice to the Agent no later than 11:00 a.m. (New York City time) two Business Days' prior to a proposed termination, terminate the Revolving Commitments or the Acquisition Commitments, or permanently reduce the Commitments by an aggregate minimum amount of \$3,000,000 or any multiple of \$1,000,000 in excess thereof; unless, after giving effect thereto and to any prepayments of Loans made on the effective date thereof subject to Sections 2.6 and 4.4, (a) the then Effective Amount of all Revolving Loans and Swing Line Loans plus the then Effective Amount of all L/C Obligations would exceed the amount of the Revolving Commitments then in effect or (b) the then Effective Amount of all outstanding Acquisition Loans would exceed the amount of the Acquisition Commitments then in effect. Once received, any notice delivered by the Borrower to the Agent under this Section 2.5 shall be irrevocable. Once reduced in accordance with this Section 2.5, the Commitments may not be increased. Any reduction of the Commitments shall be applied to each Bank according to its Pro Rata Share. All accrued facility fees to, but not including, the effective date of any reduction or termination of Commitments, shall be paid on the last day of each calendar quarter and the effective date of any such termination. The Agent will promptly notify each Bank of its receipt of a notice under this Section 2.5.

2.6 Optional Prepayments.

- Subject to Section 4.4, the Borrower may, upon notice (a) to the Agent, at any time or from time to time voluntarily prepay Loans (other than Swing Line Loans) in whole or in part without premium or penalty; provided that such notice must be received by the Agent not later than 1:00 p.m. (New York City time) (A) three Business Days prior to any date of prepayment of Eurodollar Rate Loans and (B) on the date of prepayment of Base Rate Loans. Such notice of prepayment shall be irrevocable and specify the date and amount of such prepayment and the Type(s) of Loans to be prepaid and whether the Loans to be prepaid are Acquisition Loans or Revolving Loans. The Agent will promptly notify each Bank of its receipt of any such notice, and of such Bank's Pro Rata Share of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein, together with, in the case of Eurodollar Rate Loans, accrued interest to such date on the amount prepaid and any amounts required pursuant to Section 4.4; provided, that no amount of any optional prepayment under this Section 2.6(a) may be applied to the Revolving Loans unless and until all Specified Acquisition Loans have been paid in full.
- (b) The Borrower may, upon notice to the Swing Line Bank (with a copy to the Agent), at any time or from time to time, voluntarily prepay Swing Line Loans in whole or in part without premium or penalty; provided that (i) such notice must be received by the Swing Line Bank and the Agent not later than 1:00 p.m. (New York City time) on the date of the prepayment, and (ii) any such prepayment shall be in a minimum principal amount of \$100,000. Each such notice shall be irrevocable and shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein.
- 2.7 Mandatory Prepayments of Loans; Mandatory Commitment Reductions.
- Asset Sales. In the event that any Asset Sale results (a) in Excess Sale Proceeds which are not applied as provided in Section 8.8(c)(ii)(B)(x), such Excess Sale Proceeds shall be applied to the prepayment of Senior Indebtedness on a pro rata basis based upon the aggregate principal amount of Senior Indebtedness then outstanding (assuming, with respect to revolving debt, that the maximum commitment amount is outstanding); provided, however, that the amounts that would be applicable to payments to the Banks hereunder shall be applied first to outstanding amounts under the Acquisition Commitments, then to outstanding amounts under the Revolving Commitments. Such prepayments shall be allocated among the Banks according to their respective Pro Rata Shares. The Acquisition Commitments and Revolving Commitments shall be permanently reduced by the amount of such prepayments applied to outstanding principal amounts thereunder, and any such reduction shall be applied to each Bank according to its Pro Rata Share. If the amount of such Excess Sale Proceeds applicable to payment to the Banks hereunder exceeds the amount of the outstandings under the Commitments, the Commitments shall be permanently reduced by such excess, by reduction, first to the Acquisition Commitments and then to the Revolving Commitments, and any such reduction shall be applied to each Bank in accordance with its Pro Rata Share.
- (b) Excess Outstandings. If on any date the Effective Amount of L/C Obligations exceeds the L/C Commitment, the Borrower shall Cash Collateralize on such date the outstanding Letters of Credit in an amount equal to the excess of the maximum amount then

available to be drawn under the Letters of Credit over the L/C Commitment. Subject to Section 4.4, if on any date after giving effect to any Cash Collateralization made on such date pursuant to the preceding sentence (i) the Effective Amount of all Revolving Loans and L/C Obligations exceeds the Revolving Commitments or (ii) the Effective Amount of all Acquisition Loans exceeds the Acquisition Commitments, then the Borrower shall immediately, and without notice or demand, prepay the outstanding principal amount of the Revolving Loans, L/C Advances and/or Acquisition Loans, by an amount equal to such excess.

- (c) Other Prepayments. In the event of a prepayment or mandatory repurchase of the First Mortgage Notes or other Parity Debt not governed by Section 2.7(a) after the Acquisition Loan Termination Date, the Loans shall be prepaid in a pro rata amount, such pro rata amount based upon the maximum commitment amount of Parity Debt plus the Commitments hereunder. Such prepayment shall be applied first to outstanding amounts under the Acquisition Loans, then to outstanding amounts under the Revolving Loans (with a concurrent permanent reduction of the Revolving Commitments).
- (d) Reduction of Acquisition Commitments. Upon any prepayment of the Acquisition Loans under Section 2.7(c), the respective Acquisition Commitments of the Banks shall be automatically and permanently reduced by an amount for each Bank equal to such Bank's Pro Rata Share of such prepayment of the Acquisition Loans.
- 2.8 Repayment. (a) The Acquisition Credit. The Borrower shall repay to the Banks in full on the Acquisition Loan Termination Date the aggregate principal amount of Acquisition Loans outstanding on such date, together with all accrued and unpaid interest thereon.
- (b) The Revolving Credit. The Borrower shall repay to the Agent, for the benefit of the Banks, in full on the Revolving Termination Date the aggregate principal amount of Revolving Loans outstanding on such date, together with all accrued and unpaid interest thereon.
- (c) Swing Line Loans. The Borrower shall repay to the Agent, for the benefit of the Swing Line Bank, in full the aggregate principal amount of each Swing Line Loan, together with all accrued and unpaid interest thereon, upon the earlier of (a) 5 calendar days following the date on which such Swing Line Loan was funded by the Swing Line Bank and (b) the Revolving Termination Date.
- 2.9 Interest. (a) Each Loan shall bear interest on the outstanding principal amount thereof from the applicable Borrowing Date at a rate per annum equal to the Eurodollar Rate (other than with respect to Swing Line Loans) or the Base Rate, as the case may be (and subject to the Borrower's right to convert to the other Type of Loan under Section 2.4), plus the Applicable Margin.
- (b) Interest on each Loan shall be paid in arrears on each Interest Payment Date. Interest shall also be paid on the date of any prepayment of Eurodollar Rate Loans under Section 2.6 or 2.7 for the portion of the Loans so prepaid and upon payment (including prepayment) in full thereof and, during the existence of any Event of Default, interest shall be paid on demand of the Agent at the request or with the consent of the Required Banks. Interest

on each Swing Line Loan shall be for the sole account of the Swing Line Bank (except to the extent the other Banks have funded the purchase of participations therein pursuant to subsection 2.16(c)).

- (c) Notwithstanding subsection (a) of this Section, if any amount of principal of or interest on any Loan, or any other amount payable hereunder or under any other Loan Document is not paid in full when due (whether at stated maturity, by acceleration, demand or otherwise), the Borrower agrees to pay interest on such unpaid principal or other amount, from the date such amount becomes due to the date such amount is paid in full, and after as well as before any entry of judgment thereon to the extent permitted by law, payable on demand (but not more frequently than once per week), at a fluctuating rate per annum equal to the Base Rate plus 2%.
- (d) Anything herein to the contrary notwithstanding, the obligations of the Borrower hereunder shall be subject to the limitation that payments of interest shall not be required for any period for which interest is computed hereunder, to the extent (but only to the extent) that contracting for or receiving such payment would be contrary to the provisions of any applicable law limiting the highest rate of interest that may be lawfully contracted for, charged or received by the Agent, applicable Bank, Swing Line Bank or Issuing Bank, and in such event the Borrower shall pay such Bank interest for such period at the highest rate permitted by applicable law.
- 2.10 Fees. (a) Arrangement, Agency Fees. The Borrower shall pay all fees as required by the letter agreement ("FEE LETTER") among the Borrower, the Agent and the Arranger dated June 16, 2003.
- (b) Facility Fees. The Borrower shall pay on the last Business Day of each calendar quarter to the Agent for the account of each Bank a facility fee on the daily average amount of (i) such Bank's Revolving Commitment (whether or not used) from the date hereof until the Revolving Termination Date and (ii) such Bank's Acquisition Commitment (whether or not used) from the date hereof until the Acquisition Loan Termination Date, in each case at the rate per annum set forth below for each Pricing Tier as such Pricing Tier is applicable:

Pricing Tier	Funded Debt Ratio	Facility Fee Rate
I	Less than or equal to 1.75x	0.2500%
II	Greater than 1.75 x but less than or equal to 2.75x	0.2500%
III	Greater than 2.75 x but less than or equal to 3.25x	0.3750%
IV	Greater than 3.25 x but less than or equal to 3.75x	0.3750%
V	Greater than 3.75x but less than or equal to 4.25x	0.5000%
VI	Greater than 4.25x but less than or equal to 4.75x	0.5000%
VII	Greater than 4.75	0.5000%

For the purpose of determining the applicable Pricing Tier pursuant to this Section 2.10(b) and subject to the last sentence of this paragraph, EBITDA shall be determined as at the end of each fiscal quarter for the four fiscal quarters then ending and Funded Debt shall be determined as at the end of each fiscal quarter for which such calculation is being determined. Pricing changes shall be effective on the later of (i) 45 days after the end of each of the first three fiscal quarters of each fiscal year and 90 days after each fiscal year end and (ii) the Agent's receipt of financial statements hereunder for such fiscal quarter or fiscal year; provided, however, that if the financial statements are not delivered when due in accordance with Section 7.1, then Pricing Tier VII shall apply as of the first Business Day after the date on which such financial statements were required to have been delivered until the date upon which such financial statements are delivered to the Agent. For the period from the Closing Date through December 29, 2003, the applicable Pricing Tier shall be determined by reference to the Compliance Certificate delivered pursuant to Section 5.1(k).

2.11 Computation of Fees and Interest. (a) All computations of interest for Base Rate Loans when the Base Rate is determined by the Prime Rate shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid, provided that any Loan that is repaid on the same day on which it is made shall, subject to Section 2.12(a), bear interest for one day.

- 2.12 Payments by the Borrower. (a) All payments to be made by the Borrower shall be made without set-off, recoupment or counterclaim. Except as otherwise expressly provided herein, all payments by the Borrower shall be made to the Agent for the account of the Banks at the Agent's Payment Office, and shall be made in dollars and in immediately available funds, no later than 1:00 p.m. (New York City time) on the date specified herein. The Agent will promptly distribute to each Bank its Pro Rata Share (or other applicable share as expressly provided herein) of such payment in like funds as received. Any payment received by the Agent later than 1:00 p.m. (New York City time) shall be deemed to have been received on the following Business Day and any applicable interest or fee shall continue to accrue to such Business Day.
- (b) Subject to the provisions set forth in the definition of "Interest Period" herein, whenever any payment is due on a day other than a Business Day, such payment shall be made on the following Business Day, and such extension of time shall in such case be included in the computation of interest or fees, as the case may be.
- (c) Unless the Agent receives notice from the Borrower prior to the date on which any payment is due to the Banks that the Borrower will not make such payment in full as and when required, the Agent may assume that the Borrower has made such payment in full to the Agent on such date in immediately available funds and the Agent may (but shall not be so required), in reliance upon such assumption, distribute to each Bank on such due date an amount equal to the amount then due such Bank. If and to the extent the Borrower has not made such payment in full to the Agent, each Bank shall repay to the Agent on demand such amount distributed to such Bank, together with interest thereon at the Federal Funds Rate for each day from the date such amount is distributed to such Bank until the date repaid.
- Payments by the Banks to the Agent, etc. (a) Unless the Agent receives notice from a Bank on or prior to the Closing Date or, with respect to any Borrowing after the Closing Date, at least one Business Day prior to the date of such Borrowing, that such Bank will not make available as and when required hereunder to the Agent for the account of the Borrower the amount of that Bank's Pro Rata Share of the Borrowing, the Agent may assume that each Bank has made or will make such amount available to the Agent in immediately available funds on the Borrowing Date and the Agent may (but shall not be so required), in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If and to the extent any Bank shall not have made the full amount of its Pro Rata Share of any Borrowing available to the Agent in immediately available funds and the Agent in such circumstances has made available to the Borrower such amount, that Bank shall on the Business Day following such Borrowing Date make such amount available to the Agent, together with interest at the Federal Funds Rate for each day during such period. A notice of the Agent submitted to any Bank with respect to amounts owing under this subsection (a) shall be conclusive, absent manifest error. If such amount is so made available, such payment to the Agent shall constitute such Bank's Loan on the date of Borrowing for all purposes of this Agreement. If such amount is not made available to the Agent on the Business Day following the Borrowing Date, the Agent will notify the Borrower of such failure to fund and, upon demand by the Agent, the Borrower shall pay such amount to the Agent for the Agent's account, together with interest thereon for each day elapsed since the date of such Borrowing, at a rate per annum equal to the interest rate applicable at the time to the Loans comprising such Borrowing.

- (b) The failure of any Bank to make any Loan on any Borrowing Date shall not relieve any other Bank of any obligation hereunder to make a Loan on such Borrowing Date, but no Bank shall be responsible for the failure of any other Bank to make the Loan to be made by such other Bank on any Borrowing Date. No Bank shall be entitled to take any action to protect or enforce its rights arising out of any Loan Document without the prior written consent of the Required Banks, including the exercise, or attempt to exercise, any right of set-off, banker's lien, or any similar such action, against any deposit account or property of the Borrower held by any such Bank.
- 2.14 Sharing of Payments, etc. If, other than as expressly provided elsewhere herein, any Bank shall obtain on account of the Loans made by it, or the participations in L/C Obligations or in Swing Line Loans held by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise, except pursuant to Sections 2.15, 4.7, 12.1, and 12.9) in excess of its Pro Rata Share, such Bank shall immediately (a) notify the Agent of such fact, and (b) purchase from the other Banks such participations in the Loans made by them as shall be necessary to cause such purchasing Bank to share the excess payment pro rata with each of them; provided, that if all or any portion of such excess payment is thereafter recovered from the purchasing Bank, such purchase shall to that extent be rescinded and each other Bank shall repay to the purchasing Bank the purchase price paid therefor, together with an amount equal to such paying Bank's ratable share (according to the proportion of (i) the amount of such paying Bank's required repayment to (ii) the total amount so recovered from the purchasing Bank) of any interest or other amount paid or payable by the purchasing Bank in respect of the total amount so recovered. The Borrower agrees that any Bank so purchasing a participation from another Bank may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off, but subject to Section 12.11) with respect to such participation as fully as if such Bank were the direct creditor of the Borrower in the amount of such participation. The Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of participations purchased under this Section and will in each case notify the Banks following any such purchases or repayments.
- 2.15 Revolving Termination Date. (a) The Revolving Commitments shall terminate and each Bank shall be relieved of its obligations to make any Revolving Loan on the Revolving Termination Date. The Borrower may from time to time request an extension of the Revolving Termination Date for an additional one-year period by executing and delivering to the Agent a Commitment Termination Date Extension Request at least 60 but not more than 90 days prior to the then scheduled Revolving Termination Date. The Revolving Termination Date shall be so extended if the Agent shall have received from each Bank on or prior to the 30th day preceding the then scheduled Revolving Termination Date a duly executed counterpart of such Commitment Termination Date Extension Request. Each Bank may in its sole and absolute discretion withhold its consent, or condition its consent, to any such Commitment Termination Date Extension Request.
- (b) Notwithstanding the foregoing, if the Agent shall have received duly executed counterparts of a Commitment Termination Date Extension Request from Banks representing, in the aggregate, 80% or more of the Revolving Commitments, but less than 100% of the Revolving Commitments, on or prior to the 30th day preceding the then scheduled Revolving Termination Date, the Agent shall so notify (the date of such notice being the "NOTICE"

DATE") the Borrower, and the Borrower shall have the right to seek a substitute bank or banks (the "NEW BANKS") which New Banks would meet the requirements to be Eligible Assignees, acceptable to the Agent and the Borrower (which may be one or more of the Banks) to replace the Bank or Banks which have not delivered a counterpart of such Commitment Termination Date Extension Request by such time; provided, that such New Banks shall replace such nonrenewing Banks on all such nonrenewing Banks' Commitments, Loans, L/C Obligations and L/C Advances, so the Pro Rata Share of any New Bank of the Acquisition Commitments, Revolving Commitments, Loans, L/C Obligations and L/C Advances shall be the same. If any Revolving Termination Date shall not have been extended pursuant to clause (a) above, the Borrower shall elect, by delivering to the Agent at least four Business Days' prior to the then scheduled Revolving Termination Date a written notice of election, either (i) not to extend such Revolving Termination Date, in which case such Revolving Termination Date shall not be so extended for any Bank irrespective of whether such Bank has or has not sent its duly executed counterpart of the Commitment Termination Date Extension Request or (ii) if the aggregate Revolving Commitments of the Banks who have delivered duly executed counterparts of a Commitment Termination Date Extension Request represent at least 80% of the Revolving Commitments, to extend such current Revolving Termination Date, in which case (x) the Revolving Termination Date shall be extended for an additional period of one year from the then scheduled Revolving Termination Date, and (y) the Revolving Commitments shall be reduced on the then scheduled Revolving Termination Date to an amount equal to the aggregate of the Revolving Commitments of the Banks who had delivered duly executed counterparts of a Commitment Termination Date Extension Request on or prior to the 30th day preceding the then scheduled Revolving Termination Date, plus the aggregate Revolving Commitments of the New Banks and (z) the Commitments shall be reduced on the then scheduled Revolving Termination Date to an amount equal to (1) the aggregate of the Commitments of the Banks who have delivered executed counterparts of a Commitment Termination Date Extension Request on or prior to the 30th day preceding the then scheduled Revolving Termination Date plus (2) the aggregate Commitments of the New Banks, and the Borrower shall pay (such payment to be made on such Revolving Termination Date) in full all Revolving Loans and Acquisition Loans plus all accrued interest and fees (including any amounts owed under Section 4.4) owing to each such non-renewing Bank and each such non-renewing Bank (to the extent that such Loans have not been acquired by the new Banks) shall no longer have any Commitment for purposes of this Agreement and each other Loan Document. If the Borrower shall not have delivered such a written notice of election to the Agent on or prior to the then scheduled Revolving Termination Date, such Revolving Termination Date shall not be extended.

2.16 Swing Line Loans.

(a) The Swing Line. On the terms and subject to the conditions set forth in Section 5.1 (in the case of any Swing Line Loan to be made on the Closing Date), Section 5.2 and this Section 2.16, the Swing Line Bank agrees to make loans (each such loan, a "SWING LINE LOAN") to the Borrower from time to time on any Business Day during the period from the Closing Date to the Revolving Termination Date in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit, notwithstanding the fact that such Swing Line Loans, when aggregated with the Pro Rata Share of the Effective Amount of Revolving Loans and L/C Obligations of the Bank acting as Swing Line Bank, may exceed the amount of the Swing Line Bank's Revolving Commitment; provided, however, that after giving

42

effect to any Swing Line Loan, (i) the aggregate Effective Amount of all Revolving Loans, all Swing Line Loans and all L/C Obligations shall not exceed the Revolving Commitments of all the Banks, and (ii) the aggregate Effective Amount of the Revolving Loans of any Bank, plus such Bank's Pro Rata Share of the Effective Amount of all L/C Obligations, plus such Bank's Pro Rata Share of the Effective Amount of all Swing Line Loans shall not exceed such Bank's Revolving Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.16, prepay under Section 2.6, and reborrow under this Section 2.16. Each Swing Line Loan shall be a Base Rate Loan. Immediately upon the making of a Swing Line Loan, each Bank shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Bank a risk participation in such Swing Line Loan in an amount equal to the product of such Bank's Pro Rata Share times the amount of such Swing Line Loan.

(b) Borrowing Procedures. Each Borrowing of Swing Line Loans shall be made upon the Borrower's irrevocable notice to the Swing Line Bank and the Agent, which may be given by telephone. Each such notice must be received by the Swing Line Bank and the Agent not later than 1:00 p.m. (New York City time) on the requested borrowing date, and shall specify (i) the amount to be borrowed, which shall be a minimum of \$1,000,000 or any multiple of \$1,000,000 in excess thereof and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic notice must be confirmed promptly by delivery to the Swing Line Bank and the Agent of a written Notice of Borrowing, appropriately completed and signed by a Responsible Officer of the Borrower. Promptly after receipt by the Swing Line Bank of any telephonic notice of borrowing of Swing Line Loans, the Swing Line Bank will confirm with the Agent (by telephone or in writing) that the Agent has also received such notice of borrowing and, if not, the Swing Line Bank will notify the Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Bank (x) has received notice (by telephone or in writing) from the Agent (including at the request of any Bank) prior to 2:00 p.m. (New York City time) on the date of the proposed Borrowing (A) directing the Swing Line Bank not to make such Swing Line Loan as a result of the limitations set forth in the proviso to the first sentence of Section 2.16(a), or (B) that one or more of the applicable conditions specified in Article V is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Bank will, not later than 3:00 p.m. (New York City time) on the borrowing date specified in such Notice of Borrowing, make the amount of its Swing Line Loan available to the Borrower.

(c) Refinancing of Swing Line Loans.

(i) The Swing Line Bank at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably authorizes the Swing Line Bank to so request on its behalf), that each Bank make a Base Rate Loan in an amount equal to such Bank's Pro Rata Share of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Notice of Borrowing for purposes hereof) and in accordance with the requirements of Section 2.3, without regard to the minimum and multiples specified therein for the principal amount of Base Rate Loans. The Swing Line Bank shall furnish the Borrower with a copy of the applicable Notice of Borrowing promptly after delivering such notice to the Agent. Each Bank shall make an amount equal to its Pro Rata Share of the amount specified in such Notice of Borrowing

available to the Agent in immediately available funds for the account of the Swing Line Bank at the Agent's Office not later than 1:00 p.m. (New York City time) on the day specified in such Notice of Borrowing, whereupon, subject to Section 2.16(c)(ii), each Bank that so makes funds available shall be deemed to have made a Base Rate Loan to the Borrower in such amount. The Agent shall remit the funds so received to the Swing Line Bank.

- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Borrowing in accordance with Section 2.16(c)(i), the request for Base Rate Loans submitted by the Swing Line Bank as set forth herein shall be deemed to be a request by the Swing Line Bank that each of the Banks fund its risk participation in the relevant Swing Line Loan and each Bank's payment to the Agent for the account of the Swing Line Bank pursuant to Section 2.16(c)(i) shall be deemed payment in respect of such participation.
- (iii) If any Bank fails to make available to the Agent for the account of the Swing Line Bank any amount required to be paid by such Bank pursuant to the foregoing provisions of this Section 2.16(c) by the time specified in Section 2.16(c)(i), the Swing Line Bank shall be entitled to recover from such Bank (acting through the Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Bank at a rate per annum equal to the Federal Funds Rate from time to time in effect. A certificate of the Swing Line Bank submitted to any Bank (through the Agent) with respect to any amounts owing under this clause (iii) shall be conclusive absent manifest error.
- (iv) Each Bank's obligation to make Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.16(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Bank may have against the Swing Line Bank, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Bank's obligation to make Loans pursuant to this Section 2.16(c) is subject to the conditions that the Swing Line Loan was made by the Swing Line Bank in accordance with Sections 2.16(a) and 2.16(b). No such funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.
 - (d) Repayment of Participations.
- (i) At any time after any Bank has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Bank receives any payment on account of such Swing Line Loan, the Swing Line Bank will distribute to such Bank its Pro Rata Share of such payment (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Bank's risk participation was funded) in the same funds as those received by the Swing Line Bank.
- (ii) If any payment received by the Swing Line Bank in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Bank under any of the circumstances described in Section 12.7 (including pursuant to any settlement

entered into by the Swing Line Bank in its discretion), each Bank shall pay to the Swing Line Bank its Pro Rata Share thereof on demand of the Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the Federal Funds Rate. The Agent will make such demand upon the request of the Swing Line Bank.

- (e) Interest for Account of Swing Line Bank. The Swing Line Bank shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Bank funds its Base Rate Loan or risk participation pursuant to this Section 2.16 to refinance such Bank's Pro Rata Share of any Swing Line Loan, interest in respect of such Pro Rata Share shall be solely for the account of the Swing Line Bank.
- (f) Payments Directly to Swing Line Bank. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans to the Agent for the benefit of the Swing Line Bank.

ARTICLE III

THE LETTERS OF CREDIT

- 3.1 The Letter of Credit Subfacility. (a) On the terms and subject to the conditions set forth herein (i) the Issuing Bank agrees, in reliance upon the agreements of the other Banks set forth in this Article III, (A) from time to time on any Business Day during the period from the Closing Date to the Revolving Termination Date to issue Letters of Credit for the account of the Borrower, and to amend or renew, extend the expiration of or increase the amount of Letters of Credit previously issued by it, in accordance with Sections 3.2(c) and 3.2(d), and (B) to honor drafts under the Letters of Credit; and (ii) the Banks severally agree to participate in Letters of Credit Issued for the account of the Borrower; provided, that the Issuing Bank shall not be obligated to Issue, and no Bank shall be obligated to participate in, any Letter of Credit if as of the date of Issuance of such Letter of Credit (the "ISSUANCE DATE") (x) the Effective Amount of all L/C Obligations plus the Effective Amount of all Revolving Loans plus the Effective Amount of all Swing Line Loans exceeds the amount of the Revolving Commitment or (y) the Effective Amount of all L/C Obligations exceeds \$100,000,000. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower's ability to obtain Letters of Credit shall be fully revolving, and, accordingly, the Borrower may, during the foregoing period, obtain Letters of Credit to replace Letters of Credit which have expired or which have been drawn upon and reimbursed. The Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.
- (b) The Issuing Bank shall not Issue any Letter of Credit if:
- (i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuing Bank from Issuing such Letter of Credit, or any Requirement of Law applicable to the Issuing Bank or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuing Bank shall prohibit, or request that the Issuing Bank refrain from, the Issuance of letters of credit generally or such Letter of Credit in particular or shall impose

upon the Issuing Bank with respect to such Letter of Credit any restriction, reserve or capital requirement (for which the Issuing Bank is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon the Issuing Bank any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the Issuing Bank in good faith deems material to it;

- (ii) the Issuing Bank has received written notice from any Bank, the Agent or the Borrower, on or prior to the Business Day prior to the requested date of Issuance of such Letter of Credit, that one or more of the applicable conditions contained in Article V is not then satisfied;
- (iii) the expiry date of any requested Letter of Credit is (A) more than twelve months after the date of Issuance, unless the Required Banks have approved such expiry date in writing, or (B) after the Revolving Termination Date, unless the Agent, the Issuing Bank and all of the Banks have approved such expiry date in writing;
- (iv) any requested Letter of Credit is now otherwise in form and substance acceptable to the Issuing Bank, or the Issuance of a Letter of Credit shall violate any applicable policies of the Issuing Bank;
- (v) such Letter of Credit is in a face amount less than \$500,000 or to be denominated in a currency other than Dollars.
- 3.2 Issuance, Amendment and Renewal of Letters of Credit. (a) Each Letter of Credit shall be issued upon the irrevocable written request of the Borrower received by the Issuing Bank (with a copy sent by the Borrower to the Agent) not later than 11:00 a.m. (New York City time) at least two Business Days (or such shorter time as the Issuing Bank may agree in a particular instance in its sole discretion) prior to the proposed date of issuance. Each such request for issuance of a Letter of Credit shall be by facsimile, confirmed promptly in an original writing, in the form of an L/C Application, appropriately completed and signed by a Responsible Officer of the Borrower, and shall specify in form and detail satisfactory to the Issuing Bank: (i) the proposed date of issuance of the Letter of Credit (which shall be a Business Day); (ii) the face amount of the Letter of Credit; (iii) the expiry date of the Letter of Credit; (iv) the name and address of the beneficiary thereof; (v) the documents to be presented by the beneficiary of the Letter of Credit in case of any drawing thereunder; (vi) the full text of any certificate to be presented by the beneficiary in case of any drawing thereunder; and (vii) such other matters as the Issuing Bank may require.
- (b) Promptly after receipt of any L/C Application, the Issuing Bank will confirm with the Agent (by telephone or in writing) that the Agent has received a copy of such L/C Application from the Borrower and, if not, the Issuing Bank will provide the Agent with a copy thereof. Upon receipt by the Issuing Bank of confirmation from the Agent that the requested issuance or amendment is permitted in accordance with the terms hereof, then, subject to the terms and conditions hereof, the Issuing Bank shall, on the requested date, issue a Letter of Credit for the account of the Borrower or enter into the applicable amendment, as the case may be, in each case in accordance with the Issuing Bank's usual and customary business practices. Immediately upon the Issuance of each Letter of Credit, each Bank shall be deemed to, and

hereby irrevocably and unconditionally agrees to, purchase from the Issuing Bank a risk participation in such Letter of Credit in an amount equal to the product of such Bank's Pro Rata Share times the amount of such Letter of Credit.

- (c) From time to time while a Letter of Credit is outstanding and prior to the Revolving Termination Date, the Issuing Bank will, upon the written request of the Borrower received by the Issuing Bank (with a copy sent by the Borrower to the Agent) not later than 11:00 a.m. (New York City time) at least two Business Days (or such shorter time as the Issuing Bank may agree in a particular instance in its sole discretion) prior to the proposed date of amendment, amend any Letter of Credit issued by it. Each such request for amendment of a Letter of Credit shall be made by facsimile made in the form of an L/C Amendment Application and shall specify in form and detail satisfactory to the Issuing Bank: (i) the Letter of Credit to be amended; (ii) the proposed date of amendment of the Letter of Credit (which shall be a Business Day); (iii) the nature of the proposed amendment; and (iv) such other matters as the Issuing Bank may reasonably require. The Issuing Bank shall not amend any Letter of Credit if: (A) the Issuing Bank would have no obligation at such time to issue such Letter of Credit in its amended form under the terms of this Agreement; or (B) the beneficiary of any such Letter of Credit does not accept the proposed amendment to the Letter of Credit. The Agent will promptly notify the Banks of the receipt by it of any L/C Application or L/C Amendment Application.
- If the Borrower so requests in any applicable L/C Application, the Issuing Bank may, in its sole and absolute discretion, agree to issue a Letter of Credit that has automatic renewal provisions (each, an "AUTO-RENEWAL LETTER OF CREDIT"); provided that any such Auto-Renewal Letter of Credit must permit the Issuing Bank to prevent any such renewal at least once in each twelve month period (commencing with the date of issuance of such Letter of Credit) by giving prior notice to the beneficiary thereof not later than a day in each such twelve month period (the "NONRENEWAL NOTICE DATE") to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by the Issuing Bank, the Borrower shall not be required to make a specific request to the Issuing Bank for any such renewal. Once an Auto-Renewal Letter of Credit has been issued, the Banks shall be deemed to have authorized (but may not require) the Issuing Bank to permit the renewal of such Letter of Credit at any time to an expiry date not later than the Revolving Termination Date; provided, however, that the Issuing Bank shall not permit any such renewal if (A) the Issuing Bank has determined that it would have no obligation at such time to issue such Letter of Credit in its renewed form under the terms hereof (by reason of the provisions of Section 3.1(b) or otherwise), or (B) it has received notice (which may be by telephone or in writing) on or before the day that is two Business Days before the Nonrenewal Notice Date (1) from the Agent that the Required Banks have elected not to permit such renewal or (2) from the Agent, any Bank or the Borrower that one or more of the applicable conditions specified in Section 5.2 is not then satisfied.
- (e) The Issuing Bank may, at its election (or as required by the Agent at the direction of the Required Banks), deliver any notices of termination or other communications to any Letter of Credit beneficiary or transferee, and take any other action as necessary or appropriate, at any time and from time to time, in order to cause the expiry date of such Letter of Credit to be a date not later than the Revolving Termination Date.

- (f) This Agreement shall control in the event of any conflict with any L/C-Related Document (other than any Letter of Credit).
- (g) The Issuing Bank will also deliver to the Agent, concurrently or promptly following its delivery of a Letter of Credit, or amendment to or renewal of a Letter of Credit, to an advising bank or a beneficiary, a true and complete copy of each such Letter of Credit or amendment to or renewal of a Letter of Credit.
 - 3.3 Risk Participations, Drawings and Reimbursements.
- (a) Immediately upon the Issuance of each Letter of Credit, each Bank shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Issuing Bank a participation in such Letter of Credit and each drawing thereunder in an amount equal to the product of (i) the Pro Rata Share of such Bank times (ii) the maximum amount available to be drawn under such Letter of Credit and the amount of each such drawing, respectively.
- In the event of any request for a drawing under a Letter of Credit by the beneficiary or transferee thereof, the Issuing Bank will promptly notify the Borrower and the Agent. The Borrower shall reimburse the Issuing Bank prior to 1:00 p.m. (New York City time), on each date that any amount is paid by the Issuing Bank under any Letter of Credit (each such date, an "HONOR DATE"), in an amount equal to the amount so paid by the Issuing Bank. In the event the Borrower fails to reimburse the Issuing Bank for the full amount of any drawing under any Letter of Credit by 1:00 p.m. (New York City time) on the Honor Date, the Issuing Bank will promptly notify the Agent and the Agent will promptly notify each Bank thereof, and the Borrower shall be deemed to have requested that Base Rate Loans be made by the Banks to be disbursed on the Honor Date under such Letter of Credit, subject to the amount of the unutilized portion of the Revolving Commitment. Any notice given by the Issuing Bank or the Agent pursuant to this Section 3.3(b) may be given by telephone if immediately confirmed in writing (including by facsimile); provided, that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.
- (c) Each Bank shall upon any notice pursuant to Section 3.3(b) make available to the Agent for the account of the Issuing Bank an amount in Dollars and in immediately available funds equal to its Pro Rata Share of the amount of the drawing, whereupon the participating Banks shall (subject to Section 3.3(e)) each be deemed to have made a Revolving Loan consisting of a Base Rate Loan to the Borrower in that amount. The Agent shall remit the funds so received to the Issuing Bank. If any Bank so notified fails to make available to the Agent for the account of the Issuing Bank the amount of such Bank's Pro Rata Share of the amount of the drawing by no later than 3:00 p.m. (New York City time) on the Honor Date, then interest shall accrue on such Bank's obligation to make such payment, from the Honor Date to the date such Bank makes such payment, at a rate per annum equal to the Federal Funds Rate in effect from time to time during such period. The Agent will promptly give notice of the occurrence of the Honor Date, but failure of the Agent to give any such notice in sufficient time to enable any Bank to effect such payment on such date shall not relieve such Bank from its obligations under this Section 3.3 (other than the obligation to pay interest for the period prior to the notice).

- (d) With respect to any unreimbursed drawing that is not converted into Revolving Loans consisting of Base Rate Loans to the Borrower in whole or in part for any reason, the Borrower shall be deemed to have incurred from the Issuing Bank an L/C Borrowing in the amount of such drawing, which L/C Borrowing shall be due and payable on demand (together with interest) and shall bear interest at a rate per annum equal to the Base Rate plus 2%, and each Bank's payment to the Agent for the account of the Issuing Bank pursuant to Section 3.3(c) shall be deemed payment in respect of its participation in such L/C Borrowing and shall constitute an L/C Advance from such Bank in satisfaction of its participation obligation under this Section 3.3.
- (e) Each Bank's obligation in accordance with this Agreement to make Revolving Loans or L/C Advances to reimburse the Issuing Bank for amounts drawn under Letters of Credit, as contemplated by this Section 3.3, shall be absolute and unconditional and without recourse to the Issuing Bank and shall not be affected by any circumstance, including (i) any set-off, counterclaim, recoupment, defense or other right which such Bank may have against the Issuing Bank, the Borrower or any other Person for any reason whatsoever; (ii) the occurrence or continuance of a Default, an Event of Default or a Material Adverse Effect; or (iii) any other occurrence, circumstance, happening, event or condition whatsoever, whether or not similar to any of the foregoing; provided, however, that each Bank's obligation to make Revolving Loans under this Section 3.3 is subject to the conditions set forth in Section 5.2 (other than delivery by the Borrower of a Notice of Borrowing). No such making of an L/C Advance shall relieve or otherwise impair the obligation of the Borrower to reimburse the Issuing Bank for the amount of any payment made by the Issuing Bank under any Letter of Credit, together with interest as provided herein.
- 3.4 Repayment of Participations. (a) Upon (and only upon) receipt by the Agent for the account of the Issuing Bank of immediately available funds from the Borrower (i) in reimbursement of any payment made by the Issuing Bank under the Letter of Credit with respect to which any Bank has paid the Agent for the account of the Issuing Bank for such Bank's participation in the Letter of Credit pursuant to Section 3.3 or (ii) in payment of interest thereon, the Agent will pay to each Bank, in the same funds as those received by the Agent for the account of the Issuing Bank, the amount of such Bank's Pro Rata Share of such funds, and the Issuing Bank shall receive the amount of the Pro Rata Share of such funds of any Bank that did not so pay the Agent for the account of the Issuing Bank.
- (b) If the Agent or the Issuing Bank is required at any time to return to the Borrower, or to a trustee, receiver, liquidator, custodian, or any official in any Insolvency Proceeding, any portion of the payments made by the Borrower to the Agent for the account of the Issuing Bank pursuant to Section 3.4(a) in reimbursement of a payment made under any Letter of Credit or interest or fee thereon, each Bank shall, on demand of the Agent, forthwith return to the Agent for the account of the Issuing Bank the amount of its Pro Rata Share of any amounts so returned by the Agent for the account of the Issuing Bank plus interest thereon from the date such demand is made to the date such amounts are returned by such Bank to the Agent, at a rate per annum equal to the Federal Funds Rate in effect from time to time.
- 3.5 Role of the Issuing Bank. (a) Each Bank and the Borrower agree that, in paying any drawing under a Letter of Credit, the Issuing Bank shall not have any responsibility to obtain

any document (other than any sight draft, certificates and documents expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document.

- (b) None of the Issuing Bank, the Agent-Related Persons or any of the respective correspondents, participants or assignees of the Issuing Bank or the Agent-Related Persons shall be liable to any Bank for: (i) any action taken or omitted in connection herewith at the request or with the approval of the Banks (including the Required Banks, as applicable); (ii) any action taken or omitted in the absence of negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any L/C-Related Document.
- (c) Except as otherwise provided in this clause (c), the Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuing Bank, the Agent-Related Persons or any of the respective correspondents, participants or assignees of the Issuing Bank or the Agent-Related Persons shall be liable or responsible for any of the matters described in clauses (i) through (vii) of Section 3.6; provided, that anything in such clauses to the contrary notwithstanding, the Borrower may have a claim against the Issuing Bank, and the Issuing Bank may be liable to the Borrower, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Borrower which the Borrower proves were caused by the Issuing Bank's willful misconduct or gross negligence or the Issuing Bank's willful failure to pay under any Letter of Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Letter of Credit. In furtherance and not in limitation of the foregoing: (i) the Issuing Bank may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary; and (ii) the Issuing Bank shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.
- 3.6 Obligations Absolute. Subject to the proviso contained in the second sentence of Section 3.5(c), the obligations of the Borrower under this Agreement and any L/C-Related Document to reimburse the Issuing Bank for a drawing under a Letter of Credit, and to repay any L/C Borrowing and any drawing under a Letter of Credit converted into Revolving Loans, shall be unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement and each such other L/C-Related Document under all circumstances, including the following:
- (i) any lack of validity or enforceability of this Agreement or any L/C-Related Document;
- (ii) any change in the time, manner or place of payment of, or in any other term of, all or any of the obligations of the Borrower in respect of any Letter of Credit or

any other amendment or waiver of or any consent to departure from all or any of the L/C-Related Documents:

- (iii) the existence of any claim, set-off, defense or other right that the Borrower may have at any time against any beneficiary or any transferee of any Letter of Credit (or any Person for whom any such beneficiary or any such transferee may be acting), the Issuing Bank or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by the L/C-Related Documents or any unrelated transaction;
- (iv) any draft, demand, certificate or other document presented under any Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under any Letter of Credit;
- (v) any payment by the Issuing Bank under any Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of any Letter of Credit; or any payment made by the Issuing Bank under any Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of any Letter of Credit, including any arising in connection with any Insolvency Proceeding;
- (vi) any exchange, release or non-perfection of any collateral, or any release or amendment or waiver of or consent to departure from any other guarantee, for all or any of the obligations of the Borrower in respect of any Letter of Credit; or
- (vii) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, any Obligor or a guarantor.

The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will promptly notify the Issuing Bank. The Borrower shall be conclusively deemed to have waived any such claim against the Issuing Bank and its correspondents unless such notice is given as aforesaid.

- 3.7 Cash Collateral Pledge. Upon the request of the Agent, (i) if the Issuing Bank has honored any full or partial drawing request under any Letter of Credit and such drawing has resulted in an L/C Borrowing, or (ii) if, as of the Revolving Termination Date, any Letter of Credit may for any reason remain outstanding and partially or wholly undrawn, the Borrower shall immediately Cash Collateralize the then Effective Amount of all L/C Obligations (in an amount equal to such Effective Amount determined as of the date of such L/C Borrowing or the Revolving Termination Date, as the case may be). Cash collateral shall be maintained in blocked, non-interest bearing deposit accounts at Wachovia.
- 3.8 Letter of Credit Fees. (a) The Borrower shall pay to the Agent for the account of each Bank in accordance with its Pro Rata Share a Letter of Credit fee for each Letter of

Credit equal to the Applicable Margin for Revolving Loans consisting of Eurodollar Rate Loans times the average daily maximum amount available to be drawn under such Letter of Credit during the period of determination. Such letter of credit fees shall be computed on a quarterly basis in arrears. If there is any change in the Applicable Margin during any quarter, the daily maximum amount of each Letter of Credit shall be computed and multiplied by the Applicable Margin separately for each period during such quarter that such Applicable Margin was in effect.

- (b) The Borrower shall pay, to the Issuing Bank quarterly, a letter of credit fronting fee for each Letter of Credit Issued by the Issuing Bank equal to 0.125% per annum times the average daily maximum amount available to be drawn under such Letter of Credit, as computed by the Agent.
- (c) The letter of credit fees payable under Section 3.8(a) and the fronting fees payable under Section 3.8(b) shall be due and payable on the last Business Day of each calendar quarter, commencing with the first such date to occur after the Closing Date, on the Revolving Termination Date and thereafter on demand.
- (d) In addition, the Borrower shall pay directly to the Issuing Bank for its own account the customary issuance, presentation, amendment and other processing fees, and other standard costs and charges, of the Issuing Bank relating to letters of credit as from time to time in effect. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.
- agreed by the Issuing Bank and the Borrower when a Letter of Credit is issued (including any such agreement applicable to the Existing Letters of Credit), (i) the rules of the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance) (the "ISP98") shall apply to each standby Letter of Credit, and (ii) the rules of the Uniform Customs and Practice for Documentary Credits, as most recently published by the International Chamber of Commerce (the "ICC") at the time of issuance (including the ICC decision published by the Commission on Banking Technique and Practice on April 6, 1998 regarding the European single currency (euro)) shall apply to each commercial Letter of Credit.
- 3.10 Conflict with L/C Application. In the event of any conflict between the terms hereof and the terms of any L/C Application, the terms hereof shall control.

ARTICLE IV

TAXES, YIELD PROTECTION AND ILLEGALITY

- 4.1 Taxes. (a) Except as provided in Section 4.1(c), any and all payments by the Borrower to each Bank or the Agent under this Agreement and any other Loan Document shall be made free and clear of, and without deduction or withholding for any Taxes. In addition, the Borrower shall pay all Other Taxes.
- (b) The Borrower agrees to indemnify and hold harmless each Bank and the Agent for the full amount of Taxes or Other Taxes including any Taxes or Other Taxes imposed

by any jurisdiction on amounts payable under this Section paid by the Bank or the Agent and any liability (including penalties, interest, additions to tax and expenses arising therefrom or with respect thereto, whether or not such Taxes or Other Taxes were correctly or legally asserted). Payment under this indemnification shall be made within 30 days after the date the Bank or the Agent provides written proof of payment of the related Taxes or Other Taxes to the Borrower. Such written proof shall be conclusive absent manifest error.

- (c) If the Borrower shall be required by law to deduct or withhold any Taxes or Other Taxes from or in respect of any sum payable hereunder to any Bank or the Agent, then:
- (i) the sum payable shall be increased as necessary so that after making all required deductions and withholdings (including deductions and withholdings applicable to additional sums payable under this Section) such Bank or the Agent, as the case may be, receives an amount equal to the sum it would have received had no such deductions or withholdings been made;
 - (ii) the Borrower shall make such deductions and

withholdings;

- (iii) the Borrower shall pay the full amount deducted or withheld to the relevant taxing authority or other authority in accordance with applicable law; and
- (iv) the Borrower shall also pay to each Bank or the Agent for the account of such Bank, at the time interest is paid, all additional amounts which the respective Bank specifies as necessary to preserve the after-tax yield the Bank would have received if such Taxes or Other Taxes had not been imposed.
- (d) Within 30 days after their receipt of a written request therefor by Agent, the Borrower shall furnish the Agent the original or a certified copy of a receipt evidencing any payment by the Borrower of Taxes or Other Taxes, or other evidence of payment satisfactory to the Agent.
- (e) If the Borrower is required to pay additional amounts to any Bank or the Agent pursuant to subsection (c) of this Section, then such Bank shall use reasonable efforts (consistent with legal and regulatory restrictions) to change the jurisdiction of its Lending Office so as to eliminate any such additional payment by the Borrower which may thereafter accrue, if such change in the judgment of such Bank is not illegal or otherwise disadvantageous to such Bank.
- (f) No Foreign Bank shall be entitled to claim that the provisions of this Section 4.1 apply to it with respect to Taxes unless such Foreign Bank shall have delivered to the Agent and the Borrower, prior to the time that any payments are to be made under this Agreement to such Foreign Bank, a properly completed (i) Treasury Form W-8ECI, specifying that the payments to be received by such Foreign Bank pursuant to this Agreement are effectively connected with the conduct of a United States trade or business or (ii) Treasury Form W-8BEN, specifying that the payments to be received by such Foreign Bank pursuant to this Agreement are wholly exempt from United States federal income tax pursuant to the provisions of an applicable income tax treaty with the United States and, in either case, has otherwise complied with Section 10.13 hereof. Each Foreign Bank that shall have provided a Form W-

8ECI or a Form W-8BEN to the Agent and the Borrower, if permitted by law, shall be required to provide the Borrower with a new form (also showing no withholding) no later than 3 years from the date that it provided the original form to the Agent and the Borrower in order to claim advantage of this Section 4.1 from and after such time.

- 4.2 Illegality. (a) If the introduction after the date hereof of any Requirement of Law, or any change after the date hereof in any Requirement of Law, or in the interpretation or administration of any Requirement of Law, has made it unlawful, or that any central bank or other Governmental Authority has asserted after the date hereof that it is unlawful, for any Bank or its applicable Lending Office to make Eurodollar Rate Loans, then, on notice thereof by the Bank to the Borrower through the Agent, any obligation of that Bank to make Eurodollar Rate Loans shall be suspended until the Bank notifies the Agent and the Borrower that the circumstances giving rise to such determination no longer exist.
- (b) If it is unlawful for any Bank to maintain any Eurodollar Rate Loan, the Borrower shall, upon receipt by the Borrower of notice of such fact and demand from such Bank (such notice to be delivered through the Agent), prepay in full such Eurodollar Rate Loans of that Bank then outstanding, together with interest accrued thereon, either on the last day of the Interest Period thereof, if the Bank may lawfully continue to maintain such Eurodollar Rate Loans to such day, or immediately, if the Bank may not lawfully continue to maintain such Eurodollar Rate Loan. If the Borrower is required to so prepay any Eurodollar Rate Loan, then concurrently with such prepayment, the Borrower shall borrow from the affected Bank, in the amount of such prepayment, a Base Rate Loan.
- (c) If the obligation of any Bank to make or maintain Eurodollar Rate Loans has been so terminated or suspended, the Borrower may elect, by giving notice to the Bank through the Agent that all Loans which would otherwise be made by the Bank as Eurodollar Rate Loans shall be instead Base Rate Loans.
- (d) Before giving any notice to the Agent under this Section, the affected Bank shall use reasonable efforts (consistent with legal and regulatory restrictions) to change the jurisdiction of its Lending Office with respect to its Eurodollar Rate Loans if such change will avoid the need for giving such notice or making such demand and will not, in the judgment of the Bank, be illegal or otherwise disadvantageous to the Bank.
- 4.3 Increased Costs and Reduction of Return. (a) If, due to either (i) the introduction after the date hereof of, or any change after the date hereof (other than any change by way of imposition of or increase in reserve requirements included in the calculation of the Eurodollar Rate or in respect of the assessment rate payable by any Bank to the FDIC for insuring U.S. deposits) in or in the interpretation of any law or regulation applicable to any Bank (other than any such introduction or change announced prior to the date hereof) or (ii) the compliance by any Bank with any guideline or request from any central bank or other Governmental Authority (whether or not having the force of law) not in effect prior to the date hereof, there shall be any increase in the cost to such Bank of agreeing to make or making, funding or maintaining any Eurodollar Rate Loans, or participating in Letters of Credit, or, in the case of the Issuing Bank, any increase in the cost to the Issuing Bank of agreeing to issue, issuing or maintaining any Letter of Credit or of agreeing to make or making, funding or maintaining

any unpaid drawing under any Letter of Credit, then the Borrower shall be liable for, and shall from time to time, upon demand (such demand to be delivered through the Agent), pay to the Agent for the account of such Bank or the Issuing Bank, as the case may be, additional amounts as are sufficient to compensate such Bank or the Issuing Bank, as the case may be, for such increased costs.

- If (i) the introduction of any Capital Adequacy Regulation, (ii) any change in any Capital Adequacy Regulation, (iii) any change in the interpretation or administration of any Capital Adequacy Regulation by any central bank or other Governmental Authority charged with the interpretation or administration thereof, or (iv) compliance by any Bank (or its Lending Office) or any corporation controlling the Bank with any Capital Adequacy Regulation, in each case occurring after the date hereof, affects or would affect the amount of capital required or expected to be maintained by the Bank or any corporation controlling the Bank and (taking into consideration such Bank's or such corporation's commercially reasonable policies with respect to capital adequacy and such Bank's or such corporation's desired return on capital) the amount of such capital is increased as a consequence of its Commitment, loans, credits or obligations under this Agreement, then, upon written demand of such Bank to the Borrower through the Agent, the Borrower shall pay to the Agent for the account of such Bank, from time to time as specified by the Bank or such controlling corporation, additional amounts sufficient to compensate the Bank for such increase.
- 4.4 Funding Losses. Excluding losses or expenses incurred by a Bank pursuant to Section 4.2 (other than in connection with Section 4.2(b)), the Borrower shall reimburse each Bank and hold each Bank harmless from any loss or expense (but excluding in any event all consequential or exemplary damages) which the Bank may sustain or incur as a consequence of:
- (a) the failure of the Borrower to make on a timely basis any payment of principal of any Eurodollar Rate Loan;
- (b) the failure of the Borrower to borrow, continue or convert into a Eurodollar Rate Loan after the Borrower has given (or is deemed to have given) a Notice of Borrowing or a Notice of Conversion/Continuation (except as a result of a breach by a Bank of its obligations hereunder);
- (c) the failure of the Borrower to make any prepayment in accordance with any notice delivered under Section 2.6;
- (d) the repayment or prepayment (including pursuant to Sections 2.7 and 4.2(b)) or other payment (including after acceleration thereof) of a Eurodollar Rate Loan on a day that is not the last day of the relevant Interest Period: or
- (e) the automatic conversion under Section 2.4 of any Eurodollar Rate Loan to a Base Rate Loan on a day that is not the last day of the relevant Interest Period;

including any such loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain its Eurodollar Rate Loans or from fees payable to terminate the deposits from which such funds were obtained. For purposes of calculating amounts payable by the Borrower to the Banks under this Section and under Section 4.3(a), each Eurodollar Rate

Loan made by a Bank (and each related reserve, special deposit or similar requirement) shall be conclusively deemed to have been funded at the Eurodollar Base Rate used in determining the Eurodollar Rate for such Eurodollar Rate Loan by matching deposit or other borrowing in the interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Rate Loan is in fact so funded. Each Bank shall exercise its reasonable efforts to minimize such losses, costs and expenses, except that each Bank shall not be obligated to take any action to reduce net balances due to its non-U.S. offices from its U.S. offices.

- 4.5 Inability to Determine Rates. If the Agent or the Required Banks determine that for any reason adequate and reasonable means do not exist for determining the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan, or that the Eurodollar Rate applicable pursuant to Section 2.9(a) for any requested Interest Period with respect to a proposed Eurodollar Rate Loan does not adequately and fairly reflect the cost to the Banks of funding such Loan, the Agent will promptly so notify the Borrower and each Bank. Thereafter, the obligation of the Banks to make or maintain Eurodollar Rate Loans hereunder shall be suspended until the Agent upon the instruction of the Required Banks revokes such notice in writing. Upon receipt of such notice, the Borrower may revoke any Notice of Borrowing or Notice of Conversion/Continuation then submitted by it. If the Borrower does not revoke such Notice, the Banks shall make, convert or continue the Loans, as proposed by the Borrower, in the amount specified in the applicable notice submitted by the Borrower, but such Loans shall be made, converted or continued as Base Rate Loans instead of Eurodollar Rate Loans.
- Certificates of Banks. Except as specifically provided in Section 4.1, any Bank claiming reimbursement or compensation under this Article IV shall deliver to the Borrower (with a copy to the Agent) a certificate setting forth in reasonable detail the amount payable to the Bank hereunder and the circumstances giving rise to such claim, and such certificate shall be prima facie evidence of the correctness thereof. Each Bank agrees to deliver such certificate to the Borrower within reasonable time after it determines the additional amount required to be paid under this Article IV; provided, however, that in no event shall any Bank deliver such Certificate to the Borrower more than 180 days after any vice-president of such Bank knows, or upon the discharge of such vice-president's duties in the ordinary course should have known, of the occurrence of an event giving rise to the additional amount required to be paid in respect of this Article IV and if it fails to deliver such Certificate within such 180 day period, the Borrower will not be obligated for any costs incurred prior to 180 days before such notice. The Borrower shall pay such Bank the amount shown as due on any such certificate timely delivered in accordance with the foregoing within ten days after its receipt of the same; provided, however, that the Borrower shall not be required to pay any amounts (other than with respect to Taxes under Section 4.1) which were due for any period occurring more than 90 days prior to the Borrower's receipt of such certificate (other than periods with respect to which such costs or expenses are retroactively imposed). This Article IV shall survive termination of this Agreement and payment of the outstanding Notes. Notwithstanding the foregoing provisions of this Article IV, the Borrower shall not be liable for any increased cost pursuant to this Article IV if and to the extent that such increased cost results from the change in any Bank's Lending Office and such change (x) is made solely in the discretion of such Bank and not required by any applicable Requirement of Law or Governmental Authority, (y) is made for such Bank's benefit and without any benefit to the Borrower, and (z) results, at the time of such change, in an increased cost greater than that

56

which would have been incurred had the Bank not so changed its Lending Office. Each Bank shall use its reasonable efforts to avoid or minimize increased costs under this Article IV unless, in the sole opinion of such Bank, such action would adversely affect it.

- 4.7 Substitution of Banks. Upon the receipt by the Borrower from any Bank (an "AFFECTED BANK") of a claim for compensation under Section 4.3, the Borrower may: (i) request the Affected Bank to use its reasonable efforts to obtain a replacement bank or financial institution satisfactory to the Borrower to acquire and assume all or a ratable part of all of such Affected Bank's Loans and Commitments (a "REPLACEMENT BANK"); (ii) request one or more of the other Banks to acquire and assume all or part of such Affected Bank's Loans and Commitments; or (iii) designate a Replacement Bank. Any such designation of a Replacement Bank under clause (i) or (iii) shall be subject to the prior written consent of the Agent (which consent shall not be unreasonably withheld or delayed); provided, that any Replacement Bank shall meet the requirements to be an Eligible Assignee and shall purchase the same pro rata share of the Loans, L/C Obligations, L/C Borrowings and the Acquisition Commitment and the Revolving Commitment and the replacement shall be made pursuant to an assignment subject to the provisions of Section 12.9 and shall be an expense of the Borrower.
- 4.8 Survival. The agreements and obligations of the Borrower, the Agent and the Banks in this Article IV shall survive the payment of all other Obligations.

ARTICLE V

CONDITIONS PRECEDENT

- 5.1 Conditions to Effectiveness. The effectiveness of this Credit Agreement is subject to the condition that the Agent shall have received all of the following, in form and substance satisfactory to the Agent and each Bank, and in sufficient copies for each Bank:
- (a) Credit Agreement and any Notes. This Agreement, and any Notes requested by the Banks pursuant to Section 2.2(d), duly executed by each party thereto.
 - (b) Resolutions; Incumbency.
- (i) Copies of partnership authorizations for the Borrower and resolutions of the board of directors of each of the General Partner, Petrolane and the Restricted Subsidiaries authorizing the transactions contemplated hereby to which it is a party, certified as of the Closing Date by the Secretary or an Assistant Secretary of such Person; and
- (ii) A certificate of the Secretary or Assistant Secretary of each of the General Partner, Petrolane and the Restricted Subsidiaries certifying the names and true signatures of its officers authorized to execute, deliver and perform, as applicable, on behalf of such Person the Loan Documents to which it is a party.
- (c) Organization Documents; Good Standing. Each of the following documents:

- (i) the articles or certificate of incorporation and the bylaws of the General Partner and Petrolane and the Certificate of Limited Partnership and the Partnership Agreement of the Borrower, in each case as in effect on the Closing Date, certified by the Secretary or an Assistant Secretary of the General Partner or Petrolane, as applicable, as of the Closing Date; and
- (ii) a good standing certificate for Petrolane and the General Partner (and where available, the Borrower) from the Secretary of State (or similar, applicable Governmental Authority) of its state of incorporation or organization, as applicable, and each other state, where available, the Borrower is qualified to do business as a foreign corporation, in each case as of a recent date (in no case earlier than 60 days prior to the date hereof).
- (d) Legal Opinions. An opinion of Morgan, Lewis & Bockius LLP, special counsel for the Credit Parties, in form and substance reasonably satisfactory to the Agent and the Banks.
- (e) Post-Closing Agreement. The Post-Closing and Collateral Agency Transfer Agreement, substantially in the form of Exhibit H attached hereto, duly executed by each party thereto. Each Bank hereby authorizes the Agent to enter into the Post-Closing Agreement.
- (f) Insurance. Insurance complying with the provisions of the Collateral Agency Agreement shall be in full force and effect and the Agent shall have received a certificate to that effect from independent insurance brokers or consultants as shall be reasonably satisfactory to the Agent, dated on or about the Closing Date, and the Collateral Agent shall be named as loss payee as required under the Collateral Agency Agreement.
- (g) Payment of Fees. Evidence of payment by the Borrower of all accrued and unpaid fees, costs and expenses to the extent due and payable hereunder (subject to the limitations set forth in Section 12.4) on the Closing Date to the Agent, the Arranger and the Banks, together with Attorney Costs of the Agent to the extent invoiced prior to or on the Closing Date, plus such additional amounts of Attorney Costs as shall constitute the Agent's reasonable estimate by category of Attorney Costs incurred or to be incurred by it through the closing proceedings (provided, that such estimate shall not thereafter preclude final settling of accounts between the Borrower and the Agent) including any such costs, fees and expenses arising under or referenced in Sections 2.10 and 12.4.
- (h) Ownership. UGI shall own indirectly more than 45% of the partnership interests of the Borrower.
- (i) Certificate. A certificate signed by a Responsible Officer, dated as of the Closing Date, stating that:
- (i) the representations and warranties contained in Article VI of this Agreement and in the Collateral Agency Agreement, the Subsidiary Guarantee, the General Security Agreement and the Subsidiary Security Agreement (except those certain representations and warranties contained in the General Security Agreement and the Subsidiary Security Agreement that refer to the Schedules or Annexes of the General Security Agreement and the

Subsidiary Security Agreement) are true and correct in all material respects on and as of such date, as though made on and as of such date except (x) as affected by the completion of the transactions referred to herein and (y) to the extent that such representations and warranties expressly relate to an earlier time or date, in which case such representations and warranties shall have been true and correct in all material respects as of such earlier time or date;

- (ii) there has occurred since June 30, 2003, no event or circumstance that has resulted in, or presents a reasonable likelihood of having, a Material Adverse Effect;
 - (iii) no Default of Event of Default shall exist;

and

(iv) the condition set forth in clause (h) above

shall have been satisfied.

- (j) Existing Credit Agreement. On or prior to the Closing Date, the Existing Credit Agreement shall have been paid in full (including, interest, fees and other amounts owing thereunder) and all commitments thereunder shall have been irrevocably terminated.
- $\sf (k)$ Compliance Certificate. A Compliance Certificate for the fiscal quarter ending June 30, 2003.
- (1) Certified Documents. Copies of the following documents, certified by the Secretary or an Assistant Secretary of the General Partner:
 - (i) First Mortgage Note Agreements;
 - (ii) Security Documents, as such documents existed immediately prior to the effectiveness of this Agreement;
 - (iii) National Propane Purchase Agreement;
 - (iv) Columbia Purchase Agreement;
 - (v) Intercompany Loan Agreement;
 - (vi) Intercompany Note; and
 - (vii) Keep Well Agreement.
- (m) Letters of Credit. All letters of credit (other than the Existing Letters of Credit) issued and outstanding under the Existing Credit Agreement shall have been canceled.
- (n) Resignation of Collateral Agent. Evidence satisfactory to the Agent, that BofA has resigned, effective as of the Closing Date, as collateral agent under the Collateral Agency Agreement and the other Security Documents on terms and conditions satisfactory to the Agent.
- (o) Consent of Holders of First Mortgage Notes. Evidence that the Requisite Percentage (as defined in the Collateral Agency Agreement) (i) has agreed to the designation of the Indebtedness (as defined in the Collateral Agency Agreement) of the Borrower issued or incurred under, arising out of, or in connection with, the Notes, this Agreement, the Fee Letter, the L/C Related Documents and the Security Documents, including the principal of, prepayment charge, if any, with respect to, and interest on the Notes and this Agreement (including all reimbursement obligations thereunder) as Parity Debt for purposes of, and within the meaning

of, the First Mortgage Note Agreements, the Existing Credit Agreement, the Collateral Agency Agreement and the Security Documents, entitled to the benefits and security of the Security (as defined in the Collateral Agency Agreement) and Security Documents for the payment thereof on parity with the payment of the other Obligations (as defined in the Collateral Agency Agreement) in accordance with the terms of the Collateral Agency Agreement and the Security Documents; (ii) the Requisite Percentage has directed that each Bank that is a party to this Agreement constitutes a Parity Lender for purposes of, and within the meaning of, the Collateral Agency Agreement and the Security Documents and (iii) the Requisite Holders have consented to the replacement of BofA, in its capacity as collateral agent, with Wachovia.

- (p) Mortgage Amendments. Except as provided in the Post-Closing Agreement, executed amendments to the Mortgages encumbering the real property sites set forth on Schedule 5.1(p), each in form and substance satisfactory to the Agent.
- (q) Mortgage Assignments. Except as provided in the Post-Closing Agreement, executed assignments to each Mortgage, each in form and substance satisfactory to the Agent.
- (r) Title Insurance. Except as provided in the Post-Closing Agreement, endorsements, in favor of the Collateral Agent, to each title insurance policy obtained pursuant to Section 6.24 showing the Lien of the applicable Mortgage as a first priority mortgage Lien, subject only to Permitted Encumbrances.
- (s) Flood Hazard Certificates. Except as provided in the Post-Closing Agreement, flood hazard certificates for each Mortgaged Property indicating whether the applicable Mortgaged Property is or is not in a flood zone and with respect to each Mortgaged Property that is located within a flood zone, either (a) evidence that the improvements on such Mortgaged Property are not in a flood zone or (b) evidence that the Borrower has obtained flood hazard insurance for such Mortgaged Property, in each instance in a manner satisfactory to the Agent.
- (t) Other Documents. Such other approvals, opinions, documents or materials as the Agent or any Bank may reasonably request.

At the request of the Borrower or any Bank, the Agent will confirm in writing to the Banks, with a copy to the Borrower, whether, and to what extent, the above conditions have been fulfilled.

- 5.2 Conditions to All Borrowings. The obligation of each Bank to make any Loan (including its initial Loan), the obligation of the Issuing Bank to Issue any Letter of Credit (including the initial Letter of Credit) and the obligation of the Swing Line Bank to make any Swing Line Loan (including the initial Swing Line Loan) is subject to the satisfaction of the following conditions precedent on or prior to the relevant Borrowing Date or Issuance Date:
- (a) Notice of Borrowing. The Agent shall have received a Notice of Borrowing; or in the case of any Issuance of any Letter of Credit, the Issuing Bank and the Agent

shall have received an L/C Application or L/C Amendment Application, as required under Section 3.2.

- (b) Continuation of Representations and Warranties. The representations and warranties in Article VI shall be true and correct in all material respects on and as of such Borrowing Date or Issuance Date, with the same effect as if made on and as of such Borrowing Date or Issuance Date (except to the extent such representations and warranties expressly relate to an earlier time or date, in which case they shall have been true and correct in all material respects as of such earlier time or date);
- (c) No Existing Default. No Default or Event of Default shall exist or shall result from such Borrowing or Issuance; and
- (d) Specified Acquisition Loans. If a Specified Acquisition Loan is requested by the Borrower, the sum of (i) the Effective Amount of the Revolving Loans and (ii) the Effective Amount of the L/C Obligations shall be equal to the Revolving Commitment.

Each Notice of Borrowing, L/C Application or L/C Amendment Application submitted or deemed submitted by the Borrower hereunder shall constitute a representation and warranty by the Borrower hereunder, as of the date of each such notice and as of each Borrowing Date and Issuance Date that the conditions in Section 5.2 are satisfied.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES

The Borrower represents and warrants to the Agent and each Bank as set forth below in Sections 6.1 through 6.14 and Sections 6.17 through 6.23, Petrolane represents and warrants to the Agent and each Bank as set forth below in Section 6.15, and the General Partner represents and warrants to the Agent and each Bank as set forth below in Section 6.16 that:

- 6.1 Organization, Standing, etc. The Borrower is a limited partnership duly organized, validly existing and in good standing under the Delaware Revised Uniform Limited Partnership Act and has all requisite partnership power and authority to own and operate its properties (including without limitation its Assets), to conduct its business, to enter into this Agreement and such other Loan Documents to which it is a party and to carry out the terms of this Agreement and such other Loan Documents. Each Restricted Subsidiary is a corporation or limited partnership, as the case may be, duly organized, validly existing and in good standing under the laws of its state of incorporation or organization, as the case may be, and has all requisite corporate power and authority to own and operate its properties (including without limitation its Assets), to conduct its business and to execute and deliver the Security Documents to which such Restricted Subsidiary is a party and to carry out the terms of this Agreement and such other Security Documents.
- 6.2 Partnership Interests and Subsidiaries. The sole general partner of the Borrower is the General Partner, which on the Closing Date owns a 1.0101% general partnership interest in the Borrower and is an indirect Wholly-Owned Subsidiary of UGI. On the Closing Date (a) the only limited partner of the Borrower is the Public Partnership, which owns a 98.9899% limited

partnership interest in the Borrower, and (b) the Borrower does not have any partners other than the General Partner and the Public Partnership. As of the Closing Date, the Borrower does not have any Subsidiary other than as set forth on Schedule 6.2 or any Investments in any Person (other than as set forth on Schedule 6.2 or Investments of the types described in Section 8.4(a)).

- Qualification; Corporate or Partnership Authorization. The Borrower is duly qualified or registered and is in good standing as a foreign limited partnership for the transaction of business, and each of the General Partner, Petrolane and the Restricted Subsidiaries is qualified or registered and is in good standing as a foreign corporation or foreign limited partnership for the transaction of business, in the states listed in Schedule 6.3, which are the only jurisdictions in which the nature of their respective activities or the character of the properties they own, lease or use makes such qualification or registration necessary as of the Closing Date and in which the failure so to qualify or to be so registered as of the Closing Date would have a Material Adverse Effect. Each of the Borrower, the General Partner and Petrolane has taken all necessary partnership action or corporate action, as the case may be, to authorize the execution, delivery and performance by it of this Agreement and other Loan Documents to which it is a party. Each Restricted Subsidiary has taken all necessary corporate or partnership action, as the case may be, to authorize the execution, delivery and performance by it of each of the Security Documents to which it is a party. Each of the Borrower, the General Partner and Petrolane has duly executed and delivered each of this Agreement and the other Loan Documents to which it is a party, and each of them constitutes the Borrower's, the General Partner's or Petrolane's, as the case may be, legal, valid and binding obligation enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally. Each Restricted Subsidiary has duly executed and delivered each of the Security Documents to which it is a party, and each of them constitutes such Restricted Subsidiary's legal, valid and binding obligation enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally.
- 6.4 Financial Statements. The audited consolidated financial statements of the Borrower and its consolidated Subsidiaries for the fiscal years ended September 30, 2002 and September 30, 2001, and the unaudited balance sheet, statement of operations, statement of cash flows and statement of partners capital of the Borrower and its consolidated Subsidiaries for the fiscal period ended June 30, 2003, have been prepared in accordance with GAAP applied on a consistent basis throughout the periods specified (except as described in the footnotes thereto) and present fairly, in all material respects, the financial position of the Borrower as of the respective dates specified (except for the absence of footnotes and subject to changes resulting from normal year-end audit adjustments, in the case of unaudited financial statements).
- 6.5 Changes, etc. Except as contemplated by this Agreement or the other Loan Documents, (a) for the period from June 30, 2003 to and including the Closing Date, none of the Borrower and any of its Restricted Subsidiaries has incurred any material liabilities or obligations, direct or contingent, nor entered into any material transaction, in each case other than in the ordinary course of its business, and (b) since the date of the last financial statements delivered pursuant to Section 6.4 or 7.1 there has not been any material adverse change in or effect on the financial condition or prospects of the Borrower or in the Business or Assets. Since

62

June 30, 2003, no Restricted Payment of any kind has been declared, paid or made by the Borrower other than Restricted Payments permitted by Section 8.5.

- Partner, Petrolane and the Restricted Subsidiaries has filed all material tax returns required by law to be filed by it or has properly filed for extensions of time for the filing thereof, and has paid all material taxes, assessments and other governmental charges levied upon it or any of its properties, assets, income or franchises which are shown to be due on such returns, other than those which are not past due or are presently being contested in good faith by appropriate proceedings diligently conducted for which such reserves or other appropriate provisions, if any, as shall be required by GAAP have been made. The Borrower is a limited partnership and so long as it is a limited partnership it will be treated as a pass-through entity for U.S. federal income tax purposes and as of the Closing Date is not subject to taxation with respect to its income or gross receipts under applicable state (other than Michigan, New Hampshire, Tennessee, Washington and Wisconsin) laws.
- 6.7 Indebtedness. As of the Closing Date, none of the Borrower, the General Partner, Petrolane, Borrower and its respective Subsidiaries has any secured or unsecured Indebtedness outstanding, except as set forth in Schedule 6.7 and other than the Indebtedness represented by this Agreement, the other Loan Documents and the First Mortgage Notes. As of the Closing Date, no instrument or agreement to which the Borrower or any of its Subsidiaries is a party or by which the Borrower or any such Subsidiary is bound (other than this Agreement and the agreements governing the First Mortgage Notes and other than as indicated in Schedule 6.7) contains any restriction on the incurrence by the Borrower or any of its Subsidiaries of additional Indebtedness.
- except as set forth in Schedule 6.8(a), each of the Borrower and its
 Subsidiaries is in possession of, and operating in compliance in all material
 respects with, all franchises, grants, authorizations, approvals, licenses,
 permits (other than permits required by Environmental Laws), easements,
 rights-of-way, consents, certificates and orders (collectively, the "PERMITS")
 required (i) to own, lease or use its properties (including without limitation
 to own, lease or use its Assets) and (ii) considering all such Permits in the
 possession of, and complied with by, the General Partner, Petrolane, the
 Borrower and its Subsidiaries taken together, to permit the conduct of the
 Business as now conducted and proposed to be conducted, except for those Permits
 (collectively, the "ROUTINE PERMITS") (x) which are routine or administrative in
 nature and are expected in the reasonable judgment of the Borrower to be
 obtained or given in the ordinary course of business after the Closing Date, or
 (y) which, if not obtained or given, would not, individually or in the
 aggregate, present a reasonable likelihood of having a Material Adverse Effect.
- (b) On the Closing Date, each of the Borrower and its Subsidiaries has (i) good and marketable title to substantially all of the Assets constituting owned real property and (ii) good and sufficient title to substantially all of the Assets constituting fee-owned personal property for the use and operation of such personal property as it is used on the date hereof, in each case subject to no Liens except such as are permitted by Section 8.3. The Assets of the Borrower and its Subsidiaries on the Closing Date are substantially all of the assets and properties necessary to enable the Borrower and its Subsidiaries to conduct the Business.

Subject to such exceptions as would not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect, (A) on the date hereof the Borrower and its Subsidiaries enjoy peaceful and undisturbed possession under all leases and subleases necessary in any material respect for the conduct of the Business, and (B) as of the Closing Date, all such leases and subleases are valid and subsisting and are in full force and effect. Except to perfect, preserve and protect Liens permitted by Section 8.3 and Liens which will be discharged on the Closing Date, as of the Closing Date, (x) no presently effective financing statements under the Uniform Commercial Code which name any of the Borrower, the General Partner, Petrolane or their respective Subsidiaries as debtor, and which individually or in the aggregate relate to any material part of the Assets, are on file in any jurisdiction in which any of the Assets are (or have been) located or the Borrower, the General Partner, Petrolane or any such Subsidiary is organized or has its principal place of business and (y) none of the Borrower, the General Partner, Petrolane and any such Subsidiary has signed, or authorized the filing by or on behalf of any secured party of, any presently effective financing statements which individually or in the aggregate relate to any material part of the Assets.

- (c) As of the date hereof, neither the Borrower nor any of its Subsidiaries own any Railcars (as defined in the General Security Agreement).
- there is no action, proceeding or investigation pending or, to the knowledge of the Borrower upon reasonable inquiry, threatened against the Borrower, Petrolane, the Public Partnership, the General Partner or any of their respective Subsidiaries, and there is no action proceeding or investigation pending or, to the knowledge of the Borrower upon reasonable inquiry, threatened against the Borrower or its Restricted Subsidiaries, (a) which questions the validity or enforceability of this Agreement, the other Loan Documents or any action taken or to be taken pursuant to this Agreement or the other Loan Documents, or (b) except as set forth in Schedule 6.9, which would present a reasonable likelihood of having, either in any case or in the aggregate, a Material Adverse Effect.
- 6.10 Compliance with Other Instruments, etc. (a) On the Closing Date, none of the Borrower, the General Partner, Petrolane or any of their respective Subsidiaries will be in violation of (i) any provision of its certificate or articles of incorporation or other Organization Documents, (ii) any provision of any agreement or instrument to which it is a party or by which any of its properties is bound or (iii) any applicable law, ordinance, rule or regulation of any Governmental Authority or any applicable order, judgment or decree of any court, arbitrator or Governmental Authority, except (in the case of clauses (ii) and (iii) above only) for such violations which would not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect. Neither the General Partner nor the Public Partnership is in violation of any provision of the Partnership Agreement.
- (b) The execution, delivery and performance by each of the Borrower, the General Partner, Petrolane and the Restricted Subsidiaries of this Agreement and the other Loan Documents to which it is a party, and the completion of the transactions contemplated by this Agreement will not (i) violate (x) any provision of the Partnership Agreement or the certificate or articles of incorporation or other Organization Documents of the Borrower, the General Partner, Petrolane or any of their respective Subsidiaries, (y) any applicable law, ordinance, rule

or regulation of any Governmental Authority or any applicable order, judgment or decree of any court, arbitrator or Governmental Authority, or (z) any provision of any agreement or instrument to which the Borrower, the General Partner, Petrolane or any of their respective Subsidiaries is a party or by which any of its properties is bound, except (in the case of clauses (y) and (z) above) for such violations which would not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect, or (ii) result in the creation of (or impose any express obligation on the part of the Borrower to create) any Lien not permitted by Section 8.3.

- 6.11 Governmental Consent. Except as expressly contemplated by this Agreement and the other Loan Documents, and except for Routine Permits, (i) no consent, approval or authorization of, or declaration or filing with, any Governmental Authority is required for the valid execution, delivery and performance of this Agreement or the other Loan Documents to which the Borrower or any of the Restricted Subsidiaries, Petrolane or the General Partner is a party, and (ii) no such consent, approval, authorization, declaration or filing is required for the making of Loans or Issuing Letters of Credit pursuant to this Agreement.
- 6.12 Investment Company Act. None of the Borrower, Petrolane or the General Partner is an "investment company", or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended.
- 6.13 Public Utility Holding Company Act. None of the Borrower, Petrolane or the General Partner is a "holding company" within the meaning of Section 2(a)(7)(A) of the Public Utility Holding Company Act of 1935, as amended (the "PUHCA"). Each of the Borrower, Petrolane and the General Partner is a "subsidiary company" of a "holding company", within the meaning of the PUHCA, but each of UGI (the "holding company"), the Borrower, Petrolane and the General Partner is exempt from all the provisions of the PUHCA and the rules thereunder other than Section 9(a)(2) thereof, based upon the filing by UGI with the Commission of an exemption statement on Form U-3A-2 dated February 27, 2003 pursuant to Rule 2 under PUHCA (17 C.F.R. Section 250.2).
- 6.14 Chief Executive Office. As of the Closing Date, the chief executive office of the Borrower and the office where it maintains its records relating to the transactions contemplated by this Agreement and the other Loan Documents is located at 460 North Gulph Road, King of Prussia, PA 19406.
- 6.15 Matters Relating to Petrolane. (a) As of the Closing Date, Petrolane is a Wholly-Owned Subsidiary of the General Partner, has no Subsidiaries and owns an approximate 15% limited partnership interest in the Public Partnership.
- (b) Petrolane is a corporation duly organized, validly existing and in good standing under the laws of the Commonwealth of Pennsylvania and has all requisite corporate power and authority to own and operate its properties, to conduct its business and to execute and deliver this Agreement and such other Loan Documents to which Petrolane is a party and to carry out the terms of this Agreement and such other Loan Documents.
- (c) Schedule 6.15 includes a complete description of the business and activities carried on by Petrolane and of its assets and liabilities as of the Closing Date.

- 6.16 Matters Relating to the General Partner. (a) As of the Closing Date, the General Partner is a Wholly Owned Subsidiary of AmeriGas, Inc., a Pennsylvania corporation, and owns, in addition to the interest in the Borrower described in Section 6.2, (i) a 1% general partnership interest in the Public Partnership, (ii) all of the outstanding shares of Capital /Stock of Petrolane and (iii) an approximate 32% limited partnership interest in the Public Partnership. Other than AmeriGas Technology Group, Inc. and Petrolane, the General Partner has no other direct Subsidiaries as of the Closing Date.
- (b) The General Partner is a corporation duly organized, validly existing and in good standing under the laws of the Commonwealth of Pennsylvania and has all requisite corporate power and authority to own and operate its properties, to act as the sole general partner of the Borrower and to execute and deliver in its individual capacity and in its capacity as the sole general partner of the Borrower this Agreement and such other Loan Documents to which the General Partner is a party and to carry out the terms of this Agreement and such other Loan Documents.
- (c) Schedule 6.16 includes a complete description of the business and activities carried on by the General Partner and of its assets and liabilities as of the Closing Date.
- 6.17 ERISA Compliance. Except to the extent that any of the following would not, either alone or together, present a reasonable likelihood of having a Material Adverse Effect: (i) during the twelve-consecutive-month period prior to the date of the execution and delivery of this Agreement and prior to the date of any Borrowing hereunder, no steps have been taken to terminate any Pension Plan sponsored or maintained by any Obligor or any ERISA Affiliate of any Obligor, (ii) no contribution failure has occurred with respect to any Pension Plan sponsored or maintained by any Obligor or any ERISA Affiliate of any Obligor sufficient to give rise to a Lien under section 302(f) of ERISA and (iii) with respect to each Pension Plan sponsored or maintained by any Obligor or any ERISA Affiliate of any Obligor, none of the following events has occurred: termination of the plan, failure to make a required contribution to the plan, failure to satisfy the minimum funding standard for a year, request for a waiver of the minimum funding standard for any year, withdrawal from a multiple employer plan, adoption of an amendment which results in a funded current liability percentage of less than 60%, engaging in one or more prohibited transactions, failure to comply with reporting and disclosure requirements or engaging in any breach of fiduciary responsibility.
- 6.18 Use of Proceeds; Margin Regulations. The proceeds of the Loans are to be used solely for the purposes set forth in and permitted by Section 8.9. None of the Borrower and its Subsidiaries is generally engaged in the business of purchasing or selling Margin Stock or extending credit for the purpose of purchasing or carrying Margin Stock.
- 6.19 Environmental Warranties. (a) Except as disclosed on Schedule 6.19 or where non-compliance would not present a reasonable likelihood of having a Material Adverse Effect, each of the Borrower and its Subsidiaries is in compliance with all Environmental Laws applicable to it and to the Business or Assets. Except as disclosed on Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented, the Borrower and its Subsidiaries have obtained and are in compliance with all permits, licenses and approvals required by Environmental Law. Except as disclosed in Schedule 6.19 or where the failure to

timely and properly reapply would not present a reasonable likelihood of having a Material Adverse Effect, the Borrower and its Subsidiaries have submitted timely and complete applications to renew any expired or expiring Permits required by Environmental Law. Schedule 6.17 lists all notices from Federal, state or local Governmental Authorities or other Persons received within the last five years of the date hereof by the Borrower and its Subsidiaries, alleging or threatening any liability on the part of the Borrower or any of its Subsidiaries, pursuant to any Environmental Law, that present a reasonable likelihood of having a Material Adverse Effect. All reports, documents, or other submissions required by Environmental Laws to be submitted by the Borrower to any Governmental Authority or Person have been filed by the Borrower, except where the failure to file would not present a reasonable likelihood of having a Material Adverse Effect.

- (b) Except as disclosed in Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented: (i) there is no Hazardous Substance present at any of the real property currently owned or leased by the Borrower or any of its Subsidiaries, and to the knowledge of the Borrower, there was no Hazardous Substance present at any of the real property formerly owned or leased by the Borrower or any of its Subsidiaries during the period of ownership or leasing by such Person; and (ii) with respect to such real property and subject to the same knowledge and temporal qualifiers concerning Hazardous Substances with respect to formerly owned or leased real properties, there has not occurred (x) any release, or to the knowledge of the Borrower, any threatened release of a Hazardous Substance, or (y) any discharge or, to the knowledge of the Borrower, threatened discharge of any Hazardous Substance into the ground, surface, or navigable waters which violates any Federal, state, local or foreign laws, rules or regulations concerning water pollution.
- (c) Except as set forth in Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented, none of the Borrower and its Subsidiaries has disposed of, transported, or arranged for the transportation or disposal of any Hazardous Substance where such disposal, transportation, or arrangement would give rise to liability pursuant to CERCLA or any analogous state statute.
- (d) Except as set forth in Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented: (1) no lien has been asserted by any Governmental Authority or person resulting from the use, spill, discharge, removal, or remediation of any Hazardous Substance with respect to any real property currently owned or leased by the Borrower or any of its Subsidiaries, and (2) to the knowledge of the Borrower, no such lien was asserted with respect to any of the real property formerly owned or leased by the Borrower or any its Subsidiaries during the period of ownership or leasing of the real property by such Person.
- (e) Except as set forth in Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented, (1) there are no underground storage tanks, asbestos-containing materials, polychlorinated biphenyls, or urea formaldehyde insulation at any of the real property currently owned or leased by the Borrower or any of its Subsidiaries in violation of Environmental Law and (2) to the knowledge of the Borrower, there were no underground storage tanks, asbestos-containing materials, polychlorinated biphenyls, or urea formaldehyde insulation at any of the real property formerly owned or leased by the Borrower or

any of its Subsidiaries in violation of Environmental Law during the period of ownership or leasing of such real property by such Person.

- (f) Except as set forth in Schedule 6.19 or where a reasonable likelihood of having a Material Adverse Effect would not be presented, propane has been used, handled and stored by the Borrower and its Subsidiaries during the five year period ending on the Closing Date in compliance with Environmental Laws.
- 6.20 Copyrights, Patents, Trademarks and Licenses, etc. Except to the extent that the failure to do so would not present a reasonable likelihood of having a Material Adverse Effect, the Borrower and the Restricted Subsidiaries own or are licensed or otherwise have the right to use all of the patents, trademarks, service marks, trade names, copyrights, contractual franchises, authorizations and other rights that are reasonably necessary for the operation of the Business, without conflict with the rights of any other Person. To the best knowledge of the Borrower, no slogan or other advertising device, product, process, method, substance, part or other material now employed, or now contemplated to be employed, by the Borrower or any Restricted Subsidiary infringes upon any rights held by any other Person, where such infringement would present a reasonable likelihood of having a Material Adverse Effect. Except as specifically disclosed in Schedule 6.20, no claim or litigation regarding any of the foregoing is pending or to the knowledge of the Borrower threatened, and no patent, invention, device, application, principle or any statute, law, rule, regulation, standard or code is pending or, to the knowledge of the Borrower, proposed, which, in either case, would present a reasonable likelihood of having a Material Adverse Effect.
- $\,$ 6.21 $\,$ Insurance. The Borrower and each of its Subsidiaries are in compliance with the terms and conditions contained in Sections 20 and 21 of the Collateral Agency Agreement.
- 6.22 Full Disclosure. None of the representations or warranties made by any Obligor or the Restricted Subsidiaries in the Loan Documents as of the date such representations and warranties are made or deemed made, and none of the statements contained in any document, certificate or instrument furnished by or on behalf of any Obligor in connection with the Loan Documents, as of the date of such document, instrument or certificate, contains any untrue statement of a material fact or omits any material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances under which they are made, not misleading.
- $\,$ 6.23 $\,$ Solvency. As of the Closing Date, none of the Obligors or any of the Restricted Subsidiaries:
 - (i) was insolvent;
- (ii) was engaged in business, or was about to engage in business or a transaction, for which any property remaining with such Obligor or Restricted Subsidiary was an unreasonably small capital; or
- $\hbox{(iii)} \qquad \hbox{intended to incur, or believed that it would} \\ \hbox{incur, debts that would be beyond its ability to pay as such debts matured.}$

- 6.24 Title Insurance. As of the Closing Date, Schedule 6.24 sets forth a list of all real property owned by the Borrower and the Restricted Subsidiaries (other than AEPLP and its Subsidiaries) which has an individual value in excess of \$1,000,000 (as determined in good faith by the General Partner) or which is insured by a lender's title insurance policy. The Borrower has obtained and delivered to the Collateral Agent (or its predecessor) an ALTA loan policy (10-17-92 form), or its equivalent, of title insurance for all real property owned by it which has an individual value in excess of \$1,000,000 (as determined in good faith by the General Partner) or which is, as of the date hereof, insured by a lender's title insurance policy showing the Lien of the applicable Mortgage as a first priority mortgage Lien, subject only to Permitted Encumbrances.
- 6.25 Mortgages. (a) From and after April 19, 1995, the Borrower and its Restricted Subsidiaries (other than AEPLP and its Subsidiaries), as applicable, have executed, delivered and recorded a Mortgage, in substantially the form of Exhibit E, covering each district location owned or acquired by it and any other real property thereafter acquired by it which had an individual value in excess of \$100,000 or which had an aggregate value in excess of \$500,000 (in each case as determined in good faith by the General Partner). Schedule 6.25 sets forth a true and correct list of all real property owned by the Borrower or a Restricted Subsidiary (other than AEPLP and its Subsidiaries) that is subject to a Mortgage pursuant to this Section 6.25 and sets forth for each such real property the address and county of such real property.
- (b) As of the Closing Date, each Mortgage required to be delivered pursuant to Section 6.25 (i) is in full force and effect and (ii) creates a valid first mortgage lien upon the applicable Mortgaged Property or Mortgaged Properties, as applicable, subject only to Permitted Encumbrances (as defined in the applicable Mortgages).
- 6.26 Security Interest; Parity Debt. (a) The Borrower has caused to be duly recorded, published, registered and filed all the documents set forth in paragraph (b) of the definition of Security Documents (or documents or instruments in respect thereof), in such manner and in such places as was required by law to establish, and if applicable, perfect and preserve the rights and security interests of the parties thereto and their respective successors and assigns in the General Collateral.
- (b) Pursuant to the General Security Agreement, the Collateral Agent has a first priority perfected security interest in the Collateral (as defined in the General Security Agreement) subject only to Permitted Filings (as defined in the General Security Agreement) and other liens permitted by Section 8.3 of this Agreement, and the Collateral Agent is entitled to all the rights, priorities and benefits afforded by the Uniform Commercial Code or other relevant law as enacted in the relevant jurisdiction to perfected security interests.
- (c) Pursuant to the Subsidiary Security Agreement, the Collateral Agent has a first priority perfected security interest in the Collateral (as defined in the Subsidiary Security Agreement) subject only to Permitted Filings (as defined in the Subsidiary Security Agreement) and other liens permitted by Section 8.3 of this Agreement, and the Collateral Agent is entitled to all the rights, priorities and benefits afforded by the Uniform Commercial Code or other relevant law as enacted in the relevant jurisdiction to perfected security interests.

- (d) This Agreement constitutes a Parity Debt Agreement for purposes of, and within the meaning of, the Collateral Agency Agreement and the Security Documents. The Agent constitutes a Parity Agent for purposes of, and within the meaning of, the Collateral Agency Agreement and the Security Documents. Each Bank that is a party to this Agreement constitutes a Parity Lender for purposes of, and within the meaning of, the Collateral Agency Agreement and the Security Documents.
- Evidence that (i) the Requisite Percentage (as defined in the Collateral Agency Agreement) has agreed to the designation of the Indebtedness (as defined in the Collateral Agency Agreement) of the Borrower issued or incurred under, arising out of, or in connection with, the Notes, this Agreement, the Fee Letter, the L/C Related Documents and the Security Documents, including the principal of, prepayment charge, if any, with respect to, and interest on the Notes and this Agreement (including all reimbursement obligations thereunder) as Parity Debt for purposes of, and within the meaning of, the First Mortgage Note Agreements, the Existing Credit Agreement, the Collateral Agency Agreement and the Security Documents, entitled to the benefits and security of the Security (as defined in the Collateral Agency Agreement) and Security Documents for the payment thereof on parity with the payment of the other Obligations (as defined in the Collateral Agency Agreement) in accordance with the terms of the Collateral Agency Agreement and the Security Documents; (ii) the Requisite Percentage has directed that each Bank that is a party to this Agreement constitutes a Parity Lender for purposes of, and within the meaning of, the Collateral Agency Agreement and the Security Documents and (iii) the Requisite Holders have consented to the replacement of BofA, in its capacity as collateral agent, with Wachovia.
- 6.27 PPD/GP Debt Contributions. The aggregate amount of PPD/GP Debt Contributions made by the Public Partnership and the General Partner to the Borrower during the period from August 21, 2001 to the Closing Date is in excess of \$105,000,000.
- 6.28 State Mortgage Taxes (a) Other than in connection with the amendments to the Mortgages referred to in Section 5.1(p) and the Mortgage Assignments (as defined in the Post-Closing Agreement) to be filed on the Mortgaged Properties pursuant to this Agreement and the Post-Closing Agreement, all state documentary stamp taxes, recording fees and nonrecurring intangible taxes, if any, due and payable in connection with the Secured Obligations prior to the Closing Date were paid upon recording of each Mortgage; (b) all state documentary stamp taxes, recording fees and nonrecurring intangible taxes, if any, due and payable in connection with such amendments to the Mortgages referred to in Section 5.1(p) and Mortgage Assignments have been or will be paid upon recording of the same and (c) no document (including, without limitation, this Agreement) evidencing, creating, extending, renewing or modifying any of the Secured Obligations has been executed by any person or entity (excluding the payees or other beneficiaries thereof) other than those persons and entities who were obligated to pay the Secured Obligations at the time that the Mortgages were originally recorded. As used in this Section 6.28, the term "Secured Obligations" means "Obligations" as defined in the Mortgages.
- 6.29 Tax Disclosure. Neither the Agent, the Banks nor the Borrower intend to treat the Loans as being a "reportable transaction" (within the meaning of Treasury Regulation Section 1.6011-4) or a "tax shelter" (within the meaning of Section 6111 of the Code).

ARTICLE VII

AFFIRMATIVE COVENANTS

So long as any Bank shall have any Commitment hereunder, or any Loan or other Obligation shall remain unpaid or unsatisfied, or any Letters of Credit shall remain outstanding unless the Required Banks waive compliance in writing:

- 7.1 Financial Statements. The Borrower will maintain, and will cause each of its Subsidiaries to maintain, a system of accounting established and administered in accordance with GAAP, and will accrue, and will cause each of its Subsidiaries to accrue, all such liabilities as shall be required by GAAP. The Borrower will furnish or cause to be furnished to the Agent, on behalf of Banks, and the Agent will promptly distribute to each Bank at their respective addresses as set forth on Schedule 12.2 hereto, or such other office as may be designated by the Agent and Banks from time to time:
- as soon as practicable, but in any event within 45 days after the end of each of the first three quarterly fiscal periods in each fiscal year of the Borrower, consolidated and consolidating balance sheets of the Borrower and its Subsidiaries (except, as to consolidating balance sheets only, for inactive Subsidiaries) as at the end of such period and the related consolidated (and, as to statements of income, consolidating, except for inactive Subsidiaries) statements of income, partners' capital and cash flows of the Borrower and its Subsidiaries for such period and (in the case of the second and third quarterly periods) for the period from the beginning of the current fiscal year to the end of such quarterly period, setting forth in each case in comparative form the consolidated and, where applicable, consolidating figures for the corresponding periods of the previous fiscal year, all in reasonable detail and certified by the principal financial officer of the General Partner as presenting fairly, in all material respects, the information contained therein (except for the absence of footnotes and subject to changes resulting from normal year-end adjustments), in accordance with GAAP applied on a basis consistent with prior fiscal periods except for inconsistencies resulting from changes in accounting principles and methods agreed to by the Borrower's independent accountants;
- as soon as practicable, but in any event within 90 days after the end of each fiscal year of the Borrower, consolidated and consolidating balance sheets of the Borrower and its Subsidiaries (except, as to consolidating balance sheets only, for inactive Subsidiaries) as at the end of such year and the related consolidated (and, as to statements of income, consolidating except for inactive Subsidiaries) statements of income, partners' capital and cash flows of the Borrower and its Subsidiaries for such fiscal year, setting forth in each case in comparative form the consolidated and, where applicable, consolidating figures for the previous fiscal year, all in reasonable detail and (i) in the case of such consolidated financial statements, accompanied by a report thereon of PricewaterhouseCoopers LLP or other independent public accountants of recognized national standing selected by the Borrower, which report shall not be qualified with respect to scope limitations imposed by the Borrower or any of its Restricted Subsidiaries or with respect to accounting principles followed by the Borrower or any of its Restricted Subsidiaries not in accordance with GAAP and shall state that such consolidated financial statements present fairly, in all material respects, the financial position of the Borrower and its Subsidiaries as at the dates indicated and the results of their operations and cash flows for

the periods indicated in conformity with GAAP unless otherwise disclosed, applied on a basis consistent with prior years, and that the audit by such accountants in connection with such consolidated financial statements has been made in accordance with generally accepted auditing standards then in effect in the United States, and (ii) in the case of such consolidated and consolidating financial statements, certified by the principal financial officer of the General Partner as presenting fairly, in all material respects, the information contained therein (except, in the case of such consolidating financial statements, for the absence of footnotes), in accordance with GAAP (the items in subsections (a) and (b) of this Section 7.1, the "BORROWER FINANCIALS");

- together with each delivery of financial statements of the Borrower pursuant to subsections (a) and (b) of this Section 7.1, a Compliance Certificate of the Borrower (i) stating that the signers have reviewed the terms of this Agreement and the other Loan Documents and have made, or caused to be made under their supervision, a review in reasonable detail of the transactions and condition of the Borrower and its Subsidiaries during the accounting period covered by such financial statements, and that the signers do not have knowledge of the existence and continuance as at the date of such Compliance Certificate of any Default or Event of Default, or, if any of the signers have knowledge that any Default or Event of Default then exists, specifying the nature and approximate period of existence thereof and what action the Borrower has taken or is taking or proposes to take with respect thereto, (ii) specifying the amount available at the end of such accounting period for Restricted Payments in compliance with Section 8.5 and showing in reasonable detail all calculations required in arriving at such amount, (iii) demonstrating in reasonable detail compliance at the end of such accounting period with the restrictions contained in Section 8.1 (the last two paragraphs), Sections 8.1(b), (d), (e), (f), (k) and (l), Section 8.2, Section 8.4(c), Section 8.4(h), Section 8.5, Section 8.8(a)(ii), Section 8.8(a)(iii), Section 8.8(c)(iii) (calculation of Excess Sale Proceeds), Section 8.13, Section 8.14 (first sentence), Section 8.15 and Sections 8.17(a), (b) and (d), (iv) calculating the applicable Pricing Tier, (v) if not specified in the related financial statements being delivered pursuant to subsections (a) and (b) above, specifying the aggregate amount of interest paid or accrued by, and aggregate rental expenses of, the Borrower and its Subsidiaries, and the aggregate amount of depreciation, depletion and amortization charged on the books of the Borrower and its Subsidiaries, during the fiscal period covered by such financial statements, and (vi) if at the time of the delivery of such financial statements the Borrower shall have any Unrestricted Subsidiaries, setting forth therein (or in an accompanying schedule) the adjustments required to be made to indicate the consolidated financial position, cash flows and results of operations of the Borrower and the Restricted Subsidiaries without regard to the financial position, cash flows or results of operations of such Unrestricted Subsidiaries;
- (d) together with each delivery of consolidated financial statements of the Borrower pursuant to subsection (b) of this Section 7.1, a written statement by the independent public accountants giving the report thereon stating that they have reviewed the terms of this Agreement and the Notes and that, in making the audit necessary for the certification of such financial statements, they have obtained no knowledge of the existence and continuance as at the date of such written statement of any Default or Event of Default, or, if they have obtained knowledge that any Default or Event of Default then exists, specifying, to the extent possible, the nature and approximate period of the existence thereof (such accountants, however, shall not be liable to anyone by reason of their failure to obtain knowledge of any Default or Event of Default

72

which would not be disclosed in the course of an audit conducted in accordance with generally accepted auditing standards then in effect in the United States);

- (e) promptly following the receipt and timely review thereof by the Borrower, copies of all reports submitted to the Borrower by independent public accountants in connection with each special, annual or interim audit of the books of the Borrower or any Subsidiary thereof made by such accountants, including without limitation the comment letter submitted by each such accountant to management in connection with their annual audit;
- (f) promptly upon their becoming publicly available, copies of (i) all financial statements, reports, notices and proxy statements sent or made available by the Borrower or the Public Partnership to any of its security holders in compliance with the Exchange Act, or any comparable Federal or state laws relating to the disclosure by any Person of information to its security holders, (ii) all regular and periodic reports and all registration statements and prospectuses filed by the Borrower or the Public Partnership with any securities exchange or with the Securities and Exchange Commission or any governmental authority succeeding to any of its functions (other than registration statements on Form S-8 and Annual Reports on Form 10-R), (iii) all press releases and other similar written statements made available by the . Borrower or the Public Partnership to the public concerning material developments in the business of the Borrower or the Public Partnership, as the case may be and (iv) all reports, notices and other similar written statements sent or made available by the Borrower or the Public Partnership to any holder of its Indebtedness pursuant to the terms of any agreement, indenture or other instrument evidencing such Indebtedness, including without limitation the First Mortgage Notes and the Public Partnership Indenture, except to the extent the same substantive information is already being sent to the Agent;
- (g) as soon as reasonably practicable, and in any event within five Business Days after a Responsible Officer obtains knowledge that any Default or Event of Default has occurred, a written statement of such Responsible Officer setting forth details of such Default or Event of Default and the action which the Borrower has taken, is taking and proposes to take with respect thereto;
- (h) as soon as reasonably practicable, and in any event within five Business Days after a Responsible Officer obtains knowledge of (i) the occurrence of an adverse development with respect to any litigation or proceeding involving the Borrower or any of its Subsidiaries which in the reasonable judgment of the Borrower presents a reasonable likelihood of having a Material Adverse Effect or (ii) the commencement of any litigation or proceeding involving the Borrower or any of its Subsidiaries which in the reasonable judgment of the Borrower presents a reasonable likelihood of having a Material Adverse Effect, a written notice of such Responsible Officer describing in reasonable detail such commencement of, or adverse development with respect to, such litigation or proceeding;
- (i) as soon as reasonably practicable, and in any event within five Business Days after a responsible officer of any Obligor becomes aware of the occurrence or existence of any of the events or conditions specified below, and such event or condition has resulted in, or in the opinion of the principal financial officer of the General Partner might reasonably be expected to result in, a Material Adverse Effect: (i) the institution of any steps by any Obligor or any other

Person to terminate any Pension Plan sponsored or maintained by an Obligor or any ERISA Affiliate of any Obligor, (ii) the failure to make a required contribution to any Pension Plan sponsored or maintained by any Obligor if such failure is sufficient to give rise to a Lien under section 302(f) of ERISA, or (iii) if any of the subsequently listed events have occurred with respect to any Pension Plan sponsored or maintained by any Obligor, or any ERISA Affiliate of any Obligor, the occurrence of termination of the plan, failure to make a required contribution to the plan, failure to satisfy the minimum funding standard for a year, request for a waiver of the minimum funding standard for any year, withdrawal from a multiple employer plan, adoption of an amendment which results in a funded current liability percentage of less than 60%, engaging in one or more prohibited transactions, failure to comply with reporting and disclosure requirements or engaging in any breach of fiduciary responsibility, notice thereof and copies of all documentation relating thereto;

- (j) within 15 days after being approved by the governing body of the Borrower, and in any event no later than November 15th each fiscal year, an annual operating forecast for the next fiscal year;
- (k) as soon as reasonably practicable, and in any event within five Business Days after a Responsible Officer obtains knowledge of a violation or alleged violation of Environmental Law or the presence or release of any Hazardous Substance within, on, from, relating to or affecting any property, which in the reasonable judgment of the Borrower presents a reasonable likelihood of having a Material Adverse Effect, provide notice thereof, and upon request, copies of relevant documentation, provided, however, no such notice is required with respect to matters disclosed in Schedule 6.19 or matters with respect to which notice has previously been provided pursuant to this Section 7.1(1);
- (1) with reasonable promptness (and in no event later than thirty (30) days after request), the Borrower shall provide to the Agent upon request by the Agent which request shall not be made more frequently than once every six (6) months or prior to March 31, 2004, a true and correct schedule of each parcel of real property owned in fee by the Borrower or its Restricted Subsidiaries (other than AEPLP and its Subsidiaries), as of the end of the most recent calendar quarter, which schedule shall set forth the address (or approximate address if no address is available), county, record owner and approximate value (as determined in good faith by the General Partner) of each such parcel of real property;
- (m) within 30 days of the acquisition by the Borrower or its Restricted Subsidiary (other than AEPLP or its Subsidiaries) of any parcel of real property for which the Borrower or such Restricted Subsidiary, as the case may be, does not execute and deliver a Mortgage pursuant to Section 7.10, the Borrower shall provide to the Agent evidence that (i) the individual value of such acquired parcel of real property is not in excess of \$100,000 (as determined in good faith by the General Partner) or (ii) the aggregate value of such acquired parcel of real property and the other real properties owned by the Borrower or its Restricted Subsidiary (other than AEPLP or its Subsidiaries) and located within the same district is not in excess of \$500,000 (as determined in good faith by the General Partner); and

(n) with reasonable promptness, such other information and data (financial or other) with respect to the Obligors or any of their Subsidiaries as from time to time may be reasonably requested by the Agent or any Bank.

7.2 Reserved.

- 7.3 Adequate Reserves. The Borrower will, and will cause each of its Restricted Subsidiaries to maintain, overall reserves on their respective books and records in accordance with GAAP, which overall reserves shall be adequate in the opinion of the management of the Borrower and each Restricted Subsidiary for the purposes for which they were established.
- 7.4 Partnership or Corporate Existence; Business; Compliance with Laws. (a) Except as otherwise expressly permitted in accordance with Section 8.7 or 8.8, (i) the Borrower will at all times preserve and keep in full force and effect its partnership existence and its status as a partnership not taxable as a corporation, (ii) the Borrower will cause each of the Restricted Subsidiaries to keep in full force and effect its partnership or corporate existence and (iii) the Borrower will, and will cause each Restricted Subsidiary to, at all times preserve and keep in full force and effect all of its material rights and franchises; provided, however, that the partnership or corporate existence of any Restricted Subsidiary, and any right or franchise of the Borrower or any Restricted Subsidiary, may be terminated notwithstanding this Section 7.4 if, in the good faith judgment of the Borrower, such termination (x) is in the best interest of the Borrower and the Restricted Subsidiaries, (y) is not disadvantageous to the Agent, the Issuing Bank or the Banks in any material respect and (z) would not have a reasonable likelihood of having a Material Adverse Effect.
- (b) The Borrower will, and will cause each of its Subsidiaries to, at all times comply with all laws, regulations and statutes (including without limitation any zoning or building ordinances) applicable to it, except for failures to so comply which, individually or in the aggregate, would not present a reasonable likelihood of having a Material Adverse Effect.
- (c) The Borrower will not, and will not permit any Restricted Subsidiary to, engage in any lines of business other than its current Business as defined in this Agreement and other activities incidental or related to the Business.
- 7.5 Payment of Taxes and Claims. The Borrower will, and will cause each of its Subsidiaries to, pay all material Taxes, Other Taxes, assessments and other governmental charges imposed upon it or any of its Subsidiaries, or any of its or its Subsidiaries' properties or assets or in respect of any of its or any of its Subsidiaries' franchises, business, income or profits when the same becomes due and payable, and all claims (including without limitation claims for labor, services, materials and supplies) for sums which have become due and payable and which by law have or might become a Lien upon any of its or any of its Subsidiaries' properties or assets, and promptly reimburse the Banks for any such Taxes, Other Taxes, assessments, charges or claims paid by them; provided, that no such Tax, Other Tax, assessment, charge or claim need be paid or reimbursed if it is being contested in good faith by appropriate proceedings promptly initiated and diligently conducted and if such reserves or other appropriate provision, if any, as shall be required by GAAP shall have been made therefor and be adequate in the good faith judgment of the General Partner.

- 7.6 Maintenance of Properties; Insurance. (a) The Borrower will maintain or cause to be maintained in good repair, working order and condition all properties used or useful in the business of the Borrower and the Restricted Subsidiaries and from time to time will make or cause to be made all appropriate repairs, renewals and replacements thereof.
- (b) The Borrower will maintain or cause to be maintained, with Permitted Insurers to the extent available on commercially reasonable terms from Permitted Insurers and otherwise with financially sound and reputable insurers, insurance with respect to its properties and business and the properties and business of the Restricted Subsidiaries of the types and in the amounts specified in Sections 20 and 21 of the Collateral Agency Agreement and the Collateral Agent shall be named as an additional insured party on each insurance policy maintained pursuant to this Section 7.6(b).
- 7.7 License Agreements. The Borrower will perform and comply with all of its obligations under each of the License Agreements to which it is a party, will enforce each such License Agreement against each other party thereto and will not accept the termination of any such License Agreement or any amendment or supplement thereof or modification or waiver thereunder, unless any such failure to perform, comply or enforce or any such acceptance would not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect.
- 7.8 Chief Executive Office. The Borrower will not move its chief executive office and the office at which it maintains its records relating to the transactions contemplated by this Agreement and the Loan Documents unless not less than 45 days' prior written notice of its intention to do so clearly describing the new location, shall have been given to the Collateral Agent and each Bank
- Guarantors. Promptly upon any Person becoming a Restricted Subsidiary of the Borrower, the Borrower will cause such Restricted Subsidiary to execute and deliver to the Collateral Agent such appropriate documents to become (i) a guarantor under the Subsidiary Guarantee and an assignor under the Subsidiary Security Agreement and (ii) bound by the terms and provisions of the Collateral Agency Agreement. If any Restricted Subsidiary then or thereafter shall have any interests in real property, the Borrower will, subject to and if required by the provisions of Section 7.10, cause such Restricted Subsidiary to execute and deliver to the Collateral Agent a Mortgage in substantially the form of Exhibit E with such changes, mutatis mutandis, so as to make such instrument applicable to such Restricted Subsidiary and its interests in real property, and cause the same to be recorded, published, registered and filed as provided in Section 7.10. Notwithstanding the foregoing, until the AEPLP Available Date, the Borrower shall not be required to cause AEPLP or any of AEPLP's Subsidiaries, and neither AEPLP nor any of its Subsidiaries shall be required to comply with this Section 7.9.
- 7.10 New Mortgages; Recordation. The Borrower and its Restricted Subsidiaries, if applicable, will execute and deliver a Mortgage (in substantially the form of Exhibit E with such changes, mutatis mutandis, so as to make such instrument applicable to the Borrower or such Restricted Subsidiary, as the case may be, and its interests in real property) covering each district location hereafter acquired by it and any other real property hereafter acquired by it which has an individual value in excess of \$100,000 or which has an aggregate value in excess of \$500,000 (in

each case as determined in good faith by the General Partner) and which is not already subject to the Lien of a Mortgage. The Borrower will cause to be duly recorded, published, registered and filed all the documents set forth in paragraph (b) of the definition of Security Documents (or documents or instruments in respect thereof), in such manner and in such places as is required by law to establish, and if applicable, perfect and preserve the rights and security interests of the parties thereto and their respective successors and assigns in the General Collateral. The Borrower will obtain and deliver to the Collateral Agent an ALTA loan policy (10-17-92 form), or its equivalent in any state, of title insurance for real property hereafter acquired by it which has an individual value in excess of \$1,000,000 (as determined in good faith by the General Partner) showing the Lien of a Mortgage as a first priority mortgage Lien. The Borrower will pay or cause to be paid all taxes, fees and other governmental charges in connection with the execution, delivery, recording, publishing, registration and filing of such documents or instruments in such places, together with all expenses and premiums of the title companies in connection with the issuance of any title policies or endorsements thereto. Notwithstanding the foregoing, until the AEPLP Available Date, the Borrower shall not be required to cause AEPLP or any of AEPLP's Subsidiaries, and neither AEPLP nor any of its Subsidiaries shall be required to comply with this Section 7.10.

- Further Assurances. At any time and from time to time promptly, the Borrower shall, at its expense, execute and deliver to the Agent and each Bank and to the Collateral Agent such further instruments and documents, and take such further action, as the Agent or any Bank may from time to time reasonably request, in order to further carry out the intent and purpose of this Agreement and the other Loan Documents and to establish, perfect, preserve and protect the rights, interests and remedies created, or intended to be created, in favor of the Banks and the Collateral Agent hereunder and thereunder, including without limitation the execution, delivery, recordation and filing of financing statements and continuation statements under the Uniform Commercial Code of any applicable jurisdiction, and the delivery of satisfactory opinions of counsel as to the recording, registration or filing of the Security Documents (or documents in respect thereof) and the legal, valid, binding and enforceable nature thereof and the validity of the Liens created thereby on the General Collateral. Until the AEPLP Available Date, nothing in this Section 7.11 shall require the Borrower, AEPLP or AmeriGas Eagle Parts & Service to provide to the Collateral Agent a first priority Lien in the assets of AEPLP or AmeriGas Eagle Parts & Service.
- 7.12 Covenant to Secure Notes Equally. The Borrower covenants that, if it or any Restricted Subsidiary shall create or assume any Lien upon any of its property or assets, whether now owned or hereafter acquired, other than Liens permitted by the provisions of Section 8.3 (unless prior written consent to the creation or assumption thereof shall have been obtained pursuant to Section 12.1) it will make or cause to be made effective provision whereby the Loans will be secured by such Lien equally and ratably with any and all other Indebtedness thereby secured so long as any such other Indebtedness shall be so secured, it being understood that the provision of such equal and ratable security shall not constitute a cure or waiver of any related Event of Default.
- 7.13 Designations With Respect to Subsidiaries. (a) The Borrower may designate any Restricted Subsidiary or newly acquired or formed Subsidiary as an Unrestricted Subsidiary

77

or any Unrestricted Subsidiary as a Restricted Subsidiary, in each case subject to satisfaction of the following conditions:

- (i) immediately before and after giving effect to such designation, no Default or Event of Default shall exist and be continuing; and
- (ii) after giving effect to such designation, the Borrower would be permitted to incur at least \$1 of additional Indebtedness in accordance with the provisions of clauses (i) and (ii) of Section 8.1(f);
- (iii) in the case of a designation of a Restricted Subsidiary or a newly acquired or formed Subsidiary as an Unrestricted Subsidiary, the conditions set forth in subsection (ii)(A) of Section 8.8(c) (the "SALE CONDITION") and Section 8.4(h) (the "INVESTMENT CONDITION") would be satisfied, assuming for this purpose that such designation (and all prior designations of Restricted Subsidiaries or newly acquired or formed Subsidiaries as Unrestricted Subsidiaries during the current fiscal year) constitutes a sale by the Borrower of (in the case of the Sale Condition), and an Investment by the Borrower in an amount equal to (in the case of the Investment Condition), all the assets of the Subsidiary so designated, in each case for an amount equal to (x) the net book value of such assets in the case of a Restricted Subsidiary and (y) the cost of acquisition or formation in the case of a newly acquired or formed Subsidiary (such amounts being herein referred to as "DESIGNATION AMOUNTS" and deemed to constitute Net Proceeds for the purposes of the Sale Condition); provided, however, that notwithstanding anything to the contrary contained herein, until the AEPLP Security Date, AEPLP and each of its Subsidiaries shall at all times remain Restricted Subsidiaries and in no event shall the Borrower have any right to redesignate AEPLP or any of its Subsidiaries as an Unrestricted Subsidiary.
- (b) A Subsidiary that has twice previously been designated an Unrestricted Subsidiary may not thereafter be designated as a Restricted Subsidiary.
- (c) The Borrower shall deliver to the Agent and each Bank, within 20 Business Days after any such designation, an Officer's Certificate stating the effective date of such designation and stating that the foregoing conditions contained in this Section 7.13 have been satisfied. Such certificate shall be accompanied by a schedule setting forth in reasonable detail the calculations demonstrating compliance with such conditions, where appropriate.
- (d) All Investments, Indebtedness, Liens, Guaranty Obligations and other obligations that an Unrestricted Subsidiary (the "DESIGNEE") has at the time of being designated a Restricted Subsidiary hereunder shall be deemed to have been acquired, made or incurred, as the case may be, at the time of such designation and in anticipation of such Designee becoming a Subsidiary and of acquiring its assets (except as otherwise specifically provided in Section 8.1(i) or (j) or Section 8.3(m)).
- 7.14 Covenants of the General Partner and Petrolane. (a) Petrolane covenants that it will engage only in the business and activities described in Schedule 6.16, and each of the General Partner and Petrolane covenants that it will not create any Liens on the general partnership interests in the Borrower or the Public Partnership or dispose of any assets or

properties covered by the terms of any License Agreement and will maintain and keep in effect its corporate existence and franchises.

- (b) Each of the General Partner and Petrolane will deliver to the Agent, on behalf of the Banks, and the Agent will promptly distribute to each Bank at their respective addresses as set forth on Schedule 12.2 hereto, or such other office as may be designated by the Agent and Banks from time to time, (i) financial statements as to itself of the same character described in, and at the times specified in, Sections 7.1(a) and 7.1(b) with respect to the Borrower (the "GUARANTOR FINANCIALS"), in each case certified and reported on in the same manner as the Borrower Financials (except that the financial statements of Petrolane need not be audited), and (ii) with reasonable promptness, such other information and data (financial or other) with respect to the General Partner or Petrolane, as the case may be, as may from time to time be reasonably requested by the Agent.
- (c) The General Partner will perform and comply with all of its obligations under the Partnership Agreement and each of the License Agreements to which it is a party, will enforce the Partnership Agreement and each such License Agreement against each other party thereto and will not accept the termination of the Partnership Agreement or any such License Agreement or any amendment or supplement thereof or modification or waiver thereunder, unless any such failure to perform, comply or enforce or any such acceptance would not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect.
- (d) Section 6.5 of the Partnership Agreement (the "INCORPORATED COVENANT") as in effect on the date hereof, together with all related definitions, is hereby incorporated herein in the form included in the Partnership Agreement on April 19, 1995 and without regard to any subsequent amendments or waivers of the provisions of, or any termination of, the Partnership Agreement. The General Partner agrees to fully perform and comply with the Incorporated Covenant.
- (e) The General Partner agrees to apply all distributions received by the Public Partnership from the Borrower and made with the proceeds of Indebtedness incurred pursuant to Section 8.1(1) only to make payments, purchases, prepayments, redemptions, defeasances or other repayments (scheduled or unscheduled) of Indebtedness of the Public Partnership (and to pay all fees, premiums, make whole amounts and transaction expenses incurred in connection therewith).
- 7.15 Books and Records. The Borrower will, and will cause each of its Restricted Subsidiaries to, keep books and records which accurately reflect all of its business affairs and transactions and permit the Agent and each Bank or any of their respective representatives, at reasonable times and intervals, to visit all of its offices, to discuss its financial matters with its officers and to examine (and, at the expense of the Borrower, photocopy extracts from) any of its books or other Borrower records. Upon the occurrence and during the continuance of any Default or Event of Default the Borrower hereby authorizes its independent public accountant to discuss the Borrower's financial matters with the Agent and each Bank or any of their respective representatives provided that a representative of the Borrower is present. So long as a Default has occurred and is continuing, the Borrower shall pay any fees of the Agent, each Bank and

such independent public accountant incurred in connection with the Agent's or any Bank's exercise of its rights pursuant to this Section.

- $$ 7.16 Environmental Covenant. The Borrower will, and will cause each of the Restricted Subsidiaries to:
- (a) comply with all applicable Environmental Laws and any permit, license, or approval required under any Environmental Law, except for failures to so comply which would not present a reasonable likelihood of having a Material Adverse Effect;
- (b) store, use, release, or dispose of any Hazardous Substance in compliance with Environmental Laws at any property owned or leased by the Borrower or any of its Restricted Subsidiaries, except where such non-compliance would not present a reasonable likelihood of having a Material Adverse Effect;
- (c) avoid committing any act or omission which would cause any Lien to be asserted against any property owned by the Borrower or any of its Restricted Subsidiaries pursuant to any Environmental Law, except where such Lien would not present a reasonable likelihood of having a Material Adverse Fffect.
- (d) use, handle or store propane in compliance with Environmental Laws, except where such non-compliance would not present a reasonable likelihood of having a Material Adverse Effect;
- (e) take all steps required by Environmental Law to cure any violation thereof disclosed in Schedule 6.19; and
- (f) provide such information and certificates which the Agent or any Bank may reasonably request from time to time to evidence compliance with this Section 7.16.
- $7.17\,$ $\,$ Post Closing. The Borrower shall comply with the terms of the Post-Closing Agreement.
- 7.18 Tax Disclosure. In the event that the Agent, any of the Banks or the Borrower determine to take any action inconsistent with its intention not to treat the Loans as being a "reportable transaction" (within the meaning of the Treasury Regulation Section 1.6011-4) or a "tax shelter" (within the meaning of Section 6111 of the Code), shall promptly notify the other parties to this Agreement thereof in writing.

ARTICLE VIII

NEGATIVE COVENANTS

So long as any Bank shall have any Commitment hereunder, or any Loan or other Obligation shall remain unpaid or unsatisfied, or any Letters of Credit shall remain outstanding, unless the Required Banks waive compliance in writing:

- 8.1 Indebtedness. The Borrower will not, and will not permit any Restricted Subsidiary to, create, incur, assume or otherwise become or remain directly or indirectly liable with respect to, any Indebtedness, except that:
- (a) the Borrower may become and remain liable with respect to the Indebtedness evidenced by the First Mortgage Notes and Long Term Funded Debt incurred in connection with any extension, renewal, refunding or refinancing of Indebtedness evidenced by the First Mortgage Notes, provided, that the principal amount of such Long Term Funded Debt shall not exceed the principal amount of such Indebtedness evidenced by the First Mortgage Notes, together with any accrued interest and prepayment charges with respect thereto, being extended, renewed, refunded or refinanced;
- (b) the Borrower may become and remain liable with respect to Indebtedness incurred by the Borrower (i) to finance the making of expenditures for the improvement or repair (to the extent such improvements and repairs may be capitalized on the books of the Borrower in accordance with GAAP) of or additions (including additions by way of acquisitions or capital contributions of businesses and related assets) to the General Collateral or (ii) by assumption of Indebtedness in connection with additions (including additions by way of acquisitions or capital contributions of businesses and related assets) to the General Collateral or to extend, renew, refund or refinance any such Indebtedness; provided, that (x) the amount of such assumed Indebtedness shall not exceed the purchase price of such additions and (y) any such extensions, renewals, refundings or refinancings of any such Indebtedness shall not exceed the principal amount thereof;
- (c) subject to Section 8.4(c) any Restricted Subsidiary may become and remain liable with respect to Indebtedness of such Restricted Subsidiary owing to the Borrower or to a Wholly-Owned Restricted Subsidiary, and the Borrower may become and remain liable with respect to Indebtedness owing to a Wholly-Owned Restricted Subsidiary provided it is subordinated to the Obligations and the Parity Debt at least to the extent provided in the subordination provisions set forth in Exhibit I;
- (d) the Borrower may become and remain liable with respect to unsecured Indebtedness of the Borrower owing to the General Partner or an Affiliate of the General Partner, provided, that (i) the aggregate principal amount of such Indebtedness outstanding at any time shall not be in excess of \$50,000,000 and (ii) such Indebtedness is created and is outstanding under an agreement or instrument pursuant to which such Indebtedness is subordinated to the Obligations and the Parity Debt at least to the extent provided in the subordination provisions set forth in Exhibit I;
- (e) the Borrower may become and remain liable with respect to Indebtedness incurred for any purpose permitted by the Revolving Commitment, and any Indebtedness incurred for any such permitted purpose which replaces, extends, renews, refunds or refinances any such Indebtedness, in whole or in part, in an aggregate principal amount at any time not in excess of \$175,000,000 less the aggregate principal amount of the Acquisition Loans (other than Specified Acquisition Loans) outstanding at such time;

- (f) the Borrower may become and remain liable with respect to Indebtedness, in addition to that otherwise permitted by the foregoing subsections of this Section 8.1, if on the date the Borrower becomes liable with respect to any such additional Indebtedness and immediately after giving effect thereto and to the substantially concurrent repayment of any other Indebtedness (i) the ratio of Consolidated Cash Flow to Consolidated Pro Forma Debt Service is equal to or greater than 2.50 to 1.0 and (ii) the ratio of Consolidated Cash Flow to Average Consolidated Pro Forma Debt Service is equal to or greater than 1.25 to 1.0;
- (g) the Borrower and its Restricted Subsidiaries may become and remain liable with respect to the Indebtedness described on Schedule 6.7;
- (h) the Borrower may become and remain liable with respect to obligations under Interest Rate Agreements entered into to hedge interest rate risk;
- (i) any Person that after the Closing Date becomes a Restricted Subsidiary may become and remain liable with respect to any Indebtedness to the extent such Indebtedness existed at the time such Person became a Subsidiary (and was not incurred in anticipation of such Person becoming a Subsidiary); provided, that immediately after giving effect to such Person becoming a Restricted Subsidiary, the Borrower could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8 1(f):
- (j) the Borrower and any Restricted Subsidiary may become and remain liable with respect to Indebtedness relating to any business acquired by or contributed to the Borrower or such Restricted Subsidiary or which is secured by a Lien on any property or assets acquired by or contributed to the Borrower or such Restricted Subsidiary to the extent such Indebtedness existed at the time such business or property or assets were so acquired or contributed (and was not incurred in anticipation thereof) and if such Indebtedness is secured by such property or assets, such security interest does not extend to or cover any other property of the Borrower or any of the Restricted Subsidiaries, provided, that immediately after giving effect to such acquisition or contribution, the Borrower could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8.1(f);
- (k) Capitalized Lease Liabilities not in excess of \$10,000,000 at any time outstanding; and
- (1) the Borrower may become and remain liable with respect to Indebtedness which otherwise complies with the terms of Section 8.1(f), the proceeds of which are used to make distributions permitted under Section 8.5, provided, that the aggregate principal amount of all Indebtedness incurred under this Section 8.1(l) since August 21, 2001 and outstanding at any time shall not exceed \$105,000,000, provided, further, that at the time the Borrower incurs any Indebtedness permitted under the above provisions of this Section 8.1(l), such Indebtedness shall have received (i) a Special Rating and (ii) an investment grade rating from at least two nationally recognized statistical rating organizations (as defined for purposes of Rule 436(g) under the Securities Act), such as Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., Fitch Ratings and Moody's Investors Service;

Further, notwithstanding anything in this Agreement to the contrary, until the AEPLP Security Date, the Borrower will not permit AEPLP or any of its Subsidiaries to create, incur, assume or otherwise become or remain directly or indirectly liable with respect to any Indebtedness, other than (i) Indebtedness of the type described in Section 8.1(c), (ii) the Indebtedness of AEPLP on the date of closing of the Columbia Acquisition, as disclosed in the Columbia Purchase Agreement (which amount was not in excess of \$10,000,000), and (iii) the Indebtedness of AEPLP owing to the Borrower which is evidenced by the Intercompany Note to the extent that the aggregate principal amount outstanding thereunder does not exceed \$137,997,000.

- 8.2 Minimum Interest Coverage. The Borrower will not permit the ratio of EBITDA to Consolidated Interest Expense as at any fiscal quarter end for the four fiscal quarters then ending to be less than 2.25 to 1.0.
- 8.3 Liens, etc. The Borrower will not, and will not permit any Restricted Subsidiary to, directly or indirectly create, incur, assume or permit to exist any Lien on or with respect to any property or asset (including any document or instrument in respect of goods or accounts receivable) of the Borrower or any Restricted Subsidiary, whether now owned or hereafter acquired, or any income or profits therefrom (whether or not provision is made for the equal and ratable securing of the Obligations in accordance with the provisions of Section 7.12), except:
- (a) Liens for taxes, assessments or other governmental charges the payment of which is not yet due and payable or which is being contested in compliance with Section 7.5 hereof and Section 1.18 of the Mortagues:
- (b) Liens of lessors, landlords and carriers, vendors, warehousemen, mechanics, materialmen, repairmen and other like Liens incurred in the ordinary course of business for sums not yet due or the payment of which is being contested in compliance with Section 7.5 hereof and Section 1.18 of the Mortgages, in each case (i) not incurred or made in connection with the borrowing of money, the obtaining of advances or credit or the payment of the deferred purchase price of property or (ii) incurred in the ordinary course of business securing the unpaid purchase price of property or services constituting current accounts payable; and precautionary Liens in favor of lessors under capital leases and leases of equipment in the ordinary course of business;
- (c) Liens (other than any Lien imposed by ERISA) incurred or deposits made in the ordinary course of business (i) in connection with workers' compensation, unemployment insurance and other types of social security, or (ii) to secure (or to obtain letters of credit that secure) the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, performance bonds, purchase, construction or sales contracts and other similar obligations, in each case not incurred or made in connection with the borrowing of money;
- (d) other deposits made to secure liability to insurance carriers under insurance or self-insurance arrangements;
- (e) Liens securing reimbursement obligations under letters of credit, provided in each case that such Liens cover only the title documents and related goods (and any proceeds thereof) covered by the related letter of credit;

- (f) any attachment or judgment Lien, unless the judgment it secures shall not, within 60 days after the entry thereof, have been discharged or execution thereof stayed pending appeal or review, or shall not have been discharged within 60 days after expiration of any such stay;
- (g) leases or subleases granted to others, easements, rights-of-way, restrictions and other similar charges or encumbrances, which, in each case either (i) are granted, entered into or created in the ordinary course of the business of the Borrower or any Restricted Subsidiary or (ii) do not, individually or in the aggregate, present a reasonable likelihood of having a Material Adverse Effect:
- (h) Liens on property or assets of any Restricted Subsidiary securing Indebtedness of the type described in Section 8.1(c) of such Restricted Subsidiary owing to the Borrower or a Wholly-Owned Restricted Subsidiary;
 - (i) [Intentionally omitted];
- (j) Liens created by any of the Security Documents securing Indebtedness evidenced by the First Mortgage Notes (or any extension, renewal, refunding, replacement or refinancing of any such Indebtedness) in accordance with Section 8.1(a);
- (k) Liens created by any of the Security Documents securing the Indebtedness incurred under the Acquisition Commitment (or any extension, renewal, refunding, replacement-or refinancing of any such Indebtedness) in accordance with Section 8.1(b);
- (1) Liens created by any of the Security Documents securing the Indebtedness, or Letters of Credit, incurred under the Revolving Commitment (or any extension, renewal, refunding, replacement or refinancing of any such Indebtedness) in accordance with Section 8.1(e);
- Liens (other than the Liens referred to in clauses (m) (j), (k) or (1) above) securing Indebtedness represented by the First Mortgage Notes or other Indebtedness incurred in accordance with Section 8.1(b), 8.1(e) or 8.1(1) or, to the extent incurred (i) to repay Indebtedness or letter of credit obligations incurred and outstanding under the Acquisition Commitment or the Revolving Commitment (or any extension, renewal, refunding, replacement or refinancing of any such Indebtedness), (ii) to finance the making of expenditures for the improvement or repair (to the extent such improvements and repairs may be capitalized on the books of the Borrower and the Restricted Subsidiaries in accordance with GAAP) of or additions (including additions by way of acquisitions or capital contributions of businesses and related assets) to the General Collateral, or (iii) by assumption in connection with additions (including additions by way of acquisitions or capital contributions of businesses and related assets) to the General Collateral, under Section 8.1(f), provided, that (1) such Liens are effected through an amendment to the Security Documents to the extent necessary to provide the holders of such Indebtedness equal and ratable security in the property and assets subject to the Security Documents with the holders of the Notes and the other Indebtedness secured under the Security Documents, (2) in the case of Indebtedness incurred in accordance with Section 8.1(b) or Section 8.1(f) to finance the making of additions to the General Collateral, the Borrower has delivered to

the Collateral Agent an Officers' Certificate demonstrating that the principal amount of such Indebtedness (net of transaction costs funded by the proceeds of such Indebtedness) does not exceed the lesser of the cost to the Borrower and the Restricted Subsidiaries of such additional property or assets and the fair market value of such additional property or assets at the time of the acquisition thereof (as determined in good faith by the General Partner), and (3) the Borrower has delivered to the Collateral Agent an opinion of counsel reasonably satisfactory to the Collateral Agent with regard to the attachment and perfection of the Lien of the Security Documents with respect to such additional property and assets;

- (n) Liens existing on any property of any Person at the time it becomes a Subsidiary of the Borrower, or existing at the time of acquisition upon any property acquired by the Borrower or any such Subsidiary through purchase, merger or consolidation or otherwise, whether or not assumed by the Borrower or such Subsidiary, or created to secure Indebtedness incurred under Section 8.1(f) to pay all or any part of the purchase price (a "PURCHASE MONEY LIEN") of property (including without limitation Capital Stock and other securities) acquired by the Borrower or a Restricted Subsidiary, provided, that (i) any such Lien shall be confined solely to such item or items of property and, if required by the terms of the instrument originally creating such Lien, other property which is an improvement to or is acquired for use specifically in connection with such acquired property, (ii) such item or items of property so acquired are not required to become part of the General Collateral under the terms of the Security Documents, (iii) in the case of a Purchase Money Lien, the principal amount of the Indebtedness secured by such Purchase Money Lien shall at no time exceed an amount equal to the lesser of (A) the cost to the Borrower and the Restricted Subsidiaries of such property and (B) the fair market value of such property at the time of the acquisition thereof (as determined in good faith by the General Partner), (iv) any such Purchase Money Lien shall be created not later than 30 days after the acquisition of such property and (v) any such Lien (other than a Purchase Money Lien) shall not have been created or assumed in contemplation of such Person's becoming a Subsidiary of the Borrower or such acquisition of property by the Borrower or any Subsidiary;
- (0) easements, exceptions or reservations in any property of the Borrower or any Restricted Subsidiary granted or reserved for the purpose of pipelines, roads, the removal of oil, gas, coal or other minerals, and other like purposes, or for the joint or common use of real property, facilities and equipment, which are incidental to, and do not materially interfere with, the ordinary conduct of the business of the Borrower or any Restricted Subsidiary;
- $\mbox{(p)}$ Liens arising from or constituting Permitted Encumbrances as defined under the Security Documents; and
- (q) any Lien renewing or extending any Lien permitted by subsection (h), (j), (k), (l), (m) or (n) of this Section 8.3, provided, that (i) the principal amount of the Indebtedness secured by any such Lien shall not exceed the principal amount of such Indebtedness outstanding immediately prior to the renewal or extension of such Lien (together with, in the case of Indebtedness permitted by Section 8.1(a), any accrued interest thereon and prepayment charges with respect thereto), and (ii) no assets encumbered by any such Lien other than the assets encumbered immediately prior to such renewal or extension shall be encumbered thereby.

Notwithstanding anything in this Agreement to the contrary, until the AEPLP Security Date, other than Liens permitted by subsections (a), (b), (c), (d), (f), (g), (h), (o) and (to the extent that any such Lien extends or renews a Lien permitted by subsection (h) of this Section 8.3) (q) of this Section 8.3, the Borrower will not permit AEPLP or any of its Subsidiaries to, directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any property or asset (including any document or instrument in respect of goods or accounts receivable) of AEPLP or such Subsidiary, whether such property or assets are now owned or held or hereafter acquired, or any income or profits therefrom.

- 8.4 Investments, Contingent Obligations, etc. The Borrower will not, and will not permit any Restricted Subsidiary to, directly or indirectly (i) make or own any Investment in any Person (including an Investment in a Subsidiary of the Borrower), (ii) create or become liable with respect to any Contingent Obligation with respect to any Indebtedness of a Control Affiliate, or (iii) create or become liable with respect to any Contingent Obligation (provided, however, that nothing contained in this Section 8.4, except clause (ii) above, is intended to limit the making of any Contingent Obligation which would be permitted as Indebtedness under Section 8.1), except:
- (a) the Borrower or any Restricted Subsidiary may make and own Investments in the following (collectively, "CASH EQUIVALENTS"):
- (i) marketable obligations issued or unconditionally guaranteed by the United States of America, or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing one year or less from the date of acquisition thereof,
- (ii) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof maturing within one year from the date of acquisition thereof and having as at such date the highest rating obtainable from either Standard & Poor's Rating Group or Moody's Investors Service, Inc.,
- (iii) commercial paper maturing no more than 270 days from the date of creation thereof and having as of the date of acquisition thereof one of the two highest ratings obtainable from either Standard & Poor's Rating Group or Moody's Investors Service, Inc.,
- (iv) certificates of deposit maturing one year or less from the date of acquisition thereof issued by commercial banks incorporated under the laws of the United States of America or any state thereof or the District of Columbia or Canada, (A) the commercial paper or other short term unsecured debt obligations of which are as of such date rated either A-2 or better (or comparably if the rating system is changed) by Standard & Poor's Rating Group or Prime-2 or better (or comparably if the rating system is changed) by Moody's Investors Service, Inc. or (B) the long-term debt obligations of which are as at such date rated either A or better (or comparably if the rating system is changed) by either Standard & Poor's Rating Group or A-2 or better or comparably if the rating system is changed by Moody's Investors Service, Inc. ("PERMITTED BANKS"),

- (v) Eurodollar time deposits having a maturity of less than 270 days from the date of acquisition thereof purchased directly from any Permitted Bank;
- (vi) bankers' acceptances eligible for rediscount under requirements of the FRB and accepted by Permitted Banks, and
- (vii) obligations of the type described in clause (i), (ii), (iii), (iv) or (v) above purchased from a securities dealer designated as a "primary dealer" by the Federal Reserve Bank of New York or from a Permitted Bank as counterparty to a written repurchase agreement obligating such counterparty to repurchase such obligations not later than 14 days after the purchase thereof and which provides that the obligations which are the subject thereof are held for the benefit of the Borrower or a Restricted Subsidiary by a custodian which is a Permitted Bank and which is not a counterparty to the repurchase agreement in question;
- (b) the Borrower or any Restricted Subsidiary may acquire Capital Stock or other ownership interests, whether in a single transaction or a series of related transactions, of a Person (i) located in the United States or Canada, (ii) incorporated or otherwise formed pursuant to the laws of the United States or Canada or any state or province thereof or the District of Columbia and (iii) engaged in substantially the same business as the Borrower such that, upon the completion of such transaction or series of transactions, such Person becomes a Restricted Subsidiary;
- (c) subject to the provisions of subsection (h) below, the Borrower or any Restricted Subsidiary may make and own Investments (in addition to Investments permitted by subsections (a), (b), (d), (e), (f) and (g) of this Section 8.4) in any Person incorporated or otherwise formed pursuant to the laws of the United States or Canada or any state or province thereof or the District of Columbia which is engaged in the United States or Canada in substantially the same business as the Borrower; provided, that (i) the aggregate amount of all such Investments made by the Borrower and its Restricted Subsidiaries following the Original Closing Date (including without limitation the transactions contemplated by this Agreement) and outstanding pursuant to this subsection (c) and subsection (h) below shall not at any date of determination exceed 10% of Total Assets (the "INVESTMENT LIMIT"), provided, that in addition to Investments that would be permitted under the Investment Limit, during any fiscal year the Borrower and its Restricted Subsidiaries may invest up to \$25,000,000 (the "ANNUAL LIMIT") pursuant to the provisions of this subsection (c), but the unused amount of the Annual Limit shall not be carried over to any future years and provided, further, that neither the Annual Limit nor the Investment Limit shall include the aggregate principal amount of the Intercompany Note outstanding on August 21, 2001 to the extent that such amount is not in excess of \$137,997,000 at the time of determination or the Investment by AEPI in Atlantic Energy, Inc. in existence on August 21, 2001 in the amount of approximately \$3,100,000, (ii) such Investments shall become part of the General Collateral and shall be subjected to the Lien of the Security Documents and (iii) such Investments shall not be made in Capital Stock or Indebtedness of the Public Partnership or any of its Subsidiaries (other than the Borrower and the Restricted Subsidiaries);
- (d) the Borrower or any Restricted Subsidiary may make and own Investments (x) arising out of loans and advances to employees incurred in the ordinary course of business not in excess of \$1,000,000 at any time outstanding, (y) arising out of extensions of

trade credit or advances to third parties in the ordinary course of business and (z) acquired by reason of the exercise of customary creditors' rights upon default or pursuant to the bankruptcy, insolvency or reorganization of a debtor;

- (e) the Borrower and any Restricted Subsidiary may create or become liable with respect to any Contingent Obligation constituting an obligation, warranty or indemnity, not guaranteeing Indebtedness of any Person, which is undertaken or made in the ordinary course of business;
- (f) the Borrower may create and become liable with respect to any Interest Rate Agreements;
- $\hbox{ (g)} \qquad \hbox{any Restricted Subsidiary may make Investments in the } \\ Borrower;$
- (h) the Borrower or any Restricted Subsidiary may make or own Investments in Unrestricted Subsidiaries, provided, that the Net Amount of Unrestricted Investment shall not at any time exceed \$5,000,000 (and subject to the limitations specified in subsection (c) above);
- (i) the Borrower may own Investments in Atlantic Energy, Inc. which were in existence on August 21, 2001 and as set forth in Schedule 8.4(i):
- (j) the Borrower may own Investments consisting of the Intercompany Note to the extent that the aggregate principal amount of the Intercompany Note does not exceed \$137,990,000;
- (k) AEPI, AEPH and AEPLP may remain liable for any obligations, warranties or indemnities set forth in the National Propane Purchase Agreement as such agreement is in effect on the Closing Date; and
- (1) the Borrower may remain (i) liable for its indemnification and guarantee obligations under the Columbia Purchase Agreement, as in effect on August 21, 2001, and (ii) under the Keep Well Agreement, as in effect on August 21, 2001.

Notwithstanding the foregoing, the Borrower may have outstanding undrawn letters of credit (including Letters of Credit) not in excess of \$100,000,000. Furthermore, the Agent and the Banks hereby permanently waive compliance with the obligation under the Subsidiary Security Agreement to provide to the Collateral Agent a Lien on the Capital Stock of Atlantic Energy, Toc

8.5 Restricted Payments. The Borrower will not directly or indirectly declare, order, pay, make or set apart any sum for any Restricted Payment, except that the Borrower may declare or order, and make, pay or set apart, once during each calendar quarter a Restricted Payment if (a) such Restricted Payment is in an amount not exceeding Available Cash for the immediately preceding calendar quarter, and (b) immediately after giving effect to any such proposed action no Event of Default (or Default under Sections 9.1(a), (f) or (g)) shall exist and be continuing; provided, that notwithstanding the foregoing, the Borrower may declare, order, pay, make or set apart sums for Restricted Payments to the Public Partnership at any time, and from time to time, in an aggregate amount not exceeding the proceeds of Indebtedness of the

Borrower incurred pursuant to Section 8.1(1) if immediately after giving effect to any such proposed action no Event of Default (or Default under Sections 9.1(a), (f), or (g)) shall exist and be continuing. The Borrower will comply with, and accrue on its books, the reserve provisions required under the definition of Available Cash. The Borrower will not, in any event, directly or indirectly declare, order, pay or make any Restricted Payment except in cash. The Borrower will not permit any Restricted Subsidiary to declare, order, pay or make any Restricted Payment or to set apart any sum or property for any such purpose (it being understood that nothing in this Section 8.5 shall prohibit any such Restricted Subsidiary from declaring, ordering, paying, making, or setting apart any sum or property for, any payment or other distribution or dividend to (i) the Borrower or any Wholly-Owned Restricted Subsidiary and (ii) so long as no Default or Event of Default shall occur and be continuing, all holders of Capital Stock of such Restricted Subsidiary on a pro rata basis) (with any such distribution or dividend to a Control Affiliate being subject to the limitation of the first sentence of this Section 8.5).

- Transactions with Affiliates. The Borrower will not, and will not permit any Restricted Subsidiary to, directly or indirectly, engage in any transaction with any Affiliate, including without limitation the purchase, transfer, disposition, sale, lease or exchange of assets or the rendering of any service, unless (1)(a) such transaction or series of related transactions is on fair and reasonable terms that are no less favorable to the Borrower or such Restricted Subsidiary, as the case may be, than those which would be obtained in an arm's-length transaction at the time such transaction is agreed upon between Persons which are not Affiliates, and (b) with respect to a transaction or series of transactions involving aggregate payments or value equal to or greater than \$15 million, the Borrower shall have delivered an Officers' Certificate to the Agent certifying that such transaction or series of transactions complies with the preceding clause (a) and that such transaction or series of transactions has been approved by a majority of the Board of Directors of the General Partner (including a majority of the Disinterested Directors), or (2) such transaction or series of related transactions is between the Borrower and any Wholly-Owned Restricted Subsidiary or between two Wholly-Owned Restricted Subsidiaries, provided, however, that this Section 8.6 will not restrict the Borrower, any Restricted Subsidiary or the General Partner from entering into (i) any employment agreement, stock option agreement, restricted stock agreement or other similar agreement or arrangement in the ordinary course of business, (ii) transactions permitted by Section 8.5 and (iii) transactions in the ordinary course of business in connection with reinsuring the self-insurance programs or other similar forms of retained insurable risks of the retail propane business operated by the Borrower, its Subsidiaries and its Affiliates.
 - 8.7 Subsidiary Stock and Indebtedness. The Borrower will not:
- (a) directly or indirectly sell, assign, pledge or otherwise dispose of any Indebtedness of or any shares of stock or similar interests of (or warrants, rights or options to acquire stock or similar interests of) any Restricted Subsidiary, except to a Wholly-Owned Restricted Subsidiary;
- (b) permit any Restricted Subsidiary directly or indirectly to sell, assign, pledge or otherwise dispose of any Indebtedness of the Borrower or any other Restricted Subsidiary, or any shares of stock or similar interests of (or warrants, rights or options to acquire

stock or similar interests of) any other Restricted Subsidiary, except to the Borrower or a Wholly-Owned Restricted Subsidiary;

- (c) permit any Restricted Subsidiary to have outstanding any shares of stock or similar interests which are preferred over any other shares of stock or similar interests in such Restricted Subsidiary owned by the Borrower or a Wholly-Owned Restricted Subsidiary unless such shares of preferred stock or similar interests are owned by the Borrower or a Wholly-Owned Restricted Subsidiary; or
- (d) permit any Restricted Subsidiary directly or indirectly to issue or sell (including without limitation in connection with a merger or consolidation of such Subsidiary otherwise permitted by Section 8.8(a)) any shares of its stock or similar interests (or warrants, rights or options to acquire its stock or similar interests) except to the Borrower or a Wholly-Owned Restricted Subsidiary;

provided, that (i) any Restricted Subsidiary may sell, assign or otherwise dispose of Indebtedness of the Borrower if, assuming such Indebtedness were incurred immediately after such sale, assignment or disposition, such Indebtedness would be permitted under Section 8.1 (other than Section 8.1(c)) (in which case such Indebtedness need not be subject to the subordination provisions required by Section 8.1(c)) and (ii) subject to compliance with Section 8.8(c), all Indebtedness and shares of stock or partnership interests of any Restricted Subsidiary owned by the Borrower or any other Restricted Subsidiary may be simultaneously sold as an entirety for an aggregate consideration at least equal to the fair value thereof (as determined in good faith by the General Partner) at the time of such sale if (x) such Restricted Subsidiáry does not at the time own (A) any Indebtedness of the Borrower or any other Restricted Subsidiary (other than Indebtedness which, if incurred immediately after such transaction, would be permitted under Section 8.1, other than Section 8.1(c)) (in which case such Indebtedness need not be subject to the subordination provisions required by Section 8.1(c)) or (B) any stock or other interest in any other Restricted Subsidiary which is not also being simultaneously sold as an entirety in compliance with this proviso or Section 8.8(b)(ii) and (y) at the time of such transaction and immediately after giving effect thereto, the Borrower could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8.1(f) and (iii) AEPLP may issue or sell its Capital Stock to the Special Limited Partner (as defined in the AEPLP Partnership Agreement) of AEPLP in accordance with Section 5.3 of the AEPLP Partnership Agreement, as such Section 5.3 was in effect on August 21, 2001.

- 8.8 Consolidation, Merger, Sale of Assets, etc. The Borrower will not, and will not permit any Restricted Subsidiary to, directly or indirectly,
- (a) consolidate with or merge into any other Person or permit any other Person to consolidate with or merge into it, except that:
- (i) any Restricted Subsidiary may consolidate with or merge into the Borrower or a Wholly-Owned Restricted Subsidiary if the Borrower or a Wholly-Owned Restricted Subsidiary, as the case may be, shall be the surviving Person and if, immediately after giving effect to such transaction, no Default or Event of Default shall exist and be continuing; and

(ii) any entity (other than a Restricted Subsidiary) may consolidate with or merge into the Borrower or a Wholly-Owned Restricted Subsidiary if the Borrower or a Wholly-Owned Restricted Subsidiary, as the case may be, shall be the surviving Person and if, immediately after giving effect to such transaction, (x) the Borrower (1) shall not have a Consolidated Net Worth, determined in accordance with GAAP applied on a basis consistent with the consolidated financial statements of the Borrower most recently delivered pursuant to Section 7.1(b), of less than the Consolidated Net Worth of the Borrower immediately prior to the effectiveness of such transaction, satisfaction of this requirement to be set forth in reasonable detail in an Officers' Certificate delivered to the Agent at the time of such transaction, (2) shall not be liable with respect to any Indebtedness or allow its property to be subject to any Lien which it could not become liable with respect to or allow its property to become subject to under this Agreement (including without limitation under Section 8.1 or 8.3) on the date of such transaction, and (3) could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8.1(f), (y) substantially all of the assets of the Borrower and its Restricted Subsidiaries shall be located and substantially all of their business shall be conducted within the United States and Canada and (z) no Default or Event of Default shall exist and be continuing;

subject to compliance with Section 12.1, the (iii) Borrower may consolidate with or merge into any other entity if (w) the surviving entity is a corporation or limited partnership organized and existing under the laws of the United States of America or any state thereof or the District of Columbia, with substantially all of its properties located and its business conducted (without giving effect to the properties owned by, and the business conducted by, Unrestricted Subsidiaries) within the United States and Canada, (x) such corporation or limited partnership expressly and unconditionally assumes the obligations of the Borrower under this Agreement, and the other Loan Documents and License Agreements to which the Borrower is a party, and delivers to the Agent an opinion of counsel reasonably satisfactory to the Required Banks with respect to the due authorization and execution of the related agreement of assumption and the enforceability of such agreement against such corporation or partnership and the continued effectiveness and priority of the Liens of the Security Documents, (y) immediately after giving effect to such transaction, such corporation or limited partnership (1) shall not have (without giving effect to Unrestricted Subsidiaries) a Consolidated Net Worth, determined in accordance with GAAP applied on a basis consistent with the consolidated financial statements of the Borrower most recently delivered pursuant to Section 7.1(b), of less than the Consolidated Net Worth of the Borrower immediately prior to the effectiveness of such transaction, satisfaction of this requirement to be set forth in reasonable detail in an Officers' Certificate delivered to the Agent at the time of such transaction, (2) shall not be liable with respect to any Indebtedness or allow its property to be subject to any Lien which it could not become liable with respect to or allow its property to become subject to under this Agreement (including without limitation under Section 8.1 or 8.3) on the date of such transaction and (3) could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8.1(f), and (z) immediately after giving effect to such transaction no Default or Event of Default shall exist and be continuing; or

(b) sell, lease, abandon or otherwise dispose of all or substantially all its assets, except that:

- (i) any Restricted Subsidiary may sell, lease or otherwise dispose of all or substantially all its assets to the Borrower or to a Wholly-Owned Restricted Subsidiary; and
- (ii) subject to compliance with clause (c) of this Section 8.8, any Restricted Subsidiary may sell, lease or otherwise dispose of all or substantially all its assets as an entirety for an aggregate consideration at least equal to the fair value thereof (as determined in good faith by the General Partner) at the time of such sale if (x) the assets being sold, leased or otherwise disposed of do not include (A) any Indebtedness of the Borrower or any other Restricted Subsidiary (other than Indebtedness which, if incurred immediately after such transaction, would be permitted under Section 8.1 (other than Section 8.1(c)) so long as such Indebtedness is held by a Person other than the Borrower or a Restricted Subsidiary), in which case such Indebtedness need not be subject to the subordination provisions required by Section 8.1(c) or (B) any stock of or other equity interest in any other Restricted Subsidiary which is not also being simultaneously sold as an entirety in compliance with this subsection (b)(ii) or the proviso of Section 8.7 and (y) at the time of such transaction and immediately after giving effect thereto, the Borrower could incur at least \$1 of additional Indebtedness in compliance with clauses (i) and (ii) of Section 8.1(f); and
- (iii) the Borrower may sell, lease or otherwise dispose of all or substantially all its assets to any corporation or limited partnership into which the Borrower could be consolidated or merged in compliance with subsection (a)(iii) of this Section 8.8, provided, that each of the conditions set forth in such subsection (a)(iii) shall have been fulfilled; or
- (c) (1) sell, lease, convey, abandon or otherwise dispose of any of its assets (except in a transaction permitted by subsection (a)(i), (a)(iii), (b)(i) or (b)(iii) of this Section 8.8 or sales of inventory in the ordinary course of business consistent with past practice), including by way of a Sale and Lease-Back Transaction, or (2) issue or sell Capital Stock of the Borrower or any Subsidiary (other than to the Borrower or a Wholly-Owned Restricted Subsidiary), in the case of either clause (1) or (2) above, whether in a single transaction or a series of related transactions (each of the foregoing non-excepted transactions, an "ASSET SALE"), unless:
- (ii) one of the following two conditions shall be

satisfied:

(A) the aggregate Net Proceeds of all assets so disposed of (whether or not leased back) by the Borrower and its Restricted Subsidiaries during the current fiscal year (including (x) amounts deemed to be proceeds in connection with designations of Restricted Subsidiaries as Unrestricted Subsidiaries during such fiscal year under Section 7.13, (y) Net Proceeds of dispositions of shares pursuant to Section 8.7 or sales of assets pursuant to Section 8.8(b)), less the amount of all Net Proceeds of prior dispositions of assets during such fiscal year previously applied in

accordance with subsection (ii)(B) of this Section 8.8(c), shall not exceed \$10,000,000 during such fiscal year; or

in the event that such Net Proceeds (less the amount thereof previously applied in accordance with this subsection (ii)(B) during the current fiscal year exceed \$10,000,000 (such excess Net Proceeds actually realized being herein called "EXCESS SALE PROCEEDS"), the Borrower shall within 360 days of the date of the disposal of the assets giving rise to such proceeds, cause an amount equal to such Excess Sale Proceeds to be applied (with the designation of an Unrestricted Subsidiary as a Restricted Subsidiary being deemed to be such an application to the extent of the fair value of such Restricted Subsidiary as determined in good faith by the General Partner) (x) to the acquisition of assets in replacement of the assets so disposed of or of assets which may be productively used in the United States or Canada in the conduct of the Business (and such newly acquired assets shall become part of the General Collateral and shall be subjected to the Lien of the Security Documents), or (y) to the extent not applied pursuant to the immediately preceding clause (x), to the prepayment of the Obligations and Parity Debt, if any, pursuant to Section 2.7(a) hereof, all as provided in Section 4(c) of the Collateral Agency Agreement and such Section 2.7(a); and

(iii) (A) the consideration received for such assets is at least equal to their aggregate fair market value (as determined in good faith by the Board of Directors of the General Partner) at the time of such disposition and that such consideration has been applied or is being held for application in accordance with the terms of this Agreement and (B) at least 80% of the consideration therefor received is in the form of cash; provided, however, that the amount of (1) any liabilities (as shown on the Borrower's or such Restricted Subsidiary's most recent balance sheet or in the notes thereto) of the Borrower or any Restricted Subsidiary (other than liabilities that are by their terms subordinated in right of payment to the Loans) that are assumed by the transferee of any such assets and (2) any notes or other obligations received by the Borrower or any such Restricted Subsidiary from such transferee that are immediately converted by the Borrower or such Restricted Subsidiary into cash (to the extent of the cash received), shall be deemed to be cash for purposes of this clause (B); and provided, further, that the 80% limitation referred to in this clause (B) shall not apply to any Asset Sale in which the cash portion of the consideration received therefrom, determined in accordance with the foregoing proviso, is equal to or greater than what the after-tax proceeds would have been had such Asset Sale complied with the aforementioned 80% limitation.

Notwithstanding the foregoing, Asset Sales shall not be deemed to include (1) any transfer of assets or issuance or sale of Capital Stock by the Borrower or any Restricted Subsidiary to the Borrower or a Wholly-Owned Restricted Subsidiary, (2) any transfer of assets or issuance or sale of Capital Stock by the Borrower or any Restricted Subsidiary to any Person in exchange for other assets used in a line of business permitted under Section 7.4(c) and having a fair market value (as determined in good faith by the General Partner) not less than that of the assets so transferred or Capital Stock so issued or sold (so long as such assets shall become part of the

General Collateral and shall be subjected to the Lien of the Security Documents) and (3) any transfer of assets pursuant to an Investment permitted by Section 8.4.

- 8.9 Use of Proceeds. (a) The Obligors will not, and will not suffer or permit any Subsidiary to, use any portion of the Loan proceeds or any Letter of Credit, directly or indirectly, (i) to purchase or carry Margin Stock, (ii) to repay or otherwise refinance Indebtedness of the Borrower or others incurred to purchase or carry Margin Stock, (iii) to extend credit for the purpose of purchasing or carrying any Margin Stock, or (iv) to acquire any security in any transaction that is subject to Section 13 or 14 of the Exchange
- any portion of the Loan proceeds or any Letter of Credit (i) knowingly to purchase Ineligible Securities from the Arranger during any period in which the Arranger makes a market in such Ineligible Securities, (ii) knowingly to purchase during the underwriting or placement period Ineligible Securities being underwritten or privately placed by the Arranger, or (iii) to make payments of principal or interest on Ineligible Securities underwritten or privately placed by the Arranger and issued by or for the benefit of the Borrower or any Affiliate of the Borrower. The Arranger is a registered broker-dealer and permitted to underwrite and deal in certain Ineligible Securities; and "INELIGIBLE SECURITIES" means securities which may not be underwritten or dealt in by member banks of the Federal Reserve System under Section 16 of the Banking Act of 1933 (12 U.S.C. Section 24, Seventh), as amended.
- (c) The proceeds of the Revolving Loans will be used for working capital purposes and general purposes of the Borrower and its Restricted Subsidiaries.
- (d) The proceeds of the Acquisition Loans will be used for, as selected by the Borrower in a Notice of Borrowing, (i) the acquisition by the Borrower of companies or assets in businesses similar to the Business, and may be used, without limitation, for the payment of related fees and expenses and the retirement, repayment or refinancing of any Indebtedness incurred as part of such acquisition, including any Indebtedness assumed by the Borrower in connection with an addition of assets by way of capital contribution and (ii) in the event there is no availability under the Revolving Commitments, working capital purposes and general purposes (other than the purposes described in clause (i) above) of the Borrower and its Restricted Subsidiaries.
- (e) The proceeds of the Swing Line Loans will be used for working capital purposes and general purposes of the Borrower and its Restricted Subsidiaries.
- 8.10 Change in Business. The Borrower will not, and will not suffer or permit any Restricted Subsidiary to, engage in any material line of business substantially different from the Business.
- 8.11 Accounting Changes. The Borrower will not, and will not suffer or permit any Restricted Subsidiary to, make any significant change in accounting treatment or reporting practices, except as required by GAAP, or change the fiscal year of the Borrower or of any Subsidiary.

- 8.12 Clean Down. The Borrower will not permit the sum of (a) the outstanding Revolving Loans and (b) the outstanding Specified Acquisition Loans to exceed \$30,000,000 for a period of 30 consecutive days during each fiscal year.
- 8.13 Receivables. The Borrower will not, and will not permit any Restricted Subsidiary to, discount, pledge or sell (with or without recourse) any of its accounts or notes receivable, except for sales of receivables (i) made in the ordinary course of business with a face amount not to exceed \$500,000 in the aggregate which have been sold and remain unpaid by the account debtors, (ii) without recourse which are seriously past due and which have been substantially written off as uncollectible or collectible only after extended delays, (iii) from a Restricted Subsidiary to the Borrower or (iv) made in connection with the sale of a business but only with respect to the receivables directly generated by the business so sold.
- 8.14 Leverage Ratio. The Borrower will not permit the Leverage Ratio at any time to exceed 4.75 to 1.00. For purposes of this Section 8.14, the Borrower may elect whether to calculate EBITDA (i) as at the end of any fiscal quarter for the four full consecutive fiscal quarters most recently ended or (ii) as at the end of any fiscal quarter for the eight full consecutive fiscal quarters most recently ended (in which case EBITDA shall be divided by two).
- Acquisitions. After the date hereof and until the AEPLP Security Date, the Borrower will not, and will not permit any Restricted Subsidiary to, make any Acquisition unless, after giving effect to the consummation of such Acquisition (including any substantially concurrent mergers), (a) all PP&E Assets acquired in connection with such Acquisition shall be owned by the Borrower or a Restricted Subsidiary, (b) the aggregate net book value of the PP&E Assets of AEPLP and its Subsidiaries (both prior to and after giving effect to such Acquisition) shall not exceed the sum of (i) 33-1/3% of the aggregate net book value of all PP&E Assets of the Borrower and its Restricted Subsidiaries and (ii) \$70,000,000 and (c) the aggregate net book value (as determined in good faith by the General Partner) of all PP&E Assets acquired by AEPLP or any of its Subsidiaries in any fiscal year pursuant to Acquisitions (other than PP&E Assets acquired with the proceeds of any prior or concurrent Capped Investments or PP&E Transfers) ("AEPLP ACQUISITIONS") shall not, together with any Capped Investments and any PP&E Transfers made in such fiscal year pursuant to Section 8.17(a) and Section 8.17(b)(iii), respectively, in the aggregate, exceed (i) \$35,000,000 (the "YEARLY THRESHOLD"), plus (ii) the amount of any Carryover Threshold (such sum is referred to herein as the "PP&E" ACOUISITION/INVESTMENT/TRANSFER LIMIT"). "CARRYOVER THRESHOLD" shall mean, for any fiscal year, an amount equal to the PP&E Acquisition/Investment/Transfer Limit for the prior fiscal year minus the aggregate AEPLP Acquisitions, Capped Investments and PP&E Transfers in such prior fiscal year, provided, that the Carryover Threshold shall in no event exceed \$100,000,000. As June 30, 2003, the Carryover Threshold was \$70,000,000 and the PP&E Acquisition/Investment/Transfer Limit for the period ending June 30, 2003 was \$102,000,000.
- 8.16 Limitation on Restricted Agreements. The Borrower will not, and will not permit any Subsidiary to, enter into, or suffer to exist, any agreement (other than the National Propane Purchase Agreement) with any Person which, directly or indirectly, prohibits or limits the ability of any Restricted Subsidiary to (a) pay dividends or make other distributions to the

Borrower or prepay any Indebtedness owed to the Borrower, (b) make loans or advances to the Borrower or (c) transfer any of its properties or assets to the Borrower.

- 8.17 AEPLP. Notwithstanding anything in this Agreement to the contrary (including the final paragraph of Section 8.8 hereof), until the first date as of which (i) the property and assets of AEPLP and each of its Subsidiaries have become part of the General Collateral and are subjected to the Lien of the Security Documents and (ii) AEPLP and each of its Subsidiaries have become guarantors under the Subsidiary Guarantee and assignors under the Subsidiary Security Agreement in accordance with Section 7.9 and 7.11 hereof (such date, the "AEPLP SECURITY DATE"), provided, that (A) the security interest granted by AEPLP pursuant to the Subsidiary Security Agreement may be subject and subordinate to the first priority Lien on the assets of AEPLP held by the Borrower to secure the obligations of AEPLP under the Intercompany Note and the Intercompany Loan Agreement, upon terms and conditions satisfactory to the Collateral Agent, (B) the security interest granted by any Subsidiary of AEPLP pursuant to the Subsidiary Security Agreement may be subject and subordinate to the first priority Lien on the assets of such Subsidiary held by the Borrower to secure the obligation of such Subsidiary to guarantee (the "AEPLP SUBSIDIARY GUARANTY") the obligations of AEPLP under the Intercompany Note and the Intercompany Loan Agreement, upon terms and conditions satisfactory to the Collateral Agent, (C) the Subsidiary Guarantee of each Subsidiary of AEPLP may be subject and subordinate to the guaranty of such Subsidiary in favor of the Borrower pursuant to the AEPLP Subsidiary Guaranty of such Subsidiary, upon terms and conditions satisfactory to the Collateral Agent, and (D) the Subsidiary Guarantee of AEPLP may be subject and subordinate to the obligations of AEPLP under the Intercompany Note and the Intercompany Loan, upon terms and conditions satisfactory to the Collateral Agent:
- (a) Investments. The Borrower will not, and will not permit any Restricted Subsidiary (other than AEPLP and its Subsidiaries) (each, a "NON-AEPLP RESTRICTED SUBSIDIARY") to, directly or indirectly, make or own any Investment in AEPLP or any of its Subsidiaries, except for Investments in AEPLP or its Subsidiaries permitted under Sections 8.4(b), (c), (d), (e), (i) and (j) and Section 8.17(b) hereof; provided, however, that the aggregate net book value (as determined in good faith by the General Partner) of all such Investments made pursuant to Sections 8.4(b) and (c) (the "CAPPED INVESTMENTS") in any fiscal year shall not, together with any AEPLP Acquisitions and PP&E Transfers made in such fiscal year pursuant to Section 8.15 and Section 8.17(b)(iii), respectively, in the aggregate, exceed the PP&E Acquisition/Investment/Transfer Limit for such fiscal year.
- (b) Asset Transfers. The Borrower will not, and will not permit any Non-AEPLP Restricted Subsidiary to, directly or indirectly, sell, lease, convey or otherwise transfer, directly or indirectly, any of its assets to AEPLP or any Subsidiary of AEPLP, including by way of a Sale and Lease-Back Transaction (each, a "TRANSFER"), except that:
- (i) the Borrower may, and may permit any non-AEPLP Restricted Subsidiary to, Transfer to AEPLP or any of its Subsidiaries assets, provided, that (A) such assets ("NON-PP&E ASSETS") would not, in accordance with the past practice of the Borrower, be classified and accounted for as "property, plant and equipment" on the consolidated balance sheet of the Borrower and the Restricted Subsidiaries, (B) the consideration paid by AEPLP or its Subsidiaries to the Borrower or a Non-AEPLP Restricted Subsidiary for such Non-PP&E

Assets is at least equal to the transferor's aggregate net book value therefor and (C) the aggregate amount of propane inventory (by number of gallons) of AEPLP and its Subsidiaries shall not at any time exceed 40% of the aggregate amount of propane inventory (by number of gallons) of the Borrower and the Restricted Subsidiaries:

(ii) the Borrower may, and may permit any Non-AEPLP Restricted Subsidiary to, Transfer to AEPLP or any of its Subsidiaries assets in exchange for other assets used in the line of business permitted under Section 8.10 and having a fair market value (as determined in good faith by the General Partner, and the Managing General Partner (as defined in the AEPLP Partnership Agreement) of AEPLP) not less than that of the assets so Transferred (so long as the assets Transferred to the Non-AEPLP Restricted Subsidiary or to the Borrower shall become part of the General Collateral and shall be subjected to the Lien of the Security Documents);

(iii) the Borrower may, and may permit any Non-AEPLP Restricted Subsidiary to, Transfer (a "PP&E TRANSFER") to AEPLP or any of its Subsidiaries PP&E Assets (together with associated working capital), provided, that (A) the aggregate net book value (as determined in good faith by the General Partner) of all PP&E Assets that are Transferred by the Borrower or a Non-AEPLP Restricted Subsidiary to AEPLP or any of its Subsidiaries in any fiscal year shall not, together with any AEPLP Acquisitions and Capped Investments made in such fiscal year pursuant to Section 8.15 and Section 8.17(a), respectively, in the aggregate, exceed the PP&E Acquisition/Investment/Transfer Limit for such fiscal year; (B) the consideration paid by AEPLP or its Subsidiaries to the Borrower or any Non-AEPLP Restricted Subsidiary for such PP&E Assets is at least equal to the transferor's net book value therefor; (C) the aggregate net book value of all PP&E Assets of AEPLP and its Subsidiaries shall not at any time exceed the sum of (i) 33-1/3% of the aggregate net book value of all PP&E Assets of the Borrower and its Restricted Subsidiaries and (ii) \$70,000,000;

the limitations contained in Sections (iv) 8.15(b) and (c) and Sections 8.17(b)(iii)(A) and (C) shall not apply to or prohibit or otherwise restrict (A) any Investment in AEPLP or any of its Subsidiaries permitted by Section 8.17(a), (B) any lease of real or personal property from the Borrower or a Restricted Subsidiary (other than AEPLP and its Subsidiaries), as lessor, to AEPLP or a Subsidiary of AEPLP, as lessee, where the interest of the lessee in the leased assets is expressly subject to the Liens created by the Security Documents securing Indebtedness evidenced by the Notes, (C) any Transfer of assets by the Borrower or any Non-AEPLP Restricted Subsidiary to AEPLP or any of its Subsidiaries if (1) such assets consist of the proceeds, or assets purchased or subsequently funded with the proceeds, of a sale of equity interests or debt of the Public Partnership or the General Partner to an entity other than the Borrower or any Restricted Subsidiaries. (2) such Transfer is made within one year of such equity or debt sale and (3) in the case of a subsequent funding, such proceeds are used to repay Parity Debt of the Borrower (other than Indebtedness incurred previously pursuant to Section 8.1(e) or (to the extent such Indebtedness incurred pursuant to Section 8.1(f) is used to repay Indebtedness or letter of credit obligations incurred and outstanding under the Revolving Credit Facility) 8.1(f)) or Indebtedness incurred by the Borrower to make Acquisitions of assets that have been Transferred to AEPLP, or (D) any AEPLP Acquisition (1) if the assets acquired are purchased in exchange for equity interests or debt of the Public Partnership or the General Partner or (2)(x) if the assets acquired are purchased or subsequently funded with the proceeds of

a sale of equity interests or debt by the Public Partnership or the General Partner to an entity other than the Borrower or any Restricted Subsidiary, (y) such AEPLP Acquisition is made within one year of such equity or debt sale and (z) in the case of a subsequent funding, such proceeds are used to repay Parity Debt of the Borrower (other than Indebtedness incurred pursuant to Section 8.1(e) or (to the extent such Indebtedness incurred pursuant to Section 8.1(f) is used to repay Indebtedness or letter of credit obligations incurred and outstanding under the Revolving Credit Facility) 8.1(f)) or Indebtedness incurred by AEPLP (and owing to the Borrower) or the Borrower to make AEPLP Acquisitions.

- (c) AEPLP Partnership Agreement. The Borrower will not, and will cause its Subsidiaries to not, (i) permit the AEPLP Partnership Agreement, as in effect on the date hereof, to be amended, modified or supplemented in any respect if such amendment, modification or supplement would adversely affect the rights or powers of the Managing General Partner, or any successor General Partner (each as defined in the AEPLP Partnership Agreement), with respect to the liquidation, dissolution or winding-up of the affairs of AEPLP or any disposition of assets, discharge of liabilities or distribution of assets in connection therewith (including but not limited to any modification to Section 12.1 of the Partnership Agreement) or (ii) permit AEPLP to admit any Person as a Class A Limited Partner or any Managing General Partner (as defined in the AEPLP Partnership Agreement) unless all of the capital stock of such Person has been pledged to the Collateral Agent for the benefit of the Banks and the other holders of Parity Debt.
- (d) Trade Accounts Payable. The Borrower will not permit AEPLP and its Subsidiaries to create, incur, assume or otherwise become or remain directly or indirectly liable with respect to an aggregate amount of trade accounts payable (including but not limited to amounts owed under equipment leases) in excess of \$15,000,000 at any time, provided, that the amount of any (a) AEPLP Taxes, fines or penalties owing by AEPLP and its Subsidiaries to any Governmental Authority and (b) obligations of AEPLP and its Subsidiaries owing to the Borrower or any Restricted Subsidiary, shall in each case be excluded from the calculation of the aggregate amount of trade accounts payables pursuant to this Section 8.17(d).

In addition, both prior to and after the AEPLP Security Date, the Borrower will not, and will cause its Subsidiaries to not, permit the Intercompany Note to be amended, modified or supplemented in any respect if such amendment, modification or supplement would materially and adversely affect the rights of the holder of the Intercompany Note (in its capacity as a holder of the Intercompany Note), including, without limitation, any modification of the July 19, 2009, maturity date of the outstanding principal amount thereunder.

ARTICLE IX

EVENTS OF DEFAULT

- $9.1\,$ Event of Default. Any of the following shall constitute an "EVENT OF DEFAULT":
- (a) Non-Payment. The Borrower fails to pay the Agent or any Bank or the Issuing Bank, (i) when and as required to be paid herein, any amount of principal of any Loan or L/C Borrowing, or (ii) within 5 days after the same becomes due, any interest, fee, or any other amount payable to the Agent or the Banks hereunder or under any other Loan Document; or

- (b) Representation or Warranty. Any representation or warranty made in writing by any Obligor, or any Restricted Subsidiary made or deemed made herein, in any other Loan Document, or in any License Agreements, or which is contained in any certificate, financial statement or other document of such Obligor or such Restricted Subsidiary required to be delivered hereunder, furnished at any time under this Agreement, or in or under any other Loan Document, is incorrect in any material respect on or as of the date made or deemed made; or
- (c) Specific Defaults. There shall be a default in the performance of, or compliance with, any term contained in Section 7.1(g) or any of Sections 8.1 through 8.8, inclusive, 7.4(a)(i), 8.14 (and, in the case of the first sentence of Section 8.14, such default shall continue unremedied for 30 days) and 8.17, provided, however, that with respect to (i) incurrence of Indebtedness in violation of Section 8.1 in an aggregate outstanding principal amount which is less than \$5,000,000, (ii) incurrence of a Lien in violation of Section 8.3 which secures Indebtedness which is in an aggregate outstanding principal amount of less than \$5,000,000, (iii) transactions with an Affiliate in violation of Section 8.6 involving an aggregate amount of less than \$2,000,000, (iv) the making of any Investment or creation of a Contingent Obligation in violation of Section 8.4 involving an aggregate amount of less than \$2,000,000, or (v) the entering into of any transaction in violation of Section 8.7 involving an aggregate amount of less than \$2,000,000, there shall be no Event of Default hereunder unless the aggregate amount of all violations under clauses (i) through (v) exceeds \$8,000,000 on any date of determination or any such violation shall remain uncured for 30 days after a Responsible Officer becomes aware of any such violation; or
- (d) Other Defaults. Any Obligor, or any Restricted Subsidiary fails to perform or observe any other term or covenant contained in this Agreement, any other Loan Document, or in any License Agreements, and such default shall continue unremedied for a period of 30 days after the date upon which written notice thereof is given to the Obligors by the Agent or the Required Banks; provided, however, that defaults under any Mortgage shall not constitute an Event of Default under this subsection (d) unless such default shall not have been remedied within the applicable 30-day period and when aggregated with all other defaults described in this proviso (w) applies to at least 17 Mortgages, or (x) applies to Mortgages covering Mortgaged Property having an aggregate fair market value at the time of at least \$1,000,000, or (y) would cost in excess of \$1,000,000 to cure or would present a reasonable likelihood of resulting in liability to the Borrower or the Restricted Subsidiaries in excess of \$1,000,000 or (z) would result in a Material Adverse Effect; or
- (e) Cross-Default. The Borrower, any Restricted Subsidiary, the General Partner, any of its Subsidiaries or the Public Partnership or any of its Subsidiaries (other than the Partnership Unrestricted Subsidiaries) (as principal or guarantor or other surety) shall default in the payment of any amount of principal of or premium or interest on any Parity Debt or any other Indebtedness, other than the Obligations (regardless of whether or not such payment default shall have been waived by the holders of such Indebtedness); or any event shall occur or condition shall exist in respect of any Indebtedness of the Borrower, any Restricted Subsidiary, the General Partner, any of its Subsidiaries or the Public Partnership or any of its Subsidiaries (other than the Partnership Unrestricted Subsidiaries) or under any evidence of any such Indebtedness or under any mortgage, indenture or other agreement relating thereto, and the effect of such event or

condition is to cause (or to permit one or more Persons to cause) such Indebtedness to become due or be repurchased or repaid before its stated maturity or before its regularly scheduled dates of payment (other than pursuant to mandatory prepayment provisions pursuant to a (1) Change of Control or similar transaction or (2) prepayment under circumstances and on terms substantially identical to, and not inconsistent with, Section 9.3(b) of each of the First Mortgage Note Agreements to the extent it relates to Excess Taking Proceeds, as defined therein, or Section 8.8(c)(ii) hereof to the extent it relates to Excess Sale Proceeds, in each case not involving a default) or to permit the holders thereof to cause the Borrower, any Restricted Subsidiary, the General Partner, any of its Subsidiaries or the Public Partnership or any of its Subsidiaries (other than the Partnership Unrestricted Subsidiaries) to repurchase or repay such Indebtedness (other than pursuant to mandatory prepayment provisions pursuant to a (1) Change of Control or similar transaction or (2) prepayment under circumstances and on terms substantially identical to, and not inconsistent with, Section 9.3(b) of each of the First Mortgage Note Agreements to the extent it related to Excess Taking Proceeds, as defined therein, or Section 8.3(c)(ii) to the extent it relates to Excess Sale Proceeds, in each case not involving a default), and such default, event or condition shall continue for more than the period of grace, if any, specified therein (regardless of whether or not such default, event or condition shall have been waived by the holders of such Indebtedness); provided, that the aggregate principal amount of all Indebtedness as to which such a default (payment or other), event or condition shall occur or exist exceeds \$7,500,000; or

- (f) Insolvency Voluntary Proceedings. Any Obligors, or any Significant Subsidiary Group (i) ceases or fails to be solvent, or admits in writing its inability to pay its debts as they become due, subject to applicable grace periods, if any, whether at stated maturity or otherwise; (ii) voluntarily ceases to conduct its business in the ordinary course; (iii) commences any Insolvency Proceeding with respect to itself; or (iv) takes any action to effectuate or authorize any of the foregoing; or
- Insolvency Proceeding is commenced or filed against any Obligor, or any Significant Subsidiary Group, or any writ, judgment, warrant of attachment, execution or similar process, is issued or levied against a substantial part of any Obligor's, or any such Significant Subsidiary Group's properties, and any such proceeding or petition shall not be dismissed, or such writ, judgment, warrant of attachment, execution or similar process shall not be released, vacated or fully bonded within 60 days after commencement, filing or levy; (ii) any Obligor, or any such Significant Subsidiary Group admits the material allegations of a petition against it in any Insolvency Proceeding, or an order for relief (or similar order under non-U.S. law) is ordered in any Insolvency Proceeding; or (iii) any Obligor, or any such Significant Subsidiary Group acquiesces in the appointment of a receiver, trustee, custodian, conservator, liquidator, mortgagee in possession (or agent therefor), or other similar Person for itself or a substantial portion of its property or business; or
- (h) Judgments. Any judgment or order for the payment of money in excess of \$9,000,000 and not covered by insurance shall be rendered against any of the Obligors or any Significant Subsidiary Group, and either
 - (a) enforcement proceedings shall have been commenced by any creditor upon such judgment or order; or

- (b) there shall be any period of 60 consecutive days during which a stay of enforcement of such judgment or order, by reason of a pending appeal or otherwise, shall not be in effect and prior to the expiration of such 60-day period, the judgment shall not have been discharged.
- (i) Pension Plans. Any of the following events shall occur with respect to any Pension Plan and such events, either alone or together, present a reasonable likelihood of having a Material Adverse Effect:
 - (a) the institution of any steps by any Obligor, or any other Person to terminate a Pension Plan maintained or sponsored by an Obligor, or any Subsidiary of an Obligor; or
 - (b) an ERISA Event.
 - (j) Change of Control. There occurs any Change of

Control; or

- (k) Impairment of Security, etc. Any Loan Document or any Lien granted thereunder shall (except in accordance with its terms) in whole or in part, cease to be effective or cease to be the legally valid, binding and enforceable obligation of any Obligor party thereto, except to the extent that the assets of the Borrower or any of its Restricted Subsidiaries which are secured by the Liens which are terminated, no longer effective or no longer enforceable obligations of such Credit Party are not, individually or in the aggregate, material to the business of the Borrower and its Restricted Subsidiaries taken as a whole; the Borrower, General Partner, Petrolane or any other Credit Party shall, directly or indirectly, contest in any manner such effectiveness, validity, binding nature or enforceability or any Lien on property material to the business of the Borrower and its Restricted Subsidiaries taken as a whole securing any Obligations shall, in whole or in part, cease to be a perfected first priority Lien except as permitted by this Agreement.
- 9.2 Remedies. If any Event of Default occurs and is continuing, the Agent shall, at the request of, or may, with the consent of, the Required Banks, take any or all of the following actions:
- (a) declare the commitment of each Bank to make Loans (including the Swing Ling Bank to make Swing Line Loans) and any obligation of the Issuing Bank to Issue Letters of Credit to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Obligors;
- (c) require that the Borrower Cash Collateralize the L/C Obligations (in an amount equal to the then Effective Amount thereof); and
- (d) exercise on behalf of itself and the Banks all rights and remedies available to it and the Banks under the Loan Documents or applicable law;

provided, however, that upon the occurrence of any event specified in subsection (f) or (g) of Section 9.1 (in the case of clause (i) of subsection (g) upon the expiration of the 60-day period mentioned therein), the obligation of each Bank to make Loans and any obligation of the Issuing Bank to Issue Letters of Credit shall automatically terminate, the unpaid principal amount of all outstanding Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of the Borrower to Cash Collateralize the L/C Obligations as aforesaid shall automatically become effective, in each case without further act of the Agent or any Bank.

- 9.3 Rights Not Exclusive. The rights provided for in this Agreement and the other Loan Documents are cumulative and are not exclusive of any other rights, powers, privileges or remedies provided by law or in equity, or under any other instrument, document or agreement now existing or hereafter arising.
- 9.4 Application of Funds. Subject to the terms of the Collateral Agency Agreement, after the exercise of remedies provided for in Section 9.2 (or after the Loans have automatically become immediately due and payable and the L/C Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to Section 9.2), any amounts received on account of the Obligations shall be applied by the Agent in the following order:

First, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts (including Attorney Costs and amounts payable under Article IV) payable to the Agent in its capacity as such;

Second, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Banks (including Attorney Costs and amounts payable under Article IV), ratably among them in proportion to the amounts described in this clause Second payable to them;

Third, to payment of that portion of the Obligations constituting accrued and unpaid interest on the Loans and L/C Borrowings, ratably among the Banks in proportion to the respective amounts described in this clause Third payable to them;

Fourth, to payment of that portion of the Obligations constituting unpaid principal of the Loans and L/C Borrowings, ratably among the Banks in proportion to the respective amounts described in this clause Fourth held by them;

Fifth, to the Agent for the account of the Issuing Bank, to Cash Collateralize that portion of L/C Obligations comprised of the aggregate undrawn amount of Letters of Credit; and

Last, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by law.

Subject to Section 3.3(c), amounts used to Cash Collateralize the aggregate undrawn amount of Letters of Credit pursuant to clause Fifth above shall be applied in accordance herewith and with the other Loan Documents to satisfy drawings under such Letters of Credit as they occur. If any amount remains on deposit as Cash Collateral after all Letters of Credit have either been fully

drawn or expired, such remaining amount shall be applied in accordance herewith and with the other Loan Documents to the other Obligations, if any, in the order set forth above.

ARTICLE X

THE AGENT

- Appointment and Authorization. (a) Each Bank hereby irrevocably appoints, designates and authorizes the Agent to take such action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Loan Document, the Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall the Agent have or be deemed to have any fiduciary relationship with any Bank or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Loan Documents with reference to the Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.
- (b) The Issuing Bank shall act on behalf of the Banks with respect to any Letters of Credit issued by it and the documents associated therewith, and the Issuing Bank shall have all of the benefits and immunities (i) provided to the Agent in this Article X with respect to any acts taken or omissions suffered by the Issuing Bank in connection with Letters of Credit issued by it or proposed to be issued by it and the applications and agreements for letters of credit pertaining to such Letters of Credit as fully as if the term "Agent" as used in this Article X and in the definition of "Agent-Related Person" included the Issuing Bank with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the Issuing Bank.
- 10.2 Delegation of Duties. The Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys-in-fact and shall be entitled to advice of counsel and other consultants or experts concerning all matters pertaining to such duties. The Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of gross negligence or willful misconduct.
- 10.3 Liability of Agent. No Agent-Related Person shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct in connection with its duties expressly set forth herein), or (b) be responsible in any manner to any Bank or participant for any recital, statement, representation or warranty made by any Credit Party or any officer thereof, contained herein or in any other Loan Document, or in any certificate, report, statement or other document referred to or provided for in, or received by the Agent under or in connection with, this Agreement or any other Loan

Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document, or for any failure of any Credit Party or any other party to any Loan Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Bank or participant to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the properties, books or records of any Credit Party or any Affiliate thereof.

- Reliance by Agent. (a) The Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Credit Party), independent accountants and other experts selected by the Agent. The Agent shall be fully justified in failing or refusing to take any action under any Loan Document unless it shall first receive such advice or concurrence of the Required Banks as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Banks against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of the Required Banks (or such greater number of Banks as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Banks.
- (b) For purposes of determining compliance with the conditions specified in Section 5.1, each Bank that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Bank.
- knowledge or notice of the occurrence of any Default, except with respect to defaults in the payment of principal, interest and fees required to be paid to the Agent for the account of the Banks, unless the Agent shall have received written notice from a Bank or the Borrower referring to this Agreement, describing such Default and stating that such notice is a "notice of default." The Agent will notify the Banks of its receipt of any such notice. The Agent shall take such action with respect to such Default as may be directed by the Required Banks in accordance with Article IX; provided, however, that unless and until the Agent has received any such direction, the Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default as it shall deem advisable or in the best interest of the Banks.
- 10.6 Credit Decision. Each Bank acknowledges that no Agent-Related Person has made any representation or warranty to it, and that no act by the Agent hereafter taken, including any consent to and acceptance of any assignment or review of the affairs of any Credit Party or any Affiliate thereof, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Bank as to any matter, including whether Agent-Related Persons have disclosed material information in their possession. Each Bank represents to the Agent that it has, independently and without reliance upon any Agent-Related Person and based on such

documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Credit Parties and their respective Subsidiaries, and all applicable bank or other regulatory laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to the Borrower and the other Credit Parties hereunder. Each Bank also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Borrower and the other Credit Parties. Except for notices, reports and other documents expressly required to be furnished to the Banks by the Agent herein, the Agent shall not have any duty or responsibility to provide any Bank with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any of the Credit Parties or any of their respective Affiliates which may come into the possession of any Agent-Related Person.

- Indemnification. Whether or not the transactions contemplated hereby are consummated, the Banks shall indemnify upon demand each Agent-Related Person (to the extent not reimbursed by or on behalf of any Credit Party and without limiting the obligation of any Credit Party to do so), pro rata, and hold harmless each Agent-Related Person from and against any and all Indemnified Liabilities incurred by it; provided, however, that no Bank shall be liable for the payment to any Agent-Related Person of any portion of such Indemnified Liabilities to the extent determined in a final, nonappealable judgment by a court of competent jurisdiction to have resulted from such Agent-Related Person's own gross negligence or willful misconduct; provided, however, that no action taken in accordance with the directions of the Required Banks shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section. Without limitation of the foregoing, each Bank shall reimburse the Agent upon demand for its ratable share of any costs or out-of-pocket expenses (including Attorney Costs) incurred by the Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement, any other Loan Document, or any document contemplated by or referred to herein, to the extent that the Agent is not reimbursed for such expenses by or on behalf of the Borrower. The undertaking in this Section shall survive termination of the Commitments, the payment of all other Obligations and the resignation of the Agent.
- 10.8 Agent in Individual Capacity. Wachovia and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each of the Credit Parties and their respective Affiliates as though Wachovia were not the Agent or the Issuing Bank hereunder and without notice to or consent of the Banks. The Banks acknowledge that, pursuant to such activities, Wachovia or its Affiliates may receive information regarding any Credit Party or its Affiliates (including information that may be subject to confidentiality obligations in favor of such Credit Party or such Affiliate) and acknowledge that the Agent shall be under no obligation to provide such information to them. With respect to its

105

•

Loans, Wachovia shall have the same rights and powers under this Agreement as any other Bank and may exercise such rights and powers as though it were not the Agent or the Issuing Bank, and the terms "Bank" and "Banks" include Wachovia in its individual capacity.

- Successor Agent. The Agent may resign as Agent upon 30 days' 10.9 notice to the Banks; provided, that any such resignation by Wachovia shall also constitute its resignation as Issuing Bank and Swing Line Bank. If the Agent resigns under this Agreement, the Required Banks shall appoint from among the Banks a successor administrative agent for the Banks, which successor administrative agent shall be consented to by the Borrower at all times other than during the existence of an Event of Default (which consent of the Borrower shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of the Agent, the Agent may appoint, after consulting with the Banks and the Borrower, a successor administrative agent from among the Banks. Upon the acceptance of its appointment as successor administrative agent hereunder, the Person acting as such successor administrative agent shall succeed to all the rights, powers and duties of the retiring Agent, Issuing Bank and Swing Line Bank and the respective terms "Agent," "Issuing Bank" and "Swing Line Bank" shall mean such successor administrative agent, Letter of Credit issuer and swing line bank, and the retiring Agent's appointment, powers and duties as Agent shall be terminated and the retiring Issuing Bank's and Swing Line Bank's rights, powers and duties as such shall be terminated, without any other or further act or deed on the part of such retiring Issuing Bank, Swing Line Bank or any other Bank, other than the obligation of the successor Issuing Bank to issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or to make other arrangements satisfactory to the retiring Issuing Bank to effectively assume the obligations of the retiring Issuing Bank with respect to such Letters of Credit. After any retiring Agent's resignation hereunder as Agent, the provisions of this Article X and Sections 12.4 and 12.5 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement. If no successor administrative agent has accepted appointment as Agent by the date which is 30 days following a retiring Agent's notice of resignation, the retiring Agent's resignation shall nevertheless thereupon become effective and the Banks shall perform all of the duties of the Agent hereunder until such time, if any, as the Required Banks appoint a successor agent as provided for above.
- 10.10 Agent May File Proofs of Claim. In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to any Obligor, the Agent (irrespective of whether the principal of any Loan or L/C Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:
- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, L/C Obligations and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Banks and the Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Banks and the Agent and their respective agents and counsel and all other amounts due the Banks and the Agent under Sections 2.10, 3.8 and 11.4) allowed in such judicial proceeding; and

(b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Bank to make such payments to the Agent and, in the event that the Agent shall consent to the making of such payments directly to the Banks, to pay to the Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Agent and its agents and counsel, and any other amounts due the Agent under Sections 2.10 and 12.4.

Nothing contained herein shall be deemed to authorize the Agent to authorize or consent to or accept or adopt on behalf of any Bank any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Bank or to authorize the Agent to vote in respect of the claim of any Bank in any such proceeding.

- 10.11 Collateral and Guaranty Matters. The Banks irrevocably authorize the Agent, at its option and in its discretion,
- (a) to release any Lien on any property granted to or held by the Agent under any Loan Document (i) upon termination of the Commitments and payment in full of all Obligations (other than contingent indemnification obligations) and the expiration or termination of all Letters of Credit, (ii) that is sold or to be sold as part of or in connection with any sale permitted hereunder or under any other Loan Document, or (iii) subject to Section 12.1, if approved, authorized or ratified in writing by the Required Banks:
- (b) to subordinate any Lien on any property granted to or held by the Agent under any Loan Document to the holder of any Purchase Money Lien;
- (c) to release any Restricted Subsidiary from its obligations under the Subsidiary Guarantee if such Person ceases to be a Subsidiary as a result of a transaction permitted hereunder; and
- $\mbox{\ensuremath{\mbox{(d)}}}$ to enter into the Collateral Agency Agreement and agree to be bound by the terms thereof.

Upon request by the Agent at any time, each Bank will confirm in writing the Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Restricted Subsidiary from its obligations under the Subsidiary Guarantee pursuant to this Section 10.11.

10.12 Other Agents; Arrangers and Managers. None of the Banks or other Persons identified on the facing page or signature pages of this Agreement as a "syndication agent," "co-documentation agent," "co-agent," "book manager," "lead manager," "lead arranger" or "co-arranger" shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than, in the case of such Banks, those applicable to all Banks as such. Without limiting the foregoing, none of the Banks or other Persons so identified shall have or be deemed to have any fiduciary relationship with any Bank. Each Bank acknowledges that it has

not relied, and will not rely, on any of the Banks or other Persons so identified in deciding to enter into this Agreement or in taking or not taking action hereunder.

Withholding Tax. (a) (i) Each Bank that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code (a "FOREIGN BANK") shall deliver to the Agent, prior to receipt of any payment subject to withholding under the Code (or upon accepting an assignment of an interest herein), two duly signed completed copies of either IRS Form W-8BEN or any successor thereto (relating to such Foreign Bank and entitling it to an exemption from, or reduction of, withholding tax on all payments to be made to such Foreign Bank by the Borrower pursuant to this Agreement) or IRS Form W-8ECI or any successor thereto (relating to all payments to be made to such Foreign Bank by the Borrower pursuant to this Agreement) or such other evidence satisfactory to the Borrower and the Agent that such Foreign Bank is entitled to an exemption from, or reduction of, U.S. withholding tax, including any exemption pursuant to Section 881(c) of the Code. Thereafter and from time to time, each such Foreign Bank shall (A) promptly submit to the Agent such additional duly completed and signed copies of one of such forms (or such successor forms as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to the Borrower and the Agent of any available exemption from or reduction of, United States withholding taxes in respect of all payments to be made to such Foreign Bank by the Borrower pursuant to this Agreement, (B) promptly notify the Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (C) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Bank, and as may be reasonably necessary (including the re-designation of its Lending Office) to avoid any requirement of applicable laws that the Borrower make any deduction or withholding for taxes from amounts payable to such Foreign Bank.

(ii) Each Foreign Bank, to the extent it does not act or ceases to act for its own account with respect to any portion of any sums paid or payable to such Bank under any of the Loan Documents (for example, in the case of a typical participation by such Bank), shall deliver to the Agent on the date when such Foreign Bank ceases to act for its own account with respect to any portion of any such sums paid or payable, and at such other times as may be necessary in the determination of the Agent (in the reasonable exercise of its discretion), (A) two duly signed completed copies of the forms or statements required to be provided by such Bank as set forth above, to establish the portion of any such sums paid or payable with respect to which such Bank acts for its own account that is not subject to U.S. withholding tax, and (B) two duly signed completed copies of IRS Form W-8IMY (or any successor thereto), together with any information such Bank chooses to transmit with such form, and any other certificate or statement of exemption required under the Code, to establish that such Bank is not acting for its own account with respect to a portion of any such sums payable to such Bank.

(iii) The Borrower shall not be required to pay any additional amount to any Foreign Bank under Section 4.1 (A) with respect to any Taxes required to be deducted or withheld on the basis of the information, certificates or statements of exemption such Bank transmits with an IRS Form W-8BEN, W-8ECI or W-8IMY pursuant to this Section 10.13(a) or (B) if such Bank shall have failed to satisfy the foregoing provisions of this Section 10.13(a); provided that if such Bank shall have satisfied the requirement of this Section 10.13(a) on the

date such Bank became a Bank or ceased to act for its own account with respect to any payment under any of the Loan Documents, nothing in this Section 10.13(a) shall relieve the Borrower of its obligation to pay any amounts pursuant to Section 4.1 in the event that, as a result of any change in any applicable law, treaty or governmental rule, regulation or order, or any change in the interpretation, administration or application thereof, such Bank is no longer properly entitled to deliver forms, certificates or other evidence at a subsequent date establishing the fact that such Bank or other Person for the account of which such Bank receives any sums payable under any of the Loan Documents is not subject to withholding or is subject to withholding at a reduced rate.

- (iv) The Agent may, without reduction, withhold any Taxes required to be deducted and withheld from any payment under any of the Loan Documents with respect to which the Borrower is not required to pay additional amounts under Section 4.1.
- (b) Upon the request of the Agent, each Bank that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Agent two duly signed completed copies of IRS Form W-9. If such Bank fails to deliver such forms, then the Agent may withhold from any interest payment to such Bank an amount equivalent to the applicable back-up withholding tax imposed by the Code, without reduction.
- (c) If any Governmental Authority asserts that the Agent did not properly withhold or backup withhold, as the case may be, any tax or other amount from payments made to or for the account of any Bank, such Bank shall indemnify the Agent therefor, including all penalties and interest, any taxes imposed by any jurisdiction on the amounts payable to the Agent under this Section, and costs and expenses (including Attorney Costs) of the Agent. The obligation of the Banks under this Section shall survive the termination of the Commitments, repayment of all other Obligations hereunder and the resignation of the Agent.

ARTICLE XI

GUARANTEE

- 11.1 Each Guaranteed Obligation. (a) Each Guarantor, jointly and severally, irrevocably and unconditionally guarantees the Obligations; provided, however, that each Guarantor shall be liable under this Agreement for the maximum amount of such liability that can be hereby incurred without rendering this Agreement, as it relates to such Guarantor, voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount. Each Guarantor understands, agrees and confirms that the Agent may enforce this Article XI up to the full amount of the Obligations against each Guarantor, subject as aforesaid, without proceeding against the Borrower, against any security for the Obligations, or under any other Guaranty covering the Obligations.
- (b) This Article XI shall be subject to the terms of the Collateral Agency Agreement.
- 11.2 Obligations Exclusive. The liability of each Guarantor hereunder is exclusive and independent of any security for or other Guaranty Obligation of the Obligations whether executed by such Guarantor, any other Guarantor, any other guarantor or by any other party, and

the liability of each Guarantor hereunder shall not be affected or impaired by (a) any direction as to application of payment by the Borrower or by any other party, or (b) any other continuing or other Guaranty, undertaking or maximum liability of a guarantor or of any other party as to the Indebtedness of the Borrower, or (c) any payment on or in reduction of any such other Guaranty Obligation or undertaking except to the extent such payment is applied to the Obligations or such reduction results from application of a payment to the Obligations, or (d) any dissolution, termination or increase, decrease or change in personnel by the Borrower, or (e) any payment made to any Bank or the Agent on the amounts which the Banks or the Agent repay the Borrower pursuant to a court order in any bankruptcy, reorganization, arrangement, moratorium or other debtor relief proceeding, and each Guarantor waives any right to the deferral or modification of its obligations hereunder by reason of any such proceeding.

- 11.3 Obligations Independent. The obligations of each Guarantor hereunder are independent of the obligations of any other Guarantor, any other guarantor or the Borrower, and a separate action or actions may be brought and prosecuted against each Guarantor whether or not action is brought against any other Guarantor, any other guarantor or the Borrower and whether or not any other Guarantor, any other guarantor or the Borrower be joined in any such action or actions. Each Guarantor waives, to the fullest extent permitted by law, the benefit of any statute of limitations affecting its liability hereunder or the enforcement thereof. Any payment by the Borrower or other circumstance which operates to toll any statute of limitations as to the Borrower shall operate to toll the statute of limitations as to each Guarantor.
- 11.4 Waiver of Notice. Each Guarantor hereby waives notice of acceptance of this Agreement and notice of any liability to which it may apply, and waives promptness, diligence, presentment, demand of payment, protest, notice of dishonor or nonpayment of any such liabilities, suit or taking of other action by the Agent or any Bank against, and any other notice to, any party liable thereon (including such Guarantor or any other guarantor).
- 11.5 Guarantee of Payment. This Agreement is a guarantee of payment and not of collection. The Agent or any Bank may at any time and from time to time without the consent of, or notice to, any Guarantor, without incurring responsibility to such Guarantor, without impairing or releasing the obligations of such Guarantor hereunder, upon or without any terms or conditions and in whole or in part:
 - (i) change the manner, place or terms of payment of, and/or change or extend the time of payment of, renew or alter, any of the Obligations, any security therefor, or any liability incurred directly or indirectly in respect thereof, and the guarantee made in this Agreement shall apply to the Obligations as so changed, extended, renewed or altered;
 - (ii) sell, exchange, release, surrender, realize upon or otherwise deal with, in any manner and in any order, any property by whomsoever at any time pledged or mortgaged to secure, or howsoever securing, the Obligations or any liabilities (including any of those hereunder) incurred directly or indirectly in respect thereof or hereof, and/or any offset thereagainst;

110

- (iii) exercise or refrain from exercising any rights against the Borrower or any Guarantor or others or otherwise act or refrain from acting;
- (iv) settle or compromise any of the Obligations, any security therefor or any liability (including any of those hereunder) incurred directly or indirectly in respect thereof or hereof, and may subordinate the payment of all or any part thereof to the payment of any liability (whether due or not) of the Borrower to creditors of the Borrower;
- (v) apply any sums by whomsoever paid or howsoever realized to any liability or liabilities of the Borrower to the Banks regardless of what liabilities of the Borrower remain unpaid;
- (vi) consent to or waive any breach of, or any act, omission or default under, this Agreement, any of the First Mortgage Notes, the First Mortgage Note Agreements, the Bank Notes, the Parity Debt, the Parity Debt Agreements, the Security Documents, the other Financing Agreements (as such terms not defined herein are defined in the Collateral Agency Agreement) or any of the instruments or agreements referred to therein, or otherwise amend, modify or supplement this Agreement, the First Mortgage Notes, the Note Agreements, the Bank Notes, the Parity Debt, the Parity Debt Agreements, the Security Documents, the other Financing Agreements or any of such other instruments or agreements; and/or
- (vii) fail to perfect any Lien granted to the Agent or to or for the benefit of any of the Banks to secure any of the Obligations.
- 11.6 Obligations Unconditional. (a) The obligations of each Guarantor under this Agreement are absolute and unconditional and shall remain in full force and effect without regard to, and shall not be released, suspended, discharged, terminated (except in accordance with the terms hereof) or otherwise affected by, any circumstance or occurrence whatsoever, including without limitation: (i) any action or inaction by the Agent or the Banks as contemplated in Section 11.5; (ii) any invalidity, irregularity or unenforceability of all or part of the Obligations or of any security therefor; or (iii) to the extent permitted by applicable law, any other act or circumstance that might otherwise constitute a legal or equitable discharge or defense of a surety or a guarantor. The obligations of each Guarantor hereunder are primary obligations of each Guarantor.
- (b) The obligations of each Guarantor hereunder shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of the Borrower in respect of the Obligations is rescinded or must be otherwise returned by any holder of any of the Obligations, whether as a result of any proceedings in bankruptcy or reorganization or otherwise.
- 11.7 Continuing Guarantee. This Agreement is a continuing one and all liabilities to which it applies (or may apply) under the terms hereof shall be conclusively presumed to have been created in reliance hereon. No failure or delay on the part of the Agent or any Bank in exercising any right, power or privilege hereunder shall operate as a waiver thereof; nor shall any

single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights and remedies herein expressly specified are cumulative and not exclusive of any rights or remedies which the Agent or any Bank would otherwise have. No notice to or demand on any Guarantor in any case shall (i) entitle such Guarantor to any other further notice or demand in similar or other circumstances except for any notice or demand required hereunder or under any Subsidiary Security Agreement entered into by such Guarantor or (ii) constitute a waiver of the rights of the Agent or any Bank to any other or further action in any circumstances without notice or demand. It is not necessary for the Agent or any Bank to inquire into the capacity or powers of the officers, directors, partners or agents acting or purporting to act on behalf of any Guarantor or the Borrower, and any Obligations made or created in reliance upon the professed exercise of such powers shall be guaranteed hereunder.

- 11.8 Subordination. Any Indebtedness of the Borrower now or hereafter held by any Guarantor, whether arising by subrogation, contribution or otherwise, is hereby subordinated to the Obligations as provided for below; and such Indebtedness of the Borrower to any Guarantor, if the Agent, after an Event of Default has occurred and is continuing, so requests, shall be collected, enforced and received by such Guarantor as trustee for the Banks and be paid over to the Agent on account of the Obligations, but without affecting or impairing in any manner the liability of such Guarantor under the other provisions of this Agreement. Prior to the transfer to any non-Affiliate by any Guarantor of any note or negotiable instrument evidencing any Indebtedness of the Borrower to such Guarantor, such Guarantor shall mark such note or negotiable instrument with a legend, acceptable to the Agent, that the same is subject to this subordination.
- Exhaustion of Remedies. (a) Each Guarantor waives any right (except as shall be required by applicable statute and cannot be waived) to require the Agent or the Banks to (i) proceed against the Borrower, any other Guarantor or any other Person, (ii) proceed against or exhaust any security held from the Borrower, any other Guarantor or any other Person or (iii) pursue any other remedy in the Agent's or the Banks' power whatsoever. Each Guarantor waives any defense based on or arising out of any defense of the Borrower, any other Guarantor or any other Person other than payment in full of the Obligations, including without limitation any defense based on or arising out of the disability of the Borrower, any other Guarantor or any other party, or the unenforceability of the Obligations or any part thereof from any cause, or the cessation from any cause of the liability of the Borrower other than payment in full of the Obligations. The Agent or the Collateral Agent, as the case may be, on behalf of the Banks may, at its election, foreclose on any security held by the Agent, the Collateral Agent or the Banks by one or more judicial or nonjudicial sales, whether or not every aspect of any such sale is commercially reasonable (to the extent such sale is permitted by applicable law), or exercise any other right or remedy the Agent, the Collateral Agent or the Banks may have against the Borrower or any other Person, or any security, without affecting or impairing in any way the liability of any Guarantor hereunder except to the extent the Obligations have been paid. Each Guarantor waives any defense arising out of any such election by the Agent, the Collateral Agent or the Banks, even though such election operates to impair or extinguish any right of reimbursement or subrogation or other right or remedy of such Guarantor against the Borrower or any other Person or any security.

112

- (b) Each Guarantor waives all presentments, demands for performance, protests and notices, including without limitation notices of nonperformance, notice of protest, notices of dishonor, notices of acceptance of this Agreement, and notices of the existence, creation or incurring of new or additional Indebtedness. Each Guarantor assumes all responsibility for being and keeping itself informed of the Borrower's financial condition and assets, and of all other circumstances bearing upon the risk of nonpayment of the Obligations and the nature, scope and extent of the risks which such Guarantor assumes and incurs hereunder, and agrees that the Agent and the Banks shall have no duty to advise any Guarantor of information known to them regarding such circumstances or risks.
- Each Guarantor understands, is aware and hereby acknowledges that to the extent the Obligations are secured by real property located in the State of California, such Guarantor shall be liable for the full amount of its liability hereunder notwithstanding foreclosure on such real property by trustee sale or any other reason impairing each Guarantor's or the Agent's or any Bank's right to proceed against the Borrower. Each Guarantor hereby waives, to the fullest extent permitted by law, all rights and benefits under Section 2809 of the California Civil Code purporting to reduce a guarantor's obligation in proportion to the principal obligation. Each Guarantor hereby waives all rights and benefits under Section 580a of the California Code of Civil Procedure purporting to limit the amount of any deficiency judgment which might be recoverable following the occurrence of a trustee's sale under a deed of trust and all rights and benefits under Section 580b of the California Code of Civil Procedure stating that no deficiency judgment may be recovered on a real property purchase money obligation. Each Guarantor further understands, is aware and hereby acknowledges that if the Agent (or the Collateral Agent) on behalf of the Banks elects to nonjudicially foreclose on any real property security located in the State of California, any right of subrogation of the Guarantors against the Agent, the Collateral Agent or the Banks may be impaired or extinguished and that as a result of such impairment or extinguishment of subrogation rights, each Guarantor will have a defense to a deficiency judgment arising out of the operation of (i) Section 580d of the California Code of Civil Procedure which states that no deficiency judgment may be recovered on a note secured by a deed of trust on real property in case such real property is sold under the power of sale contained in such deed of trust, and (ii) related principles of estoppel. To the fullest extent permitted by law, each Guarantor hereby waives all rights and benefits and any defense arising out of the operation of Section 580d of the California Code of Civil Procedure and related principles of estoppel, even though such election operates to impair or extinguish any right of reimbursement or subrogation or other right or remedy of such Guarantor against the Borrower or any other party or any security. In addition, each Guarantor hereby waives, to the fullest extent permitted by applicable law and without limiting the generality of the foregoing or any other provision hereof, all rights and benefits which might otherwise be available to such Guarantor under Section 726 of the California Code of Civil Procedure and all rights and benefits which might otherwise be available to such Guarantor under California Civil Code Sections 2809, 2810, 2815, 2819, 2821, 2839, 2845, 2848, 2849, 2850, 2899, 3275 and 3433 (and any analogous or successor provisions to such Sections). Furthermore, each Guarantor hereby waives, to the fullest extent permitted by law, the benefits of the provisions of Nevada Revised Statutes Sections 40.430 et seq., 40.451 et seq., and 40.465 et seq. (and any analogous or successor provisions to such Sections).

- (d) Each Guarantor agrees that, as between such Guarantor and the Agent and Banks, the Obligations may be declared to be forthwith due and payable (and shall be deemed to have become automatically due and payable) in accordance with the terms thereof for purposes of Section 11.1 hereof notwithstanding any stay, injunction or other prohibition preventing such declaration (or preventing such Obligations from becoming automatically due and payable) as against the Borrower and that, in the event of such declaration (or such Obligations being deemed to have become automatically due and payable) such Obligations (whether or not due and payable by the Borrower) shall forthwith become due and payable by each Guarantor for purposes of Section 11.1.
- 11.10 Reinstatement. If claim is ever made upon the Agent or any Bank for repayment or recovery of any amount or amounts received in payment or on account of any of the Obligations and any of the aforesaid payees repays all or part of said amount by reason of (a) any judgment, decree or order of any court or administrative body having jurisdiction over such payee or any of its property or (b) any settlement or compromise of any such claim effected by such payee with any such claimant (including the Borrower), then and in such event each Guarantor agrees that any such judgment, decree, order, settlement or compromise shall be binding upon such Guarantor, notwithstanding any revocation hereof or other instrument evidencing any liability of the Borrower, and such Guarantor shall be and remain liable to the aforesaid payees hereunder for the amount so repaid or recovered to the same extent as if such amount had never originally been received by any such payee.

ARTICLE XII

MISCELLANEOUS

- 12.1 Amendments and Waivers. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by any Obligor or any other Credit Party therefrom, shall be effective unless in writing signed by the Required Banks and the Borrower, and acknowledged by the Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall, unless in writing and signed by all the Banks and the Borrower do any of the following:
- (a) waive any condition set forth in subsection (a), (b), (c)(i), (d), (e), (f), (h), (i), (j), (k), (l), (m), (n) or (o) of Section 5.1 without the written consent of each Bank;
- (b) increase or extend the Commitment of any Bank (or reinstate any Commitment terminated pursuant to Section 9.2);
- (c) postpone or delay any date fixed by this Agreement or any other Loan Document for any payment of principal, interest, fees or other amounts due to the Banks (or any of them) hereunder or under any other Loan Document
- (d) reduce the principal of, or the rate of interest specified herein on any Loan, or (subject to clause (iv) below) any fees or other amounts payable hereunder or under any other Loan Document;

- (e) change the definition of "Required Banks" or change the percentage of the Commitments or of the aggregate unpaid principal amount of the Loans which is required for the Banks or any of them to take any action hereunder; or
- (f) amend this Section, or Section 2.14, or any provision herein providing for consent or other action by all Banks; or
- $\mbox{\ensuremath{(g)}}$ release all or substantially all the collateral securing the Obligations;

and, provided, further, that (i) no amendment, waiver or consent shall, unless in writing and signed by the Agent in addition to the Required Banks or all the Banks, as the case may be, affect the rights or duties of the Agent under this Agreement or any other Loan Document, (ii) no amendment, waiver or consent shall, unless in writing and signed by the Issuing Bank in addition to the Required Banks or all the Banks, as the case may be, affect the rights or duties of the Issuing Bank under this Agreement or any L/C-Related Documents relating to any Letter of Credit Issued or to be Issued by it, (iii) no amendment, waiver or consent shall, unless in writing and signed by the Swing Line Bank in addition to the Banks required above, affect the rights or duties of the Swing Line Bank under this Agreement and (iv) the Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed by the parties thereto.

In connection with a proposed merger, consolidation or sale of all or substantially all of the assets of the Borrower in accordance with Section 8.8(a)(iii) or (b)(iii) to a corporation, the parties agree (i) to effect, simultaneously with such transaction, all necessary and appropriate modifications to the terms and conditions of this Agreement and the other Loan Documents and License Agreements to which it is a party (including without limitation the ability of the Borrower to make payments under Section 8.5, taking into account the effect of any change in the tax status of the Borrower on its financial condition and the applicable financial covenants) to reflect the corporate existence of such successor corporation and any other matters in form acceptable to the Required Banks, provided, that such modified terms and conditions convey to the parties substantially the same rights and obligations provided under the Loan Documents and License Agreements to which it is a party immediately prior to such transaction, and (ii) that any Default described in Section 9.1(j) which would result from such transaction shall not be asserted by the Agent or any Bank if after giving effect to such transaction UGI shall own directly or indirectly at least 51% of the voting shares of the corporation that is the successor to the Borrower.

In the event a Bank or Participant (as hereinafter defined) shall refuse to enter into or consent to any amendment, waiver or other modification of any provision of this Agreement or any other Loan Document, and such Bank's or Participant's consent is necessary for such amendment, waiver or modification to become effective, the Borrower may pay Obligations (including, with respect to Letter of Credit, cash collateralization of an interest therein) outstanding to any such nonconsenting Bank or to any Originating Bank having participated interests to any such nonconsenting Participant and reduce or eliminate any such Bank's Commitment; provided, that the Borrower may take such action only if Banks representing at least 80% of the outstanding Commitments necessary therefor have entered into or consented to such amendment, waiver or modification and no Default or Event of Default then exists.

- Notices and Other Communications; Facsimile Copies. (a) 12.2 General. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission). All such written notices shall be mailed, faxed or delivered to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, if to the Borrower, the Agent, the Issuing Bank, the Swing Line Bank or any Bank, to the address, facsimile number, electronic mail address or telephone number specified for such Person on Schedule 12.2 or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties. All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered; provided, however, that notices and other communications to the Agent, the Issuing Bank and the Swing Line Bank pursuant to Articles II and III shall not be effective until actually received by such Person. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.
- (b) Effectiveness of Facsimile Documents and Signatures. Loan Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable law, have the same force and effect as manually-signed originals and shall be binding on all Credit Parties, the Agent and the Banks. The Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; provided, however, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.
- (c) Limited Use of Electronic Mail. Electronic mail and Internet and intranet websites may be used only to distribute routine communications, such as financial statements and other information as provided in Section 7.1, and to distribute Loan Documents for execution by the parties thereto, and may not be used for any other purpose.
- (d) Reliance by Agent and Banks. The Agent and the Banks shall be entitled to rely and act upon any notices (including telephonic Notices of Borrowing) purportedly given by or on behalf of the Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify each Agent-Related Person and each Bank from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower. All telephonic notices to and other communications with the Agent may be recorded by the Agent, and each of the parties hereto hereby consents to such recording.
- $12.3\,$ No Waiver; Cumulative Remedies. No failure to exercise and no delay in exercising, on the part of the Agent or any Bank, any right, remedy, power or privilege

hereunder, shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

12.4 Costs and Expenses. The Borrower shall:

- (a) whether or not the transactions contemplated hereby are consummated, pay or reimburse Wachovia (including in its capacity as Agent) within five Business Days after demand and receipt by the Borrower of reasonable supporting documentation (subject to Section 5.1(d)) for all reasonable costs and expenses incurred by Wachovia (including in its capacity as Agent) in connection with the development, preparation, delivery, administration and execution of this Agreement, any Loan Document and any other documents prepared in connection herewith or therewith, and the consummation of the transactions contemplated hereby and thereby, including Attorney Costs (excluding allocated costs of internal legal counsel) incurred by Wachovia (including in its capacity as Agent) with respect thereto; and
- (b) pay or reimburse the Agent within five Business Days after demand and receipt by the Borrower of reasonable supporting documentation for all reasonable costs and expenses incurred by the Agent in connection with any amendment, supplement, waiver or modification to (in each case, whether or not consummated), this Agreement, any Loan Document and any other documents prepared in connection herewith or therewith, and the consummation of the transactions contemplated hereby and thereby, including Attorney Costs incurred by the Agent with respect thereto (provided, that the fees of any law firm or other external counsel, and the allocated costs of internal legal services, shall not both be reimbursed with respect to any amendment, supplement, waiver or modification relating to the same or any substantially similar matter); and
- (c) pay or reimburse the Agent, the Arranger and each Bank within five Business Days after demand and receipt by the Borrower of reasonable supporting documentation (subject to Section 5.1(g)) for all reasonable costs and expenses (including Attorney Costs) incurred by them in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or any other Loan Document during the existence of an Event of Default or after acceleration of the Loans (including in connection with any "workout" or restructuring regarding the Loans, and including in any Insolvency Proceeding or appellate proceeding).

The foregoing costs and expenses shall include all search, filing, recording, title insurance and fees and taxes related thereto incurred by the Agent, and, with respect to those costs and expenses referred to in Section 12.4(b) or 12.4(c) above, the reasonable cost of independent public accountants, appraisers and other outside experts retained by the Agent. The agreements in this Section shall survive the termination of the Commitments and repayment of all other Obligations.

12.5 Indemnity. Whether or not the transactions contemplated hereby are consummated, the Obligors shall indemnify and hold harmless each Agent-Related Person, the Arranger, the Issuing Bank, the Collateral Agent, each Bank and their respective affiliates, directors, officers, employees and agents (collectively, the "INDEMNIFIED PARTIES") from and

against any and all losses, claims, damages (other than consequential or exemplary damages), liabilities and reasonable out-of-pocket expenses (including, without limitation, reasonable fees and disbursements of counsel, amounts paid in settlement and court costs) (collectively, the "INDEMNIFIED LIABILITIES") which may be incurred by any such Indemnified Party as a result of a claim by a third party or asserted by a third party against any such Indemnified Party, in each case, in connection with or arising out of or in any way relating to or resulting from any transaction or proposed transaction (whether or not consummated) contemplated to be financed with the proceeds of any Loan or other financial accommodation contemplated hereby, and the Obligors hereby agree to reimburse each such Indemnified Party for any Attorneys' Costs or other out-of-pocket expenses incurred in connection with investigating, defending or participating in any action or proceeding (whether or not such Indemnified Party is a party to such action or proceeding) out of which any such losses, claims, damages, liabilities or expenses may arise; provided, however, that the Obligors shall not be required to reimburse the expenses of more than one counsel for all Indemnified Parties except to the extent that different Indemnified Parties shall have conflicting interests. Notwithstanding anything herein to the contrary, the Obligors shall not be liable or responsible for losses, claims, damages, costs and expenses incurred by any Indemnified Party arising out of or relating to such Indemnified Party's own gross negligence or willful misconduct. If for any reason the indemnification provided for herein is unavailable to any Indemnified Party or insufficient to hold it harmless as and to the extent contemplated hereby, then the Obligors hereby agree to contribute to the amount paid or payable by such Indemnified Party as a result of such loss, claim, damage, liability or expense in such proportion as is appropriate to reflect the relative benefits received by the Obligors, on the one hand, and such Indemnified Party, on the other hand, and also the respective fault of the Obligors, on the one hand, and such Indemnified Party, on the other hand, as the case may be, as well as any other relevant equitable considerations. This Section 12.5 shall survive the termination of this Agreement.

- 12.6 Liability. (a) The liability of the Obligors hereunder and under the Loan Documents shall be absolute, unconditional and irrevocable irrespective of:
- (i) any lack of validity, legality or enforceability of this Agreement, any Note or any other Loan Document;
 - (ii) the failure of any Bank
 - (A) to enforce any right or remedy against any other Person (including any guarantor) under the provisions of this Agreement, the Note, any other Loan Document or otherwise, or
 - (B) to exercise any right or remedy against any guarantor of, or collateral securing, any Obligations;
- (iii) any change in the time, manner or place of payment of, or in any other term of, all or any of the Obligations, or any other extension, compromise or renewal of any Obligations;

(iv) any reduction, limitation, impairment or termination of any Obligations with respect to any other Credit Party for any reason including any claim of waiver, release, surrender, alteration or compromise, and shall not be subject to (and the Borrower hereby waives any right to or claim of) any defense or setoff, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality, nongenuineness, irregularity, compromise, unenforceability of, or any other event or occurrence affecting, any Obligations with respect to any other Credit Party;

(v) any addition, exchange, release, surrender or nonperfection of any collateral, or any amendment to or waiver or release or addition of, or consent to departure from, any guaranty, held by any Bank securing any of the Obligations; or

(vi) any other circumstance which might otherwise constitute a defense available to, or a legal or equitable discharge of, any other Credit Party, any surety or any guarantor.

The Borrower agrees that its liability hereunder shall continue to be effective or be reinstated, as the case may be, if at any time any payment (in whole or in part) of any of the Obligations is rescinded or must be restored by any Bank, upon the insolvency, bankruptcy or reorganization of the Borrower as though such payment had not been made.

The Obligors hereby expressly waive: (a) notice of the Banks' acceptance of this Agreement; (b) notice of the existence or creation or non-payment of all or any of the Obligations; (c) presentment, demand, notice of dishonor, protest, and all other notices whatsoever other than notices expressly provided for in this Agreement and (d) all diligence in collection or protection of or realization upon the Obligations or any thereof any obligation hereunder, or any security for or guaranty of any of the foregoing.

No delay on any of the Banks' part in the exercise of any right or remedy shall operate as a waiver thereof, and no single or partial exercise by any of the Banks of any right or remedy, shall preclude other or further exercise thereof or the exercise of any other right or remedy. No action of any of the Banks permitted hereunder shall in any way affect or impair any such Banks' rights or Obligors' obligations under this Agreement.

Each Obligor hereby represents and warrants to each of the Banks that it now has and will continue to have independent means of obtaining information concerning the other Obligors' affairs, financial condition and business. The Banks shall not have any duty or responsibility to provide any Obligor with any credit or other information concerning the Obligors' Subsidiaries' affairs, financial condition or business which may come into the Banks' possession. Each of the Obligors agrees that any action or notice which is required or authorized to be taken or given or received under this Agreement or any of the Loan Documents shall be taken, given or received by the Borrower acting on behalf of the other Credit Parties (and not by Petrolane or the General Partner), and the other Credit Parties agree to be bound by, and authorizes the Agent and each Bank to rely upon, any such action or notice as if fully authorized by each of the Obligors.

12.7 Payments Set Aside. To the extent that the Borrower makes a payment to the Agent or the Banks, or the Agent or the Banks exercise their right of set-off, and such payment

or the proceeds of such set-off or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Agent or such Bank in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any Insolvency Proceeding or otherwise, then (a) to the extent of such recovery the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, and (b) each Bank severally agrees to pay to the Agent upon demand its pro rata share of any amount so recovered from or repaid by the Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect.

- 12.8 Successors and Assigns. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns, except that no Obligors may assign or transfer any of its rights or obligations under this Agreement without the prior written consent of the Agent and each Bank. Any attempted assignment in violation of this provision shall be null and void.
- Assignments, Participations. etc. (a) Any Bank may, with the written consent of the Borrower, the Agent and the Issuing Bank, which consent of the Borrower shall not be unreasonably withheld, at any time assign and delegate to one or more Eligible Assignees (provided, that no written consent of the Borrower, the Agent or the Issuing Bank shall be required in connection with any assignment and delegation by a Bank to (x) an Eligible Assignee that is an Affiliate of such Bank or (y) another Bank (each an "ASSIGNEE")) all, or any ratable part of all, of the Loans, the Commitments and the other rights and obligations of such Bank hereunder, in a minimum amount of \$5,000,000; provided, however, that the Borrower, the Agent and the Issuing Bank may continue to deal solely and directly with such Bank in connection with the interest so assigned to an Assignee until (i) written notice of such assignment, together with payment instructions, addresses and related information with respect to the Assignee, shall have been given to the Borrower, the Agent and the Issuing Bank by such Bank and the Assignee; (ii) such Bank and its Assignee shall have delivered to the Borrower, the Agent and the Issuing Bank an Assignment and Acceptance in the form of Exhibit F (an "ASSIGNMENT AND ACCEPTANCE") and (iii) the assignor Bank or Assignee has paid to the Agent a processing fee in the amount of \$4,000; and provided, further, each Bank's Pro Rata Share shall be the same in each type of Commitment.
- (b) From and after the date that the Agent notifies the assignor Bank that it has received (and the Borrower, the Agent and the Issuing Bank have provided their consent with respect to) an executed Assignment and Acceptance and payment of the above-referenced processing fee, (i) the Assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, shall have the rights and obligations of a Bank under the Loan Documents, and (ii) the assignor Bank shall, to the extent that rights and obligations hereunder and under the other Loan Documents have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights and be released from its obligations under the Loan Documents (and, in the case of an Assignment and Acceptance covering all of the assigning Bank's rights and obligations under this Agreement, such Bank shall cease to be a party hereto but shall continue to be entitled to the

120

benefits of Sections 4.1, 4.3, 4.4, 12.4 and 12.5 with respect to facts and circumstances occurring prior to the effective date of such assignment).

- (c) Within five Business Days after its receipt of notice by the Agent that it has received an executed Assignment and Acceptance and payment of the processing fee (and provided, that the Borrower consents to such assignment in accordance with Section 12.9(a)), the Borrower shall, if requested by the Assignee or the assignor Bank thereunder, execute and deliver to the Agent new Notes evidencing such Assignee's assigned Loans and Commitments and, if the assignor Bank has retained a portion of its Loans and its Commitment, replacement Notes in the principal amount of the Loans and Commitments retained by the assignor Bank (such Notes to be in exchange for, but not in payment of, the Notes held by such Bank) and the assignor Bank shall deliver its Note or Notes marked "exchanged" or "cancelled," as applicable, to the Agent.

 Immediately upon payment of the processing fee payment under the Assignment and Acceptance and the satisfaction of the other conditions set forth in Section 12.9(a), this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to reflect the addition of the Assignee and the resulting adjustment of the Commitments arising therefrom. The Commitment allocated to each Assignee shall reduce such Commitments of the assigning Bank pro tanto.
- (d) The Agent shall maintain at its address referred to in Schedule 12.2 a copy of each Assignment and Acceptance delivered to and accepted by it and a register for the recordation of the names and addresses of the Banks and the Commitment of, and principal amount of the Loans owing to, each Bank from time to time (the "REGISTER"). The entries in the Register shall be conclusive and binding for all purposes, absent manifest error, and the Borrower, the Agent and the Banks may treat each Person whose name is recorded in the Register as a Bank hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower or any Bank at any reasonable time and from time to time upon reasonable prior notice. Any assignment of any Loan or other obligations shall be effective only upon an entry with respect thereto being made in the Register.
- Any Bank may at any time sell to one or more (e) commercial banks or other Persons not Affiliates of the Borrower (a "PARTICIPANT") participating interests in any Loans, the Commitment of that Bank and the other interests of that Bank (the "ORIGINATING BANK") hereunder and under the other Loan Documents; provided, however, that (i) the Originating Bank's obligations under this Agreement shall remain unchanged, (ii) the Originating Bank shall remain solely responsible for the performance of such obligations, (iii) the Borrower, the Agent, the Issuing Bank and the other Banks shall continue to deal solely and directly with the Originating Bank in connection with the Originating Bank's rights and obligations under this Agreement and the other Loan Documents. Any agreement or instrument pursuant to which a Bank sells such a participation shall provide that such Bank shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Bank will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 12.1 that directly affects such Participant. In the case of any such participation, the Participant shall be entitled to the benefit of Sections 4.1, 4.3, 4.4 and 12.5 as though it were also a Bank hereunder (but not in any greater amounts than would have been payable to the Bank selling the participation if no participation were sold), and not have any

rights under this Agreement, or any of the other Loan Documents, and all amounts payable by the Borrower hereunder shall be determined as if such Bank had not sold such participation; except that, if amounts outstanding under this Agreement are due and unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Bank under this Agreement, provided such Participant agrees to be subject to Section 2.14 as though it were a Bank.

- (f) Nothing contained in this Agreement shall prevent a Bank from pledging its interest in its Loans to a Federal Reserve Bank in the Federal Reserve System of the United States in accordance with applicable law.
- (g) After payment in full of, and satisfaction of all Obligations under, any Note, the Bank or other party holding such Note agrees to promptly return such Note marked "Paid in Full" to the Borrower.
- (h) Notwithstanding the foregoing provisions of this Section 12.9, no assignment or participation may be made if such assignment or participation involves, or could involve, the use of assets that constitute, or may be deemed under ERISA, the Code or any other applicable law, or any ruling or regulation issued thereunder, or any court decision, to constitute the assets of any employee benefit plan (as defined in section 3(3) of ERISA) or any plan as defined in section 4975(e)(1) of the Code).
- Confidentiality. (a) Each of the Agent and the Banks agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (i) to its and its Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (ii) to the extent requested by any regulatory authority; (iii) to the extent required by applicable laws or regulations or by any subpoena or similar legal process; (iv) to any other party to this Agreement; (v) in connection with the exercise of any remedies hereunder or any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder; (vi) subject to an agreement containing provisions substantially the same as those of this Section, to (x) any Eligible Assignee of or Participant in, or any prospective Eligible Assignee of or Participant in, any of its rights or obligations under this Agreement or (y) any direct or indirect contractual counterparty or prospective counterparty (or such contractual counterparty's or prospective counterparty's professional advisor) to any credit derivative transaction relating to obligations of the Credit Parties; (vii) with the written consent of the Borrower; (viii) to the extent such Information (ix) becomes publicly available other than as a result of a breach of this Section or (x) becomes available to the Agent or any Bank on a nonconfidential basis from a source other than a Credit Party; or (xi) to the National Association of Insurance Commissioners or any other similar organization.(b) The Agent and the Banks may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry, and service providers to the Agent and the Banks in connection with the administration and management of this Agreement, the other Loan Documents, the Commitments, and the Credit Extensions; provided, however, that

information disclosed by the Agent or any Bank to any such market data collectors or similar service providers shall be of a type generally provided to such Persons in other transactions. For the purposes of this Section 12.10, "INFORMATION" means all information received from any Credit Party relating to any Credit Party or its business. Notwithstanding anything herein to the contrary, any party subject to confidentiality obligations under this Agreement or under any other related document (and any employee, representative or other agent of such party) may disclose to any and all persons, without limitation of any kind, the U.S. federal income tax treatment of the U.S. federal income tax structure of the transactions contemplated herein and all materials of any kind (including opinions or other tax analyses) that are provided to it relating to such tax treatment and tax structure.

- (c) The Borrower acknowledges that one or more of the Banks may treat its Loans as part of a transaction that is subject to Treasury Regulation Section 1.6011-4 or Section 301.6112-1, and the Agent and such Bank or Banks, as applicable, may (i) maintain such lists or other records as they may determine are required by such Treasury Regulations and (ii) file such IRS forms as they may determine are required by such Treasury Regulations with written notice by the party making such filing to the Borrower.
- (d) Any Person required to maintain the confidentiality of Information as provided in this Section 12.10 shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information. Each of the Agent, the Banks and the Participants shall promptly notify the Borrower of its receipt of any subpoena or similar process or authority, unless prohibited therefrom by the issuing Person.
- 12.11 Set-off. In addition to any rights and remedies of the Banks provided by law, upon the occurrence and during the continuance of any Event of Default, each Bank is authorized at any time and from time to time, without prior notice to the Borrower or any other Credit Party, any such notice being waived by the Borrower (on their own behalf and on behalf of each Credit Party) to the fullest extent permitted by law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held by, and other indebtedness at any time owing by, such Bank to or for the credit or the account of the respective Credit Party against any and all Obligations owing to such Bank hereunder or under any other Loan Document, now or hereafter existing, irrespective of whether or not the Agent or such Bank shall have made demand under this Agreement or any other Loan Document and although such Obligations may be contingent or unmatured or denominated in a currency different from that of the applicable deposit or indebtedness. Each Bank agrees promptly to notify the Borrower and the Agent after any such set-off and application made by such Bank; provided, however, that the failure to give such notice shall not affect the validity of such set-off and application.
- 12.12 Notification of Addresses; etc. Each Bank shall notify the Agent in writing of any changes in the address to which notices to the Bank should be directed, of addresses of any Lending Office, of payment instructions in respect of all payments to be made to it hereunder and of such other administrative information as the Agent shall reasonably request.

- 12.13 Counterparts. This Agreement may be executed in any number of separate counterparts, each of which, when so executed, shall be deemed an original, and all of said counterparts taken together shall be deemed to constitute but one and the same instrument.
- 12.14 Severability. The illegality or unenforceability of any provision of this Agreement or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Agreement or any instrument or agreement required hereunder.
- 12.15 No Third Parties Benefited. This Agreement is made and entered into for the sole protection and legal benefit of the Obligors, the Banks, the Agent and the Agent-Related Persons, and their permitted successors and assigns, and no other Person shall be a direct or indirect legal beneficiary of, or have any direct or indirect cause of action or claim in connection with, this Agreement or any of the other Loan Documents.
- 12.16 Governing Law and Jurisdiction. (a) THIS AGREEMENT AND THE NOTES SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK; PROVIDED, THAT THE AGENT AND THE BANKS SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW. THE LETTERS OF CREDIT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE ISP98. AS TO MATTERS NOT COVERED BY THE ISP98, THE LETTER OF CREDIT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF NEW YORK, INCLUDING, TO THE EXTENT NOT INCONSISTENT WITH THE ISP98, THE UNIFORM COMMERCIAL CODE AS IN EFFECT IN THE STATE OF NEW YORK.
- (b) ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK OR OF THE UNITED STATES FOR THE SOUTHERN DISTRICT OF NEW YORK, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, EACH OF THE OBLIGORS, THE AGENT AND THE BANKS EACH CONSENT, TO THE EXTENT PERMITTED BY APPLICABLE LAW, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. EACH OF THE OBLIGORS, THE AGENT AND THE BANKS IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF THIS AGREEMENT OR ANY DOCUMENT RELATED HERETO. THE OBLIGORS, THE AGENT AND THE BANKS EACH WAIVE PERSONAL SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER PROCESS, WHICH MAY BE MADE BY ANY OTHER MEANS PERMITTED BY NEW YORK LAW.
- 12.17 Waiver of Jury Trial. EACH OF THE OBLIGORS, THE BANKS AND THE AGENT WAIVES ITS RIGHTS TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF OR RELATED TO THIS AGREEMENT, THE OTHER LOAN DOCUMENTS, OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY, IN ANY ACTION, PROCEEDING OR OTHER LITIGATION OF

ANY TYPE BROUGHT BY ANY OF THE PARTIES AGAINST ANY OTHER PARTY OR ANY AGENT-RELATED PERSON, PARTICIPANT OR ASSIGNEE, WHETHER WITH RESPECT TO CONTRACT CLAIMS, TORT CLAIMS, OR OTHERWISE. EACH OF THE OBLIGORS, THE BANKS AND THE AGENT AGREE THAT ANY SUCH CLAIM OR CAUSE OF ACTION SHALL BE TRIED BY A COURT TRIAL WITHOUT A JURY. WITHOUT LIMITING THE FOREGOING, THE PARTIES FURTHER AGREE THAT THEIR RESPECTIVE RIGHT TO A TRIAL BY JURY IS WAIVED BY OPERATION OF THIS SECTION AS TO ANY ACTION, COUNTERCLAIM OR OTHER PROCEEDING WHICH SEEKS, IN WHOLE OR IN PART, TO CHALLENGE THE VALIDITY OR ENFORCEABILITY OF THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS OR ANY PROVISION HEREOF OR THEREOF. THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS.

- 12.18 Entire Agreement. This Agreement, together with the other Loan Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document (other than the Collateral Agency Agreement or any other Security Document), the provisions of this Agreement shall control; provided, that the inclusion of supplemental rights or remedies in favor of the Agent, the Banks, the Collateral Agent or any holder of Parity Debt in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 12.19 Collateral Agency Agreement. (a) THIS AGREEMENT IS SUBJECT TO THE TERMS AND CONDITIONS CONTAINED IN THE COLLATERAL AGENCY AGREEMENT, WHICH COLLATERAL AGENCY AGREEMENT, AMONG OTHER THINGS, ESTABLISHES CERTAIN RIGHTS WITH RESPECT TO THE SECURITY FOR THIS AGREEMENT AND THE SHARING OF PROCEEDS THEREOF WITH CERTAIN OTHER SECURED CREDITORS (AS DEFINED IN THE COLLATERAL AGENCY AGREEMENT). COPIES OF SUCH COLLATERAL AGENCY AGREEMENT WILL BE FURNISHED TO ANY BANK UPON REQUEST TO THE BORROWER OR THE AGENT.
- (b) Pursuant to Section 12 of the Collateral Agency Agreement, the Banks hereby consent, without any further action or notice required, to the appointment of Wachovia as the successor Collateral Agent.
- 12.20 Ratification and Confirmation of the Security Documents. Except as specifically amended by this Agreement, and the documents executed and delivered in connection herewith, each of the Security Documents, including without limitation the General Security Agreement, shall remain in full force and effect and are hereby ratified and confirmed.
- 12.21 Interest Rate Limitation. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable law (the "MAXIMUM RATE"). If the Agent or any Bank shall receive interest in an amount that exceeds the Maximum

Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Agent or a Bank exceeds the Maximum Rate, such Person may, to the extent permitted by applicable law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

12.22 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Agent and each Bank, regardless of any investigation made by the Agent or any Bank or on their behalf and notwithstanding that the Agent or any Bank may have had notice or knowledge of any Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered by their proper and duly authorized officers as of the day and year first above written.

BORROWER:

```
WACHOVIA BANK,
NATIONAL ASSOCIATION, as Agent,
Issuing Bank, Swing Line Bank and a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

```
CITICORP USA, INC.,
as Syndication Agent and as a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

```
CREDIT SUISSE FIRST BOSTON,
acting through its Cayman Islands Branch,
as Documentation Agent and as a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

MANUFACTURERS AND TRADERS TRUST COMPANY, a successor in interest to ALLFIRST BANK, as a Bank

```
MELLON BANK, N.A.,
as a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

NATIONAL CITY BANK OF PENNSYLVANIA, as a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)

```
PNC BANK, NATIONAL ASSOCIATION, as a Bank
```

```
By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

```
SUN NATIONAL BANK, as a Bank
```

```
By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

```
UNION BANK OF CALIFORNIA,
as a Bank

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

Each of the undersigned hereby acknowledges receipt of the foregoing Credit Agreement and hereby acknowledges and reaffirms that its Subsidiary Guarantee and its Subsidiary Security Agreement shall remain in full force and effect and is hereby ratified and confirmed in all respects notwithstanding the execution of such Credit Agreement and the consummation of the transactions described or otherwise contemplated therein. Each of the undersigned further acknowledges, confirms and ratifies (i) its obligations under the Subsidiary Guarantee and the Subsidiary Security Agreement are valid and binding obligations upon it and (ii) the Subsidiary Security Agreement and all of the Collateral described therein does, and shall continue to, secure the payment of all of the Obligations under the Credit Agreement and the Notes. To the best of the each of the undersigned's knowledge, it possesses no defense, offset, counterclaim, or cross-claim whatsoever to the enforcement of such Subsidiary Guarantee and Subsidiary Security Agreement.

Date: August 28, 2003

A	MERIGAS PROPANE PARTS & SERVICE, INC.
	y:
	ame:
ı	itle:
А	MERIGAS EAGLE PROPANE, INC.
	formerly Columbia Propane Corporation)
	y:
	ame:
ı	itle:
Α	MERIGAS EAGLE HOLDINGS, INC.
	formerly CP Holdings, Inc.)
	y:
	ame:
Т	itle:
Signat	ure Page
Signat	to
Credit	Agreement
	Propane, L.P.)
(Amerigas	Propane, L.P.)

```
AMERIGAS EAGLE PROPANE, L.P.

By:
Name:
Title:

AMERIGAS EAGLE PARTS & Service, Inc.

By:
Name:
Title:
Signature Page
to
Credit Agreement
(AmeriGas Propane, L.P.)
```

NOTICE OF APPOINTMENT

August 28, 2003

Wachovia Bank, National Association One Wachovia Center 301 South College Street Charlotte, NC 28288

Ladies and Gentlemen:

Reference is made to that certain Intercreditor and Agency Agreement, dated as of April 19, 1995, as amended (the "COLLATERAL AGENCY AGREEMENT"), by and among AmeriGas Propane, L.P., AmeriGas Propane, Inc., Petrolane Incorporated, the Restricted Subsidiaries of AmeriGas Propane, L.P., Bank of America, N.A., as Agent and as Collateral Agent, Mellon Bank, N.A., as Cash Collateral Sub-Agent, and each of the Note Holders and Parity Lenders that is a party thereto. Capitalized terms not otherwise defined herein shall have the meanings set forth in the Collateral Agency Agreement.

In accordance with the terms of Section 12(a) of the Collateral Agency Agreement, we have received the consent of the Requisite Holders (the "CONSENT") to the appointment of Wachovia Bank, National Association ("WACHOVIA") as Collateral Agent under the Collateral Agency Agreement. Pursuant to the terms of the Consent, you shall assume the duties of Collateral Agent as of August 28, 2003.

We hereby indemnify Wachovia against any loss, expense, liability or obligation arising out of or in connection with the acceptance and the performance of its duties under the Collateral Agency Agreement and under the Security Documents, except for any loss, expense, liability or obligation resulting from the gross negligence or willful misconduct of the Collateral Agent.

AMERIGAS PROPANE, L.P.

By: Robert W. Krick
----Name:
Title:

AGREED AND ACCEPTED:

WACHOVIA BANK, NATIONAL ASSOCIATION

By: Lawrence P. Sullivan
.....
Name:

Title:

FINANCING AGREEMENT

This Amended and Restated Financing Agreement, dated as of August 28, 2003 (the "Agreement"), is between AmeriGas Propane, Inc., a Pennsylvania corporation ("AGP"), and AmeriGas Propane, L.P., a Delaware limited partnership ("APLP").

BACKGROUND

- A. AGP is the sole general partner of APLP.
- B. AGP has agreed to make available to APLP a revolving credit facility, upon the terms and conditions set forth in this Agreement, to finance APLP's working capital, capital expenditures and interest and distribution expenses.
- C. Capitalized terms used herein and not otherwise defined shall have the respective meanings assigned to them in that certain Credit Agreement dated as of August 28, 2003 among AGP, APLP, Petrolane Incorporated, Wachovia Bank, N. A., as Agent, Issuing Bank and Swing Line Bank, and the other financial institutions party thereto (the "Credit Agreement").

NOW, THEREFORE, in consideration of the premises and covenants contained herein, and intending to be legally bound hereby, the parties hereto agree as follows:

1. Credit Line Terms.

- (a) Credit Limit. AGP agrees, on the terms and conditions set forth herein, to make loans ("Loans") to APLP from time to time on any Business Day during the period from the date hereof to the Revolving Termination Date, in an aggregate principal amount not to exceed at anytime outstanding of \$20,000,000 (the "Commitment"). Subject to the other terms and conditions hereof, APLP may borrow under this Section 1(a), prepay under Section 1(d) and reborrow under this Section 1(a). The Commitment shall automatically be terminated (i) upon the occurrence of an Event of Default (as defined in Section 4 below) unless waived in writing by AGP or (ii) if AGP is no longer the sole general partner of APLP. Upon the termination of the Commitment, subject to the provisions of Section 15 below, APLP shall pay to AGP all amounts owing or payable under this Agreement, without presentment, demand, protest or any other notice of any kind, all of which are expressly waived by APLP.
- (b) Procedure for Borrowing. Each borrowing by APLP pursuant to Section 1(a) shall be made upon APLP's written notice delivered to AGP one Business Day prior to the requested borrowing date specifying the requested borrowing amount and date. All

advances by AGP hereunder shall be noted on the Master Promissory Note provided for in Section 3 hereof.

- (c) Voluntary Termination or Reduction of Commitment. APLP may, upon prior written notice to AGP, terminate the Commitment or permanently reduce the Commitment by an aggregate minimum amount of \$1,000,000 or any multiple of \$1,000,000 in excess thereof. At no time shall the amount of outstanding Loans exceed the amount of the Commitment.
- (d) Optional Prepayment. APLP may at anytime, upon prior notice to AGP, prepay Loans, together with all accrued and unpaid interest thereon, in whole or in part.
- (e) Repayment. APLP shall repay to AGP on the Revolving Termination Date the aggregate principal amount of Loans outstanding on such date, together with all accrued and unpaid interest thereon.
- (f) Interest. Each Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date to the one-month anniversary of the borrowing date at a rate per annum equal to the Eurodollar Rate, with an Interest Period of one month, on such borrowing date, plus the Applicable Margin for Eurodollar Rate Loans Revolving Loans (the "Interest Rate"). On each one-month anniversary of any Loan (the "Reset Date"), such Loan shall bear interest on the outstanding principal amount thereof until the next one-month anniversary date at a rate per annum equal to the Interest Rate on the Reset Date. The Interest Rate shall change during any Interest Period as a result of changes in the Applicable Margin. Interest on each Loan shall be paid in arrears on the last day of each month and on the Revolving Termination Date.
- (g) Default Interest. During the existence of any Event of Default, interest shall be paid upon demand by AGP. The foregoing notwithstanding, if any amount of principal of or interest on any Loan, or any other amount payable hereunder, is not paid in full when due, APLP agrees to pay interest on such unpaid amount, from the date such amount becomes due to the date such amount is paid in full, and after as well as before any entry of judgment thereon to the extent permitted by law, payable on demand (but not more frequently than once per week), at a fluctuating rate per annum equal to the Base Rate plus 2%.
- (h) Facility Fees. APLP shall pay to AGP on the last day of each calendar quarter a facility fee from the date hereof through the Revolving Termination Date, on the daily average amount of the Commitment (whether or not used), at the rate per annum set forth in Section 2.10(b) of the Credit Adreement.

- (i) Payments by APLP. All payments by APLP shall be made without set-off, recoupment or counterclaim. Any payment due on a day other than a Business Day shall be made on the following Business Day, and such extension of time shall in such case be included in the computation of interest or fees, as the case may be.
- (j) Waivers of Notices, Etc. APLP hereby waives presentment, demand, protest, notice of default, and any and all other notices or demands in connection with the delivery, acceptance, performance or enforcement of this Agreement.
- (k) Computation of Fees and Interest. All computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed. Interest and fees shall accrue during each period during which interest on such fees are computed from the first day thereof to the last day thereof.
- 2. Lending Authority. The President, any Vice President, the Treasurer and any other officer of AGP shall have the authority hereunder on behalf of AGP to receive requests for advances from, to lend funds to, and to give notices to, APLP, and to take such other steps on behalf of AGP as are reasonable and necessary to carry out the terms of this Agreement. The President, any Vice President, the Treasurer and any other officer of AGP shall have the authority hereunder on behalf of APLP to request advances and to borrow funds from, and to give notices and confirmations to, AGP, to make repayments of any amounts due hereunder, and to take such other steps on behalf of APLP as are reasonable and necessary to carry out the terms of this Agreement.
- 3. Master Promissory Note. Concurrent with the signing of this Agreement, APLP shall deliver to AGP a properly completed and duly executed Master Promissory Note, substantially in the form attached hereto as Exhibit A; provided, however, that notwithstanding the face amount of the Master Promissory Note, APLP's liability thereunder shall be limited at all times to APLP's actual indebtedness (principal and interest) then outstanding to AGP hereunder.
- 4. Events of Default. It shall be an "Event of Default" under this Agreement if (a) APLP shall fail to pay to AGP any amount of principal of any Loan or, within ten (10) days after the same becomes due, any interest, fee or other amount payable to AGP hereunder, or (b) an Event of Default as described in Section 9.1 of the Credit Agreement shall occur and be continuing.

If an Event of Default occurs and is continuing, APLP may, subject to the provisions of Section 15 hereof:

- (a) declare all amounts owing or payable under this Agreement due and payable, without presentment, demand, protest or other notice of any kind, all of which are expressly waived by APLP; and/or
- (b) exercise all rights and remedies available under this Agreement or applicable law; $\,$

provided, however, that upon the occurrence of any event specified in subsection (f) or (g) of Section 9.1 of the Credit Agreement (in the case of clause (i) of subsection (g) upon the expiration of the sixty (60) day period mentioned therein), the unpaid principal amount of all Loans and all interest and other amounts shall automatically become due and payable without further act of AGP.

The rights provided in this Agreement are cumulative and are not exclusive of any other rights, powers, privileges or remedies provided by law or in equity, or under any other instrument, document or Agreement now existing or hereafter arising.

- 5. Payment of Expenses. APLP shall bear all expenses incurred by either party hereto in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereby.
- 6. Further Assurances. Each party shall from time to time at the request of the other execute, acknowledge and deliver any and all such further papers, documents, powers of attorney, notices or other instruments that may reasonably be required to give full force and effect to the provisions and intent of this Agreement.
- 7. Assignment. None of the rights or obligations of APLP under this Agreement shall be assignable by APLP. AGP shall not sell, pledge, assign or otherwise transfer any of its rights hereunder without giving APLP written notice thereof.
- 8. Notices. Except where oral notice is specifically permitted by this Agreement and other than as set forth below, any notice to AGP or APLP hereunder shall be in writing and sent to the following addresses, unless and until either party notifies the other in writing to the contrary:
- If to AmeriGas Propane, Inc., to Box 965, Valley Forge, PA 19482, Attention: Treasurer.
- If to AmeriGas Propane, L. P., to Box 965, Valley Forge, PA 19482, Attention: Treasurer.
- 9. Governing Law. This Agreement, and all documents issued or delivered pursuant hereto, shall be deemed to have been ${\sf C}$

signed, accepted, completed and issued at the office of AGP at King of Prussia, Pennsylvania, and shall be governed by, and construed and enforced in accordance with, the laws of the Commonwealth of Pennsylvania.

- 10. Binding Effect. This Agreement shall inure to the benefit of, and be binding upon and enforceable by, the parties hereto and their respective successors and assigns.
- 11. Entire Agreement; Amendments. This Agreement, together with the Exhibit referred to herein, sets forth the entire agreement of the parties hereto with respect to the subject matter hereof. Any prior agreements or understandings between the parties hereto regarding the subject matter hereof, whether written or oral, are superseded by this Agreement. This Agreement may not be amended or modified except by a written instrument duly executed by each of the parties hereto.
- 12. Waiver. Any term or provision of this Agreement may be waived at any time by the party entitled to the benefit thereof by a written instrument duly executed by such party, and such waiver shall not constitute a waiver of any other provision of this Agreement or a further waiver of the provision waived.
- 13. Section Headings; Gender; Number. All section headings, and the use of a particular gender or number (plural or singular), are for convenience only and shall in no way modify or restrict any of the terms or provisions hereof.
- 14. Counterparts. This Agreement may be executed in two counterparts, each of which shall be deemed an original, and both of which taken together shall constitute but one and the same instrument. This Agreement shall become binding only when each party hereto has executed and delivered to the other party one or more counterparts.
- 15. Subordination. (a) The indebtedness ("Subordinated Debt") evidenced by this Agreement is subordinate and junior in right of payment to all Senior Debt (as defined in subdivision (b) hereof) of APLP to the extent provided herein.
- (b) For all purposes of these subordination provisions the term "Senior Debt" shall mean all principal of and Make Whole Amount, if any, and interest on (i) APLP's First Mortgage Notes, Series A through C, originally issued in the aggregate principal amount of \$518,000,000, pursuant to separate Note Agreements, dated as of April 12, 1995 as amended, between APLP, AmeriGas Propane, Inc., a Pennsylvania corporation and Petrolane Incorporated, a California corporation and the institutional investors listed on Schedule I thereto (and any notes issued in substitution therefor), (ii) APLP's

First Mortgage Notes, Series D, originally issued in the aggregate amount of \$70,000,000, pursuant to separate Note Agreements, dated as of March 31, 1999 as amended, between APLP, AmeriGas Propane, Inc., a Pennsylvania corporation and the institutional investors listed on Schedule I thereto (and any notes issued in substitution therefor), (iii) APLP's First Mortgage Notes, Series E, originally issued in the aggregate amount of \$80,000,000, pursuant to separate Note Agreements, dated as of March 15, 2000 as amended, between APLP, AmeriGas Propane, Inc., a Pennsylvania corporation and the institutional investors listed on Schedule I thereto (and any notes issued in substitution therefor), (iv) those obligations outstanding under the Credit Agreement, and (v) all other indebtedness of APLP for borrowed money unless, under the instrument evidencing the same or under which the same is outstanding, it is expressly provided that such other indebtedness is junior and subordinate to other indebtedness and obligations of APLP. The Senior Debt shall continue to be Senior Debt and entitled to the benefits of these subordination provisions irrespective of any amendment, modification or waiver of any term of or extension or renewal of the Senior Debt.

(c) Upon the happening of an event of default with respect to any Senior Debt, as defined therein or in the instrument under which the same is outstanding, which occurs at the maturity thereof or which automatically accelerates or permits the holders thereof to accelerate the maturity thereof, then, unless and until such event of default shall have been remedied or waived or shall have ceased to exist, no direct or indirect payment (in cash, property or securities or by set-off or otherwise) other than Permitted Payments shall be made on account of the principal of, or premium, if any, or interest on any Subordinated Debt, or as a sinking fund for the Subordinated Debt, or in respect of any redemption, retirement, purchase or other acquisition of any of the Subordinated Debt. For purposes of these subordination provisions, "Permitted Payments" shall mean (i) payments of in-kind interest and (ii) payments of Permitted Securities (as defined below) pursuant to paragraph (d) below.

(d) In the event of

- (i) any insolvency, bankruptcy, receivership, liquidation, reorganization, readjustment, composition or other similar proceeding relating to APLP, its creditors as such or its property,
- (ii) any proceeding for the liquidation, dissolution or other winding-up of APLP, voluntary or involuntary, whether or not involving insolvency or bankruptcy proceedings,
- (iii) any assignment by APLP for the benefit of creditors, or

all Senior Debt (including any interest thereon accruing at the legal rate after the commencement of any such proceedings and any additional interest that would have accrued thereon but for the commencement of such proceedings) shall first be paid in full before any payment or distribution, whether in cash, securities or other property (other than Permitted Payments), shall be made to any holder of any Subordinated Debt on account of any Subordinated Debt. Any payment or distribution, whether in cash, securities or other property (other than securities ("Permitted Securities") of APLP or any other entity provided for by a plan of reorganization or readjustment the payment of which is subordinate, at least to the extent provided in these subordination provisions with respect to Subordinated Debt, to the payment of all Senior Debt at the time outstanding and to any securities issued in respect thereof under any such plan of reorganization or readjustment), which would otherwise (but for these subordination provisions) be payable or deliverable in respect of this Subordinated Debt shall be paid or delivered directly to the holders of Senior Debt in accordance with the priorities then existing among such holders until all Senior Debt (including any interest thereon accruing at the legal rate after the commencement of any such proceedings and any additional interest that would have accrued thereon but for the commencement of such proceedings) shall have been paid in full.

- (e) In the event that any holder of Subordinated Debt shall have the right to declare any Subordinated Debt due and payable as a result of the occurrence of any one or more defaults in respect thereof, under circumstances when the terms of subdivision (d) above are not applicable, such holder shall not declare such Subordinated Debt due and payable or otherwise to be in default and, solely in its capacity as a holder of such Subordinated Debt, shall take no action at law or in equity in respect of any such default unless and until all Senior Debt shall have been paid in full.
- (f) If any payment or distribution of any character or any security, whether in cash, securities or other property (other than Permitted Payments), shall be received by a holder of Subordinated Debt in contravention of any of the terms hereof before all the Senior Debt shall have been paid in full, such payment or distribution or security shall be received in trust for the benefit of, and shall be paid over or delivered and transferred to, the holders of the Senior Debt at the time outstanding in accordance with the priorities then existing among such holders for application to the payment of all Senior Debt remaining unpaid, to the extent necessary to pay all such Senior Debt in full. In the event of the failure of any holder of any Subordinated Debt to endorse or assign

any such payment, distribution or security, each holder of Senior Debt is hereby irrevocably authorized to endorse or assign the same.

- (g) No present or future holder of any Senior Debt shall be prejudiced in the right to enforce subordination of Subordinated Debt by any act or failure to act on the part of APLP. Nothing contained herein shall impair, as between APLP and the holder of this Subordinated Debt, the obligation of APLP to pay to the holder hereof the principal hereof and interest hereon as and when the same shall become due and payable in accordance with the terms hereof, or prevent the holder of any Subordinated Debt from exercising all rights, powers and remedies otherwise permitted by applicable law or hereunder upon a default or event of default hereunder, all subject to the rights of the holders of the Senior Debt to receive cash, securities or other property (other than Permitted Payments) otherwise payable or deliverable to the holders of Subordinated Debt.
- (h) Upon the payment in full of all Senior Debt, the holders of Subordinated Debt shall be subrogated to all rights of any holders of Senior Debt to receive any further payments or distributions applicable to the Senior Debt until the Subordinated Debt shall have been paid in full, and, for purposes of such subrogation, no payment or distribution received by the holders of Senior Debt of cash, securities or other property to which the holders of the Subordinated Debt would have been entitled except for these subordination provisions shall, as between APLP and its creditors other than the holders of Subordinated Debt, on the one hand, and the holders of Subordinated Debt, on the other, be deemed to be a payment or distribution by APLP to or on account of Senior Debt.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

AMERIGAS PROPANE, INC.

By: Robert W. Krick
Robert W. Krick
Treasurer

AMERIGAS PROPANE, L.P.

BY: AMERIGAS PROPANE, INC. AS GENERAL PARTNER

By: Robert W. Krick
Robert W. Krick
Treasurer

FORM OF MASTER PROMISSORY NOTE

Maximum Principal Amount: \$20,000,000 Date: August 28, 2003

FOR VALUE RECEIVED, AmeriGas Propane, L.P., a Delaware limited partnership (the "Borrower"), hereby promises to pay to AmeriGas Propane, Inc., a Pennsylvania corporation (the "Lender"), the principal amount of each advance made to the Borrower under the Financing Agreement referred to below on such dates as may be determined in accordance with such Agreement. The aggregate principal amount of such advances outstanding at any one time shall not exceed the amount set forth above.

This Master Promissory Note is issued under, and subject to the terms and conditions of, the Financing Agreement, dated as of August 28, 2003 (the "Agreement"), between the Borrower and the Lender.

Interest on the outstanding principal amount of each advance evidenced by this Master Promissory Note shall accrue and be payable in accordance with the terms of the Agreement.

The Agreement provides for optional prepayment by the Borrower of the advances evidenced by this Master Promissory Note. The Agreement also provides that this Master Promissory Note is subject to subordination terms as provided in paragraph 15 thereof.

This Master Promissory Note is being made and delivered in the Commonwealth of Pennsylvania and shall be governed by, and construed and enforced in accordance with, the laws of such Commonwealth.

IN WITNESS WHEREOF, the Borrower has caused this Master Promissory Note to be duly executed and delivered as of the date first above written.

AMERIGAS PROPANE, L.P.

BY: AMERIGAS PROPANE, INC. AS GENERAL PARTNER

By: ROBERT W. KRICK

NAME: ROBERT W. KRICK TITLE: TREASURER

Master Promissory Note (cont'd) LOANS AND PAYMENTS OF PRINCIPAL

Date	Quoted Rate	Amount of Loan	Amount of Principal Repaid	Unpaid Principal Balance	Notation Made By

AmeriGas Partners, L.P. 2003 Annual Report

CONSOLIDATED BALANCE SHEETS (Thousands of dollars)

	September 30,		
		2002	
ASSETS			
Current assets: Cash and cash equivalents	\$ 45,872	\$ 47,400	
Accounts receivable (less allowances for	φ 45,672	\$ 47,400	
doubtful accounts of \$ 9,192 and \$7,588, respectively)	100,170	83,274	
Accounts receivable - related parties	2,915	6,862	
Inventories	70,171	62,496	
Prepaid expenses and other current assets	17,204	6,862 62,496 24,811	
Total current assets	236, 332	224,843	
Total current assets	230,332	224,043	
Property, plant and equipment (less accumulated depreciation and			
amortization of \$473,090 and \$408,590, respectively)	594,604	611,550	
Goodwill and excess reorganization value	602,475	589,923	
Intangible assets (less accumulated amortization	07.000	00 500	
of \$11,934 and \$8,651, respectively) Other assets	27,032	22,586	
Other assets	21,733	23,710	
Total assets	\$ 1,482,176	\$ 1,472,618	
	27,032 21,733 \$ 1,482,176 ========	=========	
LIABILITIES AND PARTNERS' CAPITAL Current liabilities: Current maturities of long-term debt	\$ 58,705	\$ 60,398	
Bank loans		10.000	
Accounts payable - trade	87,352	81,891	
Accounts payable - related parties	930	5.003	
Employee compensation and benefits accrued	26,307	22,390	
Interest accrued Customer deposits and advances	31,987 52 771	34,492 52 177	
Other current liabilities	39,996	40.473	
00.00.00.00.00.00.00.00.00.00.00.00.00.	26,307 31,987 52,771 39,996		
Total current liabilities	298,048	307,824	
Long-term debt	868,581	885,386	
Other noncurrent liabilities	54,859	44,810	
Commitments and contingencies (note 11)			
Minority interests	7,005	6,232	
Partners' capital:	,	,	
Common unitholders (units issued - 52,333,208 and 39,541,286, respectively)	255,423	201,660	
Subordinated unitholders (units issued - 9,891,072 in 2002)		17,846	
General partner	2,577	2,214	
Accumulated other comprehensive (loss) income	2,577 (4,317)	6,646	
Total partners' capital	253,683	228,366	
Total lightlities and partners! capital	ф 1 400 176	е 1 470 610	
Total liabilities and partners' capital	\$ 1,482,176 =======	\$ 1,472,618 ========	

CONSOLIDATED STATEMENTS OF OPERATIONS (Thousands of dollars, except per unit)

Year	Ended	September	30,	
------	-------	-----------	-----	--

	rear Ended September 30,			
		2002		
Revenues: Propane	\$ 1,502,564	\$ 1,191,649	\$ 1,322,934	
Other	125,860	\$ 1,191,649 116,231	95,430	
		1,307,880		
Costs and expenses:				
Cost of sales - propane	856,883	605,695	809,210	
Cost of sales - other	53,452	47,383 447,809	37,809 368,949	
Operating and administrative expenses	53, 452 488, 434	447,809	368,949	
Depreciation and amortization	14,023	00,104	74,760	
Other (income), net	(8,960)	(4, 403)	(6,154)	
		1,162,588		
Operating income	163,990			
Loss on extinguishments of debt	(3,023)	(752)		
Interest expense	(87,19)	(87,839)	(80,396)	
·		(87,839)		
Income before income taxes	73,772	56,701	53,394	
Income tax (expense) benefit	(586)	(340) (995)	327	
Minority interests	(1,228)	(995)	(706)	
Income before accounting changes				
Cumulative effect of accounting changes	, 	55,366 	53,015 12,494	
Net income	\$ 71.958	\$ 55,366	\$ 65.509	
	=======	========	=======	
General partner's interest in net income	\$ 720 ======	\$ 554 =======	\$ 655 =======	
Limited partners' interest in net income	\$ 71,238 =======	\$ 54,812 ======	\$ 64,854 ======	
Income per limited partner unit - basic and diluted:				
Income before accounting changes	\$ 1.42	\$ 1.12	\$ 1.18	
Cumulative effect of accounting changes			0.28	
Not income nor limited norther unit	т 1 40	т 1 1 2	т 1 46	
Net income per limited partner unit	\$ 1.42 =======	\$ 1.12 =======	\$ 1.46 =======	
Average limited partner units outstanding (thousands):				
Basic	50,267	48,909 ======= 48,932	44,453	
Diluted	======================================	40.000	44 450	
Diluted	50,33 <i>1</i>	48,932 ======	44,453	

CONSOLIDATED STATEMENTS OF CASH FLOWS (Thousands of dollars)

	Year Ended September 30,			
		2002		
CASH FLOWS FROM OPERATING ACTIVITIES				
Net income	\$ 71,958	\$ 55,366	\$ 65,509	
Adjustments to reconcile net income to net				
cash provided by operating activities: Cumulative effect of accounting changes			(12 404)	
Depreciation and amortization	74 625	66,104	(12,494) 74,760	
Other, net	6 747	1,321	2,920	
Net change in:	0,747	1,321	2,920	
Accounts receivable	(10 027)	0 717	4 902	
Inventories	(19,921)	10 576	2 620	
Accounts payable	1 200	0.570	5,030 (5 511)	
Other current assets and liabilities	12 021	7 057	10 206	
Other current assets and manifetes	12,021		19,290	
Net cash provided by operating activities	139,303	159,520	153,011	
		8,717 10,576 9,579 7,857 159,520		
CASH FLOWS FROM INVESTING ACTIVITIES				
Expenditures for property, plant and equipment	(52,933)	(53,472)	(37,890)	
Proceeds from disposals of assets	7,408	9,849	5,347	
Acquisitions of businesses, net of cash acquired	(27,000)	(736)	(205.571)	
		(53,472) 9,849 (736)		
Net cash used by investing activities	(72,525)	(44,359)	(238,114)	
CASH FLOWS FROM FINANCING ACTIVITIES				
Distributions	(111,462)	(108,504)	(98,435)	
Minority interest activity	(686)	(624)	2,374	
(Decrease) increase in bank loans	(10,000)	10,000	(30,000)	
Issuance of long-term debt	122,780	40,900	252,833	
Repayment of long-term debt	(144,701)	(99,149)	(110,767)	
Proceeds from issuance of Common Units	75,005	56,556	39,836	
Proceeds from sale of AmeriGas OLP interest			50,000	
Capital contributions from General Partner	758	(108,504) (624) 10,000 40,900 (99,149) 56,556 571	956	
Net cash (used) provided by financing activities	(68,306)	(100,250) \$ 14,911	106,797	
Cook and sook assistalanta (dassessa) increasa	 Φ (4 ΕΩΩ)	Ф. 44.044	т ол сол	
Cash and cash equivalents (decrease) increase	\$ (1,528) =======	\$ 14,911 =======	\$ 21,694 ======	
	_	-		
CASH AND CASH EQUIVALENTS				
End of year	\$ 45,872	\$ 47,400	\$ 32,489	
Beginning of year	47,400	32,489	10,795	
(Decrees) increes	Φ (4 500)	Ф. 44.046	т ол ссл	
(Decrease) increase	\$ (1,528) =======	\$ 14,911 ======	\$ 21,694 ======	
	_	_		

CONSOLIDATED STATEMENTS OF PARTNERS' CAPITAL (Thousands of dollars, except unit data)

	Numbe Common	er of units Subordinated	Common	Subordinated	General partner	Accumulated other comprehensive income (loss)	Total partners' capital
Balance September 30, 2000 Net income Cumulative effect of change	32,078,293	9,891,072	\$ 118,872 50,123	\$ 35,542 14,731	\$ 1,557 655	\$	\$ 155,971 65,509
in accounting principle - SFAS No. 133 Net gains on derivative						8,921	8,921
instruments Reclassification of net gains						276	276
on derivative instruments						(23,380)	(23,380)
Comprehensive income Distributions Common Units issued in			50,123 (75,691)	14,731 (21,760)	655 (984)	(14,183)	51,326 (98,435)
connection with acquisitions Common Units issued in connection with public	2,382,946		53,861		544		54,405
offering	2,300,000		39,836		402		40,238
Balance September 30, 2001	36,761,239	9,891,072	187,001	28,513	2,174	(14, 183)	203,505
Net income			43,719	11,093	554		55,366
Net losses on derivative instruments Reclassification of net losses						(10,664)	(10,664)
on derivative instruments						31,493	31,493
Comprehensive income Distributions Common Units issued in			43,719 (85,659)	11,093 (21,760)	554 (1,085)	20,829	76,195 (108,504)
connection with executive compensation plan Common Units issued in connection with public	2,000		43				43
offering	2,428,047		49,623		501		50,124
Common Units sold to General Partner	350,000		6,933		70		7,003
Balance September 30, 2002	39,541,286	9,891,072	201,660	17,846	2,214	6,646	228,366
Net income Net gains on derivative			69,859	1,379	720		71,958
instruments Reclassification of net gains						14,909	14,909
on derivative instruments						(25,872)	(25,872)
Comprehensive income Distributions Common Units issued in			69,859 (104,907)	1,379 (5,440)	720 (1,115)	(10,963)	60,995 (111,462)
connection with public offering Common Units issued in	2,900,000		75,005		758		75,763
connection with executive compensation plan	850		21				21
Conversion of Subordinated Units	9,891,072	(9,891,072)	13,785	(13,785)			
Balance September 30, 2003	52,333,208		\$ 255,423 ======	\$ ========	\$ 2,577 =======	\$ (4,317) ======	\$ 253,683 ======

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Thousands of dollars, except per unit amounts)

NOTE 1 - PARTNERSHIP ORGANIZATION AND FORMATION

AmeriGas Partners, L.P. ("AmeriGas Partners") was formed on November 2, 1994, and is a publicly traded limited partnership. AmeriGas Partners conducts a national propane distribution business through its principal operating subsidiaries AmeriGas Propane, L.P. ("AmeriGas OLP") and AmeriGas OLP's subsidiary, AmeriGas Eagle Propane, L.P. ("Eagle OLP"). AmeriGas Partners, AmeriGas OLP and Eagle OLP are Delaware limited partnerships. AmeriGas OLP and Eagle OLP are collectively referred to herein as "the Operating Partnerships," and AmeriGas Partners, the Operating Partnerships and all of their subsidiaries are collectively referred to herein as "the Partnership" or "we."

The Operating Partnerships are engaged in the distribution of propane and related equipment and supplies. The Operating Partnerships comprise the largest retail propane distribution business in the United States serving residential, commercial, industrial, motor fuel and agricultural customers from locations in 46 states, including Alaska and Hawaii.

At September 30, 2003, AmeriGas Propane, Inc. (the "General Partner"), an indirect wholly owned subsidiary of UGI Corporation ("UGI"), held a 1% general partner interest in AmeriGas Partners and a 1.01% general partner interest in AmeriGas OLP. The General Partner and its wholly owned subsidiary Petrolane Incorporated ("Petrolane," a predecessor company of the Partnership) also owned 24,525,004 Common Units of AmeriGas Partners. The remaining 27,808,204 Common Units are publicly held. The Common Units represent limited partner interests in AmeriGas Partners.

AmeriGas Partners holds a 99% limited partner interest in AmeriGas OLP. AmeriGas OLP, indirectly through subsidiaries, owns an effective 0.1% general partner interest and a direct approximate 99.8% limited partner interest in Eagle OLP. An unrelated third party (minority partner) holds an approximate 0.1% limited partner interest in Eagle OLP.

AmeriGas Partners and the Operating Partnerships have no employees. Employees of the General Partner conduct, direct and manage the activities of AmeriGas Partners and AmeriGas OLP. The General Partner also provides management and administrative services to AmeriGas Eagle Holdings, Inc. ("AEH"), the general partner of Eagle OLP, under a management services agreement. The General Partner is reimbursed monthly for all direct and indirect expenses it incurs on behalf of the Operating Partnerships.

NOTE 2 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

CONSOLIDATION PRINCIPLES. The consolidated financial statements include the accounts of AmeriGas Partners and its majority-owned subsidiaries. We eliminate all significant intercompany accounts and transactions when we consolidate. We account for the General Partner's 1.01% interest in AmeriGas OLP and the minority partner's 0.1% limited partner interest in Eagle OLP as minority interests in the consolidated financial statements. The Partnership's 50% ownership interest in Atlantic Energy, Inc., a propane storage terminal located in Chesapeake, Virginia, is accounted for by the equity method.

RECLASSIFICATIONS. In order to more appropriately classify direct costs associated with the Partnership's Prefilled Propane Xchange ("PPX(R)") program, for the year ended September 30, 2003, certain costs previously considered operating and administrative expenses have been included in cost of sales. We have reclassified \$21,008 and \$11,044 of such costs incurred during the years ended September 30, 2002 and 2001, respectively, to conform to the current-year presentation.

In January 2003, the Partnership recorded a loss of \$3,023 resulting from an early extinguishment of long-term debt. This loss has been reflected in the Consolidated Statements of Operations for the year ended September 30, 2003 as "loss on extinguishments of debt." A loss of \$752 associated with a November 2001 early extinguishment of long-term debt previously included in other income, net, in the Consolidated Statement of Operations for the year ended September 30, 2002 has been reclassified to conform to the current-year presentation (see Note 6).

We have reclassified certain other prior-year balances to conform to the current-year presentation.

USE OF ESTIMATES. We make estimates and assumptions when preparing financial statements in conformity with accounting principles generally accepted in the United States. These estimates and assumptions affect the reported amounts of assets and liabilities, revenues and expenses, as well as the disclosure of contingent assets and liabilities. Actual results could differ from these estimates.

REVENUE RECOGNITION. We recognize revenue from the sale of propane principally as product is delivered to customers. Revenue from the sale of appliances and equipment is recognized at the time of sale or installation. Revenue from repairs and maintenance is recognized upon completion of the service.

Effective October 1, 2000, we applied the guidance of Securities and Exchange Commission ("SEC") Staff Accounting Bulletin No. 101 entitled "Revenue Recognition" ("SAB 101") with respect to annually billed nonrefundable tank fees. Under this accounting method, revenues from such fees are recorded on a straight-line basis over one year. Prior to the change in accounting, such revenues were recorded when billed. For a detailed description of this change in accounting and its impact on our results, see Note 18.

INVENTORIES. Our inventories are stated at the lower of cost or market. We determine cost using an average cost method for propane, specific identification for appliances, and the first-in, first-out ("FIFO") method for all other inventories.

PROPERTY, PLANT AND EQUIPMENT AND RELATED DEPRECIATION. We record property, plant and equipment at cost. The amounts we assign to property, plant and equipment of businesses we acquire are based upon estimated fair value at date of acquisition. When plant and equipment are retired or otherwise disposed of, we remove the cost and accumulated depreciation from the appropriate accounts and any resulting gain or loss is recognized in "other income, net" in the Consolidated Statements of Operations. We compute depreciation expense on plant and equipment using the straight-line method over estimated service lives generally ranging from 15 to 40 years for buildings and improvements; 7 to 30 years for storage and customer tanks and cylinders; and 2 to 10 years for vehicles, equipment, and office furniture and fixtures.

Depreciation expense was \$70,423 in 2003, \$61,993 in 2002 and \$48,169 in 2001.

Effective October 1, 2000, we changed our method of accounting for costs to install Partnership-owned tanks at customer locations. Under the new accounting method, all costs to install such tanks, net of amounts billed to customers, are capitalized and depreciated over the estimated period of benefit not exceeding ten years. For a detailed description of this change in accounting and its impact on our results, see Note 18.

INTANGIBLE ASSETS. The Partnership's intangible assets comprise the following at September 30:

	2003	2002
SUBJECT TO AMORTIZATION: Customer relationships and noncompete agreements (a) Accumulated amortization	\$ 38,966 (11,934) \$ 27,032	\$ 31,237 (8,651) \$ 22,586
NOT SUBJECT TO AMORTIZATION: Goodwill (a) Excess reorganization value	\$ 509,155 93,320 \$ 602,475	\$ 496,603 93,320 \$ 589,923

(a) The increases in the carrying amounts of intangible assets during Fiscal 2003 resulted from Partnership business acquisitions.

We amortize customer relationship and noncompete agreement intangibles over their estimated periods of benefit which do not exceed 15 years. Prior to the adoption of SFAS 142, we amortized goodwill resulting from purchase business combinations on a straight-line basis over 40 years, and excess reorganization value (resulting from Petrolane's July 1993 reorganization under Chapter 11 of the U.S. Bankruptcy Code) on a straight-line basis over 20 years. Amortization expense of intangible assets was \$3,283 in 2003 and \$3,287 in 2002. Amortization expense of intangible assets in 2001, which includes amortization of goodwill and excess reorganization value prior to the adoption of SFAS 142, was \$25,767. Estimated amortization expense of intangible assets during the next five fiscal years is as follows: Fiscal 2004 - \$3,470; Fiscal 2005 - \$3,226; Fiscal 2006 - \$2,841; Fiscal 2007 - \$2,212; Fiscal 2008 - \$1,982.

Effective October 1, 2001, we early adopted the provisions of Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets" ("SFAS 142"). SFAS 142 addresses the financial accounting and reporting for acquired goodwill and other intangible assets and supersedes Accounting Principles Board ("APB") Opinion No. 17, "Intangible Assets." SFAS 142 addresses the financial accounting and reporting for intangible assets acquired individually or with a group of other assets (excluding those acquired in a business combination) at acquisition and also addresses the financial accounting and reporting for goodwill and other intangible assets subsequent to their acquisition. Under SFAS 142, an intangible asset is amortized over its useful life unless that life is determined to be indefinite. Goodwill, including excess reorganization value, and other intangible assets with indefinite lives are not amortized but are subject to tests for impairment at least annually. In accordance with the provisions of SFAS 142, the Partnership ceased the amortization of goodwill and excess reorganization value effective October 1, 2001.

The following table provides reconciliations of reported and adjusted net income and basic and diluted net income per limited partner unit as if SFAS 142 had been adopted as of October 1, 2000:

year	Ended September :	30,
2003	2002	2001

NET INCOME: Reported income before accounting changes

Add back goodwill and excess reorganization value amortization, net of adjustment to minority interests				23,570
Adjusted income before accounting changes Cumulative effect of	71,958	55,366		76,585
accounting changes				12,494
Adjusted net income		55,366		89,079
BASIC AND DILUTED INCOME PER LIMITED PARTNER UNIT: Reported income before accounting changes Add back goodwill and excess reorganization value amortization, net of adjustment to minority interests	\$ 1.42	\$ 1.12	·	1.18
milliority interests	 	 		0.53
Adjusted income per unit before accounting	 	 		
changes	1.42	1.12		1.71
Cumulative effect of accounting changes	 	 		0.27
Adjusted net income				
per unit	\$ 1.42	\$ 1.12	\$	1.98

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Thousands of dollars, except per unit amounts)

SFAS 142 requires that we perform an impairment test annually or more frequently if events or circumstances indicate that the value of goodwill might be impaired. No provisions for goodwill impairments were recorded during 2003 and 2002.

DEFERRED DEBT ISSUANCE COSTS. Included in other assets are net deferred debt issuance costs of \$14,022 and \$13,673 at September 30, 2003 and 2002, respectively. We are amortizing these costs over the term of the related debt.

COMPUTER SOFTWARE COSTS. We include in property, plant and equipment costs associated with computer software we develop or obtain for use in our business. We amortize computer software costs on a straight-line basis over expected periods of benefit not exceeding seven years once the installed software is ready for its intended use.

CUSTOMER DEPOSITS. We offer certain of our customers prepayment programs which require customers to pay a fixed periodic amount, or to otherwise prepay a portion of their anticipated propane purchases. Customer prepayments, which exceed associated billings, are classified as customer deposits and advances on the Consolidated Balance Sheets.

ENVIRONMENTAL LIABILITIES. We accrue environmental investigation and clean-up costs when it is probable that a liability exists and the amount or range of amounts can be reasonably estimated. Our estimated liability for environmental contamination is reduced to reflect anticipated participation of other responsible parties but is not reduced for possible recovery from insurance carriers. We do not discount to present value the costs of future expenditures for environmental liabilities.

INCOME TAXES. AmeriGas Partners and the Operating Partnerships are not directly subject to federal income taxes. Instead, their taxable income or loss is allocated to their individual partners. The Operating Partnerships have corporate subsidiaries which are directly subject to federal income taxes. Accordingly, our Consolidated Financial Statements reflect income taxes related to these corporate subsidiaries. Net income for financial statement purposes may differ significantly from taxable income reportable to unitholders. This is a result of (1) differences between the tax basis and financial reporting basis of assets and liabilities and (2) the taxable income allocation requirements of the Second Amended and Restated Agreement of Limited Partnership of AmeriGas Partners, L.P. ("Partnership Agreement") and the Internal Revenue Code.

UNIT-BASED COMPENSATION. As permitted by SFAS No. 123, "Accounting for Stock-Based Compensation" ("SFAS 123"), we apply the provisions of APB Opinion No. 25, "Accounting for Stock Issued to Employees" ("APB 25"), in recording compensation expense for grants of equity instruments to employees. We recorded unit-based compensation expense of \$2,151 in 2003, \$1,018 in 2002 and \$497 in 2001. Our unit-based compensation expense under the provisions of APB 25 for all periods presented was not materially different from amounts determined under the provisions of SFAS 123. For a description of unit-based compensation and related disclosure, see Note 10.

NET INCOME PER UNIT. Net income per unit is computed by dividing net income, after deducting the General Partner's 1% interest, by the weighted average number of limited partner units outstanding. Potentially dilutive Common Units included in the diluted limited partner units outstanding computation of 70,000 in 2003 and 23,000 in 2002 reflect the effects of Common Unit awards issued under AmeriGas Propane, Inc. incentive compensation plans. There were no potentially dilutive securities in 2001.

DERIVATIVE INSTRUMENTS. Effective October 1, 2000, we adopted SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities" ("SFAS 133"). SFAS 133, as amended, establishes accounting and reporting standards for derivative instruments and for hedging activities. It requires that all derivative instruments be recognized as either assets or liabilities and measured at fair value. The accounting for changes in fair value depends upon the purpose of the derivative instrument and whether it is designated and qualifies for hedge accounting. The adoption of SFAS 133 resulted in a cumulative effect charge to net income of \$736 and a cumulative effect increase to accumulated other comprehensive income of \$8,921. The increase in accumulated other comprehensive income is attributable to net gains on derivative instruments designated and qualifying as cash flow hedges on October 1, 2000.

For a detailed description of the derivative instruments we use, our objectives for using them, and related supplemental information required by SFAS 133, see Note 14.

CONSOLIDATED STATEMENTS OF CASH FLOWS. We define cash equivalents as all highly liquid investments with maturities of three months or less when purchased. We record cash equivalents at cost plus accrued interest, which approximates market value. We paid interest totaling \$89,157 in 2003, \$86,556 in 2002 and \$79,302 in 2001.

COMPREHENSIVE INCOME. Comprehensive income comprises net income and other comprehensive income (loss). Other comprehensive income (loss) results from gains and losses on derivative instruments qualifying as cash flow hedges.

SEGMENT INFORMATION. We have determined that we have a single reportable operating segment which engages in the distribution of propane and related equipment and supplies. No single customer represents ten percent or more of consolidated revenues. In addition, virtually all of our revenues are derived

from sources within the United States and virtually all of our long-lived assets are located in the United States.

16

RECENTLY ISSUED ACCOUNTING PRONOUNCEMENTS. In December 2002, the Financial Accounting Standards Board ("FASB") issued SFAS No. 148, "Accounting for Stock-Based Compensation -Transition and Disclosure" ("SFAS 148"). SFAS 148 provides alternative methods of transition for an entity that voluntarily changes to a fair value based method of accounting for stock-based employee compensation. In addition, SFAS 148 amends SFAS No. 123, "Accounting for Stock-Based Compensation" ("SFAS 123"), to require more prominent disclosure about the effects on reported net income of stock-based employee compensation. As permitted by SFAS 148 and SFAS 123, the Partnership expects to continue to account for unit-based compensation in accordance with APB 25 and will continue to provide the required disclosures in its annual and interim financial statements.

In April 2003, the FASB issued SFAS No. 149, "Amendment of Statement 133 on Derivative Instruments and Hedging Activities" ("SFAS 149"). SFAS 149 is effective for contracts entered into or modified after June 30, 2003 and for hedging relationships designated after June 30, 2003. SFAS 149 (i) clarifies under what circumstances a contract with an initial net investment meets the characteristic of a derivative, (ii) clarifies when a derivative contains a financing component, (iii) amends the definition of an underlying-rate, price or index to conform it to language used in FASB Interpretation No. 45, "Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness of Others," and (iv) amends certain other existing pronouncements. SFAS 149 did not change the methods the Partnership uses to account for and report its derivatives and hedging activities.

In May 2003, the FASB issued SFAS No. 150, "Accounting for Certain Financial Instruments with Characteristics of Both Liabilities and Equity" ("SFAS 150"). SFAS 150 is effective at the beginning of the first interim period beginning after June 15, 2003. SFAS 150 establishes guidelines on how an issuer classifies and measures certain financial instruments with characteristics of both liabilities and equity. SFAS 150 further defines and requires that certain instruments within its scope be classified as liabilities on the financial statements. The adoption of SFAS 150 did not affect the Partnership's financial position or results of operations.

In January 2003, the FASB issued Financial Interpretation No. 46, "Consolidation of Variable Interest Entities" ("FIN 46"), which clarifies Accounting Research Bulletin No. 51, "Consolidated Financial Statements." FIN 46 is effective immediately for variable interest entities created or obtained after January 31, 2003. For variable interests created or acquired before February 1, 2003, FIN 46 is effective for the first fiscal or interim period beginning after December 15, 2003. If certain conditions are met, FIN 46 requires the primary beneficiary to consolidate certain variable interest entities in which the other equity investors lack the essential characteristics of a controlling financial interest or their investment at risk is not sufficient to permit the variable interest entity to finance its activities without additional subordinated financial support from other parties. The Partnership has not created or obtained any variable interest entities after January 31, 2003, and is currently in the process of evaluating the impact of FIN 46, which is not expected to have a material effect on its financial position or results of operations.

NOTE 3 - ACQUISITIONS

During 2003, AmeriGas OLP acquired several retail propane distribution businesses for total cash consideration of \$27,000. In conjunction with these acquisitions, liabilities of \$1,469 were incurred. The operating results of these businesses have been included in our operating results from their respective dates of acquisition. The total purchase price of these acquisitions has been allocated to the assets and liabilities acquired as follows:

Net current assets	\$ 2,260
Property, plant and equipment	6,095
Goodwill	12,552
Customer relationships and noncompete agreements	
(estimated useful life of 10 and 5 years,	7,729
respectively)	
Other assets and liabilities	(167
Total	\$ 28,469
	=======

The pro forma effect of these transactions was not material to the Partnership's results of operations.

On August 21, 2001, AmeriGas Partners, through AmeriGas OLP, acquired the propane distribution businesses of Columbia Energy Group ("Columbia Propane Businesses") in a series of equity and asset purchases pursuant to the terms of the Purchase Agreement dated January 30, 2001, and Amended and Restated August 7, 2001 ("Columbia Purchase Agreement") by and among Columbia Energy Group ("CEG"), Columbia Propane Corporation ("Columbia Propane"), Columbia Propane, L.P. ("CPLP"), CP Holdings, Inc. ("CPH"), AmeriGas Partners, AmeriGas OLP, and the General Partner. The acquired businesses comprised the seventh largest retail marketer of propane in the United States with annual sales of over 300 million gallons from locations in 29 states. The acquired businesses were principally conducted through Columbia Propane and its approximate 99% owned subsidiary, CPLP (referred to after the acquisition as "Eagle OLP"). AmeriGas

OLP acquired substantially all of the assets of Columbia Propane, including an indirect 1% general partner interest and an approximate 99% limited partnership interest in Eagle OLP.

The purchase price of the Columbia Propane Businesses consisted of \$201,750 in cash. In addition, AmeriGas OLP agreed to pay CEG for the amount of working capital, as defined, in excess of \$23,000. In April 2002, the Partnership's management and CEG agreed upon the amount of working capital acquired by AmeriGas OLP and AmeriGas OLP made an additional payment for working capital and other adjustments totaling \$736. The Columbia Purchase Agreement also provided for the purchase by CEG of limited partnership interests in AmeriGas OLP valued at \$50,000 for \$50,000 in cash, which interests were exchanged for 2,356,953 Common Units of AmeriGas Partners having an estimated fair

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Thousands of dollars, except per unit amounts)

value of \$54,422. Concurrently with the acquisition, AmeriGas Partners issued \$200,000 of 8.875% Senior Notes due May 2011, the net proceeds of which were contributed to AmeriGas OLP to finance the acquisition of the Columbia Propane Businesses, to fund related fees and expenses, and to repay debt outstanding under AmeriGas OLP's bank credit agreement.

The operating results of the Columbia Propane Businesses are included in our consolidated results from August 21, 2001.

The following table identifies the components of the purchase price of the Columbia Propane Businesses:

Cash paid	\$202,486
Cash received from sale of AmeriGas OLP limited partner interests Fair value of AmeriGas Partners' Common Units	(50,000)
issued in exchange for the AmeriGas OLP	
limited partner interests	54,422
Transaction costs and expenses	8,140
Involuntary employee termination benefits and	
relocation costs	2,618
Other liabilities and obligations incurred	1,031
Total	\$218,697
	=======

The purchase price of the Columbia Propane Businesses was allocated to the assets and liabilities acquired as follows:

Net current assets	\$ 16,741
Property, plant and equipment	182,761
Customer relationships and noncompete agreement	
(estimated useful life of 15 and 5 years,	
respectively)	19,878
Other assets and liabilities	(683)
Total	\$218,697
	=======

The following table presents unaudited pro forma income statement and per unit data for 2001 as if the acquisition of the Columbia Propane Businesses had occurred as of the beginning of the year:

		2001
Barrana	•	4 700 507
Revenues	\$	1,788,567
Income before accounting changes	\$	51,637
Net income	\$	64,131
Income per limited partner unit - basic and diluted:		
Income before accounting changes	\$	1.10
Net income	\$	1.36

The pro forma results of operations reflect the Columbia Propane Businesses' historical operating results after giving effect to adjustments directly attributable to the transaction that are expected to have a continuing impact. They are not adjusted for, among other things, the impact of normal weather conditions, operating synergies and anticipated cost savings. In our opinion, the unaudited pro forma results are not necessarily indicative of the actual results that would have occurred had the acquisition of the Columbia Propane Businesses occurred as of the beginning of the year presented or of future operating results under our management.

During 2001, in addition to the acquisition of the Columbia Propane Businesses, we acquired several other small propane distribution businesses for \$147 in cash and 25,993 Common Units. These transactions did not have a material effect on the Partnership's results of operations.

On October 1, 2003, AmeriGas OLP acquired substantially all of the retail propane distribution assets and business of Horizon Propane LLC ("Horizon Propane") for total cash consideration of \$31,044. In addition, AmeriGas OLP agreed to pay Horizon for the amount of working capital, as defined in the Asset Purchase Agreement, in excess of \$2,600. During its 2003 fiscal year, Horizon Propane sold over 30 million gallons of propane from ninety locations in twelve states.

NOTE 4 - CONVERSION OF SUBORDINATED UNITS

In December 2002, the General Partner determined that the cash-based performance and distribution requirements for the conversion of the then-remaining 9,891,072 Subordinated Units, all of which were held by the General Partner, had been met

in respect of the quarter ended September 30, 2002. As a result, the Subordinated Units were converted to an equivalent number of Common Units effective November 18, 2002. The conversion of the Subordinated Units did not result in an increase in the total number of AmeriGas Partners limited partner units outstanding.

NOTE 5 - QUARTERLY DISTRIBUTIONS OF AVAILABLE CASH

The Partnership makes distributions to its partners approximately 45 days after the end of each fiscal quarter in a total amount equal to its Available Cash for such quarter. Available Cash generally means:

- all cash on hand at the end of such quarter,
- 2. plus all additional cash on hand as of the date of determination resulting from borrowings after the end of such quarter,
- less the amount of cash reserves established by the General Partner in its reasonable discretion.

The General Partner may establish reserves for the proper conduct of the Partnership's business and for distributions during the next four quarters. In addition, certain of the Partnership's debt agreements require reserves be established for the payment of debt principal and interest.

Distributions of Available Cash are made 98% to limited partners and 2% to the General Partner. The Partnership may pay an incentive distribution if Available Cash exceeds the Minimum Quarterly Distribution of \$0.55 ("MQD") on all units.

NOTE 6 - DEBT

Long-term debt comprises the following at September 30:

	2003	2002
AmeriGas Partners Senior Notes: 8.875%, due May 2011 (including unamortized premium of \$ 6,426 and \$1,550, respectively, effective		
rate - 8.56%) 10%, due April 2006 (less unamortized discount of \$167 and \$223,	\$ 366,426	\$ 241,550
respectively, effective rate - 10.125%) 10.125%, due April 2007 AmeriGas OLP First Mortgage Notes: Series A, 9.34% - 11.71%, due April 2003 through April 2009 (including unamortized premium of \$6,581 and \$7,884, respectively,	59, 833 	59,777 85,000
effective rate - 8.91%) Series B, 10.07%, due April 2003 through April 2005 (including unamortized premium of \$ 1,082 and \$2,304, respectively, effective	166,581	167,884
rate - 8.74%) Series C, 8.83%, due April 2003	81,082	122,304
through April 2010 Series D, 7.11%, due March 2009 (including unamortized premium of \$1,890 and \$2,168, respectively,	96,250	110,000
effective rate - 6.52%) Series E, 8.50%, due July 2010 (including unamortized premium of \$137 and \$149,	71,890	72,168
respectively, effective rate - 8.47%) Other	80,137 5,087	80,149 6,952
Total long-term debt Less current maturities (including net unamortized premium of \$ 3,082	927, 286	945,784
and \$2,887, respectively)	(58,705)	(60,398)
Total long-term debt due after one year	\$ 868,581 =======	\$ 885,386 ======

Scheduled principal repayments of long-term debt for each of the next five fiscal years ending September 30 are as follows: 2004 - \$55,623; 2005 - \$55,512; 2006 - \$114,427; 2007 - \$54,105; 2008 - \$54,069.

AMERIGAS PARTNERS SENIOR NOTES. The 8.875% Senior Notes generally cannot be redeemed at our option prior to May 20, 2006. A redemption premium applies thereafter through May 19, 2009.

However, prior to May 20, 2004, AmeriGas Partners may use the proceeds of a public offering of Common Units to redeem up to 33% of the 8.875% Senior Notes at 108.875% plus accrued and unpaid interest. The 10% Senior Notes generally cannot be redeemed at our option prior to their maturity. AmeriGas Partners prepaid \$15,000 of 10.125% Senior Notes in November 2001 at a redemption price of 103.375% and prepaid the remaining \$85,000 of 10.125% Senior Notes in January 2003 at a redemption price of 102.25%, in each instance, including accrued interest. AmeriGas Partners recognized losses of \$3,023 and \$752 associated with these prepayments which amounts are reflected in "Loss on extinguishments of debt" in the 2003 and 2002 Consolidated Statements of Operations, respectively. AmeriGas Partners may, under certain circumstances following the disposition of assets or a change of control, be required to offer to prepay the Senior Notes.

AMERIGAS OLP FIRST MORTGAGE NOTES. AmeriGas OLP's First Mortgage Notes are collateralized by substantially all of its assets. The General Partner and Petrolane are co-obligors of the Series A, B, and C First Mortgage Notes, and the General Partner is co-obligor of the Series D and E First Mortgage Notes. AmeriGas OLP may prepay the First Mortgage Notes, in whole or in part. These prepayments include a make whole premium. Following the disposition of assets or a change of control, AmeriGas OLP may be required to offer to prepay the First Mortgage Notes, in whole or in part.

AMERIGAS OLP CREDIT AGREEMENT. AmeriGas OLP's Credit Agreement ("Credit Agreement") consists of (1) a Revolving Credit Facility and (2) an Acquisition Facility. AmeriGas OLP's obligations under the Credit Agreement are collateralized by substantially all of its assets. The General Partner and Petrolane are guarantors of amounts outstanding under the Credit Agreement.

Under the Revolving Credit Facility, AmeriGas OLP may borrow up to \$100,000 (including a \$100,000 sublimit for letters of credit) subject to restrictions in the AmeriGas Partners Senior Notes indentures (see "Restrictive Covenants" below). The Revolving Credit Facility may be used for working capital and general purposes of AmeriGas OLP. The Revolving Credit Facility expires on

October 15, 2006, but may be extended for additional one-year periods with the consent of the participating banks representing at least 80% of the commitments thereunder. There were no borrowings outstanding under Americas OLP's Revolving Credit Facility at September 30, 2003. Americas OLP had borrowings under the Revolving Credit Facility totaling \$10,000 at September 30, 2002, which we classify as bank loans. Issued and outstanding letters of credit, which reduce available borrowings under the Revolving Credit Facility, totaled \$33,363 and \$19,781 at September 30, 2003 and 2002, respectively.

The Acquisition Facility provides AmeriGas OLP with the ability to borrow up to \$75,000 to finance the purchase of propane businesses or propane business assets or, to the extent it is not so

used, may be used for working capital and general purposes. The Acquisition Facility operates as a revolving facility through October 15, 2006, at which time amounts then outstanding will be immediately due and payable. There were no amounts outstanding under the Acquisition Facility at September 30, 2003 and 2002.

The Revolving Credit Facility and the Acquisition Facility permit AmeriGas OLP to borrow at prevailing interest rates, including the base rate, defined as the higher of the Federal Funds rate plus 0.50% or the agent bank's prime rate (4.00% at September 30, 2003), or at a two-week, one-, two-, three-, or six-month Eurodollar Rate, as defined in the Credit Agreement, plus a margin. The margin on Eurodollar Rate borrowings (which ranges from 1.00% to 2.25%), and the Credit Agreement facility fee rate (which ranges from 0.25% to 0.50%) are dependent upon AmeriGas OLP's ratio of funded debt to earnings before interest expense, income taxes, depreciation and amortization ("EBITDA"), each as defined in the Credit Agreement.

GENERAL PARTNER FACILITY. AmeriGas OLP also has a Revolving Credit Agreement with the General Partner under which it may borrow up to \$20,000 for working capital and general purposes. This agreement is coterminous with, and generally comparable to, AmeriGas OLP's Revolving Credit Facility except that borrowings under the General Partner Facility are unsecured and subordinated to all senior debt of AmeriGas OLP. Interest rates on borrowings are based upon one-month offshore interbank offering rates. Facility fees are determined in the same manner as fees under the Revolving Credit Facility. UGI has agreed to contribute up to \$20,000 to the General Partner to fund such borrowings.

RESTRICTIVE COVENANTS. The Senior Notes of AmeriGas Partners restrict the ability of the Partnership to, among other things, incur additional indebtedness, make investments, incur liens, issue preferred interests, prepay subordinated indebtedness, and effect mergers, consolidations and sales of assets. Under the Senior Notes indentures, AmeriGas Partners is generally permitted to make cash distributions equal to available cash, as defined, as of the end of the immediately preceding quarter, if certain conditions are met. These conditions include:

- no event of default exists or would exist upon making such distributions and
- 2. the Partnership's consolidated fixed charge coverage ratio, as defined, is greater than 1.75-to-1.

If the ratio in item 2 above is less than or equal to 1.75-to-1, the Partnership may make cash distributions in a total amount not to exceed \$24,000 less the total amount of distributions made during the immediately preceding 16 fiscal quarters. At September 30, 2003, such ratio was 2.79-to-1.

The Credit Agreement and the First Mortgage Notes restrict the incurrence of additional indebtedness and also restrict certain liens, guarantees, investments, loans and advances, payments, mergers, consolidations, asset transfers, transactions with affiliates, sales of assets, acquisitions and other transactions. The Credit Agreement and First Mortgage Notes require the ratio of total indebtedness, as defined, to EBITDA, as defined (calculated on a rolling four-quarter basis or eight-quarter basis divided by two), to be less than or equal to 4.75-to-1 with respect to the Credit Agreement and 5.25-to-1 with respect to the First Mortgage Notes. In addition, the Credit Agreement requires that AmeriGas OLP maintain a ratio of EBITDA to interest expense, as defined, of at least 2.25-to-1 on a rolling four-quarter basis. Generally, as long as no default exists or would result, AmeriGas OLP is permitted to make cash distributions not more frequently than quarterly in an amount not to exceed available cash, as defined, for the immediately preceding calendar quarter. At September 30, 2003, the Partnership was in compliance with its financial covenants.

NOTE 7 - EMPLOYEE RETIREMENT PLANS

The General Partner sponsors a 401(k) savings plan for eligible employees. Participants in the savings plan may contribute a portion of their compensation on a before-tax basis. Generally, employee contributions are matched on a dollar-for-dollar (100%) basis up to 5% of eligible compensation. In conjunction with a short-term cost savings program, for the period March 1, 2002 to September 30, 2002, employee contributions were matched at a rate of 25% up to 5% of eligible compensation. The cost of benefits under our savings plan was \$5,555 in 2003, \$2,904 in 2002 and \$4,765 in 2001.

NOTE 8 - INVENTORIES

Inventories comprise the following at September 30:

	2003	2002
Propane gas Materials, supplies and other Appliances for sale	\$51,691 13,199 5,281	\$38,283 18,586 5,627
Total inventories	\$70,171 ======	\$62,496 ======

In addition to inventories on hand, we also enter into contracts to purchase propane to meet a portion of our supply requirements. Generally, these contracts are one- to three-year agreements subject to annual review and call for payment based on either fixed prices or market prices at date of delivery.

NOTE 9 - PROPERTY, PLANT AND EQUIPMENT Property, plant and equipment comprise the following at September 30:

	2003	2002
Land Buildings and improvements Transportation equipment Storage facilities Equipment, primarily cylinders and tanks Other	\$ 62,419 78,729 71,874 88,472 749,867 16,333	\$ 64,672 75,949 72,300 83,528 709,308 14,383
Gross property, plant and equipment Less accumulated depreciation and	1,067,694	1,020,140
amortization	(473,090)	(408,590)
Net property, plant and equipment	\$ 594,604 ======	\$ 611,550 ======

NOTE 10 - PARTNERS' CAPITAL AND INCENTIVE COMPENSATION PLANS

In accordance with the Partnership Agreement, the General Partner may, in its sole discretion, cause the Partnership to issue an unlimited number of additional Common Units and other equity securities of the Partnership ranking on a parity with the Common Units.

In June 2003, AmeriGas Partners sold 2,900,000 Common Units in an underwritten public offering at a public offering price of \$27.12 per unit. The net proceeds of the public offering totaling \$75,005 and the associated capital contributions from the General Partner totaling \$1,531 were contributed to AmeriGas OLP and used to reduce indebtedness under its bank credit agreement and for general partnership purposes.

In December 2001 and January 2002, AmeriGas Partners sold an aggregate 2,428,047 Common Units in conjunction with an underwritten public offering at a public offering price of \$21.50 per unit. The net proceeds of the public offering and the associated capital contributions from the General Partner totaling \$50,635 were contributed to AmeriGas OLP and used to reduce indebtedness under its bank credit agreement and for working capital. In October 2001, AmeriGas Partners closed the sale of 350,000 Common Units to the General Partner at the trade-date market price of \$19.81 per unit. The proceeds of this sale and the associated capital contributions from the General Partner totaling \$7,075 were contributed to AmeriGas OLP which used the proceeds to reduce borrowings under its bank credit agreement and for working capital. In October 2000, we sold 2,300,000 Common Units in conjunction with a public offering at a public offering price of \$18.44 per unit. The net proceeds from the Common Unit offering and the associated capital contributions from the General Partner totaling \$40,649 were used to reduce indebtedness under its bank credit agreement and for working capital purposes.

Under the AmeriGas Propane, Inc. 2000 Long-Term Incentive Plan ("2000 Incentive Plan"), the General Partner may grant to key employees the rights to receive a total of 500,000 Common Units, or cash equivalent to the fair market value of such Common Units, upon the achievement of performance goals. In addition, the 2000 Incentive Plan may provide for the crediting of Partnership distribution equivalents to participants' accounts. Distribution equivalents will be paid in cash and such payment may, at the participant's request, be deferred. The actual number of Common Units (or their cash equivalents) ultimately issued, and the actual amount of distribution equivalents paid, is dependent upon the achievement of performance goals. Generally, each grant, unless paid, will terminate when the participant ceases to be employed by the General Partner. We also have a nonexecutive Common Unit plan under which the General Partner may grant awards of up to a total of 200,000 Common Units to key employees who do not participate in the 2000 Incentive Plan. Generally, awards under the nonexecutive plan vest at the end of a three-year period and will be paid in Common Units and cash. During 2003, 2002 and 2001, the General Partner made awards under the 2000 Incentive Plan and the nonexecutive plan representing 112,500, 43,250 and 66,075 Common Units, respectively. We recorded compensation expense of \$2,151 in 2003, \$1,018 in 2002 and \$497 in 2001 relating to these plans. At September 30, 2003, and 2002, awards representing 209,336 and 105,825 Common Units, respectively, were outstanding.

NOTE 11 - COMMITMENTS AND CONTINGENCIES

We lease various buildings and other facilities and transportation, computer and office equipment under operating leases. Certain of the leases contain renewal and purchase options and also contain escalation clauses. Our aggregate rental expense for such leases was \$43,179 in 2003, \$41,974 in 2002 and \$32,709 in 2001.

 $\label{thm:minimum future payments under noncancelable operating leases are as follows: \\$

2005					32,158
2006					27,196
2007					22,736
2008					18,613
Thereafter					39,620
otal minimum	operating	lease	payments	\$	177,526
				==	

The Partnership enters into fixed price contracts with suppliers to purchase a portion of its propane supply requirements. These contracts generally have terms of less than one year. As of September 30, 2003, contractual obligations under these contracts totaled \$16,749.

The Partnership also enters into contracts to purchase propane to meet additional supply requirements. Generally, these contracts are one- to three-year agreements subject to annual review and call for payment based on either fixed prices or market prices at the date of delivery.

The Partnership has succeeded to certain lease guarantee obligations of Petrolane relating to Petrolane's divestiture of non-propane operations before its 1989 acquisition by QFB Partners. Future lease payments under these leases total approximately \$15,000 at September 30, 2003. The leases expire through 2010 and some of them are currently in default. The Partnership has succeeded to the indemnity agreement of Petrolane by which Texas Eastern Corporation ("Texas Eastern"), a prior owner of Petrolane, agreed to indemnify Petrolane against any liabilities arising out of the conduct of businesses that do not relate to, and are not a part of, the propane business, including lease guarantees. In December 1999, Texas Eastern filed for dissolution under the Delaware General Corporation Law. In May 2001, Petrolane filed a declaratory judgment action in the Delaware Chancery Court seeking confirmation of Texas Eastern's indemnification obligations and judicial supervision of Texas Eastern's dissolution to ensure that its indemnification obligations to Petrolane are paid or adequately provided for in accordance with law. Those proceedings are pending. Pursuant to a Liquidation and Winding Up Agreement dated September 17, 2002, PanEnergy Corporation ("PanEnergy"), Texas Eastern's sole stockholder, assumed all of Texas Eastern's liabilities as of December 20, 2002, to the extent of the value of Texas Eastern's assets transferred to PanEnergy as of that date (which was estimated to exceed \$94,000), and to the extent that such liabilities arise within ten years from Texas Eastern's date of dissolution. Notwithstanding the dissolution proceeding, and based on Texas Eastern previously having satisfied directly defaulted lease obligations without the Partnership's having to honor its guarantee, we believe that the probability that the Partnership will be required to directly satisfy the lease obligations subject to the indemnification agreement is remote.

On August 21, 2001, AmeriGas Partners, through AmeriGas OLP, acquired the propane distribution businesses of Columbia Energy Group (the "2001 Acquisition") pursuant to the terms of a purchase agreement (the "2001 Acquisition Agreement") by and among Columbia Energy Group ("CEG"), Columbia Propane Corporation ("Columbia Propane"), Columbia Propane, L.P. ("CPLP"), CP Holdings, Inc. ("CPH," and together with Columbia Propane and CPLP, the "Company Parties"), AmeriGas Partners, AmeriGas OLP and the General Partner (together with AmeriGas Partners and AmeriGas OLP, the "Buyer Parties"). As a result of the 2001 Acquisition, AmeriGas OLP acquired all of the stock of Columbia Propane and CPH and substantially all of the partnership interests of CPLP. Under the terms of an earlier acquisition agreement (the "1999 Acquisition Agreement"), the Company Parties agreed to indemnify the former general partners of National Propane Partners, L.P. (a predecessor company of the Columbia Propane businesses) and an affiliate (collectively, "National General Partners") against certain income tax and other losses that they may sustain as a result of the 1999 acquisition by CPLP of National Propane Partners, L.P. (the "1999 Acquisition") or the operation of the business after the 1999 Acquisition ("National Claims"). At September 30, 2003, the potential amount payable under this indemnity by the Company Parties was approximately \$65,000. These indemnity obligations will expire on the date that CPH acquires the remaining outstanding partnership interest of CPLP, which is expected to occur on or after July 19, 2009.

Under the terms of the 2001 Acquisition Agreement, CEG agreed to indemnify the Buyer Parties and the Company Parties against any losses that they sustain under the 1999 Acquisition Agreement and related agreements ("Losses"), including National Claims, to the extent such claims are based on acts or omissions of CEG or the Company Parties prior to the 2001 Acquisition. The Buyer Parties agreed to indemnify CEG against Losses, including National Claims, to the extent such claims are based on acts or omissions of the Buyer Parties or the Company Parties after the 2001 Acquisition. CEG and the Buyer Parties have agreed to apportion certain losses resulting from National Claims to the extent such losses result from the 2001 Acquisition itself.

We also have other contingent liabilities, pending claims and legal actions arising in the normal course of our business. We cannot predict with certainty the final results of these matters.

However, it is reasonably possible that some of them could be resolved unfavorably to us. Although management currently believes, after consultation with counsel, that damages or settlements, if any, recovered by the plaintiffs in such claims or actions will not have a material adverse effect on our financial position, damages or settlements could be material to our operating results or cash flows in future periods depending on the nature and timing of future developments with respect to these matters and the amounts of future operating results and cash flows.

NOTE 12 - RELATED PARTY TRANSACTIONS

Pursuant to the Partnership Agreement and a Management Services Agreement among AEH, the general partner of Eagle OLP, and the General Partner, the General Partner is entitled to reimbursement for all direct and indirect expenses incurred or payments it makes on behalf of the Partnership. These costs, which totaled \$284,266 in 2003, \$262,398 in 2002 and \$208,910 in 2001, include employee compensation and benefit expenses of employees of the General Partner and general and administrative expenses.

UGI provides certain financial and administrative services to the General Partner. UGI bills the General Partner for these direct and indirect corporate expenses and the General Partner is reimbursed by the Partnership for these expenses. Such corporate expenses totaled \$8,295 in 2003, \$6,341 in 2002 and \$5,276 in 2001. In addition, UGI and certain of its subsidiaries provide office space and automobile liability insurance to the Partnership. These expenses totaled \$1,683 in 2003, \$1,502 in 2002 and \$1,348 in 2001.

Subsequent to the Columbia Propane acquisition, the Partnership purchases propane on behalf of Atlantic Energy. Atlantic Energy reimburses AmeriGas OLP for its purchases plus interest as Atlantic Energy sells such propane to third parties or to AmeriGas OLP itself. The total dollar value of propane purchased on behalf of Atlantic Energy was \$17,225 and \$11,370 in 2003 and 2002, respectively. Purchases of propane by AmeriGas OLP from Atlantic Energy during 2003 and 2002 totaled \$23,940 and \$12,131, respectively.

The General Partner also provides other services to Atlantic Energy including accounting, insurance and other administrative services and is reimbursed for the related costs. Such costs were not material during 2003 or 2002. In addition, AmeriGas OLP enters into product cost hedging contracts on behalf of Atlantic Energy. When these contracts are settled, AmeriGas OLP is reimbursed the cost of any losses, or distributes the proceeds of any gains, to Atlantic Energy.

Amounts due from Atlantic Energy at September 30, 2003 and 2002 totaled \$2,042 and \$5,243, respectively, which amounts are included in accounts receivable - related parties in the Consolidated Balance Sheets.

NOTE 13 - OTHER CURRENT LIABILITIES Other current liabilities comprise the following at September 30:

	2003	2002
Self-insured property and casualty liability	\$12,672	\$11,658
Taxes other than income taxes Fair value of derivative instruments Propane exchange liability Deferred tank fee revenue Other	4,994 2,970 8,689 7,563 3,108	5,344 3,330 4,627 6,755 8,759
Total other current liabilities	\$39,996 ======	\$40,473 ======

NOTE 14 - FINANCIAL INSTRUMENTS

In accordance with its propane price risk management policy, the Partnership uses derivative instruments, including price swap and option contracts and contracts for the forward sale of propane, to manage the cost of a portion of its forecasted purchases of propane and to manage market risk associated with propane storage inventories. These derivative instruments have been designated by the Partnership as cash flow or fair value hedges under SFAS 133. The fair values of these derivative instruments are affected by changes in propane product prices. In addition to these derivative instruments, the Partnership may also enter into contracts for the forward purchase of propane as well as fixed price supply agreements to manage propane market price risk. These contracts generally qualify for the normal purchases and normal sales exception of SFAS 133 and therefore are not adjusted to fair value.

On occasion, we enter into interest rate protection agreements ("IRPAS") designed to manage interest rate risk associated with planned issuances of fixed-rate long-term debt. We designate these IRPAs as cash flow hedges. Gains or losses on IRPAs are included in other comprehensive income and are reclassified to interest expense as the interest expense on the associated debt issue affects earnings.

During the years ended September 30, 2003, 2002 and 2001, the net loss recognized in earnings representing cash flow hedge ineffectiveness was \$3,146, \$2,123 and \$326, respectively. Gains and losses included in accumulated other comprehensive income at September 30, 2003 relating to cash flow hedges will be reclassified into (1) cost of sales when the forecasted purchase of propane subject to the hedges impacts net income and (2) interest expense when interest on anticipated issuances of fixed-rate long-term debt is reflected in net income. Included in accumulated other comprehensive income at September 30, 2003 are net losses of approximately \$2,271 from IRPAs associated with forecasted issuances of ten-year debt generally anticipated to occur during the next two years. The amount of net loss which is expected to be reclassified into net

other comprehensive income at September 30, 2003 of \$2,046 is principally associated with future purchases of propane generally anticipated to occur during the next twelve months. The actual amount of gains or losses on unsettled derivative instruments that ultimately is reclassified into net income will depend upon the value of such derivative contracts when settled. The fair value of derivative instruments is included in other current assets, other current liabilities and other noncurrent liabilities in the Consolidated Balance Sheets.

The carrying amounts of financial instruments included in current assets and current liabilities (excluding unsettled derivative instruments and current maturities of long-term debt) approximate their fair values because of their short-term nature. The carrying amounts and estimated fair values of our remaining financial instruments (including unsettled derivative instruments) at September 30 are as follows:

	Carrying Amount	Estimated Fair Value		
2003: Propane swap and option contracts Interest rate protection agreements Long-term debt	\$ (637) (188) 927,286	\$ (637) (188) 1,024,717		
2002: Propane swap, option and forward sales contracts	\$ 9,771	\$ 9,771		
Interest rate protection agreements Long-term debt	(2,758) 945,784	(2,758) 982,869		

We estimate the fair value of long-term debt by using current market prices and by discounting future cash flows using rates available for similar type debt. Fair values of derivative instruments reflect the estimated amounts that we would receive or (pay) to terminate the contracts at the reporting date based upon quoted market prices of comparable contracts.

We have financial instruments such as short-term investments and trade accounts receivable which could expose us to concentrations of credit risk. We limit our credit risk from short-term investments by investing only in investment-grade commercial paper and U.S. Government securities. The credit risk from trade accounts receivable is limited because we have a large customer base which extends across many different U.S. markets. We attempt to minimize our credit risk associated with our derivative financial instruments through the application of credit policies.

NOTE 15 - OTHER INCOME, NET Other income, net, comprises the following:

	2003	2002	2001
Gain on sales of fixed assets	\$(2,173)	\$(1,719)	\$(2,413)
Finance charges	(3,942)	(2,169)	(2,435)
Derivative loss		2,123	326
Other	(2,845)	(2,638)	(1,632)
Total other income, net	\$(8,960)	\$(4,403)	\$(6,154)
	======	======	======

Derivative loss of \$3,146 is included in "cost of sales - propane" in the 2003 Consolidated Statement of Operations.

NOTE 16 - MANAGEMENT REALIGNMENT

In June 2003, the General Partner announced a plan to realign its management structure. Pursuant to the plan, the Partnership closed its seven regional offices located across the country and relocated four regional vice presidents to its Valley Forge, Pennsylvania headquarters. In addition, the Partnership reconfigured its eighty geographically-based market areas into approximately sixty market areas.

The new management structure is expected to further streamline business processes, eliminate duplication and reduce overhead expenses. As a result of the management realignment, the Partnership incurred charges for severance, lease termination and other expenses totaling \$3,756 which are reflected as operating and administrative expenses in the 2003 Consolidated Statement of Operations. As of September 30, 2003, \$736 of costs associated with the management realignment, principally comprising employee severance and lease termination costs, are included in employee compensation and benefits accrued and other current liabilities in the Consolidated Balance Sheet. Additional future costs to be incurred in conjunction with the management realignment are not expected to have a material effect on our results of operations.

NOTE 17 - QUARTERLY DATA (UNAUDITED)

The following unaudited quarterly data includes all adjustments (consisting only of normal recurring adjustments) which we consider necessary for a fair presentation. Our quarterly results fluctuate because of the seasonal nature of our propane business.

	December 31,		March 31,			June 30,				September 30,			30,			
		2002		2001		2003		2002		2003		2002		2003		2002
Revenues Operating income (loss)		445,031		371,385		625,546 115,547		460,122		287,136		254,469	- :	270,711		221,904
Net income (loss)		64,414 40,912	\$	42,132 18,397		89,876		104,644 81,989		(6,467) (27,414)	\$ \$		\$ \$	(9,504) (31,416)		(13,853) (35,075)
Net income (loss) per limited partner unit - basic and																
diluted	\$	0.82	\$	0.38	\$	1.80	\$	1.64	\$	(0.54)	\$	(0.20)	\$	(0.59)	\$	(0.70)
			-		-		-		-							

NOTE 18 - CHANGES IN ACCOUNTING

TANK FEE REVENUE RECOGNITION. In order to apply the guidance of SAB 101, effective October 1, 2000, we changed our method of accounting for annually billed nonrefundable tank fees. Prior to the change in accounting, nonrefundable tank fees for installed Partnership-owned tanks were recorded as revenue when billed. Under the new accounting method, revenues from such fees are being recorded on a straight-line basis over one year. As a result of this change in accounting, on October 1, 2000, we recorded a charge of \$5,984 representing the cumulative effect of the change in accounting on prior years. The change in accounting for nonrefundable tank fees did not have a material impact on reported revenues in 2003, 2002 and 2001.

ACCOUNTING FOR TANK INSTALLATION COSTS. Effective October 1, 2000, we changed our method of accounting for tank installation costs which are not billed to customers. Prior to the change in accounting, costs to install Partnership-owned tanks at customer locations were expensed as incurred. Under the new accounting method, all such costs, net of amounts billed to customers, are capitalized in property, plant and equipment and amortized over the estimated period of benefit not exceeding ten years. We believe that this accounting method better matches the costs of installing Partnership-owned tanks with the periods benefited. As a result of this change in accounting, on October 1, 2000, we recorded an increase of \$19,214 in net income representing the cumulative effect of the change in accounting on prior years. The change in accounting for tank installation costs did not have a material effect on 2001 net income.

CUMULATIVE EFFECT OF ACCOUNTING CHANGES AND PRO FORMA DISCLOSURE. The cumulative effect and related per limited partner unit amounts reflected on the 2001 Consolidated Statement of Operations resulting from the above changes in accounting principles, as well as the cumulative effect resulting from the adoption of SFAS 133 (see Note 2), comprise the following:

	Cumulative Effect 	Cumulative Effect Per Limited Partner Unit
Tank fees Tank installation costs SFAS 133	\$ (5,984) 19,214 (736)	\$(0.13) 0.43 (0.02)
Total	\$ 12,494 ======	\$ 0.28 =====

REPORT OF INDEPENDENT AUDITORS

TO THE PARTNERS OF AMERIGAS PARTNERS, L.P. AND THE BOARD OF DIRECTORS OF AMERIGAS PROPANE, INC.:

In our opinion, the accompanying consolidated balance sheets and the related consolidated statements of operations, of partners' capital and of cash flows present fairly, in all material respects, the financial position of AmeriGas Partners, L.P. and its subsidiaries at September 30, 2003 and 2002, and the results of their operations and their cash flows for each of the two years in the period ended September 30, 2003 in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the management of AmeriGas Propane, Inc.; our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion. The consolidated financial statements of AmeriGas Partners, L.P. and its subsidiaries for the year ended September 30, 2001, prior to the revisions discussed in Note 2, were audited by other independent auditors who have ceased operations. Those independent auditors expressed an unqualified opinion on those financial statements in their report dated November 16, 2001.

As discussed in Note 2 to the consolidated financial statements, the Partnership adopted Statement of Financial Accounting Standards No. 142, "Goodwill and Other Intangible Assets" in fiscal 2002.

As discussed above, the consolidated financial statements of AmeriGas Partners, L.P. and its subsidiaries for the year ended September 30, 2001, were audited by other independent auditors who have ceased operations. As described in Note 2, these financial statements have also been revised to include the transitional disclosures required by Statement of Financial Accounting Standards No. 142, "Goodwill and Other Intangible Assets" which was adopted by the Partnership as of October 1, 2001. We audited the transitional disclosures described in Note 2. In our opinion, the transitional disclosures for 2001 in Note 2 are appropriate. However, we were not engaged to audit, review or apply procedures to the 2001 consolidated financial statements of the Partnership other than with respect to such adjustments and, accordingly, we do not express an opinion or any other form of assurance on the 2001 consolidated financial statements taken as a whole.

/s/ PricewaterhouseCoopers LLP

Philadelphia, Pennsylvania November 17, 2003

THIS REPORT IS A COPY OF THE PREVIOUSLY ISSUED ACCOUNTANT'S REPORT OF ARTHUR ANDERSEN LLP AND HAS NOT BEEN REISSUED BY ARTHUR ANDERSEN LLP.

REPORT OF INDEPENDENT PUBLIC ACCOUNTANTS

TO THE PARTNERS OF AMERIGAS PARTNERS, L.P. AND THE BOARD OF DIRECTORS OF AMERIGAS PROPANE, INC.:

We have audited the accompanying consolidated balance sheets of AmeriGas Partners, L.P. and subsidiaries as of September 30, 2001 and 2000, and the related consolidated statements of operations, partners' capital and cash flows for each of the three years in the period ended September 30, 2001. These financial statements are the responsibility of the management of AmeriGas Propane, Inc. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of AmeriGas Partners, L.P. and subsidiaries as of September 30, 2001 and 2000, and the results of their operations and their cash flows for each of the three years in the period ended September 30, 2001, in conformity with accounting principles generally accepted in the United States.

As explained in Notes 2 and 4 to the financial statements, effective

October 1, 2000, the Partnership changed its methods of accounting for tank installation costs and nonrefundable tank fees and also adopted the provisions of SFAS No. 133.

/s/ Arthur Andersen LLP

Philadelphia, Pennsylvania November 16, 2001

GENERAL PARTNER'S REPORT

The Partnership's consolidated financial statements and other financial information contained in this Annual Report are prepared by the management of the General Partner, Americas Propane, Inc., which is responsible for their fairness, integrity and objectivity. The consolidated financial statements and related information were prepared in accordance with accounting principles generally accepted in the United States of America and include amounts that are based on management's best judgments and estimates.

The General Partner maintains a system of internal controls. Management of the General Partner believes the system provides reasonable, but not absolute, assurance that assets are safeguarded and that transactions are executed in accordance with management's authorization and are properly recorded to permit the preparation of reliable financial information. There are limits in all systems of internal control, based on the recognition that the cost of the system should not exceed the benefits to be derived. We believe that the internal control system is cost effective and provides reasonable assurance that material errors or irregularities will be prevented or detected within a timely period. The internal control system and compliance therewith are monitored by UGI Corporation's internal audit staff.

The Audit Committee of the Board of Directors of the General Partner is composed of three members, none of whom is an employee of the General Partner. This Committee is responsible for overseeing the financial reporting process and the adequacy of controls, and for monitoring the independence of the Partnership's independent accountants and the performance of the independent accountants and internal audit staff. The Committee appoints the independent accountants to conduct the annual audit of the Partnership's consolidated financial statements. The Committee is also responsible for maintaining direct channels of communication between the Board of Directors and both the independent accountants and internal auditors.

The independent accountants perform certain procedures, including an evaluation of internal controls to the extent required by auditing standards generally accepted in the United States of America, in order to express an opinion on the consolidated financial statements and to obtain reasonable assurance that such financial statements are free of material misstatement.

/s/ Martha B. Lindsay
-----Martha B. Lindsay
Chief Financial Officer

/s/ Richard R. Eynon
Richard R. Eynon
Chief Accounting Officer

CODE OF ETHICS

FOR

CHIEF EXECUTIVE OFFICER AND SENIOR FINANCIAL OFFICERS

0F

AMERIGAS PROPANE, INC., THE GENERAL PARTNER OF AMERIGAS PARTNERS, L.P. The reputation for integrity of AmeriGas Propane, Inc. (the "Company"), the General Partner of AmeriGas Partners, L.P., is a valuable asset that is vital to the Company's success. To enhance the Company's ability to maintain its reputation for integrity, the Board of Directors of the Company (the "Board") has adopted this Code of Ethics, which applies to the Company's Chief Executive Officer (the "CEO"), Chief Financial Officer and Principal Accounting Officer (together the "Senior Financial Officers"). This Code of Ethics is designed to give the CEO and Senior Financial Officers a broad understanding of the conduct that we expect from them to protect and enhance our reputation for integrity.

Each employee of the Company is responsible for conducting his or her duties in a manner that demonstrates a commitment to the highest standards of integrity. This integrity is critical to ensure that the Company's business is conducted in an honest and ethical manner and in compliance with the law. The purpose of this Code is to focus the CEO and Senior Financial Officers on areas of ethical risk, help them to recognize and deal with ethical issues, explain the mechanisms available to them to report unethical conduct and foster a culture of honesty and accountability.

Implementation and Oversight of This Code

The Board, acting through its Audit Committee, is ultimately responsible for the implementation of this Code. The Board has designated the Vice President and General Counsel to be the compliance officer (the "Compliance Officer") for the implementation and administration of this Code. Statements in this Code to the effect that certain actions may be taken only with the "Company's approval" mean that the Audit Committee of the Board or the Board of Directors acting through its independent members must give prior written approval before the proposed action may be undertaken. You should feel free to direct questions to the Compliance Officer.

You should read this Code in conjunction with the Company's Code of Business Conduct and Ethics for Directors, Officers and Employees. For simplicity sake, we will refer to both codes together as the "Code." We will ask you to certify on an annual basis that you are in compliance with the Code.

Requests for Waiver of Any Provision of this Code

You must submit any requests for a waiver of any provision of this Code in writing to the Compliance Officer for review. Only the independent members of the Board or the Audit Committee of the Board have the authority to waive any provision of this Code. If a waiver of any provision of this Code is granted to you, the Company must publicly disclose the nature of the granted waiver, including any implicit waiver, your name, the date of the waiver and any other disclosures as and to the extent required by any SEC rule or applicable stock exchange listing standard.

Compliance with Law and Regulations

A variety of laws apply to the Company and its operations, and some laws carry criminal penalties. These laws include, but are not limited to, federal environmental, securities, antitrust and occupational safety laws, and federal and state laws governing the Company's status as a public company. Examples of criminal violations of the law include making false or misleading disclosures in documents filed with the Securities and Exchange Commission (the "SEC"); trading on inside information; stealing, embezzling or misapplying the Company's funds; making a payment for an expressed purpose on the Company's behalf to an individual who intends to use it for a different purpose; or making payments whether from your funds or the Company's funds, of cash or other items of value that are intended improperly to influence the judgment or actions of political candidates, government officials or businesspersons in connection with any of the Company's activities. The Company will investigate, address and report, as appropriate, violations of law. It is your responsibility to comply with the laws, rules and regulations applicable to you personally as a senior executive of the Company. You cannot delegate that responsibility to another person or to the Company.

Avoiding Actual, Potential or Apparent Conflicts of Interest

The Company requires you to conduct your outside associations and personal business, financial and other relationships in a manner that will avoid any actual, potential or apparent conflict of interest between you and the Company. The term "outside association" refers to any affiliation, association, interest or employment that you have with an entity other than with the Company. It is impractical to set forth rules that cover all situations in which a conflict of interest may arise. The basic factor in all conflict of interest situations is, however, the division of loyalty or the appearance of a division of loyalty, between the Company's or partnerships' best interests and your interests. Guidelines with respect to several of the more sensitive areas in which actual, potential or apparent conflicts of interest are likely to occur are set forth below. We emphasize, however, that the following is not an exhaustive list of problem areas, but rather a guide in applying the Company's basic conflict of interest policy to any situation.

Business Relationships

You may have a conflict of interest if you, a member of your immediate family or your business or financial partner owns or has a substantial direct or indirect investment in an entity with which the Company has or is likely to have a business relationship or with which the Company competes. Investments in small amounts of the stocks or bonds of a publicly held company should not alone give rise to any conflict of interest. The question of when an investment may become so substantial as to possibly affect or appear to affect your judgment is largely dependent on the particular circumstances and must be considered on a case-by-case basis.

A conflict of interest may also arise when you, a member of your immediate family or your business or financial partner holds a position as director, officer, employee or partner of or consultant, broker, finder or intermediary with an entity with which the Company has or is likely to have a business relationship or with which the Company competes or is likely to compete. In addition, a conflict of interest may arise if you, a member of your immediate family or your business or financial partner incurs significant indebtedness to an entity whose business may be affected by your actions on behalf of the Company.

Any associations, interests and business or financial relationships that you have that might cause you to act in ways that are not in the best interests of the Company, or that create the appearance of divided loyalties, will be permitted only with the Company's approval. In some circumstances, a relationship will only be permitted if the proposed transaction is competitive and/or fairly bargained for. Notwithstanding the foregoing, a transaction between the Company and any of your outside associations will be permitted if it is first reported, reviewed and approved in the manner prescribed by the Board or Audit Committee and set forth below.

Acceptance of Gifts

You may not, without the Company's approval, accept, either directly or indirectly, gifts, favors, entertainment or anything else of more than nominal value from persons or entities with which the Company has or is likely to have a business relationship. Payment of expenses for attendance at appropriate business meals and conferences will not ordinarily be considered the receipt of a gratuity under this section. (The Company's Code of Business Conduct and Ethics for Directors, Officers and Employees provides some examples of other permissible gifts and should be consulted for a discussion on this subject.)

Outside Activities/Employment

Unless expressly authorized by the Company, any outside association, including activities with other entities, should not encroach on the time and attention you are expected to devote to your Company duties and responsibilities, adversely affect the quality or quantity of your work product for the Company or entail your use of any Company assets, including its real and personal property, or imply (without the Company's approval) the Company's sponsorship or support. In addition, under no circumstances are you permitted to compete with the Company or take for yourself, your family members or any business or financial partner any business or financial opportunity belonging to the Company that you discover or that is made available to you by virtue of your position with the Company.

Civic/Political Activities

The Company supports your participation in civic, charitable and political activities so long as such participation does not encroach on the time and attention that you are expected to devote to Company duties and responsibilities. Unless the Company expressly sanctions the activity, you are to conduct any such activities in a manner that does not involve the Company or its assets or create an appearance of Company involvement or endorsement.

Reporting Procedure

You must report promptly to the Compliance Officer the existence of any outside association, interest, relationship or activity, as it arises, that actually, potentially or apparently involves a conflict of interest. Failure to report such relationships, activities and interests will be a ground for disciplinary action. The Compliance Officer will notify the Board or Audit Committee of any actual, potential or apparent conflict of interest. You must cooperate fully in the review process by providing all information that the Compliance Officer or the Audit Committee deems necessary to its review.

You must sign annually a statement reflecting your continuing awareness and understanding of this Code, including its conflicts of interest policy. At the same time, you must report either the absence or presence of actual, potential or apparent conflicts of interest, including such conflicts involving others within the Company of which you are aware.

All interests, relationships or participation in transactions disclosed by any employee in accordance with this policy shall be held in confidence unless disclosure is required by law or applicable listing standards or is in the best interests of the Company.

Resolution of Conflicts

In all cases, actual, potential or apparent conflicts of interest must be handled in an ethical manner; in particular, they must be fully disclosed and considered prior to being resolved. The Compliance Officer or, where appropriate, the Audit Committee will handle all questions of actual, potential or apparent conflicts of interest that involve you. Actual, potential or apparent conflicts of interest may be approached in the following ways, among others:

- In the case of an offer of a gift, the appropriate resolution may be for you to accept or reject the gift.
- Where the nature of the conflict of interest is such that you are unable to disclose the details thereof without breaching other confidences, you may, upon notice to the Audit Committee, remove yourself from all meetings, discussions and actions at which such conflict of interest is considered.
- Where you can disclose the details of the conflict of interest, the Audit Committee may determine the proper action on its own or in consultation with the Board.
- Any outside association, interest, relationship or participation in a transaction that is fully disclosed in writing to, and is approved in writing by, the Audit Committee or the Board shall not be deemed to involve a conflict of interest for purposes of this Code.
- When it is concluded that there is an actual, apparent or potential conflict of interest, the Audit Committee or the Board may suspend you from all or some of your duties or

require you to perform other duties with the Company for such period of time as is deemed appropriate or may request that you resign from your position with the Company.

 In the event that the conflict of interest involves an outside association, the Company may permanently cease doing business with the association.

Full, Fair, Accurate and Timely Disclosures by the Company to the Public

Because you participate, directly or indirectly, in the preparation of the financial and other disclosures that the Company makes to the public, including disclosures in its filings with the SEC or by press releases, you must, in addition to complying with all applicable laws, rules and regulations, follow these guidelines:

- Ensure full, fair, timely, accurate and understandable disclosure in the Company's filings with the SEC to the best of your ability.
- Through leadership, including communication, make sure that employees of the Company understand the Company's obligations to the public and under the law with respect to its disclosures.
- Encourage employees to raise questions and concerns regarding the Company's public disclosures and ensure that such questions and concerns are appropriately addressed.
- Provide the Company's directors, employees, consultants and advisors involved in the preparation of the Company's disclosures to the public with information that is accurate, complete, objective, relevant, timely and understandable.
- Act in good faith, responsibly, and with due care, competence and diligence, without misrepresenting material facts or allowing your independent judgment to be compromised by others.
- Ensure that the recording of entries in the Company's or partnerships' books and records is accurate to the best of your knowledge.
- Comply with the Company's disclosure controls and procedures and internal controls and procedures for financial reporting.

Prompt Internal Reporting of Violations of This Code

If you violate or think you have violated any provision of this Code, or if you observe, learn of, or, in good faith, suspect that another person subject to this Code has violated any of its provisions, you must immediately report the actual or suspected violation to the Compliance Officer who will promptly notify the Chairman of the Audit Committee of the Board of such violation. If you report an actual or suspected violation in good faith, the Company may not

subject you to retaliation of any kind. A violation of the requirement to report violations or to cooperate in a Code investigation may result in disciplinary action.

Accountability for Complying With This Code

Reported violations of this Code will be investigated and addressed promptly and the identity of the reporting person treated confidentially to the extent possible. We strive to impose discipline for each Code violation that fits the nature and particular facts of the violation. We generally will issue warnings or letters of reprimand for less significant, first-time violations. Violations of a more serious nature may result in suspension without pay, demotion, loss or reduction of bonus or option awards, or any combination of such disciplinary violations. Termination of employment generally is reserved for violations amounting to a breach of trust, such as fraud or theft, or for cases where a person has engaged in multiple violations.

The SEC treats violations of the Code that go unaddressed as implicit waivers of the Code. Accordingly, any violation that is discovered and not addressed will have to be disclosed in accordance with the rules and regulations of the SEC or applicable stock exchange listing standards. In such cases, the SEC's rules currently require disclosure of the nature of any violation, the date of the violation and the name of the person who committed the violation. Such disclosure could be harmful to the Company. Moreover, depending on the nature of the violation, the violator may be dismissed or her or his duties and responsibilities with the Company changed significantly.

EXHIBIT 21

SUBSIDIARIES OF AMERIGAS PARTNERS, L.P.

SUBSIDIARY OWNERSHIP INCORPORATION AmeriGas Finance Corp. 100% DE AmeriGas Propane, L.P. 98.99% DE AmeriGas Eagle Propane, L.P. 99% DE AmeriGas Eagle Parts Eagle Parts & Service, Inc. 100% PA AmeriGas Propane Parts & Service, Inc. 100% PA AmeriGas Eagle Propane, Inc. 100% DE AmeriGas Eagle Holdings, Inc. 100% DE Active Propane of Wisconsin LLC 100% DE AmeriGas Eagle Finance Corp. 100% DE AP Eagle

Finance Corp. 100% DE

STATE OF

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in the Registration Statements on Form S-3 (Nos. 333-45902, 333-83942 and 333-110425) and Form S-8 (No. 333-104939) of AmeriGas Partners, L.P. of our report dated November 17, 2003 relating to the financial statements and financial statement schedules, and our reports dated November 17, 2003, relating to AmeriGas Finance Corp., AmeriGas Eagle Finance Corp., and AP Eagle Finance Corp., which appear in this Form 10-K.

PricewaterhouseCoopers LLP

Philadelphia, Pennsylvania December 23, 2003

CERTIFICATIONS

- I, Eugene V. N. Bissell, certify that:
- I have reviewed this annual report on Form 10-K of AmeriGas Partners, L.P., AmeriGas Finance Corp., AmeriGas Eagle Finance Corp. and AP Eagle Finance Corp.;
- Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this annual report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(d)) for the registrants and we have:
 - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including their consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - (b) evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect the registrants' internal control over financial reporting.
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of each registrant's board of directors (or persons performing the equivalent functions):
 - (a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and
 - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants' internal controls over financial reporting.

Date: December 16, 2003

Eugene V. N. Bissell

Fugene V. N. Bissell

President and Chief Executive Officer

AmeriGas Propane, Inc.

AmeriGas Finance Corp.

AmeriGas Eagle Finance Corp.

AP Eagle Finance Corp.

- I, Martha B. Lindsay, certify that:
- I have reviewed this annual report on Form 10-K of AmeriGas Partners, L.P., AmeriGas Finance Corp., AmeriGas Eagle Finance Corp. and AP Eagle Finance Corp.;
- 2. Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this annual report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(d)) for the registrants and we have:
 - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including their consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
 - (b) evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect the registrants' internal control over financial reporting.
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of each registrant's board of directors (or persons performing the equivalent functions):
 - (a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and
 - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants' internal controls over financial reporting.

Date: December 16, 2003

Martha B. Lindsay

Martha B. Lindsay Vice President - Finance and Chief Financial Officer AmeriGas Propane, Inc. AmeriGas Finance Corp. AmeriGas Eagle Finance Corp.

AP Eagle Finance Corp.

CERTIFICATION BY THE CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER RELATING TO A PERIODIC REPORT CONTAINING FINANCIAL STATEMENTS

I, Eugene V. N. Bissell, Chief Executive Officer, and I, Martha B. Lindsay, Chief Financial Officer, of each of AmeriGas Propane, Inc., a Pennsylvania corporation and the General Partner of AmeriGas Partners, L.P., a Delaware limited partnership (the "Partnership"), AmeriGas Finance Corp. ("Finance Corp."), AmeriGas Eagle Finance Corp. ("Eagle Finance Corp.") and AP Eagle Finance Corp. ("AP Finance Corp.") and collectively with the Partnership, Finance Corp. and Eagle Finance Corp., (the "Registrant") hereby certify that:

- The Registrant's periodic report on Form 10-K for the period ended September 30, 2003 (the "Form 10-K") fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Form 10-K fairly presents, in all material respects, the financial condition and results of operations of the Registrant.

CHIEF EXECUTIVE OFFICER

CHIEF FINANCIAL OFFICER

/s/ Eugene V. N. Bissell

/s/ Martha B.Lindsay

Eugene V. N. Bissell

Martha B.Lindsay

Date: December 16, 2003

Date: December 16, 2003

A signed original of this written statement required by Section 906 has been provided to AmeriGas Partners, L.P., AmeriGas Finance Corp., AmeriGas Eagle Finance Corp. and AP Eagle Finance Corp. and will be retained by AmeriGas Partners, L.P., AmeriGas Finance Corp., AmeriGas Eagle Finance Corp. and AP Eagle Finance Corp. and furnished to the Securities and Exchange Commission or its staff upon request.